UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, DC 20549

FORM 10-Q

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the Quarterly Period Ended September 30, 2024

□ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Commission file number 001-07349

BALL CORPORATION

State of Indiana

35-0160610

(State or other jurisdiction of incorporation or organization)

(I.R.S. Employer Identification No.)

9200 West 108th Circle Westminster, CO

80021

(Address of registrant's principal executive office)

(Zip Code)

Registrant's telephone number, including area code: 303/469-3131

Securities registered pursuant to section 12(b) of the Act:

Class	Trading Symbol	Name of Exchange	Outstanding at October 28, 2024
Common Stock, without par value	BALL	NYSE	298.425.962 shares

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \boxtimes No \square

Indicate by check mark whether the registrant has submitted electronically every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T (\S 232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes \boxtimes No \square

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer ⊠ Non-accelerated filer □

Accelerated filer □
Smaller reporting company □
Emerging growth company □

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes \square No \bowtie

Ball Corporation QUARTERLY REPORT ON FORM 10-Q For the period ended September 30, 2024 INDEX

		Page Number
_	INANCIAL INFORMATION	1
F	inancial Statements	1
	Unaudited Condensed Consolidated Statements of Earnings for the Three and Nine Months Ended September 30, 024 and 2023	1
	Unaudited Condensed Consolidated Statements of Comprehensive Earnings (Loss) for the Three and Nine Months Ended September 30, 2024 and 2023	2
L	Unaudited Condensed Consolidated Balance Sheets at September 30, 2024, and December 31, 2023	3
L	Unaudited Condensed Consolidated Statements of Cash Flows for the Nine Months Ended September 30, 2024 and 023	4
N	Jotes to the Unaudited Condensed Consolidated Financial Statements	•
	Note 1. Basis of Presentation	5
	Note 2. Accounting Pronouncements	7
	Note 3. Business Segment Information	8
	Note 4. Acquisitions and Dispositions	9
	Note 5. Revenue from Contracts with Customers	12
	Note 6. Business Consolidation and Other Activities	13
	Note 7. Supplemental Cash Flow Statement Disclosures	13
	Note 8. Receivables, Net	14
	Note 9. Inventories, Net	14
	Note 10. Property, Plant and Equipment, Net	14
	Note 11. Goodwill	15
	Note 12. Intangible Assets, Net	15
	Note 13. Other Assets	15
	Note 14. Leases	16
	Note 15. Debt	16
	Note 16. Taxes on Income	17
	Note 17. Employee Benefit Obligations	17
	Note 18. Equity and Accumulated Other Comprehensive Earnings (Loss)	19
	Note 19. Earnings and Dividends Per Share	21
	Note 20. Financial Instruments and Risk Management	21
	Note 21. Contingencies	26
	Note 22. Indemnifications and Guarantees	27
N	Management's Discussion and Analysis of Financial Condition and Results of Operations	29
<u>C</u>	Quantitative and Qualitative Disclosures About Market Risk	38
C	Controls and Procedures	38
C	OTHER INFORMATION	39

PART I. FINANCIAL INFORMATION

Item 1. FINANCIAL STATEMENTS

BALL CORPORATION UNAUDITED CONDENSED CONSOLIDATED STATEMENTS OF EARNINGS

	T	hree Months En	ded Se	ptember 30,		Nine Months Ended September 30,				
(\$ in millions, except per share amounts)		2024		2023	_	2024		2023		
Net sales	\$	3,082	\$	3,111	\$	8,915	\$	9,159		
Cost of sales (excluding depreciation and amortization)		(2,425)		(2,512)		(7,065)		(7,450)		
Depreciation and amortization		(150)		(152)		(460)		(449)		
Selling, general and administrative		(142)		(133)		(518)		(409)		
Business consolidation and other activities		(85)		(29)		(171)		(43)		
Interest income		14		12		58		23		
Interest expense		(67)		(122)		(228)		(351)		
Debt refinancing and other costs						(3)				
Earnings before taxes		227		175		528		480		
Tax (provision) benefit		(42)		(45)		(118)		(107)		
Equity in results of affiliates, net of tax		8		3		21		13		
Earnings from continuing operations		193		133		431		386		
Discontinued operations, net of tax		6		71		3,613		171		
Net earnings		199		204		4,044		557		
Net earnings attributable to noncontrolling interests		2		1		4		4		
Net earnings attributable to Ball Corporation	\$	197	\$	203	\$	4,040	\$	553		
Earnings per share:										
Basic - continuing operations	\$	0.63	\$	0.42	\$	1.38	\$	1.22		
Basic - discontinued operations		0.02		0.22		11.70		0.54		
Total basic earnings per share	\$	0.65	\$	0.64	\$	13.08	\$	1.76		
Diluted - continuing operations	\$	0.63	\$	0.42	\$	1.37	\$	1.20		
Diluted - discontinued operations		0.02		0.22		11.59		0.54		
Total diluted earnings per share	\$	0.65	\$	0.64	\$	12.96	\$	1.74		
Weighted average shares outstanding: (000s)		202.405		214.002		200.051		214.506		
Basic		302,406		314,983		308,851		314,596		
Diluted		305,219		317,296		311,674		316,938		

BALL CORPORATION UNAUDITED CONDENSED CONSOLIDATED STATEMENTS OF COMPREHENSIVE EARNINGS (LOSS)

	ee Months End	led Sept		Nine Months Ended September 30,					
(\$ in millions)	 2024		2023		2024		2023		
Net earnings	\$ 199	\$	204	\$	4,044	\$	557		
Other comprehensive earnings (loss):									
Currency translation adjustment	43		(40)		(96)		18		
Pension and other postretirement benefits	(14)		(1)		134		8		
Derivatives designated as hedges	(39)		(30)		(6)		14		
Total other comprehensive earnings (loss)	(10)		(71)		32		40		
Tax (provision) benefit	19		11		(28)		(3)		
Total other comprehensive earnings (loss), net of tax	9		(60)		4		37		
Total comprehensive earnings	208		144		4,048		594		
Comprehensive earnings attributable to noncontrolling									
interests	 2		1		4		4		
Comprehensive earnings attributable to Ball	• • •								
Corporation	\$ 206	\$	143	\$	4,044	\$	590		

BALL CORPORATION UNAUDITED CONDENSED CONSOLIDATED BALANCE SHEETS

(§ in millions)	September 30, 2024		De	cember 31, 2023
Assets				
Current assets				
Cash and cash equivalents	\$	1,440	\$	695
Receivables, net		2,655		2,057
Inventories, net		1,385		1,531
Other current assets		113		231
Current assets held for sale		14		369
Total current assets		5,607		4,883
Noncurrent assets				
Property, plant and equipment, net		6,550		6,715
Goodwill		4,244		4,250
Intangible assets, net		1,138		1,248
Other assets		1,285		1,354
Noncurrent assets held for sale		_		853
Total assets	\$	18,824	\$	19,303
Liabilities and Equity				
Current liabilities				
Short-term debt and current portion of long-term debt	\$	452	\$	1.065
Accounts payable		3,261		3,661
Accrued employee costs		308		245
Other current liabilities		1,103		779
Current liabilities held for sale		´ —		435
Total current liabilities		5.124		6,185
Noncurrent liabilities		- ,		-,
Long-term debt		5,353		7,504
Employee benefit obligations		630		735
Deferred taxes		592		421
Other liabilities		370		384
Noncurrent liabilities held for sale		_		237
Total liabilities		12,069		15,466
Equity				
Common stock (684,049,413 shares issued - 2024; 683,241,401 shares issued - 2023)		1,384		1,312
Retained earnings		11.618		7,763
Accumulated other comprehensive earnings (loss)		(912)		(916)
Treasury stock, at cost (384,006,576 shares - 2024; 367,551,366 shares - 2023)		(5,407)		(4,390)
Total Ball Corporation shareholders' equity		6,683	_	3,769
Noncontrolling interests		72		68
Total equity	_	6.755		3,837
Total liabilities and equity	\$	18,824	\$	19,303
total natinues and equity	Φ	10,024	Ψ	17,505

BALL CORPORATION UNAUDITED CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS

	Nine Months Ended Septembe						
(\$ in millions)		2024		2023			
Cash Flows from Operating Activities							
Net earnings	\$	4,044	\$	557			
Adjustments to reconcile net earnings to cash provided by (used in) operating activities:	Ψ	7,077	Ψ	331			
Depreciation and amortization		469		509			
Business consolidation and other activities		171		43			
Deferred tax provision (benefit)		201		(87)			
Gain on Aerospace disposal		(4,694)		18			
Pension contributions		(24)		(13)			
Other, net		78		71			
Changes in working capital components, net of dispositions		(630)		29			
Cash provided by (used in) operating activities		(385)		1,127			
Cash Flows from Investing Activities							
Capital expenditures		(377)		(830)			
Business dispositions, net of cash sold		5,422		`—			
Other, net		136		4			
Cash provided by (used in) investing activities		5,181		(826)			
• • • • • • • • • • • • • • • • • • • •							
Cash Flows from Financing Activities							
Long-term borrowings		450		1,700			
Repayments of long-term borrowings		(3,279)		(913)			
Net change in short-term borrowings		51		(135)			
Acquisitions of treasury stock		(1,061)		(3)			
Common stock dividends		(185)		(189)			
Other, net		26		30			
Cash provided by (used in) financing activities		(3,998)		490			
Effect of exchange rate changes on cash		(64)					
Effect of exchange rate changes on easi		(04)	_	_			
Change in cash, cash equivalents and restricted cash		734		791			
Cash, cash equivalents and restricted cash - beginning of period	_	710		558			
Cash, cash equivalents and restricted cash - end of period	\$	1,444	\$	1,349			

Ball Corporation

Notes to the Unaudited Condensed Consolidated Financial Statements

1. Basis of Presentation

The accompanying unaudited condensed consolidated financial statements (consolidated financial statements) include the accounts of Ball Corporation and its controlled affiliates, including its consolidated variable interest entities (collectively Ball, the company, we or our), and have been prepared by the company. Certain information and footnote disclosures, including critical and significant accounting policies normally included in consolidated financial statements prepared in accordance with generally accepted accounting principles, have been condensed or omitted for this quarterly presentation.

Results of operations for the periods shown are not necessarily indicative of results for the year, particularly in view of the seasonality in the packaging segments. These consolidated financial statements and accompanying notes should be read in conjunction with the consolidated financial statements and the notes thereto included in the company's 2023 Annual Report on Form 10-K filed on February 20, 2024, pursuant to the Securities Exchange Act of 1934 for the fiscal year ended December 31, 2023 (annual report).

The preparation of consolidated financial statements in conformity with accounting principles generally accepted in the United States of America (U.S. GAAP) requires Ball's management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent liabilities at the date of the consolidated financial statements and reported amounts of revenues and expenses during the reporting periods. These estimates are based on historical experience and various assumptions believed to be reasonable under the circumstances. Ball's management evaluates these estimates on an ongoing basis and adjusts or revises the estimates as circumstances change. As future events and their impacts cannot be determined with precision, actual results may differ from these estimates. In the opinion of management, the consolidated financial statements reflect all adjustments that are of a normal recurring nature and are necessary to fairly state the results of the periods presented.

On February 16, 2024, the company completed the divestiture of its aerospace business. The transaction represents a strategic shift; therefore, the company's consolidated financial statements reflect the aerospace business' financial results as discontinued operations for all periods presented. The aerospace business was historically presented as a reportable segment. Effective as of the first quarter of 2024, the company reports its financial performance in three reportable segments: (1) beverage packaging, North and Central America; (2) beverage packaging, Europe, Middle East and Africa (beverage packaging, EMEA) and (3) beverage packaging, South America. See Note 3 for additional segment information.

Unless otherwise specified, these notes to the unaudited condensed consolidated financial statements reflect continuing operations only.

Certain prior year amounts, including amounts related to discontinued operations, have been reclassified in order to conform to the current year presentation. See Note 4 for additional discontinued operations information.

Risks and Uncertainties

Global Economic Environment

Recent data has indicated continued high inflation in the regions where we operate. Current and future inflationary effects may continue to be impacted by, among other things, supply chain disruptions, governmental stimulus or fiscal and monetary policies, changes in interest rates, and changing demand for certain goods and services. We cannot predict with any certainty the impact that interest rates, a global or any regional recession, or higher inflation may have on our customers or suppliers. Additionally, we are unable to predict the potential effects that any future pandemic, or the continuation or escalation of global conflicts, including the conflict between Russia and Ukraine and the rising instability in the Middle East and Myanmar, and related sanctions or market disruptions, may have on our business. It remains uncertain how long any of these conditions may last or how severe any of them may become.

Ball Corporation

Notes to the Unaudited Condensed Consolidated Financial Statements

Ball management has reviewed the estimates used in preparing the company's consolidated financial statements and the following have a reasonably possible likelihood of being affected, to a material extent, by the direct and indirect impacts of the current global economic environment in the near-term.

- Estimates regarding the future financial performance of the business used in the impairment tests for goodwill, long-lived assets, equity method investments, recoverability of deferred tax assets and estimates regarding cash needs and associated indefinite reinvestment assertions;
- Estimates of recoverability for customer receivables;
- Estimates of net realizable value for inventory; and
- Estimates regarding the likelihood of forecasted transactions associated with hedge accounting positions at September 30, 2024, which could impact the company's ability to satisfy hedge accounting requirements and result in the recognition of income and/or expenses.

In addition to the above potential impacts on the estimates used in preparing the consolidated financial statements, the current global economic environment has the potential to increase Ball's vulnerabilities to near-term severe impacts related to certain concentrations in its business. In line with other companies in the packaging industry, Ball makes the majority of its sales and significant purchases to or from a relatively small number of global, or large regional, customers and suppliers. Furthermore, Ball makes the majority of its sales from a small number of product lines. The potential of the current global economic environment to affect a significant customer or supplier, or to affect demand for certain products to a significant degree, heightens the vulnerability of Ball to these concentrations.

Argentina

Although Ball's functional currency in Argentina is the U.S. dollar, a portion of its transactions are denominated in pesos. During the fourth quarter of 2023, Argentina suddenly devalued its peso relative to the U.S. dollar as one of the economic policies implemented by the new government with the goal of stabilizing and growing the economy. The government has implemented additional policies with the same goal in mind, including additional taxes on the importation of certain goods. The currency devaluation, economic conditions and policies in Argentina make it difficult to manage currency exchange rate risk and may lead to additional adverse effects on the company's results of operations. Ball's Argentinean business is presented in its beverage packaging, South America, reportable operating segment. Ball's pesodenominated net monetary assets in Argentina were approximately \$6 million at September 30, 2024. As of September 30, 2024, Ball's Argentinean business had net asset exposure of \$388 million, which consisted primarily of working capital and property, plant and equipment.

Aluminum Cups

The Ball Aluminum Cups business was launched during 2019 to serve the growing demand for innovative, sustainable beverage packaging among customers and consumers. The aluminum cups business has allowed us to use our years of experience and specialized expertise to serve an unmet need with an environmentally friendly addition to our portfolio of aluminum packages. At this time, the growth of this business has not been at the level we initially expected. As a result, the company is currently evaluating various options for this business. If the company decides to pursue a different path, it could present a risk of future impairment loss.

As of September 30, 2024, the Ball Aluminum Cups business had net assets of \$213 million, which consisted primarily of working capital and property, plant and equipment. The business, which is a non-reportable operating segment, represented less than one percent of the company's total net sales for the nine months ended September 30, 2024.

Ball Corporation Notes to the Unaudited Condensed Consolidated Financial Statements

2. Accounting Pronouncements

Recently Adopted Accounting Standards

Supplier Finance Programs

In 2022, new guidance was issued by the Financial Accounting Standards Board (FASB) with the goal of enhancing transparency around supplier finance programs. On January 1, 2023, Ball adopted all required disclosures effective for 2023, on a retrospective basis. The company will adopt the rollforward disclosure requirements, on a prospective basis, in its 2024 annual report.

The company has several regional supplier finance programs, all of which have substantially similar characteristics, with various financial institutions that act as the paying agent for certain payables of the company. The company establishes these programs through agreements with the financial institutions to enable more efficient payment processing to our suppliers while also providing our suppliers a potential source of liquidity to the extent they enter into a factoring agreement with the financial institutions. Our suppliers' participation in the programs is voluntary, and the company is not involved in negotiations of the suppliers' arrangements with the financial institutions to sell their receivables, and our rights and obligations to our suppliers are not impacted by our suppliers' decisions to sell amounts under these programs. Under these supplier finance programs, the company pays the financial institutions the stated amount of confirmed invoices from its participating suppliers on the original maturity dates of the invoices, which vary based on the negotiated terms with each supplier. All payment terms are short-term in nature and are not dependent on whether the suppliers participate in the supplier finance programs or if the suppliers elect to receive early payment from the financial institutions. Our supplier finance programs do not include any of the following: guarantees to the financial institutions, assets pledged as securities or interest accruing on the obligation prior to the due date.

Based on the review of the facts and circumstances of our supplier finance programs, including but not limited to those noted above, the company has concluded that the characteristics of the obligations due under our supplier finance programs have not changed and remain those of standard accounts payable, rather than indicative of debt.

The amount of obligations outstanding that the company confirmed as valid to the financial institutions under the company's programs was \$432 million and \$703 million at September 30, 2024 and December 31, 2023, respectively. These amounts are classified within accounts payable in the unaudited condensed consolidated balance sheets, and the associated payments are reflected in the cash flows from operating activities section of the unaudited condensed consolidated statements of cash flows.

New Accounting Guidance and Disclosure Requirements

Climate Disclosures

In 2024, the Securities and Exchange Commission (SEC) adopted final rules to require disclosures about material climate-related risks, the actual and potential impact of the risks and additional related disclosures. The final rules are currently under a stay by the SEC and the effective dates for the rules are uncertain. The company is continuing to assess the impact that the adoption of this new guidance will have on its consolidated financial statements and expects to meet the disclosure requirements on a prospective basis when the rules become effective for Ball.

Income Tax Disclosures

In 2023, new guidance was issued by the FASB with the goal of providing financial statement users with more information in the income tax rate reconciliation table and regarding income taxes paid. The company is assessing the impact that the adoption of this new guidance will have on its consolidated financial statements and expects to meet the disclosure requirements on a prospective basis in its 2025 annual report.

Ball Corporation

Notes to the Unaudited Condensed Consolidated Financial Statements

Segment Reporting

In 2023, new guidance was issued by the FASB with the goal of providing financial statement users with more information about reportable segments, including more disaggregated expense information. The company is assessing the impact that the adoption of this new guidance will have on its consolidated financial statements and expects to meet the disclosure requirements on a retrospective basis in its 2024 annual report and interim periods thereafter.

3. Business Segment Information

Ball's operations are organized and reviewed by management along its product lines and geographical areas and presented in the three reportable segments outlined below.

<u>Beverage packaging, North and Central America</u>: Consists of operations in the U.S., Canada and Mexico that manufacture and sell aluminum beverage containers throughout those countries.

<u>Beverage packaging, EMEA</u>: Consists of operations in numerous countries throughout Europe, as well as Egypt and Turkey, that manufacture and sell aluminum beverage containers throughout those countries.

<u>Beverage packaging, South America</u>: Consists of operations in Brazil, Argentina, Paraguay and Chile that manufacture and sell aluminum beverage containers throughout most of South America.

As presented in the table below, Other consists of a non-reportable operating segment (beverage packaging, other) that manufactures and sells aluminum beverage containers in India, Saudi Arabia and Myanmar; a non-reportable operating segment that manufactures and sells extruded aluminum aerosol containers and recloseable aluminum bottles across multiple consumer categories as well as aluminum slugs (aerosol packaging) throughout North America, South America, Europe, and Asia; a non-reportable operating segment that manufactures and sells aluminum cups (aluminum cups); undistributed corporate expenses; and intercompany eliminations and other business activities.

The accounting policies of the segments are the same as those used in the consolidated financial statements, as discussed in Note 1. The company also has investments in operations in Guatemala, Panama, the U.S. and Vietnam that are accounted for under the equity method of accounting and, accordingly, those results are not included in segment sales or earnings.

Ball Corporation

Notes to the Unaudited Condensed Consolidated Financial Statements

Summary of Business by Segment

		Three Months En	ded S	eptember 30,	Nine Months Ended September 30,				
(\$ in millions)		2024	_	2023	2024		2023		
Net sales									
Beverage packaging, North and Central America	\$	1,456	\$	1,541	\$ 4,328	\$	4,582		
Beverage packaging, EMEA		950		902	2,640		2,656		
Beverage packaging, South America		484		489	1,388		1,344		
Reportable segment sales		2,890		2,932	8,356		8,582		
Other		192		179	559		577		
Net sales	\$	3,082	\$	3,111	\$ 8,915	\$	9,159		
	_		_						
Comparable segment operating earnings									
Beverage packaging, North and Central America	\$	203	\$	196	\$ 605	\$	554		
Beverage packaging, EMEA		128		103	326		274		
Beverage packaging, South America		78		61	170		141		
Reportable segment comparable operating earnings		409		360	1,101		969		
Reconciling items									
Other (a)		4		_	(66)		7		
Business consolidation and other activities		(85)		(29)	(171)		(43)		
Amortization of acquired intangibles		(34)		(34)	(105)		(102)		
Interest expense		(67)		(122)	(228)		(351)		
Debt refinancing and other costs		<u> </u>			(3)		_		
Earnings before taxes	\$	227	\$	175	\$ 528	\$	480		

⁽a) Includes undistributed corporate expenses, net, of \$32 million and \$18 million for the three months ended September 30, 2024 and 2023, respectively, and \$149 million and \$60 million for the nine months ended September 30, 2024 and 2023, respectively. For the nine months ended September 30, 2024, undistributed corporate expenses, net, includes \$82 million of incremental compensation cost from the successful sale of the aerospace business consisting of cash bonuses and stock based compensation. For the three and nine months ended September 30, 2024, undistributed corporate expenses, net, include \$7 million and \$36 million of corporate interest income, respectively.

The company does not disclose total assets by segment as such information is not provided to the chief operating decision maker.

4. Acquisitions and Dispositions

Aerospace

In the third quarter of 2023, Ball entered into a Stock Purchase Agreement (Agreement) with BAE Systems, Inc. (BAE) and, for the limited purposes set forth therein, BAE Systems plc, to sell all outstanding equity interests in Ball's aerospace business. On February 16, 2024, the company completed the divestiture of the aerospace business for a purchase price of \$5.6 billion, subject to working capital adjustments and other customary closing adjustments under the terms of the Agreement. The company is in the process of finalizing the working capital adjustments and other customary closing adjustments with BAE, which is currently expected to be completed in 2024 and may adjust the final cash proceeds and gain on sale amounts. The divestiture resulted in a pre-tax gain of \$4.67 billion, which is net of \$20 million of costs to sell incurred and paid in 2023 related to the disposal. Cash proceeds received at close from the sale of \$5.42 billion, net of the cash disposed, are presented in business dispositions, net of cash sold, in the unaudited condensed consolidated statement of cash flows for the nine months ended September 30, 2024. The company expects to pay approximately \$950 million in income taxes related to the transaction throughout 2024, of which \$484 million has been paid as of September 30, 2024. The remaining amount of income taxes related to the transaction is recorded in other

Ball Corporation

Notes to the Unaudited Condensed Consolidated Financial Statements

current liabilities in the unaudited condensed consolidated balance sheet. Additionally, the completion of the divestiture resulted in the removal of the aerospace business from the company's obligor group, as the business no longer guarantees the company's senior notes and senior credit facilities.

The sale of the aerospace business represents a strategic shift that will have a major effect on Ball's operations and financial results, including the removal of the aerospace reportable segment. Due to this shift, for all periods presented, the consolidated financial statements reflect the aerospace business' financial results as discontinued operations in the unaudited condensed consolidated statements of earnings, and its assets and liabilities are presented as assets and liabilities held for sale in the unaudited condensed consolidated balance sheet as of December 31, 2023. See Note 1 for further information on the basis of presentation.

The following table presents components of discontinued operations, net of tax for the three and nine months ended September 30, 2024 and 2023:

	Thr	ee Months End	ded Sept	ember 30,	Nine Months Ended September 30,				
(\$ in millions)		2024		2023		2024	2023		
Net sales	\$	_	\$	460	\$	261	\$	1,467	
Cost of sales (excluding depreciation and amortization)		_		(382)		(214)		(1,205)	
Depreciation and amortization		_		(21)		(9)		(60)	
Selling, general and administrative		_		(11)		(11)		(42)	
Interest expense		_		_		_		1	
Gain (loss) on disposition		(1)		(18)		4,694		(18)	
Tax (provision) benefit		7		43		(1,108)		28	
Discontinued operations, net of tax	\$	6	\$	71	\$	3,613	\$	171	

Ball Corporation

Notes to the Unaudited Condensed Consolidated Financial Statements

The following table presents assets and liabilities that are classified as held for sale in the unaudited condensed consolidated balance sheet as of December 31, 2023:

(S in millions)	De	ecember 31, 2023
Assets		
Current assets		
Receivables, net	\$	277
Other current assets		56
Total current assets		333
Noncurrent assets		
Property, plant and equipment, net		665
Other assets		188
Total assets of discontinued operations	\$	1,186
Liabilities		
Current liabilities		
Accounts payable	\$	92
Accrued employee costs		88
Deferred revenue		221
Other current liabilities		34
Total current liabilities		435
Noncurrent liabilities		
Employee benefit obligations		163
Other liabilities		74
Total liabilities of discontinued operations	\$	672

The following table presents depreciation and amortization, capital expenditures and significant operating and investing noncash items from discontinued operations for the nine months ended September 30, 2024 and 2023 included within the consolidated statements of cash flows. Amounts include adjustments to reconcile net earnings to cash provided by (used in) operating activities:

		Nine Months Ended September 30,							
(\$ in millions)		2023							
Provided by (used in)									
Depreciation and amortization	\$	9	\$	60					
Gain on Aerospace disposal		(4,694)		18					
Capital expenditures		(13)		(75)					

Noncash investing activities include the acquisition of property, plant and equipment (PP&E) for which payment has not been made. These noncash capital expenditures are excluded from the consolidated statements of cash flows. A summary of the PP&E acquired but not yet paid for from discontinued operations is as follows:

	Nine Months Ended September 30,								
(\$ in millions)	2024		2023						
Supplemental cash flow information:									
PP&E acquired but not yet paid	\$	7 \$	3 18						

Aerosol Packaging

In late-October 2024, the company acquired the entire share capital of Alucan Entec, S.A, an impact extruded aluminum packaging business with a manufacturing facility in Lummen, Belgium and Llinars del Vallés, Spain, for the purchase price of €82 million, subject to customary closing adjustments. Using the exchange rate on the date of close, the initial

Ball Corporation

Notes to the Unaudited Condensed Consolidated Financial Statements

cash consideration of \$80 million (or €75 million) was paid at close, with an additional \$8 million (or €7 million) to be paid over the next four years, less any potential obligations covered by the holdback arrangement. The business is part of Ball's aerosol packaging operating segment. The transaction broadens the geographic reach and expands the product portfolio of Ball's aerosol packaging business, serving the growing personal, home care and beverage bottle markets.

5. Revenue from Contracts with Customers

The following table disaggregates the company's net sales based on the timing of transfer of control:

		Three Months Ended September 30,						Nine Months Ended September 30,					
(\$ in millions)		Point in Time		Over Time		Total		Point in Time		Over Time		Total	
	·												
2024	\$	678	\$	2,404	\$	3,082	\$	1,861	\$	7,054	\$	8,915	
2023		629		2,482		3,111		1,728		7,431		9,159	

Contract Balances

The company did not have any contract assets at either September 30, 2024, or December 31, 2023. Unbilled receivables, which are not classified as contract assets, represent arrangements in which sales have been recorded prior to billing and right to payment is unconditional.

The opening and closing balances of the company's current and noncurrent contract liabilities are as follows:

	C	ontract		Contract
(§ in millions)		Liabilities (Current)		Liabilities (Noncurrent)
Balance at December 31, 2023	\$	114	\$	3
Increase (decrease)		(30)		(1)
Balance at September 30, 2024	\$	84	\$	2

During the nine months ended September 30, 2024, contract liabilities decreased by \$31 million, which is net of cash received of \$101 million and amounts recognized as sales of \$132 million, the majority of which related to current contract liabilities. The amount of sales recognized in the nine months ended September 30, 2024, that was included in the opening contract liabilities balance, was \$114 million, all of which related to current contract liabilities. The difference between the opening and closing balances of the company's contract liabilities primarily results from timing differences between the company's performance and the customer's payments. Current contract liabilities are classified within other current liabilities in the unaudited condensed consolidated balance sheets and noncurrent contract liabilities are classified within other liabilities.

Ball Corporation

Notes to the Unaudited Condensed Consolidated Financial Statements

6. Business Consolidation and Other Activities

Business consolidation and other activities resulted in charges of \$85 million and \$171 million for the three and nine months ended September 30, 2024, respectively, in the unaudited condensed consolidated statements of earnings. During the three and nine months ended, the charges were primarily related to facility closure costs and costs for employee severance, employee benefits and other related items resulting from the company restructuring its operating model. The charges for the three and nine months ended September 30, 2024, were partially offset by income from the receipt of insurance proceeds for replacement costs related to the 2023 fire at the company's Verona, Virginia extruded aluminum slug manufacturing facility.

Business consolidation and other activities resulted in charges of \$29 million and \$43 million for the three and nine months ended September 30, 2023, respectively, in the unaudited condensed consolidated statements of earnings. These charges were primarily composed of facility closure costs. Due to the sale of the aerospace business, the company reclassed \$18 million of costs to sell incurred and paid in 2023 previously reported as business consolidation and other activities to discontinued operations, net of tax. See Note 4 for further details on these costs and the aerospace sale.

7. Supplemental Cash Flow Statement Disclosures

		September 30,						
(\$ in millions)	2	024	2023					
Beginning of period:								
Cash and cash equivalents	\$	695	\$	548				
Current restricted cash (included in other current assets)		15		10				
Total cash, cash equivalents and restricted cash	\$	710	\$	558				
End of period:								
Cash and cash equivalents	\$	1,440	\$	1,335				
Current restricted cash (included in other current assets)		4		14				
Total cash, cash equivalents and restricted cash	\$	1,444	\$	1,349				

The company's restricted cash is primarily related to receivables factoring programs and represents amounts collected from customers that have not yet been remitted to the banks as of the end of the reporting period.

Noncash investing activities include the acquisition of property, plant and equipment (PP&E) for which payment has not been made. These noncash capital expenditures are excluded from the unaudited condensed consolidated statements of cash flows. A summary of the PP&E acquired but not yet paid, inclusive of amounts related to the historical aerospace business, is as follows:

		September 30,								
n millions) 202				2023						
Beginning of period:										
PP&E acquired but not yet paid	\$	204	\$		392					
End of period:										
PP&E acquired but not yet paid	¢	101	•		207					
11 &E acquired but not yet paid	φ	101	Ф		207					

Ball Corporation

Notes to the Unaudited Condensed Consolidated Financial Statements

8. Receivables, Net

(§ in millions)	Sept	ember 30, 2024	December 31, 2023				
Trade accounts receivable	\$	1,707	\$	1,165			
Unbilled receivables		501		520			
Less: Allowance for doubtful accounts		(13)		(15)			
Net trade accounts receivable	·	2,195		1,670			
Other receivables		460		387			
	\$	2,655	\$	2,057			

The company has entered into several regional accounts receivable factoring programs with various financial institutions for certain receivables of the company. The programs are accounted for as true sales of the receivables and had combined limits of approximately \$1.63 billion and \$2.00 billion at September 30, 2024, and December 31, 2023, respectively. A total of \$702 million and \$350 million were available for sale under these programs as of September 30, 2024, and December 31, 2023, respectively. The combined limit and available for sale amount as of December 31, 2023, included \$160 million and \$97 million, respectively, associated with receivable factoring programs included within the historical aerospace reportable segment. The company has recorded expense related to its factoring programs of \$12 million and \$29 million for the three months ended September 30, 2024 and 2023, respectively, and \$35 million and \$69 million for the nine months ended September 30, 2024 and 2023 respectively, and has presented these amounts in selling, general and administrative in its unaudited condensed consolidated statements of earnings.

Other receivables include income and indirect tax receivables, aluminum scrap sale receivables and other miscellaneous receivables.

9. Inventories, Net

(\$ in millions)	ember 30, 2024	 December 31, 2023			
Raw materials and supplies	\$ 1,041	\$ 1,182			
Finished goods	438	440			
Less: Inventory reserves	(94)	(91)			
	\$ 1,385	\$ 1,531			

10. Property, Plant and Equipment, Net

(S in millions)	Sept	September 30, 2024		December 31, 2023		
Land	\$	205	\$	215		
Buildings		1,831		1,792		
Machinery and equipment		7,824		7,636		
Construction-in-progress		1,004		1,179		
		10,864		10,822		
Accumulated depreciation		(4,314)		(4,107)		
	\$	6,550	\$	6,715		

Depreciation expense was \$113 million and \$115 million for the three months ended September 30, 2024 and 2023, respectively, and \$345 million and \$336 million for the nine months ended September 30, 2024 and 2023, respectively.

Ball Corporation

Notes to the Unaudited Condensed Consolidated Financial Statements

11. Goodwill

(§ in millions)	P Nort	Beverage ackaging, th & Central America	 Beverage Packaging, EMEA	Beverage Packaging, outh America	 Other	 Total
Balance at December 31, 2023	\$	1,277	\$ 1,378	\$ 1,298	\$ 297	\$ 4,250
Effects of currency exchange		_	6	_	(14)	(8)
Other		_	_	2	_	2
Balance at September 30, 2024	\$	1,277	\$ 1,384	\$ 1,300	\$ 283	\$ 4,244

12. Intangible Assets, Net

(§ in millions)	September 30, 2024		_	December 31, 2023
Acquired customer relationships and other intangibles (net of accumulated amortization and impairment losses of \$1.13 billion at September 30, 2024, and \$1.06 billion at December 31, 2023)	\$	1,095	\$	1,197
Capitalized software (net of accumulated amortization of \$169 million at September 30, 2024, and \$162 million at December 31, 2023)		33		37
Other intangibles (net of accumulated amortization of \$14 million at September 30, 2024, and \$49 million at December 31, 2023)		10		14
	\$	1,138	\$	1,248

Total amortization expense of intangible assets was \$37 million for the three months ended September 30, 2024 and 2023, and \$115 million and \$113 million for the nine months ended September 30, 2024 and 2023, respectively.

13. Other Assets

(§ in millions)	September 30, 2024		December 31, 2023	
Long-term pension assets	\$	44	\$	41
Right-of-use operating lease assets		318		365
Investments in affiliates		228		212
Long-term deferred tax assets		64		114
Other		631		622
	\$	1,285	\$	1,354

Investments in affiliates primarily includes the company's 50 percent ownership interest in an entity in Guatemala, a 50 percent ownership interest in an entity in Vietnam and a 50 percent ownership interest in an entity in the U.S.

Ball Corporation

Notes to the Unaudited Condensed Consolidated Financial Statements

14 Leases

The company enters into operating leases for buildings, warehouses, office equipment, production equipment, aircraft, land and other types of equipment. The company also enters into finance leases for certain plant equipment and an aircraft. Supplemental balance sheet information related to the company's leases follows:

(S in millions)	ns) Balance Sheet Location		2024	December 31, 2023		
Operating leases:						
Operating lease ROU asset	Other assets	\$	318	\$	365	
Current operating lease liabilities	Other current liabilities		78		83	
Noncurrent operating lease liabilities	Other liabilities		249		287	
Finance leases:						
Finance lease ROU assets, net	Property, plant and equipment, net		32		8	
	Short-term debt and current portion of long-term					
Current finance lease liabilities	debt		27		3	
Noncurrent finance lease liabilities	Long-term debt		5		7	

15. Debt

Long-term debt consisted of the following:

(\$ in millions)	 September 30, 2024	December 31 2023	
Senior Notes			
0.875%, euro denominated, due March 2024	\$ _	\$	828
5.25% due July 2025	189		1,000
4.875% due March 2026	256		750
1.50%, euro denominated, due March 2027	612		607
6.875% due March 2028	750		750
6.00% due June 2029	1,000		1,000
2.875% due August 2030	1,300		1,300
3.125% due September 2031	850		850
Senior Credit Facility (at variable rates)			
U.S. dollar revolver due June 2027	_		_
Term A loan due June 2027 (5.95% - 2024)	625		1,325
Finance lease obligations	7		10
Other (including debt issuance costs)	(44)		(60)
	5,545		8,360
Less: Current portion	(192)		(856)
	\$ 5,353	\$	7,504

The company's senior credit facilities include long-term multi-currency revolving facilities that mature in June 2027, which provide the company with up to the U.S. dollar equivalent of \$1.75 billion. At September 30, 2024, \$1.72 billion was available under these revolving credit facilities. In addition to these facilities, the company had \$193 million of committed short-term loans outstanding and a \$25 million short-term finance lease outstanding. The company also had approximately \$1.04 billion of short-term uncommitted credit facilities available at September 30, 2024, of which \$42 million was outstanding and due on demand. At December 31, 2023, the company had \$196 million of committed short-term loans outstanding and \$13 million outstanding under short-term uncommitted credit facilities.

Ball Corporation

Notes to the Unaudited Condensed Consolidated Financial Statements

On February 14, 2024, Ball announced a public tender of the \$1.00 billion 5.25% senior notes due July 2025 and the \$750 million 4.875% senior notes due March 2026. On March 14, 2024, \$811 million of the \$1.00 billion 5.25% senior notes and \$494 million of the \$750 million 4.875% senior notes were validly tendered and accepted. Additionally, in the first quarter of 2024, Ball repaid at maturity the outstanding 0.875% euro denominated senior notes due in the amount of \$817 million and prepaid \$700 million of the Term A loan outstanding balance.

The fair value of Ball's long-term debt was estimated to be \$5.37 billion and \$8.07 billion at September 30, 2024 and December 31, 2023, respectively. The fair value reflects the market rates at each period end for debt with credit ratings similar to the company's ratings and is classified as Level 2 within the fair value hierarchy. Rates currently available to the company for loans with similar terms and maturities are used to estimate the fair value of long-term debt based on discounted cash flows.

The U.S. note agreements and bank credit agreement contain certain restrictions relating to dividend payments, share repurchases, investments, financial ratios, guarantees and the incurrence of additional indebtedness. The company's most restrictive debt covenant requires it to maintain a leverage ratio (as defined) of no greater than 5.0 times, which will change to 4.5 times as of September 30, 2025. The company was in compliance with the leverage ratio requirement at September 30, 2024, and for all prior periods presented, and has met all debt payment obligations.

16. Taxes on Income

The company's effective tax rate was 18.5 percent and 22.3 percent for the three and nine months ended September 30, 2024, respectively. As compared to the statutory U.S. tax rate, the effective tax rate for the three and nine months ended September 30, 2024, increased by 0.7 and 0.8 percentage points, respectively, for Pillar Two Global Minimum Taxes, increased by 0.2 and 1.0 percentage points, respectively, for non-U.S. rate differences and withholding taxes net of credits and decreased by 2.8 and 0.5 percentage points, respectively, for state and local taxes.

The company's effective tax rate was 25.7 percent and 22.3 percent for the three and nine months ended September 30, 2023, respectively. As compared to the statutory U.S. tax rate, the effective tax rate for the three and nine months ended September 30, 2023, increased by 3.3 and 0.6 percentage points, respectively, for items related to share-based compensation, increased by 1.3 and 0.9 percentage points, respectively, for Global Intangible Low-Taxed Income, increased by 1.0 and 0.8 percentage points, respectively, for state and local taxes, increased by 0.9 and 0.7 percentage points, respectively, for other U.S. permanent differences and decreased by 2.9 and 1.8 percentage points, respectively, for non-U.S. rate differences and withholding taxes net of credits.

17. Employee Benefit Obligations

(§ in millions)	Sept	tember 30, 2024	 December 31, 2023
Underfunded defined benefit pension liabilities	\$	286	\$ 323
Less: Current portion		(21)	(21)
Long-term defined benefit pension liabilities		265	302
Long-term retiree medical liabilities		82	90
Deferred compensation plans		222	280
Other		61	63
	\$	630	\$ 735

Ball Corporation

Notes to the Unaudited Condensed Consolidated Financial Statements

Components of net periodic benefit cost associated with the company's defined benefit pension plans were as follows:

Three Months Ended September 3												
	2024								2023			
(\$ in millions)	1	U.S.	No	n-U.S.		Total		U.S.	No	n-U.S.		Total
Ball-sponsored plans:												
Service cost	\$	4	\$	1	\$	5	\$	4	\$	1	\$	5
Interest cost		15		21		36		16		22		38
Expected return on plan assets		(22)		(21)		(43)		(22)		(26)		(48)
Amortization of prior service cost		_		1		1		1		1		2
Recognized net actuarial loss		1		4		5		1		_		1
Settlement losses and other charges (a)								3		_		3
Total net periodic benefit cost	\$	(2)	\$	6	\$	4	\$	3	\$	(2)	\$	1

(a) The charges in 2023 were plant-related closure charges. These amounts were recorded in business consolidation and other activities.

	Nine Months Ended September 30,												
	2024								2023				
(\$ in millions)		U.S.	No	n-U.S.		Total	=	U.S.	No	1-U.S.		Total	
Ball-sponsored plans:													
Service cost	\$	12	\$	2	S	14	S	12	\$	4	\$	16	
Interest cost	Ψ	45	Ψ	62	Ψ	107	Ψ	48	Ψ	64	Ψ	112	
Expected return on plan assets		(66)		(60)		(126)		(65)		(76)		(141)	
Amortization of prior service cost		_		2		2		1		2		3	
Recognized net actuarial loss		3		11		14		2		1		3	
Settlement losses and other charges (a)		_		_		_		3		_		3	
Total net periodic benefit cost	\$	(6)	\$	17	\$	11	\$	1	\$	(5)	\$	(4)	

(a) The charges in 2023 were plant-related closure charges. These amounts were recorded in business consolidation and other activities

Non-service pension income of \$1 million and \$7 million for the three months ended September 30, 2024 and 2023, respectively, and income of \$3 million and \$23 million for the nine months ended September 30, 2024 and 2023 respectively, is included in selling, general and administrative in the unaudited condensed consolidated statements of earnings.

Contributions to the company's defined benefit pension plans were \$24 million for the first nine months of 2024 compared to \$13 million for the first nine months of 2023, and such contributions are expected to be approximately \$30 million for the full year of 2024. This estimate may change based on changes in the Pension Protection Act, actual plan asset performance and available company cash flow, among other factors.

In November 2023, the Trustee Board of the U.K. defined benefit pension plan entered into an agreement with an insurance company for a bulk annuity purchase, or "buy-in", for its U.K. defined benefit pension plan to reduce retirement plan risk, while delivering promised benefits to plan participants. This transaction allows the company to reduce volatility by removing investment, longevity, mortality, interest rate and inflation risk upon the transfer of substantially all of the pension plan assets to the insurer in exchange for the group annuity insurance contract. At this time the company retains both the fair value of the annuity contract within plan assets and the pension benefit obligations related to these participants. The plan was frozen on April 5, 2024, and future service accruals were replaced with enhanced defined contribution benefits for the impacted employees. The company anticipates the "buy-out" may occur within three years of the plan freeze, which will trigger a pension settlement that will result in all plan balances, including accumulated pension components within other comprehensive income, being charged to expense as a noncash settlement charge.

Ball Corporation

Notes to the Unaudited Condensed Consolidated Financial Statements

18. Equity and Accumulated Other Comprehensive Earnings (Loss)

The following tables provide additional details of the company's equity activity, inclusive of activity related to the aerospace business impacting the company's equity:

(S in millions; share amounts in thousands)	Number of Shares	Stock Amount	Treasury Number of Shares	Stock Amount	Retained Earnings	Accumulated Other Comprehensive Earnings (Loss)	Noncontrolling Interest	Total Equity
Balance at June 30, 2024	683,801	\$ 1,370	(377,841)	\$ (5,017)	\$ 11,481	\$ (921)	\$ 70	\$ 6,983
Net earnings				_	197	9	2	199
Other comprehensive earnings (loss), net of tax Common dividends	_	_	_	_	(60)	9	_	(60
Treasury stock purchases	_		(6,216)	(394)	(00)			(394
Treasury shares reissued	_	_	50	3	_	_	_	3
Shares issued and stock compensation for stock options and other stock plans, net of shares exchanged Distributions from deferred compensation plans and other	248	14	_	_	_	_	_	14
activity	_	_		1	_	_	_	1
Balance at September 30, 2024	684,049	\$ 1,384	(384,007)	\$ (5,407)	\$ 11,618	\$ (912)	\$ 72	\$ 6,755
(S in millions; share amounts in thousands)	Common Number of Shares	Stock Amount	Treasury Number of Shares	Stock Amount	Retained Earnings	Accumulated Other Comprehensive Earnings (Loss)	Noncontrolling Interest	Total Equity
Balance at June 30, 2023	682,728	\$ 1,291	(367,793)	\$ (4,406)	\$ 7,533	\$ (582)	\$ 69	\$ 3,905
Net earnings	002,720	- 1,2/1	(307,793)	ψ (1,100)	203	(382)	1	204
Other comprehensive earnings (loss), net of tax	_	_	_	_		(60)		(60
Common dividends	_	_	_	_	(63)		_	(63
Treasury stock purchases	_	_	(8)	_	_	_	_	-
Treasury shares reissued Shares issued and stock compensation for stock options and		_	139	8		_		8
other stock plans, net of shares exchanged Other activity	168	13		<u> </u>	=	=	_	13 1
Balance at September 30, 2023	682,896	\$ 1,304	(367,662)	\$ (4,397)	\$ 7,673	\$ (642)	\$ 70	\$ 4,008
(S in millions; share amounts in thousands)	Number of Shares	Amount	Number of Shares	Amount	Retained Earnings	Accumulated Other Comprehensive Earnings (Loss)	Noncontrolling Interest	Total Equity
Balance at December 31, 2023	Number of		Number of		Earnings \$ 7,763	Comprehensive	Interest 68	Equity \$ 3,837
Balance at December 31, 2023 Net earnings	Number of Shares	Amount	Number of Shares	Amount	Earnings	Comprehensive Earnings (Loss)	Interest	Equity \$ 3,837 4,044
Balance at December 31, 2023	Number of Shares	Amount	Number of Shares	Amount	Earnings \$ 7,763	Comprehensive Earnings (Loss) \$ (916)	Interest 68	Equity \$ 3,837
Net earnings Other comprehensive earnings (loss), net of tax Common dividends Treasury stock purchases	Number of Shares 683,241	* 1,312 — —	Number of Shares (367,551) — — — — — — — — — — — — — — — — — — —	Amount \$ (4,390)	\$ 7,763 4,040 — (185)	Comprehensive Earnings (Loss) \$ (916) 4	\$ 68 4	\$ 3,837 4,044 (185 (1,075
Balance at December 31, 2023 Net earnings Other comprehensive earnings (loss), net of tax Common dividends Treasury stock purchases Treasury shares reissued	Number of Shares 683,241	* 1,312 — — — —	Number of Shares (367,551)	Amount \$ (4,390)	\$ 7,763 4,040	Comprehensive Earnings (Loss) \$ (916) 4	\$ 68 4 —	\$ 3,837 4,044 (185
Net earnings Other comprehensive earnings (loss), net of tax Common dividends Treasury stock purchases	Number of Shares 683,241	Amount \$ 1,312	Number of Shares (367,551) — — — — — — — — — — — — — — — — — — —	Amount \$ (4,390)	\$ 7,763 4,040 — (185)	Comprehensive Earnings (Loss) \$ (916) 4	\$ 68 4 —	\$ 3,837 4,044 (185 (1,075
Net earnings Other comprehensive earnings (loss), net of tax Common dividends Treasury shock purchases Treasury shares reissued Shares issued and stock compensation for stock options and other stock plans, net of shares exchanged	Number of Shares	** Amount ** 1,312 **	(367,551) (16,530) 74	Amount \$ (4,390)	\$ 7,763 4,040 — (185) — — —	Comprehensive Earnings (Loss) \$ (916) 4	Interest S 68 4	Equity \$ 3,837 4,044 (185 (1,075 13 72
Net earnings Other comprehensive earnings (loss), net of tax Common dividends Treasury stock purchases Treasury stares reissued Shares issued and stock compensation for stock options and other stock plans, net of shares exchanged Distributions from deferred compensation plans and other	Number of Shares 683,241	* 1,312 — — — — — — — — — — — — — — — — — — —	Number of Shares (367,551) — — — — — — — — — — — — — — — — — — —	Amount \$ (4,390)	\$ 7,763 4,040 — (185)	Comprehensive Earnings (Loss) \$ (916) 4	\$ 68 4 —	\$ 3,837 4,044 (185 (1,075
Balance at December 31, 2023 Net earnings Other comprehensive earnings (loss), net of tax Common dividends Treasury stock purchases Treasury shares reissued Shares issued and stock compensation for stock options and other stock plans, net of shares exchanged Distributions from deferred compensation plans and other activity Balance at September 30, 2024	Number of Shares 683,241	Amount \$ 1,312	Number of Shares (367,551)	Amount \$ (4,390)	\$ 7,763 4,040 (185) 	Comprehensive Earnings (Loss) \$ (916) 4	Interest	Equity \$ 3,837 4,044 4 (188 (1,075 13 72 45 \$ 6,755
Net earnings Other comprehensive earnings (loss), net of tax Common dividends Treasury stock purchases Treasury stock purchases Treasury shares reissued Shares issued and stock compensation for stock options and other stock plans, net of shares exchanged Distributions from deferred compensation plans and other activity	Number of Shares 683,241	Amount \$ 1,312	Number of Shares (367,551)	Amount \$ (4,390)	\$ 7,763 4,040 (185) ————————————————————————————————————	Comprehensive Earnings (Loss) \$ (916) 4	Interest S 68 4	Equity \$ 3,837 4,044 4,048 (1885 (1,075 13 72 45 \$ 6,755
Net earnings Other comprehensive earnings (loss), net of tax Common dividends Treasury stock purchases Treasury stock purchases Treasury shares reissued Shares issued and stock compensation for stock options and other stock plans, net of shares exchanged Distributions from deferred compensation plans and other activity Balance at September 30, 2024 (S in millions; share amounts in thousands) Balance at December 31, 2022 Net earnings	Number of Shares 683,241	Amount \$ 1,312	Number of Shares (367,551)	Amount \$ (4,390)	\$ 7,763 4,040 (185) 	Comprehensive Earnings (Loss) \$ (916) 4 4	S 68 4 4 S 72	Equity \$ 3,837 4,044 4 (185 (1,075 175 77 45 \$ 6,755 Total Equity \$ 3,527 555
Balance at December 31, 2023 Net earnings Other comprehensive earnings (loss), net of tax Common dividends Treasury stock purchases Treasury shares reissued Shares issued and stock compensation for stock options and other stock plans, net of shares exchanged Distributions from deferred compensation plans and other activity Balance at September 30, 2024 (S in millions; share amounts in thousands) Balance at December 31, 2022 Net earnings Other comprehensive earnings (loss), net of tax	Number of	Amount \$ 1,312	Number of Shares (367,551)	Amount \$ (4,390)	Total Content	Comprehensive Earnings (Loss) \$ (916) 4	Interest S 68 4 4	Equity \$ 3,837 4,044 4 (188 (1,075 137 72 4 45 \$ 6,755 Total Equity \$ 3,527 557
Net earnings Other comprehensive earnings (loss), net of tax Common dividends Treasury shock purchases Treasury shares reissued Shares issued and stock compensation for stock options and other stock plans, net of shares exchanged Distributions from deferred compensation plans and other activity Balance at September 30, 2024 (S in millions; share amounts in thousands) Balance at December 31, 2022 Net earnings Other comprehensive earnings (loss), net of tax Common dividends	Number of	Amount \$ 1,312	Number of Shares (367,551)	Amount \$ (4,390)	\$ 7,763 4,040 (185) 	Comprehensive Earnings (Loss) \$ (916) 4 4	Interest S 68 4 4	Equity \$ 3,837 4,044 4 (185 (1,075 12 72 45 \$ 6,755 Total Equity \$ 3,527 557 37 (188)
Net earnings Other comprehensive earnings (loss), net of tax Common dividends Treasury stock purchases Treasury stock purchases Treasury stock purchases Treasury stock purchases Treasury shares reissued Shares issued and stock compensation for stock options and other stock plans, net of shares exchanged Distributions from deferred compensation plans and other activity Balance at September 30, 2024 (S in millions; share amounts in thousands) Balance at December 31, 2022 Net earnings Other comprehensive earnings (loss), net of tax Common dividends Treasury stock purchases	Number of	Amount \$ 1,312	Number of Shares (367,551)	Amount \$ (4,390)	Total Content	Comprehensive Earnings (Loss) \$ (916) 4 4	Interest S 68 4 4	Equity \$ 3,837 4,044 4 (188 (1,075 123 72 4 45 \$ 6,755 Total Equity \$ 3,527 537 (188
Ralance at December 31, 2023 Net earnings Other comprehensive earnings (loss), net of tax Common dividends Treasury stock purchases Treasury stares reissued Shares issued and stock compensation for stock options and other stock plans, net of shares exchanged Distributions from deferred compensation plans and other activity Balance at September 30, 2024 (S in millions; share amounts in thousands) Balance at December 31, 2022 Net earnings Other comprehensive earnings (loss), net of tax Common dividends Treasury stock purchases Treasury stock purchases	Number of	Amount \$ 1,312	Number of Shares (367,551)	Amount \$ (4,390)	S	Comprehensive Earnings (Loss) \$ (916) 4 4 S (912) Accumulated Other Comprehensive Earnings (Loss) \$ (679) 37 37	Interest S 68	Equity \$ 3,837 4,044 4 (185 (1,075 12 72 45 \$ 6,755 Total Equity \$ 3,527 557 37 (188)
Net earnings Other comprehensive earnings (loss), net of tax Common dividends Treasury stock purchases Treasury stock purchases Treasury stock purchases Treasury stock purchases Treasury shares reissued Shares issued and stock compensation for stock options and other stock plans, net of shares exchanged Distributions from deferred compensation plans and other activity Balance at September 30, 2024 (S in millions; share amounts in thousands) Balance at December 31, 2022 Net earnings Other comprehensive earnings (loss), net of tax Common dividends Treasury stock purchases	Number of	Amount \$ 1,312	Number of Shares (367,551)	Amount \$ (4,390)	S	Comprehensive Earnings (Loss) \$ (916) 4 4 S (912) Accumulated Other Comprehensive Earnings (Loss) \$ (679) 37 37	Interest S 68	Equity \$ 3,837 4,044 4 (188 (1,075 137 72 4 (188 5 6,755 Total Equity \$ 3,527 557 37 (188 (3) 22
Net earnings Other comprehensive earnings (loss), net of tax Common dividends Treasury shores reissued Shares issued and stock compensation for stock options and other stock plans, net of shares exchanged Distributions from deferred compensation plans and other activity Balance at September 30, 2024 (S in millions; share amounts in thousands) Balance at December 31, 2022 Net earnings Other comprehensive earnings (loss), net of tax Common dividends Treasury shares reissued Shares issued and stock compensation for stock options and	Number of	Amount \$ 1,312	Number of Shares (367,551)	Amount \$ (4,390)	S	Comprehensive Earnings (Loss) \$ (916) 4 4 S (912) Accumulated Other Comprehensive Earnings (Loss) \$ (679) 37 37	Interest S 68	Equity \$ 3,837 4,044 4 (188 (1,075 137 72 44 \$ 6,755 Total Equity \$ 3,527 (188 (3) 2 23

Ball Corporation

Notes to the Unaudited Condensed Consolidated Financial Statements

On April 24, 2024, Ball's Board of Directors approved the repurchase by the company of up to a total of 40 million shares of its common stock. This repurchase authorization replaced all previous authorizations.

Accumulated Other Comprehensive Earnings (Loss)

The activity related to accumulated other comprehensive earnings (loss) was as follows:

(\$ in millions)	Tra	nrency nslation t of Tax)	Po Be	Pension and Other ostretirement enefits Net of Tax)	Desi I	rivatives ignated as Hedges et of Tax)	Accumulated Other Comprehensive Earnings (Loss)		
Balance at December 31, 2023	\$	(380)	\$	(537) <i>(a)</i>	\$	1	\$	(916)	
Other comprehensive earnings (loss) before reclassifications		(91)		(2)		7		(86)	
Amounts reclassified into earnings		_		7		(11)		(4)	
Aerospace disposal		_		94		_		94	
Balance at September 30, 2024	\$	(471)	\$	(438)	\$	(3)	\$	(912)	

⁽a) Includes amounts associated with the Salaried Employees of Ball Aerospace & Technologies Corp. Pension Plan.

The following table provides additional details of the amounts reclassified into net earnings from accumulated other comprehensive earnings (loss):

(0: 111:)	Tì	ree Months En	ded September 30,			Nine Months End	led September 30,	
(\$ in millions)		2024		2023	_	2024	_	2023
Gains (losses) on cash flow hedges:								
Commodity contracts recorded in net sales	\$	1	\$	29	\$	(5)	\$	40
Commodity contracts recorded in cost of sales		7		(31)		(3)		(47)
Currency exchange contracts recorded in selling, general and administrative		(36)		31		12		34
Interest rate contracts recorded in interest expense		3		3		9		5
Total before tax effect		(25)		32	_	13		32
Tax benefit (expense) on amounts reclassified into earnings		7		(9)		(2)		(9)
Recognized gain (loss), net of tax	\$	(18)	\$	23	\$	11	\$	23
Amortization and disposal of pension and other postretirement								
benefits: (a)								
Actuarial gains (losses) (b)	\$	(3)	\$	_	\$	(9)	\$	3
Prior service income (expense) (b)		_		_		(1)		(1)
Aerospace disposal		_		_		(127)		_
Total before tax effect		(3)				(137)		2
Tax benefit (expense) on amounts reclassified into earnings		1		_		36		_
Recognized gain (loss), net of tax	\$	(2)	\$		\$	(101)	\$	2

⁽a) Includes amounts associated with the Salaried Employees of Ball Aerospace & Technologies Corp. Pension Plan

⁽b) These components are included in the computation of net periodic benefit cost detailed in Note 17.

Ball Corporation

Notes to the Unaudited Condensed Consolidated Financial Statements

19. Earnings and Dividends Per Share

	Th	ree Months End	led Sep	tember 30,	N	Nine Months Ended September 30,				
(\$ in millions, except per share amounts; shares in thousands)		2024		2023		2024		2023		
Earnings from continuing operations attributable to Ball										
Corporation, net of tax	\$	191	\$	132	\$	427	\$	382		
Discontinued operations, net of tax	Ψ	6	Ψ	71	Ψ	3,613	Ψ	171		
Net earnings attributable to Ball Corporation	\$	197	\$	203	\$	4,040	\$	553		
Basic weighted average common shares		302,406		314,983		308,851		314,596		
Effect of dilutive securities		2,813		2,313		2,823		2,342		
Weighted average shares applicable to diluted earnings per share		305,219		317,296		311,674		316,938		
Basic - continuing operations	\$	0.63	\$	0.42	\$	1.38	\$	1.22		
Basic - discontinued operations		0.02		0.22		11.70		0.54		
Per basic share	\$	0.65	\$	0.64	\$	13.08	\$	1.76		
Diluted - continuing operations	\$	0.63	\$	0.42	\$	1.37	\$	1.20		
Diluted - discontinued operations		0.02		0.22		11.59		0.54		
Per diluted share	\$	0.65	\$	0.64	\$	12.96	\$	1.74		

Certain outstanding options were excluded from the diluted earnings per share calculation because they were anti-dilutive. The excluded options totaled approximately 5 million and 4 million for the three months ended September 30, 2024 and 2023, respectively, and 5 million and 4 million for the nine months ended September 30, 2024 and 2023, respectively.

The company declared and paid dividends of \$0.20 per share for the three months ended September 30, 2024 and 2023, and \$0.60 per share for the nine months ended September 30, 2024 and 2023.

20. Financial Instruments and Risk Management

Policies and Procedures

The company employs established risk management policies and procedures, which seek to reduce the company's commercial risk exposure to fluctuations in commodity prices, interest rates, currency exchange rates, net investments in foreign operations and prices of the company's common stock with regard to common share repurchases and the company's deferred compensation stock plan. However, there can be no assurance these policies and procedures will be successful. Although the instruments utilized involve varying degrees of credit, market and interest risk, the counterparties to the agreements are expected to perform fully under the terms of the agreements. The company monitors counterparty credit risk, including lenders, on a regular basis, but Ball cannot be certain that all risks will be discerned or that its risk management policies and procedures will always be effective. Additionally, in the event of default under the company's master derivative agreements, the non-defaulting party has the option to offset any amounts owed with regard to open derivative positions.

Commodity Price Risk - The company manages commodity price risk in connection with market price fluctuations of aluminum through two different methods. First, the company enters into container sales contracts that include aluminum-based pricing terms which generally reflect the same price fluctuations under commercial purchase contracts for aluminum sheet. The terms include fixed, floating or pass through aluminum component pricing. Second, the company uses certain derivative instruments, including option and forward contracts, as economic and cash flow hedges of commodity price risk where there are material differences between sales and purchase contracted pricing and volume.

Interest Rate Risk - The company's objective in managing exposure to interest rate changes is to minimize the impact of interest rate changes on earnings and cash flows and to lower its overall borrowing costs. To achieve these objectives,

Ball Corporation

Notes to the Unaudited Condensed Consolidated Financial Statements

the company may use a variety of interest rate swaps, collars and options to manage its mix of floating and fixed-rate debt.

Currency Exchange Rate Risk - The company's objective in managing exposure to currency fluctuations is to limit the exposure of cash flows and earnings from changes associated with currency exchange rate changes through the use of various derivative contracts. In addition, at times the company manages earnings translation volatility through the use of currency option strategies, and the change in the fair value of those options is recorded in the company's net earnings.

Net Investments in Foreign Operations Risk – The company is exposed to changes in foreign currencies impacting its net investments held in foreign subsidiaries. The company's objective in managing exposure to net investments in foreign operations is to limit the foreign exchange translation risk associated with its net investments in non-U.S. Dollar foreign entities. The company uses fixed-for-fixed cross currency swaps to achieve this objective.

The following table provides additional information related to the commercial risk management derivative instruments described above:

(\$ in millions)	September 30, 2024											
Commercial risk area		Commodity		Currency	Interest Rate							
Notional amount of contracts	\$	1,088	\$	2,613	\$	600						
Net gain (loss) included in AOCI, after-tax		(2)		(1)		_						
Net gain (loss) included in AOCI, after-tax, expected to												
be recognized in net earnings within the next 12 months		(2)		8		2						
Longest duration of forecasted cash flow hedge												
transactions in years		1		1		3						

In July 2024, we entered into and designated two net investment hedges against the net assets of our euro denominated operations. We utilized cross-currency interest rate swaps for which the notional amounts of \in 250 million and \in 600 million mature in the first quarter of 2027 and the second quarter of 2029, respectively. As we have designated each of these cross currency swaps as net investment hedges, we record changes in fair value due to fluctuations in the spot rate to accumulated other comprehensive earnings (AOCI). Gains and losses remain in AOCI until a sale or upon complete or substantially complete liquidation of the respective underlying net investment in the foreign entity. As of September 30, 2024, a \$15 million net loss remained in AOCI, after tax. The changes in fair value of the swaps attributable to changes other than those due to fluctuations in the spot rate are excluded from the assessment of hedge effectiveness and are recorded as a reduction to interest expense over the life of the hedge in the unaudited condensed consolidated statements of earnings. The reduction in interest expense from these excluded components for the three and nine months ended September 30, 2024, was \$2 million.

Common Stock Price Risk

The company's deferred compensation stock program is subject to variable plan accounting and, accordingly, is marked to fair value using the company's closing stock price at the end of the related reporting period. The company entered into total return swaps to reduce the company's earnings exposure to these fair value fluctuations that will be outstanding through March 2025, and which have a combined notional value of 1.4 million shares. Based on the current number of shares in the program, each \$1 change in the company's stock price would have an insignificant impact on pretax earnings, net of the impact of related derivatives.

Ball Corporation

Notes to the Unaudited Condensed Consolidated Financial Statements

Fair Value Measurements

Ball has classified all applicable financial derivative assets and liabilities as Level 2 within the fair value hierarchy as of September 30, 2024, and December 31, 2023, and presented those values in the tables below. The company's assessment of the significance of a particular input to the fair value measurement requires judgment and may affect the valuation of fair value assets and liabilities and their placement within the fair value hierarchy levels.

		September 30, 2024									
(\$ in millions)	Balance Sheet Location		Derivatives Designated as Hedging Instruments		Derivatives not Designated as Hedging Instruments		Total				
Assets:											
Commodity contracts		\$	24	\$	_	\$	24				
Currency contracts			_		5		5				
Interest rate and other contracts			3		5		8				
Total current derivative contracts	Other current assets	\$	27	\$	10	\$	37				
Commadity contracts		\$	1	\$		\$	1				
Commodity contracts		D	1			3	1				
Total noncurrent derivative contracts	Other noncurrent assets	\$	1	\$		3	1				
Liabilities:											
Commodity contracts		\$	24	\$	_	\$	24				
Currency contracts			_		14		14				
Interest rate and other contracts			1		_		1				
Total current derivative contracts	Other current liabilities	\$	25	\$	14	\$	39				
Commodity contracts		\$	1	\$	_	\$	1				
Currency contracts			4		_		4				
Interest rate and other contracts			3		_		3				
Net investment hedge			20				20				
Total noncurrent derivative contracts	Other noncurrent liabilities	\$	28	\$		\$	28				

Ball Corporation

Notes to the Unaudited Condensed Consolidated Financial Statements

(\$ in millions) Balance Sheet Location			Derivatives Designated as Hedging Instruments		Derivatives not Designated as Hedging Instruments	 Total
Assets:						
Commodity contracts		\$	20	\$	_	\$ 20
Currency contracts			65		13	78
Interest rate and other contracts			9		2	11
Total current derivative contracts	Other current assets	\$	94	\$	15	\$ 109
Currency contracts		\$	1	\$	_	\$ 1
Total noncurrent derivative contracts	Other noncurrent assets	\$	1	\$	_	\$ 1
Liabilities:						
Commodity contracts		\$	19	\$	_	\$ 19
Currency contracts			_		30	30
Interest rate and other contracts			3		_	3
Total current derivative contracts	Other current liabilities	\$	22	\$	30	\$ 52
		_		_		
Currency contracts		\$	1	\$	_	\$ 1
Total noncurrent derivative contracts	Other noncurrent liabilities	\$	1	\$	_	\$ 1

The company uses closing spot and forward market prices as published by the London Metal Exchange, the Chicago Mercantile Exchange, Reuters and Bloomberg to determine the fair value of any outstanding aluminum, currency, energy, cross currency swaps and interest rate spot and forward contracts. Option contracts are valued using a Black-Scholes model with observable market inputs for aluminum, currency and interest rates. The company values each of its financial instruments either internally using a single valuation technique, from a reliable observable market source or from third-party software. The present value discounting factor is based on the comparable time period Secured Overnight Financing Rate (SOFR), London Inter-Bank Offered Rate (LIBOR) or 12-month LIBOR. Ball performs validations of the company's internally derived fair values reported for the company's financial instruments on a quarterly basis utilizing counterparty valuation statements. The company additionally evaluates counterparty creditworthiness and, as of September 30, 2024, has not identified any circumstances requiring the reported values of the company's financial instruments be adjusted.

Ball Corporation

Notes to the Unaudited Condensed Consolidated Financial Statements

The following tables provide the effects of derivative instruments in the unaudited condensed consolidated statements of earnings and on accumulated other comprehensive earnings (loss):

		Three Months Ended September 30,								
(\$ in millions)	Location of Gain (Loss) Recognized in Earnings on Derivatives		1 Flow dge - ssified nt from nulated ther ehensive gs (Loss)	Gain (Lo Derivati Designa Hed Instrur	ves not ted as ge	He Recl Amo Accu C Comp	th Flow edge - assified unt from mulated other rehensive ngs (Loss)	Gair Deri Des	n (Loss) on vatives not ignated as Hedge truments	
Commodity contracts - manage exposure to customer pricing	Net sales	\$	1	\$	_	\$	29	\$	_	
Commodity contracts - manage exposure to supplier pricing	Cost of sales		7		9		(31)		13	
Interest rate contracts - manage exposure for outstanding debt	Interest expense		3		_		3		(2)	
Currency contracts - manage currency exposure	Selling, general and administrative		(36)		(29)		31		51	
Equity contracts	Selling, general and administrative				9				(20)	
Total		\$	(25)	\$	(11)	\$	32	\$	42	
			201		nths End	ed Septei		••		
(\$ in millions)	Location of Gain (Loss) Recognized in Earnings on Derivatives	Cash Hed Reclas Amoun Accum Otth Compre	ge - sified t from ulated ner hensive		ss) on res not ted as	Cas He Recl Amou Accu O Comp	mber 30, 20: h Flow edge - assified unt from mulated other rehensive egs (Loss)	Gain Deri Desi	n (Loss) on vatives not ignated as Hedge truments	
(§ in millions) Commodity contracts - manage exposure to customer pricing	Recognized in Earnings	Hed Reclas Amoun Accum Oth Compre	Flow ge - sified t from ulated eer hensive s (Loss)	Gain (Lo Derivativ Designa Hed	ss) on res not ted as	Cas He Recl Amou Accu O Comp	h Flow edge - assified unt from mulated other rehensive	Gain Deri Desi	vatives not ignated as Hedge	
Commodity contracts - manage exposure to customer pricing Commodity contracts - manage exposure to supplier pricing	Recognized in Earnings on Derivatives	Hed Reclas Amoun Accum Oth Compre Earning	Flow ge - sified t from ulated eer hensive s (Loss)	Gain (Le Derivativ Designa Hed Instrun	ss) on res not ted as	Cas He Recl Amou Accu C Comp Earnin	20. h Flow edge - assified unt from mulated other rehensive ugs (Loss)	Gain Deri Desi Ins	vatives not ignated as Hedge	
Commodity contracts - manage exposure to customer pricing Commodity contracts - manage exposure to supplier pricing Interest rate contracts - manage exposure for outstanding debt	Recognized in Earnings on Derivatives Net sales Cost of sales Interest expense	Hed Reclas Amoun Accum Oth Compre Earning	Flow ge - sified t from ulated her hensive s (Loss)	Gain (Le Derivativ Designa Hed Instrun	ess) on ees not ied as ge eents	Cas He Recl Amou Accu C Comp Earnin	20. h Flow edge - assified unt from mulated other rehensive ags (Loss)	Gain Deri Desi Ins	vatives not ignated as Hedge truments	
Commodity contracts - manage exposure to customer pricing Commodity contracts - manage exposure to supplier pricing Interest rate contracts - manage exposure for outstanding debt Currency contracts - manage currency exposure	Recognized in Earnings on Derivatives Net sales Cost of sales Interest expense Selling, general and administrative	Hed Reclas Amoun Accum Oth Compre Earning	Flow ge - siffied t from ulated ter hensive s (Loss) (5)	Gain (Le Derivativ Designa Hed Instrun	ess) on ees not ied as ge eents	Cas He Recl Amou Accu C Comp Earnin	200 h Flow edge - assified unt from mulated other rehensive ngs (Loss)	Gain Deri Desi Ins	vatives not ignated as Hedge truments —	
Commodity contracts - manage exposure to customer pricing Commodity contracts - manage exposure to supplier pricing Interest rate contracts - manage exposure for outstanding debt Currency contracts - manage currency	Recognized in Earnings on Derivatives Net sales Cost of sales Interest expense Selling, general and	Hed Reclas Amoun Accum Oth Compre Earning	Flow ge - siffed t from ulated ter hensive s (Loss) (5) (3)	Gain (Le Derivativ Designa Hed Instrun	ess) on es not ried as ge enents	Cas He Recl Amou Accu C Comp Earnin	200 h Flow edge - assified unt from mulated other rehensive ngs (Loss)	Gain Deri Desi Ins	vatives not gnated as Hedge truments 12 (7)	

Ball Corporation

Notes to the Unaudited Condensed Consolidated Financial Statements

The changes in accumulated other comprehensive earnings (loss) for derivatives designated as hedges were as follows:

	Th	ree Months En	ded Se	ptember 30,		Nine Months Ended September 30,			
(\$ in millions)	2024			2023	2024			2023	
Amounts reclassified into earnings:									
Commodity contracts	\$	(8)	\$	2	\$	8	\$	7	
Interest rate contracts		(3)		(3)		(9)		(5)	
Currency exchange contracts		36		(31)		(12)		(34)	
Change in fair value of hedges:									
Commodity contracts		(16)		(35)		(6)		(11)	
Interest rate contracts		(14)		6		2		28	
Currency exchange contracts		(34)		31		11		29	
Net investment contracts		(20)		_		(20)		_	
Currency and tax impacts		16		11		7		(1)	
	\$	(43)	\$	(19)	\$	(19)	\$	13	

21. Contingencies

Ball is subject to numerous lawsuits, claims or proceedings arising out of the ordinary course of business, including actions related to product liability; personal injury; the use and performance of company products; warranty matters; patent, trademark or other intellectual property infringement; contractual liability; the conduct of the company's business; tax reporting in domestic and non-U.S. jurisdictions; workplace safety and environmental and other matters. The company has also been identified as a potentially responsible party (PRP) at several waste disposal sites under U.S. federal and related state environmental statutes and regulations and may have joint and several liability for any investigation and remediation costs incurred with respect to such sites. In addition, the company has received claims alleging that employees in certain plants have suffered damages due to exposure to alleged workplace hazards. Some of these lawsuits, claims and proceedings involve substantial amounts, including as described below, and some of the environmental proceedings involve potential monetary costs or sanctions that may be material. Ball has denied liability with respect to many of these lawsuits, claims and proceedings and is vigorously defending such lawsuits, claims and proceedings. The company carries various forms of commercial, property and casualty, and other forms of insurance; however, such insurance may not be applicable or adequate to cover the costs associated with a judgment against Ball with respect to these lawsuits, claims and proceedings. The company estimates that potential liabilities for all currently known and estimable environmental matters are approximately \$26 million in the aggregate, and such amounts have been included in other current liabilities and other noncurrent liabilities at September 30, 2024. Based on the information available at the present time, any reasonably possible loss that may be incurred in excess of the recorded accruals cannot be estimated.

On February 1, 2012, Ball Metal Beverage Container Corp. ("BMBCC") filed suit against Crown Technology Holding, Inc. ("Crown") in the United States District Court for the Southern District of Ohio seeking a declaratory judgment that the CDL beverage can end made and sold by BMBCC did not infringe certain U.S. patents held by Crown. In response, Crown filed a counterclaim alleging that the CDL ends made and sold by BMBCC infringed the subject patents and seeking damages. On September 25, 2019, the District Court granted BMBCC's motion for summary judgment holding that the patents at issue were invalid due to indefiniteness. On October 20, 2019, Crown appealed this decision to the Court of Appeals for the Federal Circuit ("CAFC"). On December 31, 2020, the CAFC in a non-precedential decision, vacated the decision of the District Court finding that the District Court had not considered an additional factor under a novel position advanced by the CAFC, and remanded the case to the District Court for further proceedings. On August 2, 2023, the District Court again granted summary judgment to Ball finding that patent claims at issue are invalid due to invalidity under the revised analytical framework specified by the CAFC. On August 4, 2023, Crown appealed this decision to the CAFC. Briefing for this appeal concluded on February 20, 2024. Oral argument is expected to be scheduled during 2024 with a decision to follow. Based on the information available at the present time, the Company is unable to predict the ultimate outcome of this claim including the amount of any reasonably possible loss and we intend to vigorously defend this matter.

Ball Corporation

Notes to the Unaudited Condensed Consolidated Financial Statements

The company's operations in Brazil are involved in various governmental assessments, which have historically mainly related to claims for taxes on the internal transfer of inventory, gross revenue taxes, and indirect tax incentives and deductibility of goodwill. In addition, one of the company's Brazilian subsidiaries received an income tax assessment focused on the disallowance of deductions associated with the acquisition price paid to a third party for a portion of its operations. Based on the information available at the present time, the Company is unable to predict the ultimate outcome of these claims including the amount of reasonably possible loss and intends to vigorously defend these matters.

22. Indemnifications and Guarantees

General Guarantees

The company or its appropriate consolidated direct or indirect subsidiaries have made certain indemnities, commitments and guarantees under which the specified entity may be required to make payments in relation to certain transactions. These indemnities, commitments and guarantees are in contracts to which the company or its subsidiaries are a party, including agreements with customers of the subsidiaries in connection with the sales of their packaging products and services; guarantees to suppliers of subsidiaries of the company guaranteeing the performance of the respective entity under a purchase agreement, construction contract, renewable energy purchase contract or other commitment; guarantees in respect of certain non-U.S. subsidiaries' pension plans; indemnities for liabilities associated with the infringement of third-party patents, trademarks or copyrights under various types of agreements; indemnities to various lessors in connection with facility, equipment, furniture and other personal property leases for certain claims arising from such leases; indemnities pursuant to agreements relating to certain joint ventures; indemnities in connection with the sale of businesses or substantially all of the assets and specified liabilities of businesses; and indemnities to directors, officers and employees of the company to the extent permitted under the laws of the State of Indiana and the United States of America. The duration of these indemnities, commitments and guarantees varies and, in certain cases, is indefinite.

In addition, many of these indemnities, commitments and guarantees do not provide for any limitation on the maximum potential future payments the company could be obligated to make. As such, the company is unable to reasonably estimate its potential exposure under these items.

The company has not recorded any material liabilities for these indemnities, commitments and guarantees in the accompanying unaudited condensed consolidated balance sheets. The company does, however, accrue for payments under promissory notes and other evidences of incurred indebtedness and for losses for any known contingent liability, including those that may arise from indemnifications, commitments and guarantees, when future payment is both reasonably estimable and probable. Finally, the company carries specific and general liability insurance policies and has obtained indemnities, commitments and guarantees from third-party purchasers, sellers and other contracting parties, which the company believes would, in certain circumstances, provide recourse to certain claims arising from these indemnifications, commitments and guarantees.

Debt Guarantees

The company's and its subsidiaries' obligations under the senior notes and senior credit facilities (or, in the case of U.S. domiciled non-U.S. subsidiaries under the senior credit facilities, the obligations of non-U.S. credit parties only) are guaranteed on a full, unconditional and joint and several basis by certain of the company's domestic subsidiaries and the domestic subsidiary borrowers, and obligations of other guarantors and the subsidiary borrowers under the senior credit facilities are guaranteed by the company, in each case with certain exceptions. These guarantees are required in support of the senior notes and senior credit facilities referred to above, are coterminous with the terms of the respective note indentures, senior notes and credit agreement, and they could be enforced by the holders of the obligations thereunder during the continuation of an event of default under the note indentures, the senior notes and/or the credit agreement. The maximum potential amounts which could be required to be paid under such guarantees are essentially equal to then-outstanding obligations under the respective senior notes or the credit agreement (or, in the case of U.S. domiciled non-U.S. subsidiaries under the senior credit facilities, the obligations of non-U.S. credit parties only), with certain exceptions. All obligations under the guarantees of the senior credit facilities are secured, with certain exceptions, by a valid first priority perfected lien or pledge on (i) 100 percent of the capital stock of each of the company's material wholly owned domestic subsidiaries directly owned by the company or any of its wholly owned domestic subsidiaries. In addition, the obligations of certain

Ball Corporation

Notes to the Unaudited Condensed Consolidated Financial Statements

non-U.S. borrowers and non-U.S. pledgors under the loan documents will be secured, with certain exceptions, by a valid first priority perfected lien or pledge on 100 percent of the capital stock of certain of the company's material wholly owned non-U.S. subsidiaries and material wholly owned U.S. domiciled non-U.S. subsidiaries directly owned by the company or any of its wholly owned material subsidiaries. The company is not in default under the above-referenced senior notes or senior credit facilities.

Item 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

Management's discussion and analysis should be read in conjunction with the unaudited condensed consolidated financial statements (consolidated financial statements) and accompanying notes included in Item-1 of this Quarterly Report on Form 10-Q, which include additional information about our accounting policies, practices and the transactions underlying our financial results. The preparation of our consolidated financial statements in conformity with accounting principles generally accepted in the United States of America (U.S. GAAP) requires us to make estimates and assumptions that affect the reported amounts in our consolidated financial statements and the accompanying notes, including various claims and contingencies related to lawsuits, taxes, environmental and other matters arising during the normal course of business. We apply our best judgment, our knowledge of existing facts and circumstances and actions that we may undertake in the future in determining the estimates that affect our consolidated financial statements. We evaluate our estimates on an ongoing basis using our historical experience, as well as other factors we believe appropriate under the circumstances, such as current economic conditions, and adjust or revise our estimates as circumstances change. As future events and their effects cannot be determined with precision, actual results may differ from these estimates. Ball Corporation and its subsidiaries are referred to collectively as "Ball Corporation," "Ball," "the company," "we" or "our" in the following discussion and analysis.

OVERVIEW

Business Overview and Industry Trends

Ball Corporation is one of the world's leading aluminum packaging suppliers. With a growth mindset and by pursuing operational excellence, we lean on our competitive strengths to reach our financial goals. We are focused on maintaining our strong financial position by listening to and partnering with our global customers, delivering operational efficiencies and an innovative product portfolio from our best-in-class manufacturing facilities and returning value to shareholders via share repurchases and dividends. In the aluminum packaging industry, sales and earnings can be increased by reducing costs, increasing prices, developing new products, expanding volumes and making strategic acquisitions.

We sell our aluminum packaging products mainly to large, multinational beverage, personal care and household products companies with which we have developed long-term relationships. This is evidenced by our high customer retention and our large number of long-term supply contracts. While we have a diversified customer base, we sell a significant portion of our packaging products to major companies and brands, as well as to numerous regional customers. The overall global aluminum beverage and aerosol container industries are growing and are expected to continue to grow in the medium to long term.

We purchase our raw materials from relatively few suppliers. We also have exposure to inflation, in particular the rising costs of raw materials, as well as other direct cost inputs. We mitigate our exposure to the changes in the costs of aluminum through the inclusion of provisions in contracts covering the majority of our volumes to pass through aluminum price changes, as well as through the use of derivative instruments. The pass through provisions generally result in proportional increases or decreases in sales and costs with a greatly reduced impact, if any, on net earnings; however, there may be timing differences of when the costs are passed through. Because of our customer and supplier concentration, our business, financial condition and results of operations could be adversely affected by the loss, insolvency or bankruptcy of a major customer or supplier or a change in a supply agreement with a major customer or supplier, although our contract provisions generally mitigate the risk of customer loss, and our long-term relationships represent a known, stable customer base.

From time to time, we have evaluated and expect to continue to evaluate possible transactions that we believe will benefit the company and our shareholders, which may include strategic acquisitions, divestitures of parts of our company or equity investments. At any time, we may be engaged in discussions or negotiations at various stages of development with respect to one or more possible transactions or may have entered into non-binding letters of intent. As part of any such initiatives, we may participate in processes being run by other companies or leading our own activities.

RESULTS OF CONSOLIDATED OPERATIONS

Management's discussion and analysis for our results of operations on a consolidated and segment basis include a quantification of factors that had a material impact. Other factors that did not have a material impact, but that are significant to understand the results, are qualitatively described.

On February 16, 2024, the company completed the divestiture of its aerospace business. Effective as of the first quarter of 2024, the company reports its financial performance in three reportable segments: (1) beverage packaging, North and Central America; (2) beverage packaging, Europe, Middle East and Africa (beverage packaging, EMEA) and (3) beverage packaging, South America. See Note 1 for further information on the basis of presentation.

Global Economic Environment

Recent data has indicated continued high inflation in the regions where we operate. Current and future inflationary effects may continue to be impacted by, among other things, supply chain disruptions, governmental stimulus or fiscal and monetary policies, changes in interest rates, and changing demand for certain goods and services. We cannot predict with any certainty the impact that interest rates, a global or any regional recession, or higher inflation may have on our customers or suppliers. Additionally, we are unable to predict the potential effects that any future pandemic, or the continuation or escalation of global conflicts, including the conflict between Russia and Ukraine and the rising instability in the Middle East and Myanmar, and related sanctions or market disruptions, may have on our business. It remains uncertain how long any of these conditions may last or how severe any of them may become.

Consolidated Sales and Earnings

	Th	ree Months Er	ded S	eptember 30,	N	Nine Months Ended September 30,			
(\$ in millions)		2024		2023		2024		2023	
Net sales	\$	3,082	\$	3,111	\$	8,915	\$	9,159	
Net earnings attributable to Ball Corporation		197		203		4,040		553	
Net earnings attributable to Ball Corporation as a % of net									
sales	6 %		Ď	7 9	6	45 %	ó	6 %	

Sales in the three months ended September 30, 2024, decreased \$29 million compared to the same period in 2023 primarily due to a decrease of \$17 million from price/mix. Sales in the nine months ended September 30, 2024, decreased \$244 million compared to the same period in 2023 primarily due to decreases of \$232 million resulting mainly from lower aluminum prices and \$70 million from lost volumes as a result of the 2023 fire at the company's Verona, Virginia extruded aluminum slug manufacturing facility, partially offset by an increase of \$33 million from higher volumes.

Net earnings attributable to Ball Corporation for the three months ended September 30, 2024, decreased \$6 million compared to the same period in 2023 primarily due to a decrease of \$65 million from lower discontinued operations, net of tax and an increase in costs of \$56 million from business consolidation and other activities, partially offset by increases of \$55 million from lower interest expense and \$49 million from the results of the reportable segments discussed below. Net earnings attributable to Ball Corporation for the nine months ended September 30, 2024, increased \$3.49 billion compared to the same period in 2023 primarily due to increases of \$3.44 billion from discontinued operations, net of tax, \$132 million from the results of the reportable segments discussed below, \$123 million from lower interest expense and \$36 million from higher interest income in corporate undistributed expenses, net, partially offset by increases in costs of \$128 million from business consolidation and other activities and \$82 million from incremental compensation cost from the successful sale of the aerospace business.

When analyzing net earnings attributable to Ball Corporation as a percentage of net sales, it is important to note that net earnings attributable to Ball Corporation includes discontinued operations, net of tax resulting from the net sales attributable to the historical aerospace reportable segment through the date of the divestiture on February 16, 2024, that are now reported as discontinued operations. However, net sales attributable to the historical aerospace reportable segment are not included in the net sales figures in the table above.

Cost of Sales (Excluding Depreciation and Amortization)

Cost of sales, excluding depreciation and amortization, was \$2,425 million and \$2,512 million for the three months ended September 30, 2024 and 2023, respectively, and \$7,065 million and \$7,450 million for the nine months ended September 30, 2024 and 2023, respectively. These amounts represented 79 percent and 81 percent of consolidated net sales for the three months ended September 30, 2024 and 2023, respectively, and 79 percent and 81 percent of consolidated net sales for the nine months ended September 30, 2024 and 2023, respectively. The decrease for the three months ended September 30, 2024, was primarily due to lower manufacturing costs, including lower aluminum costs of \$41 million, and other items discussed in the reportable segment sections below. The decrease for the nine months ended September 30, 2024, was primarily due to lower manufacturing costs, including lower aluminum costs of \$269 million and lower freight costs of \$53 million, and other items discussed in the reportable segment sections below.

Depreciation and Amortization

Depreciation and amortization expense was \$150 million and \$152 million for the three months ended September 30, 2024 and 2023, respectively, and \$460 million and \$449 million for the nine months ended September 30, 2024 and 2023, respectively. These amounts represented 5 percent of consolidated net sales for the three and nine months ended September 30, 2024 and 2023. The increase for the nine months ended September 30, 2024, was primarily due to the company's larger depreciable asset base.

Selling, General and Administrative

Selling, general and administrative was \$142 million and \$133 million for the three months ended September 30, 2024 and 2023, respectively, and \$518 million and \$409 million for the nine months ended September 30, 2024 and 2023, respectively. These amounts represented 5 percent and 4 percent of consolidated net sales for the three months ended September 30, 2024 and 2023, respectively, and 6 percent and 4 percent of consolidated net sales for the nine months ended September 30, 2024 and 2023, respectively. The increase for the nine months ended September 30, 2024, was primarily due to increased compensation costs of \$98 million, including incremental compensation cost from the successful sale of the aerospace business consisting of cash bonuses and stock based compensation.

Business Consolidation and Other Activities

Business consolidation and other activities resulted in charges of \$85 million and \$29 million for the three months ended September 30, 2024 and 2023, respectively, and charges of \$171 million and \$43 million for the nine months ended September 30, 2024 and 2023, respectively. The amounts in 2024 and 2023 primarily included facility shutdown costs. Further details regarding business consolidation and other activities are provided in Note 6.

Interest Income

Interest income was \$14 million and \$12 million for the three months ended September 30, 2024 and 2023, respectively, and \$58 million and \$23 million for the nine months ended September 30, 2024 and 2023, respectively. The increases in interest income for the three and nine months ended September 30, 2024, were primarily due to the higher amount of cash on hand in 2024 from the aerospace sale.

Interest Expense

Interest expense was \$67 million and \$122 million for the three months ended September 30, 2024 and 2023, respectively, and \$228 million and \$351 million for the nine months ended September 30, 2024 and 2023, respectively. Interest expense as a percentage of average borrowings decreased by approximately 50 basis points from 5.1 percent for the three months ended September 30, 2023, to 4.6 percent for the three months ended September 30, 2024, and remained constant at 4.9 percent for the nine months ended September 30, 2024 and 2023. The interest expense decrease for the three months ended September 30, 2024, was primarily driven by a decrease of \$29 million from a smaller amount of weighted average principal outstanding during the quarter, resulting mainly from the use of proceeds from the aerospace disposal, and a decrease of \$26 million from lower weighted average interest rates on outstanding debt during the quarter. The interest expense decrease for the nine months ended September 30, 2024, was driven by a smaller amount of weighted average principal outstanding during the year, resulting mainly from the use of proceeds from the

aerospace disposal. The company expects to carry a smaller amount of weighted average principal throughout 2024 compared to 2023, with nearly all amounts outstanding having fixed rates. See Note 15 for further details.

Income Taxes

The effective tax rate for the three and nine months ended September 30, 2024, was 18.5 and 22.3 percent, respectively, compared to 25.7 percent and 22.3 percent for the same periods in 2023. The decrease of 7.2 percentage points for the three months ended September 30, 2024, was primarily due to increased tax benefits from state and local tax credits. Similar impacts may occur in future periods, but given their inherent uncertainty, the company is unable to reasonably estimate their potential future impacts.

RESULTS OF BUSINESS SEGMENTS

Segment Results

Ball's operations are organized and reviewed by management along its product lines and geographical areas, and its operating results are presented in the three reportable segments discussed below.

Beverage Packaging, North and Central America

	Three Months Ended September 30,					Nine Months Ended September 30,			
(\$ in millions)		2024 2023			2024		2023		
Net sales	\$	1,456	\$	1,541	\$	4,328	\$	4,582	
Comparable operating earnings		203		196		605		554	
Comparable operating earnings as a % of segment net sales		14 %	6	13 9	%	14 %	6	12 %	

Ball permanently ceased production at its aluminum beverage can manufacturing facility in St. Paul, Minnesota in the first quarter of 2023, permanently ceased production at its aluminum beverage can manufacturing facility in Wallkill, New York in the third quarter of 2023, permanently discontinued plans to construct a beverage can plant in North Las Vegas in the third quarter of 2023 and permanently ceased production at its aluminum beverage can manufacturing facility in Kent, Washington in the first quarter of 2024.

Segment sales for the three and nine months ended September 30, 2024, were \$85 million lower and \$254 million lower, respectively, compared to the same periods in 2023. The decrease for the three months ended September 30, 2024, was primarily due to decreases of \$49 million from lower volumes and \$37 million from price/mix. The decrease for the nine months ended September 30, 2024, was primarily due to decreases of \$203 million from price/mix and \$51 million from lower volumes.

Comparable operating earnings for the three and nine months ended September 30, 2024, were \$7 million higher and \$51 million higher, respectively, compared to the same periods in 2023. The increase for the three months ended September 30, 2024, was primarily due to increases of \$30 million from price/mix and \$16 million from lower costs, partially offset by a decrease of \$21 million from lower volumes. The increase for the nine months ended September 30, 2024, was primarily due to increases of \$64 million from price/mix and \$50 million from lower costs, partially offset by decreases of \$32 million from income recognized in 2023 from the termination of a long term power supply contract that offset higher energy costs and \$12 million from lower volumes.

Beverage Packaging, EMEA

	T	hree Months End	ded Sep	otember 30,	Nine Months Ended September 30,					
(\$ in millions)	2024			2023		2024		2023		
Net sales	\$	950	\$	902	\$	2,640	\$	2,656		
Comparable operating earnings		128		103		326		274		
Comparable operating earnings as a % of segment										
net sales		13 %		11 %		12 %		10 %		

Segment sales for the three and nine months ended September 30, 2024, were \$48 million higher and \$16 million lower, respectively, compared to the same periods in 2023. The increase for the three months ended September 30, 2024, was primarily due to higher volumes. The decrease for the nine months ended September 30, 2024, was primarily due to a decrease of \$70 million resulting mainly from lower aluminum prices, partially offset by an increase of \$56 million from higher volumes.

Comparable operating earnings for the three and nine months ended September 30, 2024, were \$25 million higher and \$52 million higher, respectively, compared to the same periods in 2023. The increase for the three months ended September 30, 2024, was primarily due to higher volumes. The increase for the nine months ended September 30, 2024, was primarily due to increases of \$32 million from price/mix and \$19 million from higher volumes, partially offset by a decrease of \$20 million from higher costs.

Beverage Packaging, South America

	7	Three Months E	nded S	eptember 30,	Nine Months Ended September 30,				
(\$ in millions)		2024 2023		2023	2023 20			2023	
Net sales	\$	484	\$	489	\$	1,388	\$	1,344	
Comparable operating earnings		78		61		170		141	
Comparable operating earnings as a % of segment									
net sales		16 %	Ó	12 %	6	12 %)	10 %	

Ball permanently ceased production at its aluminum beverage can manufacturing facility in Pouso Alegre, Brazil in the third quarter of 2024.

Segment sales for the three and nine months ended September 30, 2024, were \$5 million lower and \$44 million higher, respectively, compared to the same periods in 2023. The decrease for the three months ended September 30, 2024, was primarily due to a decrease of \$23 million from lower volumes, partially offset by an increase of \$18 million from price/mix and higher aluminum prices. The increase for the nine months ended September 30, 2024, was primarily due to increases of \$31 million from price/mix and \$13 million from higher volumes.

Comparable operating earnings for the three and nine months ended September 30, 2024, were \$17 million higher and \$29 million higher, respectively, compared to the same periods in 2023. The increases for the three and nine months ended September 30, 2024, were primarily due to increases from price/mix.

Management Performance Measures

Management internally uses various measures to evaluate company performance such as comparable operating earnings (earnings before interest expense, taxes and business consolidation and other non-comparable items); comparable net earnings (net earnings attributable to Ball Corporation before business consolidation and other non-comparable items after tax); comparable diluted earnings per share (comparable net earnings divided by diluted weighted average shares outstanding); return on average invested capital (net operating earnings after tax over the relevant performance period divided by average invested capital over the same period); economic value added (EVA®) dollars (net operating earnings after tax less a capital charge on average invested capital employed); earnings before interest expense, taxes, depreciation and amortization (EBITDA); and diluted earnings per share. In addition, management uses operating cash flows, free cash flow (cash flows from operating activities less capital expenditures; and, it may be adjusted for additional items that affect comparability between periods) and adjusted free cash flow (free cash flow adjusted for payments made for income tax liabilities related to the aerospace disposition and other material dispositions) as measures to evaluate the company's liquidity. We believe this information is also useful to investors as it provides insight into the earnings and cash flow criteria that management uses to make strategic decisions. These financial measures may be adjusted at times for items that affect comparability between periods, including business consolidation and other non-comparable items.

Nonfinancial measures used in the packaging businesses include production efficiency and spoilage rates; quality control figures; environmental, health and safety statistics; production and sales volume data; asset utilization rates and measures of sustainability. References to sales volume data represent units shipped.

Many of the above noted financial measurements are presented on a non-U.S. GAAP basis and should be considered in connection with the consolidated financial statements included within Item 1 of this report. Non-U.S. GAAP measures should not be considered in isolation and should not be considered superior to, or a substitute for, financial measures calculated in accordance with U.S. GAAP. A presentation of earnings in accordance with U.S. GAAP is available in Item 1 of this report.

NEW ACCOUNTING PRONOUNCEMENTS

For information regarding recent accounting pronouncements, see Note 2 to the consolidated financial statements included within Item 1 of this report on Form 10-Q.

FINANCIAL CONDITION, LIQUIDITY AND CAPITAL RESOURCES

Cash Flows and Capital Expenditures

Our primary sources of liquidity are cash provided by operating activities and external borrowings. We believe that cash flows from operating activities, even in the absence of operating cash flows from the historical aerospace reportable segment, and cash provided by short-term, long-term and committed revolver borrowings, when necessary, will be sufficient to meet our ongoing operating requirements, scheduled principal and interest payments on debt, dividend payments, anticipated share repurchases and anticipated capital expenditures. We have limited near-term debt maturities and our senior credit facilities are in place until 2027. The following table summarizes our cash flows:

	Nine	Nine Months Ended September 30,					
(\$ in millions)		2024					
Cash flows provided by (used in) operating activities	\$	(385)	\$	1,127			
Cash flows provided by (used in) investing activities		5,181		(826)			
Cash flows provided by (used in) financing activities		(3,998)		490			

Cash flows from the historical aerospace reportable segment are presented within each cash flow statement category in the consolidated statements of cash flows. Depreciation and amortization, capital expenditures and significant operating and investing noncash items of the aerospace discontinued operation are presented in Note 4.

Cash flows used in operating activities were \$385 million in 2024, primarily driven by working capital outflows of \$630 million. On February 16, 2024, the company completed the divestiture of the aerospace business. We currently estimate a cash tax of \$950 million to be recorded as a cash outflow from operations in 2024, of which \$484 million has been paid as of September 30, 2024. See Note 4 for further details. In an elevated interest rate environment, payment terms with our customers and vendors become a more important element of total mix of information used to negotiate our contract terms. At September 30, 2024, days sales outstanding, net of factored receivables, was 78 days; therefore, a change of one day in days sales outstanding will impact cash flows provided by (used in) operating activities by \$34 million. At September 30, 2024, days payable outstanding was 117 days; therefore, a change of one day in days payable outstanding will impact cash flows provided by (used in) operating activities by \$27 million. At September 30, 2024, days inventory outstanding was 51 days; therefore, a change of one day in days inventory outstanding will impact cash flows provided by (used in) operating activities by \$27 million.

Cash flows provided by investing activities were \$5.18 billion in 2024, primarily driven by the initial cash proceeds received at close from the sale of the aerospace business of \$5.42 billion, which is subject to further customary closing adjustments.

Cash flows used in financing activities were \$4.00 billion in 2024, primarily driven by net repayments of long-term borrowings of \$2.83 billion and repurchases of common stock of \$1.06 billion. See Note 15 for further details on the company's borrowings, and additional amounts available.

We have entered into several regional accounts receivable factoring programs with various financial institutions for certain of our accounts receivable. The programs are accounted for as true sales of the receivables, with limited recourse to Ball, and had combined limits of approximately \$1.63 billion and \$2.00 billion at September 30, 2024 and December 31, 2023, respectively. A total of \$702 million and \$350 million were available for sale under these programs as of September 30, 2024, and December 31, 2023, respectively. The combined limit and available for sale amount as of December 31, 2023, included \$160 million and \$97 million, respectively, associated with receivable factoring programs included within the historical aerospace reportable segment. The company has recorded expense related to its factoring programs of \$12 million and \$29 million for the three months ended September 30, 2024 and 2023, respectively, and \$35 million and \$69 million for the nine months ended September 30, 2024 and 2023, respectively, and has presented these amounts in selling, general and administrative in its unaudited condensed consolidated statements of earnings.

The company has several regional supplier finance programs with various financial institutions that act as the paying agent for certain payables of the company. The amount of obligations outstanding that the company confirmed as valid to the financial institutions under the company's programs was \$432 million and \$703 million at September 30, 2024 and December 31, 2023, respectively. Our payment terms are not dependent on whether the suppliers participate in the supplier finance programs or if the suppliers decide to factor their receivables with the financial institutions; therefore, we do not believe that future changes in the availability of supplier finance programs will have a significant impact on our liquidity.

Contributions to the company's defined benefit pension plans were \$24 million in the first nine months of 2024 compared to \$13 million in the same period of 2023, and such contributions are expected to be approximately \$30 million for the full year of 2024. This estimate may change based on changes in the Pension Protection Act, actual plan asset performance and available company cash flow, among other factors.

The company expects that 2024 capital expenditures for property, plant and equipment will likely be in the range of \$650 million. Approximately \$167 million of capital expenditures for property, plant and equipment were contractually committed as of September 30, 2024, and the company intends to return approximately \$245 million to shareholders in the form of dividends for the full year 2024, inclusive of the cash dividend of 20 cents per share, payable December 16, 2024, to shareholders of record as of December 2, 2024.

Table of Contents

As of September 30, 2024, approximately \$820 million of our cash was held outside of the U.S. In the event that we would need to utilize any of the cash held outside of the U.S. for purposes within the U.S., there are no material legal or other economic restrictions regarding the repatriation of cash from any of the countries outside the U.S. where we have cash. The company believes its U.S. operating cash flows and cash on hand, as well as availability under its long-term, revolving credit facilities, uncommitted short-term credit facilities and accounts receivable factoring programs, will be sufficient to meet the cash requirements of the U.S. portion of our ongoing operations, scheduled principal and interest payments on U.S. debt, dividend payments, capital expenditures and other U.S. cash requirements. If non-U.S. funds are needed for our U.S. cash requirements and we are unable to provide the funds through intercompany financing arrangements, we may be required to repatriate funds from non-U.S. locations where the company has previously asserted indefinite reinvestment of funds outside the U.S.

Based on its indefinite reinvestment assertion, the company has not provided deferred taxes on earnings in certain non-U.S. subsidiaries because such earnings are intended to be indefinitely reinvested in its international operations. It is not practical to estimate the additional taxes that might become payable if these earnings were remitted to the U.S.

Share Repurchases

The company's share repurchases totaled \$1.06 billion during the nine months ended September 30, 2024, compared to \$3 million of repurchases during the same period of 2023. The repurchases were completed using cash on hand, cash provided by operating activities, proceeds from the sale of businesses and available borrowings. The company plans to continue capital return to shareholders via an estimated \$1.4 billion in share repurchases in 2024 using cash from the aerospace divestiture and operating activities.

On April 24, 2024, Ball's Board of Directors approved the repurchase by the company of up to a total of 40 million shares of its common stock. This repurchase authorization replaced all previous authorizations.

Debt Facilities and Refinancing

Given our cash flow projections and unused credit facilities that are available until June 2027, our liquidity is strong and is expected to meet our ongoing cash and debt service requirements. Total interest-bearing debt of \$5.82 billion and \$8.62 billion was outstanding at September 30, 2024, and December 31, 2023, respectively.

On February 14, 2024, Ball announced a public tender of the \$1.00 billion 5.25% senior notes due July 2025 and the \$750 million 4.875% senior notes due March 2026. On March 14, 2024, \$811 million of the \$1.00 billion 5.25% senior notes and \$494 million of the \$750 million 4.875% senior notes were validly tendered and accepted. Additionally, in the first quarter of 2024, Ball repaid at maturity the outstanding 0.875% euro denominated senior notes due in the amount of \$817 million and prepaid \$700 million of the Term A loan outstanding balance.

The company's senior credit facilities include a \$1.35 billion term loan and long-term, multi-currency revolving facilities that mature in June 2027, which provide the company with up to the U.S. dollar equivalent of \$1.75 billion. At September 30, 2024, approximately \$1.72 billion was available under the company's long-term, multi-currency committed revolving credit facilities. In addition to these facilities, the company had \$193 million of committed short-term loans outstanding and a \$25 million short-term finance lease outstanding. The company also had approximately \$1.04 billion of short-term uncommitted credit facilities available at September 30, 2024, of which \$42 million was outstanding and due on demand. At December 31, 2023, the company had \$196 million of committed short-term loans outstanding and \$13 million outstanding under short-term uncommitted credit facilities.

While ongoing financial and economic conditions in certain areas may raise concerns about credit risk with counterparties to derivative transactions, the company mitigates its exposure by allocating the risk among various counterparties and limiting exposure to any one party. We also monitor the credit ratings of our suppliers, customers, lenders and counterparties on a regular basis.

Table of Contents

We were in compliance with the leverage ratio requirement at September 30, 2024, and for all prior periods presented, and have met all debt payment obligations. The U.S. note agreements and bank credit agreement contain certain restrictions relating to dividend payments, share repurchases, investments, financial ratios, guarantees and the incurrence of additional indebtedness. The most restrictive of our debt covenants requires us to maintain a leverage ratio (as defined) of no greater than 5.0 times, which will change to 4.5 times as of September 30, 2025. As of September 30, 2024, the company could borrow an additional \$2.39 billion under its long-term multi-currency committed revolving facilities and short-term uncommitted credit facilities. Additional details about our debt are available in Note 15 accompanying the consolidated financial statements within Item 1 of this report. In July 2024, we entered into and designated two net investment hedges against the net assets of our euro denominated operations. See Note 20 for further details.

Argentina

See <u>Note 1</u> for information relevant to economic and other government policies that may have an impact on the financial condition, liquidity and capital resources of the company's Argentina operations.

Aluminum Cups

See Note 1 for considerations regarding the Ball Aluminum Cups business as the company assesses options for how to best position the business for future success.

Defined Benefit Pension Plans

In November 2023, the Trustee Board of the U.K. defined benefit pension plan entered into an agreement with an insurance company for a bulk annuity purchase, or "buy-in," for its U.K. defined benefit pension plan to reduce retirement plan risk, while delivering promised benefits to plan participants. The plan was frozen on April 5, 2024. See Note 17 for further details.

CONTINGENCIES, INDEMNIFICATIONS AND GUARANTEES

Details of the company's contingencies, legal proceedings, indemnifications and guarantees are available in Note 21 and Note 22 accompanying the consolidated financial statements within Lem 1 of this report. The company is routinely subject to litigation incidental to operating its businesses and has been designated by various federal, state, and international environmental agencies as a potentially responsible party, along with numerous other companies, for the clean-up of several hazardous waste sites. The company believes the matters identified will not have a material adverse effect upon its liquidity, results of operations or financial condition.

Guaranteed Securities

The company's senior notes are guaranteed on a full and unconditional, joint and several basis by the issuer of the company's senior notes and the subsidiaries that guarantee the notes (the obligor group). The entities that comprise the obligor group are 100 percent owned by the company. As described in the supplemental indentures governing the company's existing senior notes, the senior notes are guaranteed by any of the company's domestic subsidiaries that guarantee any other indebtedness of the company.

The following summarized financial information relates to the obligor group as of September 30, 2024, and December 31, 2023. Intercompany transactions, equity investments and other intercompany activity between obligor group subsidiaries have been eliminated from the summarized financial information. Investments in subsidiaries not forming part of the obligor group have also been eliminated. The results and balance sheet information of the historical aerospace reportable segment are included in the following summarized financial information of the obligor group as of and for the year ended December 31, 2023, as the guarantees of the aerospace business legal entities were in effect through that date. On February 16, 2024, the company completed the divestiture of the aerospace business. As such, the following summarized financial information of the obligor group as of and for the nine months ended September 30, 2024, does not include results and balance sheet information of the historical aerospace reportable segment.

	Nine Months Ended		Year Ended
(\$ in millions)	 September 30, 2024	_	December 31, 2023
Net sales	\$ 5,156	\$	8,962
Gross profit (a)	639		1,074
Net earnings	3,905		493
Net earnings attributable to Ball Corporation	3,905		493

⁽a) Gross profit is shown after depreciation and amortization related to cost of sales of \$142 million for the nine months ended September 30, 2024, and \$272 million for the year ended December 31, 2023.

For the nine months ended September 30, 2024, and the year ended December 31, 2023, the obligor group recorded the following transactions with other subsidiary companies: sales to them of \$946 million and \$1.13 billion, respectively, net credits from them of \$54 million and \$38 million, respectively, and net interest income from them of \$252 million and \$344 million, respectively. The obligor group received dividends from other subsidiary companies of \$38 million and \$814 million, during the nine months ended September 30, 2024, and the year ended December 31, 2023, respectively.

(\$ in millions)	Sep	tember 30, 2024	 December 31, 2023
Current assets	\$	2,562	\$ 2,339
Noncurrent assets		15,308	15,955
Current liabilities		4,245	5,163
Noncurrent liabilities		8,453	10,857

Included in the amounts disclosed in the table above, at September 30, 2024, and December 31, 2023, the obligor group held receivables due from other subsidiary companies of \$472 million and \$768 million, respectively, long-term notes receivable due from other subsidiary companies of \$10.53 billion and \$10.20 billion, respectively, payables due to other subsidiary companies of \$1.80 billion and \$1.83 billion, respectively, and long-term notes payable due to other subsidiary companies of \$2.17 billion and \$2.32 billion, respectively.

A description of the terms and conditions of the company's debt guarantees is located in Note 22 of Item 1 of this report.

Item 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

The company employs established risk management policies and procedures which seek to reduce the company's commercial risk exposure to fluctuations in commodity prices, interest rates, currency exchange rates, net investments in foreign operations and prices of the company's common stock with regard to common share repurchases and the company's deferred compensation stock plan. However, there can be no assurance that these policies and procedures will be successful. Although the instruments utilized involve varying degrees of credit, market and interest risk, the counterparties to the agreements are expected to perform fully under the terms of the agreements. The company monitors counterparty credit risk, including lenders, on a regular basis, but Ball cannot be certain that all risks will be discerned or that its risk management policies and procedures will always be effective. Additionally, in the event of default under the company's master derivative agreements, the non-defaulting party has the option to set off any amounts owed with regard to open derivative positions. Further details are available in Item 7A within Ball's 2023 Annual Report on Form 10-K filed on February 20, 2024, and in Note 20 accompanying the consolidated financial statements included within Item 1 of this report.

Item 4. CONTROLS AND PROCEDURES

Our chief executive officer and chief financial officer participated in management's evaluation of our disclosure controls and procedures, as defined by the Securities and Exchange Commission (SEC), as of the end of the period covered by this report and concluded that our controls and procedures were effective. There were no changes to internal controls during the company's third quarter of 2024 that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

FORWARD-LOOKING STATEMENTS

This report contains "forward-looking" statements concerning future events and financial performance. Words such as "expects." "anticipates," "estimates," "believes," and similar expressions typically identify forward looking statements, which are generally any statements other than statements of historical fact. Such statements are based on current expectations or views of the future and are subject to risks and uncertainties, which could cause actual results or events to differ materially from those expressed or implied. You should therefore not place undue reliance upon any forward-looking statements, and they should be read in conjunction with, and qualified in their entirety by, the cautionary statements referenced below. Ball undertakes no obligation to publicly update or revise any forward-looking statements, whether as a result of new information, future events or otherwise. Key factors, risks and uncertainties that could cause actual outcomes and results to be different are summarized in filings with the Securities and Exchange Commission, including Exhibit 99 in Ball's Form 10-K, which are available on Ball's website and at www.sec.gov. Additional factors that might affect: a) Ball's packaging segments include product capacity, supply, and demand constraints and fluctuations and changes in consumption patterns; availability/cost of raw materials, equipment, and logistics; competitive packaging, pricing and substitution; changes in climate and weather and related events such as drought, wildfires, storms, hurricanes, tornadoes and floods; footprint adjustments and other manufacturing changes, including the opening and closing of facilities and lines; failure to achieve synergies, productivity improvements or cost reductions; unfavorable mandatory deposit or packaging laws; customer and supplier consolidation; power and supply chain interruptions; changes in major customer or supplier contracts or loss of a major customer or supplier; inability to pass through increased costs; war, political instability and sanctions, including relating to the situation in Russia and Ukraine and its impact on Ball's supply chain and its ability to operate in Europe, the Middle East and Africa regions generally; changes in foreign exchange or tax rates; and tariffs, trade actions, or other governmental actions, including business restrictions and orders affecting goods produced by Ball or in its supply chain, including imported raw materials; and b) Ball as a whole include those listed above plus: the extent to which sustainability-related opportunities arise and can be capitalized upon; changes in senior management, succession, and the ability to attract and retain skilled labor; regulatory actions or issues including those related to tax, environmental, social and governance reporting, competition, environmental, health and workplace safety, including U.S. Federal Drug Administration and other actions or public concerns affecting products filled in Ball's containers, or chemicals or substances used in raw materials or in the manufacturing process; technological developments and innovations; the ability to manage cyber threats; litigation; strikes; disease; pandemic; labor cost changes; inflation; rates of return on assets of Ball's defined benefit retirement plans; pension changes; uncertainties surrounding geopolitical events and governmental policies; reduced cash flow; interest rates affecting Ball's debt; successful or unsuccessful joint ventures, acquisitions and divestitures, and their effects on Ball's operating results and business generally.

PART II. OTHER INFORMATION

Item 1. Legal Proceedings

There were no events required to be reported under Item 1 for the three months ended September 30, 2024, except as discussed in Note 21 to the consolidated financial statements included within Part I, Item 1 of this report.

Item 1A. Risk Factors

There were no changes required to be reported under Item 1A for the three months ended September 30, 2024; however, upon the sale of the aerospace business on February 16, 2024, the risk factors related to the aerospace business are no longer relevant.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

The following table summarizes the company's repurchases of its common stock during the third quarter of 2024.

Purchases of Securities						
(\$ in millions)	Total Number of Shares Purchased (a)		erage Price d per Share	Total Number of Shares Purchased as Part of Publicly Announced Plans or Programs (a)	Maximum Number of Shares that May Yet Be Purchased Under the Plans or Programs (b)	
July 1 to July 31, 2024	2,447,182	\$	61.17	2,447,182	32,779,905	
August 1 to August 31, 2024	2,210,899		63.79	2,210,899	30,569,006	
September 1 to September 30, 2024	1,557,765		66.46	1,557,765	29,011,241	
Total	6,215,846			6,215,846		

⁽a) Includes any open market purchases (on a trade-date basis), share repurchase agreements and/or shares retained by the company to settle employee withholding tax liabilities.

Item 3. Defaults Upon Senior Securities

There were no events required to be reported under Item 3 for the three months ended September 30, 2024.

Item 4. Mine Safety Disclosures

Not applicable.

Item 5. Other Information

There were no events required to be reported under Item 5 for the three months ended September 30, 2024.

⁽b) The company has an ongoing repurchase program for which 40 million shares were authorized for repurchase by Ball's Board of Directors

Table of Contents

Item 6. Exhibits

- 2.1 Stock Purchase Agreement, dated as of August 16, 2023, by and among Ball Corporation, BAE Systems, Inc., and, solely for the purposes set forth therein, BAE Systems plc
- 3(ii) Bylaws of Ball Corporation as amended, filed herewith.
 - 22 Obligor group subsidiaries of Ball Corporation
- 31.1 Certification pursuant to Rule 13a-14(a) or Rule 15d-14(a) by Daniel W. Fisher, Chairman and Chief Executive Officer of Ball Corporation.
- 31.2 Certification pursuant to Rule 13a-14(a) or Rule 15d-14(a) by Howard H. Yu, Executive Vice President and Chief Financial Officer of Ball Corporation.
- 32.1 Certification pursuant to Rule 13a-14(b) or Rule 15d-14(b) and Section 1350 of Chapter 63 of Title 18 of the United States Code by Daniel W. Fisher, Chairman and Chief Executive Officer of Ball Corporation.
- 32.2 Certification pursuant to Rule 13a-14(b) or Rule 15d-14(b) and Section 1350 of Chapter 63 of Title 18 of the United States Code by Howard H. Yu, Executive Vice President and Chief Financial Officer of Ball Corporation.
- 99 <u>Cautionary statement for purposes of the safe harbor provisions of the Private Securities Litigation Reform Act of 1995, as amended.</u>
- 101.INS XBRL Instance Document the instance document does not appear in the Interactive Data File because its XBRL tags are embedded within the Inline XBRL document.
- 101.SCH Inline XBRL Taxonomy Extension Schema Document.
- 101.CAL Inline XBRL Taxonomy Extension Calculation Linkbase Document
- 101.DEF Inline XBRL Taxonomy Extension Definitions Linkbase Document
- 101.LAB Inline XBRL Taxonomy Extension Label Linkbase Document
- 101.PRE Inline XBRL Taxonomy Extension Presentation Linkbase Document
 - The cover page of the company's quarterly report on Form 10-Q for the quarter ended September 30, 2024, formatted in Inline XBRL (contained in Exhibit 101), the: (i) Unaudited Condensed Consolidated Statement of Earnings, (ii) Unaudited Condensed Statement of Comprehensive Earnings (Loss), (iii) Unaudited Condensed Consolidated Balance Sheet, (iv) Unaudited Condensed Consolidated Statement of Cash Flows and (v) Notes to the Unaudited Condensed Consolidated Financial Statements.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Ball Corporation (Registrant)

By: /s/ Howard H. Yu

Howard H. Yu

Executive Vice President and Chief Financial Officer

Date: October 31, 2024

STOCK PURCHASE AGREEMENT

BY AND AMONG

BAE SYSTEMS, INC.,

BALL CORPORATION

AND, SOLELY FOR PURPOSES OF <u>SECTION 12.21</u> HEREOF,

BAE SYSTEMS PLC

Dated as of August 16, 2023

TABLE OF CONTENTS

ARTICLE I

DEFINITIONS

1.1	Definitions	1
	ARTICLE II	
	THE TRANSACTIONS	
2.1	Sale and Purchase of the Shares	18
2.2	Purchase Price.	
2.3	Closing Purchase Price.	
2.4	Post-Closing Adjustment	
2.5	Withholding	
	ARTICLE III	
	CLOSING AND CLOSING DELIVERIES	
3.1	Closing; Time and Place	23
3.2	Deliveries by Seller	
3.3	Deliveries by Purchaser	
3.4	Payment Mechanics.	
	ARTICLE IV	
	REPRESENTATIONS AND WARRANTIES OF SELLER	
4.1	Authority; Enforceability	24
4.2	Non-Contravention; Consents.	
4.3	Organization; Acquired Companies.	
4.4	Title; Shares.	
4.5	Financial Information; Liabilities	
4.6	Absence of Certain Changes.	
4.7	Compliance with Legal Requirements	
4.8	Material Contracts	_29
4.9	Litigation	32
4.10	Insurance	32
4.11	Intellectual Property	32
4.12	Real Property.	
4.13	Labor Matters	
4.14	Employee Benefits	36
4.15	Taxes	38
4.16	Sufficiency of Assets	39

4.17	Environmental Matters	39			
4.18	Certain Business Practices.	40			
4.19	Government Contracts	41			
4.20	Brokers	42			
4.21	Related Party Transactions	42			
4.22	Intercompany Arrangements				
4.23	Disclaimer of Seller.	43			
4.24	No Other Representations.	44			
	ARTICLE V				
	REPRESENTATIONS AND WARRANTIES OF PURC	HASER			
5.1	Authority; Enforceability	45			
5.2	Non-Contravention; Consents				
5.3	Organization	46			
5.4	Litigation	46			
5.5	Sufficiency of Funds	46			
5.6	Solvency	47			
5.7	Brokers	47			
5.8	Pending Transactions.	47			
5.9	NTIB Entity	47			
5.10	ITAR	47			
5.11	Inspection; No Other Representations.	47			
5.12	Disclaimer of Purchaser.	49			
	ARTICLE VI				
	COVENANTS OF THE PARTIES				
6.1	Conduct of the Business Prior to the Closing.	49			
6.2	Pre-Closing Access to Information.				
6.3	Cooperation				
6.4	Shared Contracts and Consents	56			
6.5	Termination of Intercompany Agreements; Release of Guarantees	58			
6.6	Seller Debt Facilities Releases	60			
6.7	Confidentiality	60			
6.8	Reasonable Best Efforts; Cooperation; Regulatory Filings	61			
6.9	Financing	65			
6.10	Financing Cooperation	65			
6.11	Insurance	68			
6.12	R&W Insurance Policy	69			
6.13	Litigation Support	70			
6.14	Registered Office Addresses				
6.15	Segregation of Email and Messaging Accounts				
6.16	DDTC 60-Day.				
6.17	Resignations	72			

	ARTICLE VII	
	ADDITIONAL COVENANTS OF THE PARTIES	
7.1	Transitional Trademark Rights	73
7.2	Closing and Post-Closing Access to Information	75
7.3	D&O Indemnification.	76
7.4	Non-Solicit	78
7.5	Further Assurances; Wrong Pockets.	78
7.6	Notifications	80
	ARTICLE VIII	
	TAX MATTERS	
8.1	Section 338(h)(10) Elections	80
8.2	Tax Returns; Allocation of Taxes	
8.3	Prohibited Actions.	83
8.4	Consolidated Returns and Purchaser Consolidated Returns; Tax Proceedings	83
8.5	Tax Matters Cooperation.	83
8.6	Transfer Taxes	84
8.7	Indemnified Taxes	
8.8	Deferred Revenue.	84
8.9	Survival	84
	ARTICLE IX	
	EMPLOYEES	
9.1	Transferred Employees.	85
9.2	Continuation Period.	85
9.3	Seller Benefit Plan Participation; M&A Qualified Beneficiaries; Certain Beneficiaries	fits for Transferred
	Employees	
9.4	Qualified Retirement Plans	
9.5	FSAs.	
9.6	Annual Cash Bonuses; Similar Benefits	
9.7	Vacation and Paid Time Off	
9.8	Communications.	
9.9	Seller Long-Term Incentive Awards	
9.10	Deferred Compensation Plans	
9.11	Employee Liabilities	
9 12	No Third-Party Beneficiaries	90

6.18 Pre-Closing Reorganization 72

ARTICLE X

CONDITIONS TO THE CLOSING

10.1	Conditions of Purchaser.	
10.2	Conditions of Seller	
10.3	Mutual Conditions.	
10.4	Waiver of Conditions	92
	ARTICLE XI	
	TERMINATION	
11.1	Termination	93
11.2	Notice of Termination.	94
11.3	Effect of Termination	94
11.4	Purchaser Termination Fee	94
	ARTICLE XII	
	MISCELLANEOUS PROVISIONS	
12.1	Expenses	96
12.2	Survival	96
12.3	Interpretation	96
12.4	Entire Agreement.	97
12.5	Amendment and Waivers	97
12.6	Successors and Assigns	98
12.7	Governing Law.	98
12.8	Jurisdiction; Venue; Service of Process.	98
12.9	Waiver of Jury Trial	99
12.10	Specific Performance	99
12.11	Severability	100
12.12	Certain Releases	
12.13	The Seller Disclosure Schedule, Schedules, Annexes and Exhibits	103
12.14	Notices	103
12.15	No Third-Party Beneficiaries.	104
12.16	Provision Regarding Legal Representation.	
12.17	No Other Duties	105
12.18	Reliance on Counsel and Other Advisors	
	Public Announcements	
12.20	Counterparts	106
12 21	Purchaser Guarantor	106

<u>Annexes</u>

<u>Annex A</u>: Certain Financial Definitions and Matters

Annex B: Real Estate Reorganization Plan

Exhibits

Exhibit A: Form of Transition Services Agreement

INDEX

Section

Accounting Principles	1.1
Acquired Companies	
Acquired Company Benefit Plan	
Affiliate Affiliate	
Agreement	
Anti-Corruption Laws	
Antitrust Laws	
Assumed Deferred Compensation	
Assumed District Compensation Assumed Disability Health Benefits	
Assumed Incentive Amount	
Axinn	
Base Price	
Benefit Plan	
BTEI	
Business	
Business Day	
Business Pension Plan	
Business Portion	
Business Systems	
Business Trademarks	
Cash	
CFIUS	
CFIUS Authorities	
CFIUS Clearance	
Clean Team Agreement	
Closing	
Closing Conditions	
Closing Date	
Closing Purchase Price	
Code	
Collective Bargaining Agreement	
Combination Mark	
Commercial Tax Agreement	
Company	
Company Employee	
Company Registered IP	
Company Subsidiary	
Consent	
Consolidated Return	
Consultation Period	
Contagion Event	
Continuation Period	
Contract	
Contract	

Controlled Affiliate	1.1
Copyrights	
COVID-19	
Credit Facilities	
Customs & Trade Laws	
Cybersecurity Incident	
Cybersecurity Measures	
D&O Indemnification Agreements	
D&O Indemnitees	7.3(a)
D&O Insurance	
Data Protection Laws	
Data Room	
DCSA	
DCSA Approval	
DDTC	
Debt Commitment Letter	
Debt Financing	
e e e e e e e e e e e e e e e e e e e	
Debt Financing Sources DFS Provisions	
Direct Employee	
Disputed Items	
DOJ	
Election Allocation	* /
Encumbrance	
Environmental Law	
ERISA	
ERISA Affiliate	
Estimated Closing Statement	
Exchange Act	
Excluded Benefits	
Excluded Emails and Messages	
Excluded Shared Contracts	
Final Closing Statement	2.4(d)
Final Overage	2.4(e)
Final Purchase Price	2.4(d)
Final Underage	2.4(f)
Financial Statements	4.5(a)
FOCI Mitigation Plan	1.1
Former Company Employees	1.1
Former Direct Employee	1.1
Former Internal Transfer Employee	
Fraud	
FTC	
GAAP	
Government Contract	
Governmental Approvals	

Governmental Authority	1.1
Hazardous Materials	
HSR Act	
Inactive Employee	
Incentive-Based Programs	
Income Tax	
Income Tax Amount	
Income Tax Return	
Indebtedness	
Indemnified Taxes	
Insurance Policies	
Intellectual Property	()
Intercompany Agreements	
Intercompany Guarantees	
Internal Transfer Employee	
IP Assignment Agreement	
IRS	
ITAR	
Key Customer	
Key Vendor	
Latest Balance Sheet	
Leased Real Property	` '
Legal Requirement	
Liabilities	
Listed Insurance Policies	
Losses	
Malicious Code	
Material Adverse Effect	
Material Contract Waiver	
Material Contracts	
Material Government Contract	· /
Maximum Amount	
Minimum Closing Cash	
Mirror Plan	
Net Working Capital	
Net Working Capital Overage	
Net Working Capital Underage	
NISPOM Rule	
Non-Business Confidential Material	
Non-Business Portion	
Non-Disclosure Agreement	
Obligations	12.21(a)
Occurrence-Based Policies	
OFAC	
Open Source Software	
Order	

Outside Counsel Only	6.8(b)
Outside Date	* /
Owned IP	()
Owned Real Property	
Pandemic Measures	
Patents	
Pending Bid Contract	
Pension Plan	* / * /
Permit	
Permitted Compensation Action	
Permitted Encumbrances	
Person	
Personal Information	
Post-Closing Statement	
Pre-Closing Tax Period	* /
Preliminary Cash	
Preliminary Closing Purchase Price	
Preliminary Indebtedness	
Preliminary Net Working Capital	* /
Preliminary Transaction Expenses	
President	
Privileged Communications	
Proceeding	
Purchase Price	
Purchaser	
Purchaser 401(k) Plan	9.4(a)
Purchaser Consolidated Return	i.í.í
Purchaser Designee	1.1
Purchaser Fundamental Representations	1.1
Purchaser Guarantor	
Purchaser Material Adverse Effect	1.1
Purchaser Releasee	12.12(a)
Purchaser Releasing Party	12.12(a)
Purchaser Termination Fee	11.4(a)
Purchaser-Filed Tax Return	8.2(a)(ii)
R&W Insurance Policy	6.12
Real Estate Reorganization Plan	6.18(c)
Real Property	1.1
Real Property Lease	4.12(b)
Released Matters	12.12(a)
Representatives	1.1
Restricted Cash	1.1
Retained Business	1.1
Retained Liabilities	1.1
Review Period	2.4(b)
Sanctioned Person	1.1

Sanctioned Territory	1.1
Sanctions Laws	
SEC	
Section 338(h)(10) Elections	
Seller	` '
Seller 401(k) Plan	
Seller Benefit Plan	· /
Seller Debt Facilities	
Seller Debt Facilities Releases	
Seller Deferred Compensation Plan	
Seller Disclosure Schedule	
Seller Fundamental Representations	
Seller Indemnitees	
Seller Mark	
Seller Releasee	
Seller Releasing Party	
Seller Transitional Trademarks	
Seller's Knowledge	
Settlement Accountant	
Shared Contracts	
Shares	
Skadden	
Software	
Specified Litigation	
Specified Losses	
Specified Sales Taxes	· ·
Standalone Go Beyond Mark	
Statement of Objections	
Straddle Period	
Subsidiary	
Target Net Working Capital	
Tax	
Tax Authority	
Tax Proceeding	
Tax Return	
TI LLC	
Topaz	
Topaz Reorganization	
Trade Secrets	
Trademarks	
Transaction Agreements	
Transaction Expenses	
Transactions Transactions	
Transfer Taxes	
Transferred Employee	
Transition Services Agreement	
Transmon screwes Agreement	3.2(0)

Transitional Trademark End Date7	.1(t	<u>)</u>
Treasury Regulations	1.	. 1
WARN Act	1.	. 1
Willful Breach	1.	. 1

STOCK PURCHASE AGREEMENT

This Stock Purchase Agreement is dated as of August 16, 2023, by and among BAE Systems, Inc., a Delaware corporation ("<u>Purchaser</u>"), Ball Corporation, an Indiana corporation ("<u>Seller</u>") and, solely for purposes of <u>Section 12.21</u> hereof, BAE Systems plc, a United Kingdom public limited company ("<u>Purchaser Guarantor</u>"). The capitalized terms used in this Agreement are defined in <u>Article I</u>, unless otherwise defined herein.

RECITALS

- **WHEREAS**, Seller owns 100% of the issued and outstanding shares of capital stock of Ball Technologies Holdings Corp., a Colorado corporation (the "<u>Company</u>");
- WHEREAS, Seller desires to sell, transfer, convey, assign and deliver to Purchaser (or one or more Purchaser Designees), and Purchaser desires to (and to cause any Purchaser Designee, as applicable, to) purchase from Seller, all of Seller's rights, title and interest in and to all of the issued and outstanding shares of capital stock of the Company (the "Shares"), subject to the terms and the conditions set forth in this Agreement;
- **WHEREAS**, Seller will, and will cause its Controlled Affiliates to, and Purchaser will, and will cause any Purchaser Designee and Purchaser's Controlled Affiliates to, at or prior to the Closing, execute and deliver each of the other Transaction Agreements to which they are a party; and
- **WHEREAS**, the parties desire to make certain representations, warranties, covenants and agreements in connection with this Agreement.
- **NOW, THEREFORE**, in consideration of the foregoing recitals and the mutual representations, warranties, covenants and promises contained herein and other good and valuable consideration, the adequacy and sufficiency of which are hereby acknowledged, and intending to be legally bound, the parties agree as follows:

ARTICLE I

DEFINITIONS

- 1.1 <u>Definitions</u>. Capitalized terms used in this Agreement shall have the meanings set forth in this Agreement. The following terms, whenever used herein, shall have the following meanings for all purposes of this Agreement.
 - "Accounting Principles" shall have the meaning set forth on Annex A, Part VI.
- "Acquired Companies" shall mean, each of and collectively, (a) the Company, (b) the Company's direct Subsidiary, Ball Aerospace & Technologies Corp., a Delaware corporation (the "Company Subsidiary") and (c) assuming the consummation of the pre-Closing reorganization contemplated by Section 6.18(a), Topaz Intelligence, LLC, a Delaware limited

liability company ("<u>TI LLC</u>") and Ball Topaz Environmental Intelligence, LLC, a Delaware limited liability company ("<u>BTEI</u>", and, together with TI LLC, "<u>Topaz</u>").

"Acquired Company Benefit Plan" shall mean each Benefit Plan that is (a) sponsored, maintained or contributed to solely by the Acquired Companies or (b) exclusively for the benefit of the Company Employees or Former Company Employees.

"Affiliate" as to any Person, shall mean any other Person that, directly or indirectly, is in control of, is controlled by, or is under common control with, such Person, through one or more intermediaries or otherwise. For purposes of this definition, "control" of a Person shall mean the power, directly or indirectly, either to (a) vote 10% or more of the securities having ordinary voting power for the election of directors of such Person or (b) direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by Contract or otherwise. The Acquired Companies shall be deemed, for purposes of this Agreement, Affiliates of Seller prior to the Closing and Affiliates of Purchaser at and after the Closing.

"Agreement" shall mean this Stock Purchase Agreement (including the Seller Disclosure Schedule and all other schedules, annexes and exhibits attached hereto), as it may be amended from time to time.

"Anti-Corruption Laws" shall mean the U.S. Foreign Corrupt Practices Act of 1977, the U.K. Bribery Act 2010, applicable anti-bribery legislation enacted by member states of the European Union and signatories implementing the OECD Convention on Combating Bribery of Foreign Public Officials in International Business Transactions and other similar Legal Requirement applicable to any Acquired Company or the Business from time to time.

"Antitrust Laws" shall mean any Legal Requirements applicable to Purchaser, Seller, any Acquired Company or the Business under any applicable jurisdiction that are designed to prohibit, restrict or regulate actions or transactions having the purpose or effect of monopolization, restraint of trade or lessening competition, including any applicable United States or foreign antitrust or competition Legal Requirements.

"Axinn" shall mean Axinn, Veltrop & Harkrider LLP.

"Benefit Plan" shall mean each (a) "employee benefit plan" within the meaning of Section 3(3) of ERISA, whether or not subject to ERISA, (b) benefit or compensation plan, policy, program, practice, arrangement or agreement, including any equity, equity-based, retirement, profit sharing, bonus, commission, incentive, severance, separation, change in control, retention, transaction-based compensation, deferred compensation, tax gross up, fringe benefit, vacation, paid time off, medical, dental, life or disability plan, program, policy or arrangement and (c) employment, consulting or other similar individual agreement, plan, policy, arrangement or program, in each case of any of the foregoing described in clauses (a), (b) and (c), that is sponsored, maintained or contributed to by Seller or any of its ERISA Affiliates for the benefit of any Company Employee, Former Company Employee or other current or former individual

service provider of the Business, with respect to which Seller or any of its ERISA Affiliates has any liability with respect to any Company Employee, Former Company Employee or other individual service provider of the Business, or otherwise with respect to which any of the Acquired Companies has or could reasonably be expected to have a liability or obligation, other than any of the foregoing described in clauses (a), (b) or (c) that is sponsored and maintained (including if required to be maintained) by a Governmental Authority.

"Business" shall mean the business of designing, developing, manufacturing and selling systems, solutions, products, equipment, parts and components and providing services used in the intelligence surveillance and reconnaissance, civil, commercial and national security aerospace markets and the defense, civil space and commercial industries, including (a) the design, development, production and manufacture of spacecraft, instruments and sensors, radio frequency systems and components, national defense hardware, antenna and video tactical solutions, civil and operational space hardware, data exploitation solutions, advanced technologies and products that enable weather prediction and climate change monitoring as well as deep space missions and other advanced technologies and products, (b) the provision of the solutions, products and services provided by the national defense, tactical solutions, civil space and advanced technology and information solutions divisions of Seller, (c) the provision of systems engineering or any other services related to the foregoing and (d) all other business activities of the Acquired Companies, in each case of the foregoing clauses (a) – (c), as conducted by the Acquired Companies, the Seller or its other Controlled Affiliates as of the date hereof or as of the Closing, as applicable.

"Business Day" shall mean any day other than (a) a Saturday or a Sunday or (b) a day on which banking and savings and loan institutions are authorized or required to be closed in New York, New York or London, U.K.

"<u>Business Pension Plan</u>" shall mean the Ball Corporation Pension Plan as it Applies to Certain Salaried Employees of Ball Aerospace & Technologies Corp.

"Business Systems" shall mean all Software, computer hardware (whether general or special purpose) and systems, including electronic data processing, information, record keeping, communications and telecommunications networks, interfaces, platforms, equipment, servers, peripherals and systems, including any outsourced systems and processes, that are owned or used by or for the Business.

"Cash" shall have the meaning set forth on Annex A, Part I.

"<u>CFIUS</u>" shall mean the Committee on Foreign Investment in the United States and each member agency thereof, acting in such capacity.

"CFIUS Authorities" shall mean the Defense Production Act of 1950, as codified at 50 U.S.C. § 4565, and its implementing regulations located at 31 C.F.R. Parts 800 and 802.

"CFIUS Clearance" shall mean that: (a) the parties have received written notice from CFIUS that either (i) CFIUS has determined that the Transactions are not a "covered transaction" within the meaning of the CFIUS Authorities or (ii) CFIUS's review (or, if applicable, investigation) under the CFIUS Authorities of the Transactions in response to a joint voluntary notice submitted by the parties has concluded, and CFIUS has determined that there are no unresolved national security concerns with respect to the Transactions, and advised that all action under the CFIUS Authorities has concluded with respect to the Transactions or (b) CFIUS shall have sent a report to the President of the United States ("President") requesting the President's decision on the joint voluntary notice submitted by the parties and either (i) the period under the CFIUS Authorities during which the President may announce a decision to take action to suspend, prohibit or place any limitations on the Transactions shall have expired or (ii) the President shall have announced a decision not to take any action to suspend, prohibit or place any limitations on the Transactions.

"Clean Team Agreement" shall mean that certain Clean Team Confidentiality Agreement, dated June 13, 2023, by and between Seller and Purchaser as it has been or may be supplemented, modified or amended from time to time.

"Closing Conditions" shall mean the conditions to the respective obligations of the parties to consummate the Transactions, as set forth in Article X.

"Code" shall mean the United States Internal Revenue Code of 1986.

"Collective Bargaining Agreement" shall mean any collective bargaining agreement and any other labor-related agreement with any labor or trade union, works council, employee representative or association or other labor organization.

"Combination Mark" shall mean the "GO BEYOND WITH BALL" Trademark and any registrations or applications for such Trademark, including U.S. Reg. No. 5214969.

"Commercial Tax Agreement" shall mean any commercial Contract entered into in the ordinary course of business the principal purpose of which does not pertain to Taxes.

"Company Employee" shall mean each (a) Direct Employee and (b) Internal Transfer Employee; in each case including each such employee who is on leave of absence (including medical leave, parental leave, personal leave, extended COVID-19-related leave, military leave, workers' compensation leave, short-term disability and long-term disability) or paid or unpaid time off.

"Consent" shall mean any consent, waiver, approval or authorization.

"Consolidated Return" shall mean any consolidated, combined, unitary or similar Tax Return that includes Seller or any of its Controlled Affiliates (other than the Acquired Companies), on the one hand, and any Acquired Company, on the other hand.

"Contagion Event" shall mean the outbreak and ongoing effects of any epidemic or pandemic (including COVID-19).

"Contract" shall mean any legally binding agreement, contract, obligation, promise, understanding, arrangement, instrument, commitment or undertaking of any nature.

"Controlled Affiliate" shall mean each Affiliate of a Person that is directly or indirectly controlled by such Person. For purposes of this definition, "control" of a Person shall mean the power, directly or indirectly, to direct or cause the direction of the management and policies of such Person, whether by ownership of equity interests, by contract or otherwise. The Acquired Companies shall be deemed, for purposes of this Agreement, Controlled Affiliates of Seller prior to the Closing and Controlled Affiliates of Purchaser at and after the Closing.

"COVID-19" shall mean the novel coronavirus (SARS-CoV-2 or COVID-19), any evolutions or mutations thereof and any associated public health emergency, epidemic, pandemic or outbreak and any treatments, therapies or vaccines for, or in connection with, any of the foregoing.

"Credit Facilities" shall mean the facilities set forth on Schedule 1.1(a) of the Seller Disclosure Schedule.

"Customs & Trade Laws" shall mean all applicable export, import, customs, anti-boycott and other trade programs and Legal Requirements administered, enacted or enforced by any Governmental Authority, including: (a) the U.S. Export Administration Regulations, the U.S. International Traffic in Arms Regulations ("ITAR"), and the import Legal Requirements administered by U.S. Customs and Border Protection; (b) the anti-boycott Legal Requirements administered by the U.S. Departments of Commerce and Treasury and (c) any other similar export, import, customs, anti-boycott or other trade programs or Legal Requirements in any relevant jurisdiction to the extent they are applicable to the Acquired Companies.

"Cybersecurity Incident" shall mean any (a) unauthorized interference with security safeguards of, or unauthorized access to, any Business System, including any phishing incident, ransomware or malware attack, denial-of-service attack, breach of information technology or any stored information, (b) unauthorized access to, or acquisition, destruction, damage, disclosure, loss, corruption, alteration or use of, any Personal Information or sensitive data or (c) other cybersecurity, data or systems breach, attack or incident, in each case ((a) through (c)), to the extent such incident impacts Personal Information or data owned or controlled by or in the possession of an Acquired Company, or to the extent relating to the conduct of the Business by Seller or any of its Controlled Affiliates.

"Cybersecurity Measures" shall mean (a) any regulations promulgated by a Governmental Authority relating to cybercrime, cyberterrorism, ransomware, malware, privacy or the protection of Personal Information and (b) any reasonable measures, changes in business operations or other practices, affirmative or negative, adopted in good faith by any Acquired

Company in response to a cybersecurity attack, breach or incident, for the protection of its information technology or any stored information.

"<u>Data Protection Laws</u>" shall mean all applicable Legal Requirements administered, enacted or enforced by a relevant Governmental Authority in any jurisdiction in which any Acquired Company or, to the extent relating to the conduct of the Business, Seller or any of its Controlled Affiliates, conducts business relating to data privacy or data security or the processing or protection of Personal Information.

"<u>Data Room</u>" shall mean all electronic and in-person data rooms created in connection with the Transactions and set forth on <u>Schedule 1.1(b)</u> of the Seller Disclosure Schedule.

"DCSA" shall mean the Defense Counterintelligence and Security Agency.

"DCSA Approval" shall mean (a) receipt by the parties of written acknowledgement (including by email) from DCSA that it has accepted a foreign ownership, control or influence mitigation plan ("FOCI Mitigation Plan") proposed by Purchaser or (b) the entry into a written commitment notice or commitment letter executed by the parties and acknowledged by DCSA to mitigate the foreign ownership, control or influence over the Business or the Acquired Companies arising as a result of the transactions contemplated by this Agreement.

"<u>Debt Commitment Letter</u>" shall mean an executed commitment letter, including all schedules, annexes and exhibits thereto, dated as of the date hereof, from the Debt Financing Sources parties thereto.

"<u>Debt Financing</u>" shall mean the debt financing committed to be provided to Purchaser by the Debt Financing Sources pursuant to the Debt Commitment Letter or other agreements entered in connection therewith in the cash amounts set forth therein with respect to each such Debt Financing Source, for purposes of funding the Transactions.

"<u>Debt Financing Sources</u>" shall mean the Persons that have committed to provide or otherwise entered into any agreements in connection with any Debt Financing or alternative debt financing in connection with the Transactions, including the parties named in the Debt Commitment Letter and any joinder agreements, note purchase agreements, indentures or credit agreements entered into pursuant thereto or relating thereto.

"<u>Direct Employee</u>" shall mean each individual who is directly employed by an Acquired Company.

"Encumbrance" shall mean any lien, security interests, license, option, pledge, hypothecation, mortgage, deed, easement, claim, encroachment, servitude, right-of-way, preemption, collateral assignment, lease, right of first offer or first refusal, buy/sell agreement, defects in title or survey, or encumbrances of a similar kind (other than, in the case of a security,

any restriction on the transfer of such security arising solely under applicable Legal Requirements).

"Environmental Law" shall mean any Legal Requirement in effect as of or prior to the Closing Date, relating to pollution, public or worker health or safety (as it relates to exposure to Hazardous Materials), or protection of the environment or natural resources, including the use, handling, transportation, treatment, storage, disposal, release or threat of release or discharge of, or exposure to, Hazardous Materials.

"ERISA" shall mean the Employee Retirement Income Security Act of 1974.

"ERISA Affiliate" shall mean any Person which is treated at a relevant time as a single employer with another Person pursuant to Subsections (b), (c), (m) or (o) of Section 414 of the Code.

"<u>Former Company Employees</u>" shall mean (a) the Former Direct Employees and (b) the Former Internal Transfer Employees.

"Former Direct Employee" shall mean each individual who was, but no longer is, as of the date of this Agreement or the Closing Date, as applicable, directly employed by an Acquired Company.

"Former Internal Transfer Employee" shall mean each individual who was, but no longer is, as of the date of this Agreement or the Closing Date, as applicable, employed by Seller or one of its Controlled Affiliates (other than an Acquired Company) and whose regular employment duties or responsibilities were primarily dedicated or primarily related to the Business.

"Fraud" shall mean an actual and intentional fraud by a party in the making of the express representations and warranties in Article IV or the certificate delivered pursuant to Section 10.1(c) (in the case of Seller) and Article V or the certificate delivered pursuant to Section 10.2(c) (in the case of Purchaser). For the avoidance of doubt, "Fraud" shall not include any cause of action based on constructive or imputed knowledge, equitable fraud or any tort based on negligence, recklessness or any similar theory.

"GAAP" shall mean generally accepted accounting principles in the United States as in effect at the relevant time(s) for purposes of this Agreement.

"Government Contract" shall mean any Contract between an Acquired Company or, to the extent related to the Business, Seller or any of its other Controlled Affiliates and (a) a Governmental Authority, (b) any prime contractor of a Governmental Authority or (c) any subcontractor at any tier with respect to any Contract of a type described in clause (a) or (b) above (in each case of the foregoing, other than any Real Property Lease).

"<u>Governmental Authority</u>" shall mean any United States federal, state or local or any supra-national or non-United States government, political subdivision, governmental,

regulatory or administrative authority, instrumentality, agency, body or commission, self-regulatory organization or any court, tribunal or judicial or arbitral body, in each case, exercising executive, legislative, judicial, regulatory, taxing or administrative functions.

"<u>Hazardous Materials</u>" shall mean (a) petroleum, petroleum products, by-products or breakdown products, radioactive materials, asbestos or asbestos-containing materials, per- and polyfluoroalkyl substances or polychlorinated biphenyls and (b) any chemical, material or substance defined or regulated as hazardous, toxic, a pollutant or a contaminant, or for which liability or standards of conduct may be imposed under any Environmental Law.

"HSR Act" shall mean the Hart-Scott-Rodino Antitrust Improvements Act of 1976.

"<u>Inactive Employee</u>" shall mean each Internal Transfer Employee who is on a long-term disability leave or has qualified for long-term disability benefits as of the Closing Date.

"Income Tax" shall mean any Taxes imposed on or determined with reference to gross or net income, profits or receipts (including any franchise or withholding Taxes imposed in lieu thereof).

"Income Tax Amount" shall mean the amount (which shall not be less than zero in the aggregate or in respect of any jurisdiction or any type of Income Tax) of all unpaid Income Taxes of the Acquired Companies attributable to or payable with respect to any Pre-Closing Tax Period, calculated (a) as of the end of the Closing Date and on an entity-by-entity basis, such that the deductions and losses of one Acquired Company may not be used to offset the income and gain of another Acquired Company unless such offset is actually permitted by applicable Legal Requirements, (b) by including in taxable income any adjustment as a result of a change in or use of an improper method of accounting on or prior to the Closing Date and any prepaid amounts or deferred revenue that, in each case, would not otherwise be included in taxable income on or prior to the Closing Date, (c) by disregarding any transactions entered into by the Acquired Companies outside the ordinary course of business on the Closing Date after the Closing not otherwise contemplated by this Agreement and (d) by taking into account the Topaz Reorganization. In the case of any Straddle Period, the Income Tax Amount shall include an amount of Income Taxes allocable to the portion of the Straddle Period ending on and including the Closing Date as determined applying the conventions set forth in Section 8.2(b). For the avoidance of doubt, the Income Tax Amount shall not include any Income Taxes reported on a Consolidated Return, which Income Taxes shall be borne by Seller and its Affiliates.

"Income Tax Return" shall mean any Tax Return in respect of Income Taxes.

"Indebtedness" shall mean those items of indebtedness listed on Annex A, Part II.

"Indemnified Taxes" shall mean, without duplication, (a) Taxes of Seller and its Subsidiaries (other than any Acquired Company) for any taxable period, (b) Taxes of any consolidated, combined, unitary or similar group of which any Acquired Company is or was a member on or prior to the Closing Date, including Taxes arising under Treasury Regulations

Section 1.1502-6 (or any corresponding provision of state, local or foreign Legal Requirements), (c) Taxes of Seller or any of its Affiliates (other than another Acquired Company) imposed on an Acquired Company as a transferee or successor, or by assumption, Contract or the operation of any Legal Requirement, which Taxes relate to an event or transaction occurring prior to the Closing, (d) Specified Sales Taxes and (e) Taxes arising from or related to the consummation of the Topaz Reorganization, the transactions contemplated by the IP Assignment Agreement, the Real Estate Reorganization Plan and any transaction described in Section 6.5 or Section 6.6; provided, however, that if the exclusion in respect of Specified Sales Taxes in the R&W Insurance Policy is eliminated prior to the Closing Date such that Specified Sales Taxes are fully covered by the R&W Insurance Policy in the same manner as other Taxes that are not subject to an exclusion, clause (d) of "Indemnified Taxes" shall be deemed deleted; provided, further, that Seller and Purchaser shall use commercially reasonable efforts to cooperate to cause such exclusion to be eliminated from the R&W Insurance Policy. For the avoidance of doubt, "Indemnified Taxes" shall not include the amount of any Taxes taken into account in the final determination of the Income Tax Amount or Net Working Capital.

"Intellectual Property" shall mean all intellectual property and proprietary rights throughout the world, including (a) patents, patent applications, patent disclosures, inventions, statutory invention registrations, registered designs and similar or equivalent rights in inventions, designs, utility models, industrial models, industrial designs and all related divisionals, continuations, continuations-in-part, reissues, extensions, substitutions and reexaminations, certificates of invention and design patents, applications for any of the foregoing and all rights therein provided by international treaties and conventions ("Patents"), (b) trade secrets and rights in confidential and proprietary information, including know-how, ideas, patent disclosures, inventions, processes, formulae, models and methodologies, techniques, protocols, source code, algorithms, layouts, specifications, data and databases, processes, designs, technical information, drawings, blueprints, quality assurance and control procedures, design tools, simulation capability, manuals and technical information and research data and records, in each case excluding any rights in respect of any of the foregoing that comprise or are protected by issued Patents ("Trade Secrets"), (c) trademarks, service marks, trade names, brand names, logos, trade dress, Internet domain names and other indicia of source or origin and all registrations and applications for registration of the foregoing, together with the goodwill symbolized by any of the foregoing ("Trademarks"), (d) copyrights and all works of authorship (whether or not copyrightable), all registrations and applications for registration of such copyrights and all issuances, extensions and renewals of such registrations and applications ("Copyrights"), (e) Software and rights therein, (f) moral rights and (g) rights of privacy and publicity.

"Internal Transfer Employee" shall mean each employee of Seller or its Controlled Affiliates (other than the Acquired Companies) whose job duties or services are primarily related to the Acquired Companies or the Business and is set forth by employee identification number on Schedule 1.1(c) of the Seller Disclosure Schedule.

"IRS" shall mean the United States Internal Revenue Service.

"<u>Key Customer</u>" shall mean each of the customers or programs set forth on <u>Schedule 1.1(d)</u> of the Seller Disclosure Schedule.

"Key Vendor" shall mean each of the vendors set forth on Schedule 1.1(e) of the Seller Disclosure Schedule.

"<u>Legal Requirement</u>" shall mean any statute, law, ordinance, regulation, rule, code, Order or other requirement or rule of law (including common law) promulgated by a Governmental Authority.

"<u>Liabilities</u>" shall mean any direct or indirect debt, liability, obligation, expense, deficiency, guaranty or endorsement of or by such Person of any type, whether accrued, absolute, contingent, matured, unmatured, liquidated, unliquidated, known or unknown, asserted or unasserted, billed or unbilled, fixed or variable, secured or unsecured, choate or inchoate, perfected or unperfected, due or to become due, or determined or determinable.

"Losses" shall mean all losses, damages, costs, expenses, penalties, judgments, settlements, interest and fines actually suffered or incurred (including reasonable and documented out-of-pocket attorneys' fees).

"Malicious Code" shall mean any surreptitious computer code or other software routines or hardware components intentionally designed to permit unauthorized access to, disable or erase software, hardware or data, or to perform any other similar type of unauthorized activities (including viruses, Trojan horses, worms or other code, designs or routines (as these terms are commonly used in the computer software industry)).

"Material Adverse Effect" shall mean any event, change, development or effect that has had or would reasonably be expected to have, individually or in the aggregate, a material adverse effect on the business, operations, financial condition or results of operations of the Business, taken as a whole; provided that no event, change, development or effect resulting or arising from or in connection with any of the following matters shall be deemed, either alone or in combination, to constitute or contribute to, or be taken into account in determining whether there has been, a Material Adverse Effect: (a) any national, international, foreign, domestic or regional economic, financial, social or political conditions (including changes therein), (b) hostilities, acts of war, protests, riots, looting, unrest, terrorism, nuclear attack, cyberterrorism or military actions or any escalation or worsening of, or other changes or developments with respect to, any of the foregoing, (c) changes in any financial, debt, credit, capital or banking markets or conditions, including any changes in inflation or any disruption thereof, (d) changes in interest, currency or exchange rates, commodity prices, tariffs or any trade wars, (e) any act of God, hurricane, flood, tornado, fire, explosion, nuclear incident, weather event, earthquake, landslide, other natural disaster, any Contagion Event or other outbreak of illness or public health event (whether human or animal) or worsening of, or other changes or developments with respect to, any of the foregoing, (f) changes in legal or regulatory conditions, including changes in Legal Requirements (or standards, official interpretations or enforcement thereof, including Pandemic Measures), including in connection with a Contagion Event or the conflict between the

interpretations or enforcement thereof, (h) changes in the industries in which the Acquired Companies operate, (i) the failure of the Acquired Companies to meet any internal or published (A) financial projections. (B) estimates. (C) budgets or (D) forecasts of revenues, goals, earnings or other measures of financial or operating performance for any future periods (provided that any event, change, development or effect or combination thereof underlying such failure may be taken into account in determining whether a Material Adverse Effect has occurred or would reasonably be expected to occur (to the extent such event, change, development or effect or combination thereof is not otherwise excluded from this definition of Material Adverse Effect)), (j) any effect resulting from (A) the negotiation, execution, pendency, announcement, performance or consummation of the Transactions or compliance with any requirements under the terms of this Agreement (other than with respect to Section 6.1(a)(1)), (B) any breach by Purchaser of any of its representations, warranties and obligations under this Agreement or the other Transaction Agreements or (C) the identity of Purchaser or its Affiliates (provided that the foregoing clause (j)(A) shall not apply to any representation or warranty that is expressly intended to address the consequences of the negotiation, execution, pendency, announcement, performance or consummation of the Transactions or with respect to the condition to the Closing to the extent it relates to any such representation or warranty). (k) the effect of any action taken or omission to act by Purchaser, including any communication or disclosure by Purchaser or any of its Affiliates of its plans or intentions with respect to the Business or the Acquired Companies, including (solely to the extent arising therefrom) losses or threatened losses of, or any adverse change in the relationship with, employees, customers, suppliers, vendors, resellers, distributors, financing sources, licensors, licensees or others having relationships with the Business, (I) the effect of any event or action taken or omission to act by Seller or its Affiliates to the extent such action or omission is at the express written request of Purchaser, (m) the failure, in and of itself, to obtain any Consents in connection with the Transactions or (n) the initiation of a Proceeding by any Person with respect to this Agreement or any of the Transactions; provided that to the extent that any event, change, development or effect in the foregoing clauses (a) through (h) disproportionately has a greater adverse impact on the Business, taken as a whole, as compared to the adverse impact such event, change, development or effect has on other Persons operating in the same industries as the Business operates, then the incremental effect of such event, change, development or effect shall be taken into account in determining whether a Material Adverse Effect has occurred.

Russian Federation and Ukraine, (g) changes in GAAP or other applicable accounting standards or

"Net Working Capital" shall have the meaning set forth on Annex A, Part V.

"Net Working Capital Overage" shall have the meaning set forth on Annex A, Part IV.

"Net Working Capital Underage" shall have the meaning set forth on Annex A, Part IV.

"Non-Disclosure Agreement" shall mean the non-disclosure agreement between Seller and Purchaser, dated May 3, 2023, as modified by the supplement effective as of June 27, 2023.

"Open Source Software" shall mean any Software that is licensed pursuant to a license now or in the future approved by the Open Source Initiative and listed at http://www.opensource.org/licenses/alphabetical or that is considered "free" or "open source software" by the Free Software Foundation.

"Order" shall mean any order, writ, judgment, injunction, temporary restraining order, decree, stipulation, determination or award entered by or with any Governmental Authority.

"Owned IP" shall mean the Intellectual Property (a) owned or purported to be owned by the Acquired Companies as of the date of this Agreement, which shall include all Intellectual Property set forth or required to be set forth in Schedule 4.11(a) of the Seller Disclosure Schedule or (b) owned by Seller or its Controlled Affiliates (other than the Acquired Companies) and transferred to an Acquired Company at or prior to the Closing.

"Pandemic Measures" shall mean any quarantine, "shelter in place," "stay at home," social distancing, curfew, shutdown, closure, sequester, safety or any other Legal Requirement, Proceeding, directive, pronouncement, guideline or recommendation by any applicable Governmental Authority, including the Centers for Disease Control and Prevention and the World Health Organization, in each case in connection with or in response to any pandemic.

"Permit" shall mean any permit, order, license, registration, certificate, identification numbers, authorization or approval issued or required by any Governmental Authority under any applicable Legal Requirement.

"Permitted Encumbrances" shall mean (a) Encumbrances for Taxes, assessments or other governmental charges or levies not yet due and payable or the amount or the validity of which is being contested in good faith by appropriate Proceedings or that may thereafter be paid without material penalty, in each case, for which adequate reserves have been established and recorded on the Financial Statements in accordance with GAAP. (b) Encumbrances of carriers, warehousemen. mechanics, materialmen, workmen, repairmen and other similar Encumbrances for labor, materials or supplies imposed or permitted by Legal Requirements in the ordinary course of business, (c) Encumbrances incurred or deposits made in the ordinary course of business in connection with workers' compensation, unemployment insurance or other types of social security, (d) defects or imperfections of title, easements, covenants, rights of way, restrictions and similar charges or other matters of record which do not, or would not reasonably be expected to, materially impair the use or occupancy of the Real Property in the operation of the Business as conducted thereon, (e) zoning, entitlement, building codes and other generally applicable land use and environmental restrictions imposed on the Real Property by a Governmental Authority having jurisdiction over such Real Property which are not violated, in any material respect, by the current use or occupancy of such Real Property or the operation of

the Business thereon, (f) Encumbrances imposed on the underlying fee interest (or any other superior interest) of any real property leased or subleased by any Acquired Company or over which any Acquired Company has easement or other similar property rights, (g) Encumbrances incurred in the ordinary course of business since the date of the Last Balance Sheet securing liabilities that are not material to the Business, taken as a whole, (h) Encumbrances arising out of, relating to or resulting from this Agreement or the other Transaction Agreements, (i) Encumbrances affecting the assets or property of any Acquired Company that are discharged or released at or prior to the Closing, (i) any set of facts that an accurate up-to-date survey or inspection would show, which do not, or would not reasonably be expected to, materially impair the use or occupancy of the Real Property in the operation of the Business as conducted thereon, (k) rights of any landlord (or similar capacity) of any real estate lease or sublease (and related terms and conditions) under which an Acquired Company is a lessee or sublessee, (I) non-exclusive licenses of, non-exclusive covenants not to sue under and other non-exclusive grants of rights to use or obligations with respect to Intellectual Property granted by an Acquired Company to its customers in the ordinary course of business consistent with past practice, including under the Defense Federal Acquisition Regulation Supplement and any other Legal Requirements applicable to the Acquired Companies and the Business, (m) Encumbrances on securities created under federal, state or foreign securities Legal Requirements, (n) deposits to secure the performance of bids, trade Contracts, leases, statutory obligations, surety and appeal bonds, performance bonds and other obligations of a like nature, in each case in the ordinary course of business, (o) Encumbrances expressly disclosed in the Financial Statements, (p) non-monetary encumbrances or imperfections of title that, individually or in the aggregate, do not materially impair the continued use and operation of the properties and assets to which they relate in the conduct of the Business as conducted and (q) Encumbrances created by or at the written request of, or resulting from the action of or the unreasonably withholding of consent in violation of the terms of this Agreement by, Purchaser or any of its Affiliates.

"Person" shall mean any individual, corporation (including any non-profit corporation), general partnership, limited partnership, limited liability partnership, joint venture, estate, trust or company (including any limited liability company or joint stock company) or other similar entity, including a Governmental Authority.

"Personal Information" shall mean any information identifying, relating to, describing, that is reasonably capable of being associated with, or that could reasonably be linked, directly or indirectly, with a particular natural person or household and any data that constitutes personal information or personal data under any Data Protection Law.

"<u>Pre-Closing Tax Period</u>" shall mean any taxable period (or portion thereof through the end of the Closing Date in the case of a Straddle Period) ending on or prior to the Closing Date.

"Proceeding" shall mean any action, claim, suit, charge, complaint, litigation, arbitration, investigation or proceeding (whether civil or criminal) by or before any Governmental Authority.

"<u>Purchaser Consolidated Return</u>" shall mean any consolidated, combined, unitary or similar Tax Return that includes Purchaser or any of its Affiliates (other than the Acquired Companies), on the one hand, and any Acquired Company, on the other hand.

"<u>Purchaser Designee</u>" shall mean a wholly owned Subsidiary of Purchaser Guarantor (a) designated in writing by Purchaser to Seller at least five (5) Business Days prior to the Closing Date, (b) formed in a jurisdiction that would not delay consummation of the Transactions and (c) treated as a corporation (or an entity disregarded as separate from a corporation) for United States federal income tax purposes.

"<u>Purchaser Fundamental Representations</u>" shall mean the representations and warranties of Purchaser set forth in <u>Section 5.1</u> (Authority; Enforceability), <u>Section 5.3</u> (Organization) and <u>Section 5.7</u> (Brokers).

"<u>Purchaser Material Adverse Effect</u>" shall mean any event, change, development or effect that is or would reasonably be expected to be, individually or in the aggregate, materially adverse to the ability of Purchaser to perform its obligations under this Agreement or to consummate the Transactions.

"Real Property" shall mean the Owned Real Property and the Leased Real Property, collectively.

"Representatives" shall mean, in relation to a Person, its officers, directors, managers, members, employees, agents, advisors, other representatives and Affiliates.

"Restricted Cash" shall have the meaning set forth in Annex A, Part I.

"<u>Retained Business</u>" shall mean any business conducted by Seller and its Controlled Affiliates, whether undertaken prior to or after the date hereof, other than the Business.

"Retained Liabilities" shall mean all Liabilities to the extent arising out of, or relating to, the Retained Business, whether any such Liability arises before or after the Closing, is known or unknown or is contingent or accrued, other than to the extent this Agreement or any of the other Transaction Agreements provides that such Liabilities shall remain or become the Liabilities of, or otherwise become the responsibility of, Purchaser, its Affiliates or the Acquired Companies (and not of Seller or any of its Affiliates (other than the Acquired Companies)).

"Sanctioned Person" shall mean (a) any Person listed in any Sanctions Law-related list of designated Persons maintained by OFAC or the United States Department of State, the United Nations Security Council, the European Union, any European Union member state or His Majesty's Treasury of the United Kingdom, (b) any Person located, organized or resident in a Sanctioned Territory or (c) any Person directly or indirectly owned or controlled (as such terms, including any applicable ownership and control requirements, are defined and construed in the

applicable Sanctions Law or in any related official guidance) by any such Person or Persons described in the foregoing clauses (a) or (b).

"Sanctioned Territory" shall mean, at any time, a country or territory which is itself the subject or target of any country-wide or territory-wide Sanctions Laws (as of the date of this Agreement, Cuba, Iran, North Korea, Syria, the Crimea and so-called Donetsk People's Republic and Luhansk People's Republic regions of Ukraine).

"Sanctions Laws" shall mean economic or financial sanctions or trade embargoes imposed, administered or enforced from time to time by the Office of Foreign Assets Control of the United States Department of the Treasury ("OFAC") or the United States Department of State, the United Nations Security Council, the European Union, any European Union member state or His Majesty's Treasury of the United Kingdom.

"SEC" shall mean the United States Securities and Exchange Commission.

"Seller Benefit Plan" shall mean each Benefit Plan that is not an Acquired Company Benefit Plan.

"Seller Debt Facilities" shall mean the documents in respect of Indebtedness of the Acquired Companies set forth on Schedule 1.1(f) of the Seller Disclosure Schedule.

"Seller Disclosure Schedule" shall mean the disclosure schedules dated as of the date of this Agreement and delivered by Seller to Purchaser in connection with the execution of this Agreement.

"Seller Fundamental Representations" shall mean the representations and warranties of Seller set forth in Section 4.1 (Authority; Enforceability), Section 4.2(b)(i) (Non-Contravention; Consents) (solely with respect to the organizational documents of Seller and the Acquired Companies), Section 4.3(b) (Organization; Acquired Companies), Section 4.4 (Title; Shares) and Section 4.20 (Brokers).

"Seller Transitional Trademarks" shall mean the Trademarks set forth on Schedule 7.1(b) of the Seller Disclosure Schedule.

"Seller's Knowledge" and similar phrases shall mean the actual knowledge of each individual set forth on Schedule 1.1(g) of the Seller Disclosure Schedule after due inquiry by each such individual of such individual's direct reports.

"Shared Contracts" shall mean the Contracts under which the Business and at least one other business unit of Seller or any of its Affiliates (other than the Acquired Companies) purchases or sells goods or services on a joint basis or uses goods or services on a joint basis; provided that, in no event shall the term "Shared Contracts" include any Contracts for general corporate functions furnished by Seller or its Affiliates (other than the Acquired Companies) on an enterprise-wide basis to Seller and its Controlled Affiliates, including finance, accounting, tax, human resources, legal, information technology, facilities, facilities security, procurement and

other ancillary or corporate shared services provided by Seller or its Affiliates (other than the Acquired Companies) on an enterprise-wide basis to Seller and its Controlled Affiliates or other enterprise-wide corporate centralized functional organizations within or controlled by Seller or its Affiliates (other than the Acquired Companies), in each case to the extent such functions will be provided to Purchaser under the Transition Services Agreement or are Excluded Services (as defined in the Transition Services Agreement) (collectively in this proviso, "Excluded Shared Contracts"). Seller and Purchaser may, by mutual written consent, elect to include, or exclude from, this definition any Contract.

"Software" shall mean all computer software, data and databases, operating systems, tools, interfaces, firmware, middleware, modules, models, algorithms and routines (in each case, as applicable, in source code and object code form) and all documentation and materials relating to any of the foregoing.

"Specified Sales Taxes" shall mean any sales or similar Taxes required to be collected by any Acquired Company, or required to be remitted or reported by any Acquired Company to any Governmental Authority, in each case, with respect to transactions involving government contractors or subcontractors occurring on or prior to the Closing Date.

"Standalone Go Beyond Mark" shall mean "GO BEYOND" as a standalone phrase or as used in combination with any other word, phrase or Trademark other than a Seller Mark.

"Straddle Period" shall mean any taxable period that includes (but does not end on) the Closing Date.

"Subsidiary" shall mean, with respect to any Person, whether incorporated or unincorporated, of which (a) such first Person directly or indirectly owns or controls at least a majority of the securities or other interests having by their terms ordinary voting power to elect a majority of the board of directors or others performing similar functions (or, if there are no such securities or other interests, a majority of the equity interests) or (b) such first Person is a general partner or managing or operating member. The Acquired Companies shall be deemed, for purposes of this Agreement, Subsidiaries of Seller prior to the Closing and Subsidiaries of Purchaser at and after the Closing.

"Target Net Working Capital" shall have the meaning set forth in Annex A, Part IV.

"Tax" shall mean (a) all forms of taxation imposed by any Governmental Authority, including all United States federal, state or local and foreign taxation (including income, value added, occupation, real and personal property, social security (or similar), gross receipts, sales, use, ad valorem, franchise, profits, license, withholding, payroll, employment, unemployment, excise, severance, occupation, premium, windfall profits, estimated, alternative, add-on minimum and other taxes, charges, levies, duties, impositions, or other governmental assessments of any kind whatsoever, including stamp duty, customs and other import or export duties), (b) any interest, penalties or additions to tax with respect to any item described in the preceding clause and (c) any liability to indemnify, assume or succeed to the liability of another

Person with respect to any item described in clauses (a) or (b), including as a result of being a member of any consolidated, combined, unitary or other similar Tax group or under any Legal Requirements or by equity, Contract, assumption, transferee or successor liability.

"<u>Tax Authority</u>" shall mean a Governmental Authority responsible for the imposition, assessment or collection of any Tax (domestic or foreign).

"<u>Tax Proceeding</u>" shall mean any federal, state, local or foreign audit, examination, litigation or other administrative proceeding or court proceeding relating to Taxes.

"<u>Tax Return</u>" shall mean any report, return, statement, declaration, notice, certificate or other document filed or required to be filed with any Tax Authority in connection with the determination, assessment, collection or payment of any Tax, including any schedule or attachment thereto and any amendment thereof.

"<u>Transaction Agreements</u>" shall mean this Agreement and the Transition Services Agreement, in each case including all exhibits, annexes and schedules thereto and all amendments thereto made in accordance with the respective terms hereof and thereof.

"Transaction Expenses" shall have the meaning set forth on Annex A, Part III.

"<u>Transactions</u>" shall mean the transactions contemplated by this Agreement and the other Transaction Agreements.

"<u>Transfer Taxes</u>" shall mean any sales, use, stock transfer, real property transfer, transfer, indirect transfer, goods and services, value-added, stamp, registration, documentary, conveyancing, recording or similar Taxes.

"<u>Treasury Regulations</u>" shall mean the United States Treasury regulations promulgated under the Code.

"<u>WARN Act</u>" shall mean the Worker Adjustment and Retraining Notification Act of 1988 or any similar applicable state or local Legal Requirements requiring notice to employees in the event of a plant closing or mass layoff.

"Willful Breach" shall mean (a) an intentional action or failure to act by one of the parties that constitutes a material breach of this Agreement, and such action was taken or such failure to act occurred with such party's knowledge, or in circumstances where such party should reasonably have known, that such action or failure to act constituted a material breach of this Agreement, and such breach (i) resulted in, or contributed to, the failure of any of the Closing Conditions to be satisfied or (ii) resulted in, or contributed to, the Closing not being consummated at the time the Closing would have otherwise occurred pursuant to Section 3.1 or (b) the failure of Purchaser to deliver the full consideration payable pursuant to Article III substantially concurrently with the Closing (and, in any event, on the Closing Date).

ARTICLE II

THE TRANSACTIONS

- 2.1 <u>Sale and Purchase of the Shares</u>. Subject to the terms and conditions of this Agreement, at the Closing, Seller shall sell, transfer, convey, assign and deliver to Purchaser (or a Purchaser Designee), and Purchaser (or a Purchaser Designee) shall purchase from Seller, all of Seller's rights, title and interest in and to the Shares, free and clear of all Encumbrances (other than those arising under applicable securities Legal Requirements).
- 2.2 <u>Purchase Price</u>. The aggregate consideration to be paid by Purchaser to Seller for the purchase of the Shares shall be an amount in cash equal to (a) \$5,555,000,000 (the "<u>Base Price</u>"), plus (b) the Net Working Capital Overage (if any), minus (c) the Net Working Capital Underage (if any), plus (d) Cash as of immediately prior to the Closing, minus (e) Indebtedness as of immediately prior to the Closing (except that the Income Tax Amount shall be determined as of the end of the Closing Date), minus (f) Transaction Expenses unpaid as of immediately prior to the Closing (the amount calculated pursuant to this sentence, the "<u>Purchase Price</u>").

2.3 Closing Purchase Price.

- (a) Not less than five (5) Business Days prior to the anticipated Closing Date, Seller shall prepare and deliver to Purchaser a written statement (the "Estimated Closing Statement") setting forth Seller's good faith estimate of (i) Cash as of immediately prior to the Closing, (ii) Indebtedness as of immediately prior to the Closing (except that the Income Tax Amount shall be determined as of the end of the Closing Date), (iii) Net Working Capital as of the Closing, (iv) Transaction Expenses unpaid as of immediately prior to the Closing and (v) the resulting calculation of the Purchase Price (such amount, the "Closing Purchase Price"), together with reasonable supporting detail and documentation, in each case, prepared in accordance with the terms and conditions of this Agreement, including the Accounting Principles as applicable.
- (b) Following delivery of the Estimated Closing Statement to Purchaser, Purchaser and its Representatives, accountants, advisors and other representatives shall be permitted reasonable access to review the books, records and work papers of the Acquired Companies and Seller to the extent reasonably related to the Estimated Closing Statement (in each case, upon reasonable notice to Seller and during normal business hours and only to the extent that such access does not interfere with the normal business operations of the Acquired Companies and Seller; provided that the accountants of Seller and the Acquired Companies shall not be obliged to make any work papers available except in accordance with such accountants' normal disclosure procedures and then only after, as applicable, Purchaser and its applicable Representatives have signed a customary agreement relating to such access to work papers), and Seller shall consider in good faith any potential adjustments to the calculation of the Closing Purchase Price (or any of the components thereof) raised by Purchaser in good faith not less than two (2) Business Days prior to the anticipated Closing Date; provided that (i) nothing in this Section 2.3(b) shall obligate Seller to accept any such comments and in the event Seller does not accept such comments, Seller's calculation of the Closing Purchase Price (and components thereof) shall control for purposes of the Closing and (ii) Seller's obligations to consider in good faith any such comments shall not in any event require that the contemplated Closing Date be postponed or otherwise delayed.

2.4 <u>Post-Closing Adjustment</u>.

- As soon as practicable after the Closing Date but in no event later than one-hundred twenty (120) days after the Closing Date, Purchaser shall deliver to Seller a written statement (the "Post-Closing Statement") setting forth Purchaser's good faith calculation of (i) Cash as of immediately prior to the Closing (the "Preliminary Cash"), (ii) Indebtedness as of immediately prior to the Closing (except that the Income Tax Amount shall be determined as of the end of the Closing Date) (the "Preliminary Indebtedness"), (iii) Net Working Capital as of the Closing (the "Preliminary Net Working Capital"), (iv) Transaction Expenses unpaid as of immediately prior to the Closing ("Preliminary Transaction Expenses") and (v) the resulting calculation of the Purchase Price (such amount, the "Preliminary Closing Purchase Price"), together with reasonable supporting detail and documentation, in each case, prepared in accordance with the terms and conditions of this Agreement, including the Accounting Principles as applicable. Purchaser shall not amend, supplement or modify the Post-Closing Statement following delivery to Seller. Notwithstanding anything herein to the contrary, but subject to the Accounting Principles and the terms and conditions of this Agreement, (A) the Post-Closing Statement shall be based solely on facts and circumstances as they exist as of the Closing and shall exclude the effect of any event, change, circumstance, development, occurrence, condition, effect or state of facts occurring after the Closing (except, with respect to the Income Tax Amount, to the extent taken or occurring in the ordinary course of business) and (B) the parties acknowledge and agree that the purpose of preparing the calculations under this Section 2.4 is to determine only the arithmetic difference between the items in the Estimated Closing Statements and the Post-Closing Statement applicable items in accordance with the Accounting Principles and the terms and conditions of this Agreement. If Purchaser fails to deliver the Post-Closing Statement in accordance with this Section 2.4(a) within such one-hundred twenty (120) day period, then the Estimated Closing Statement delivered by Seller to Purchaser pursuant to Section 2.3 shall be deemed to be the Post-Closing Statement.
- Following the receipt of the Post-Closing Statement, Seller shall have sixty (60) days (the "Review Period") to review such Post-Closing Statement and related computations of the Preliminary Cash, the Preliminary Indebtedness, the Preliminary Net Working Capital, the Preliminary Transaction Expenses and the Preliminary Closing Purchase Price. Following the Closing through the date that the Final Closing Statement becomes final and binding in accordance with Section 2.4(d), Seller, its Controlled Affiliates and its and their respective Representatives, accountants, advisors and other representatives shall be permitted reasonable access to review the books, records and work papers of the Acquired Companies and Purchaser to the extent reasonably related to the Post-Closing Statement or calculations of Cash, Indebtedness, Net Working Capital or Transaction Expenses, and Purchaser shall, and shall use commercially reasonable efforts to cause its Affiliates and its and their respective Representatives, accountants, advisors and other representatives to, cooperate with and assist Seller, its Controlled Affiliates and its and their respective Representatives, accountants, advisors and other representatives in connection with such review, including by providing access to such books, records and work papers and making available personnel to the extent requested, in each case, upon reasonable notice and during normal business hours and only to the extent that such access does not interfere with the normal business operations of the Acquired Companies and Purchaser; provided that the accountants of Purchaser shall not be obliged to make any work papers available except in accordance with such accountants' normal disclosure procedures and then only after, as applicable, Seller, its applicable

Controlled Affiliates and its and their applicable Representatives have signed a customary agreement relating to such access to work papers. Purchaser agrees that, following the Closing through the date that the Final Closing Statement becomes final and binding in accordance with Section 2.4(d), it will not take, or permit to be taken, any actions with respect to its or the Acquired Companies' accounting books, records, policies or procedures on which the Financial Statements or the Post-Closing Statement are based, or upon which the Final Closing Statement is to be based, that would be reasonably expected to impede or materially delay the determination of the amount of Cash, Indebtedness, Net Working Capital or the preparation of any Statement of Objections or the Final Closing Statement in the manner and utilizing the methods provided by this Agreement, including the Accounting Principles. If Seller has accepted the Post-Closing Statement in writing or has not given written notice to Purchaser setting forth any objection of Seller to such Post-Closing Statement, specifying in reasonable detail the nature and basis for such objection, Seller's alternative calculation of each disputed item (together with reasonable supporting documentation to support Seller's alternative calculation), and Seller's proposed modifications to the Post-Closing Statement (such notice, the "Statement of Objections") prior to the expiration of the Review Period, then such Post-Closing Statement shall be final and binding upon the parties, and shall be deemed the Final Closing Statement for purposes of Section 2.4(d).

In the event that Seller delivers a Statement of Objections to Purchaser prior to the expiration of the Review Period, Seller and Purchaser shall negotiate in good faith to resolve any such objection within thirty (30) days (or, if mutually agreed by the parties, such longer period) following the receipt by Purchaser of the Statement of Objections (the "Consultation Period"). If Seller and Purchaser reach an agreement in writing as to any such objections within the Consultation Period, the amounts so agreed upon shall be final and such agreement shall be deemed to be included in the Final Closing Statement for purposes of Section 2.4(d). If Seller and Purchaser are unable to reach an agreement in writing as to any such objections within the Consultation Period, then Seller and Purchaser shall jointly submit such matter to Deloitte Touche Tohmatsu LLC, or if Deloitte Touche Tohmatsu LLC is unable or unwilling to serve in such capacity, such other independent accounting firm of national reputation as shall be agreed upon in writing by Seller and Purchaser (the "Settlement Accountant"), for resolution of those items on the Statement of Objections that remain in dispute (the "Disputed Items"). The Settlement Accountant shall act as an expert and not as an arbitrator, and shall only consider the Disputed Items. Any items or amounts that have not been disputed in a Statement of Objections delivered prior to the expiration of the Review Period shall be final and binding upon Seller and Purchaser, If any Disputed Item is referred to the Settlement Accountant, Seller, on the one hand, and Purchaser, on the other hand, shall prepare separate written reports of each such Disputed Item and deliver such reports to the Settlement Accountant and each other within fifteen (15) Business Days after the date the Settlement Accountant is retained. Each of Seller and Purchaser shall have ten (10) Business Days after receipt of the other party's written report to deliver to the Settlement Accountant and each other one written rebuttal thereto (if applicable). Seller and Purchaser shall not make any further submissions to the Settlement Accountant unless otherwise agreed in writing by Seller and Purchaser; provided, that the Settlement Accountant may also reasonably request either Seller or Purchaser to answer questions that it deems relevant to the resolution of the dispute, and Seller and Purchaser, as applicable, shall reasonably cooperate with such request. The Settlement Accountant may not assign a value to any Disputed Item greater than the greatest value for such Disputed Item claimed by either Seller in the Statement of Objections or Purchaser in the Post-Closing Statement or less than the smallest value for such Disputed Item claimed by either Seller in the Statement of

shall be (i) limited only to the Disputed Items, (ii) based solely on such reports, rebuttals and supporting information submitted by Seller and Purchaser and the terms of this Agreement including the Accounting Principles (i.e., not on the basis of an independent review) and (iii) in accordance with the terms and procedures set forth in this Agreement, including the Accounting Principles, and consistent with the definitions of Cash, Indebtedness, Net Working Capital and Transaction Expenses contained herein. During the review by the Settlement Accountant, each of Seller and Purchaser shall, and shall cause its respective Subsidiaries and its and their respective Representatives, accountants, advisors and other representatives to, each make available to the Settlement Accountant reasonable access to personnel and such information, books, records and work papers as may be reasonably requested by the Settlement Accountant to fulfill its obligations under this Section 2.4(c); provided that the accountants of Seller or Purchaser shall not be obliged to make any work papers available to the Settlement Accountant except in accordance with such accountants' normal disclosure procedures and then only after such Settlement Accountant has signed a customary agreement relating to such access to work papers. A copy of all materials submitted to the Settlement Accountant shall be provided by Seller or Purchaser, as applicable, to the other party in the dispute concurrently with the submission thereof to the Settlement Accountant; provided that the accountants of Seller or Purchaser, as applicable, shall not be obliged to make any work papers available to the other party except in accordance with such accountants' normal disclosure procedures and then only after such other party has signed a customary agreement relating to such access to work papers. Subject to Section 2.4(h), the Settlement Accountant shall have exclusive jurisdiction over, and resort to the Settlement Accountant as provided in this Section 2.4(c) shall be the only recourse and remedy of the parties against one another with respect to, any disputes arising out of or relating to the calculation of, and any adjustments to, the Purchase Price. The final determination with respect to all Disputed Items, including the Settlement Accountant's basis for such determination, shall be set forth in a written statement by the Settlement Accountant delivered to Seller and Purchaser and, absent mathematical or manifest error raised within five (5) Business Days of the Settlement Accountant's determination and promptly resolved by the Settlement Accountant in its sole discretion, the resolution of the dispute by the Settlement Accountant shall be final, binding and non-appealable on the parties and such determination may be entered and enforced in any court of competent jurisdiction in accordance with Section 12.8. The costs and expenses of the Settlement Accountant shall be borne by Seller and Purchaser in inverse proportion to the difference between the Settlement Accountant's determination of the Purchase Price and the determination of the Purchase Price claimed by Seller and Purchaser. For example, if Seller claims that the Purchase Price is, in the aggregate, \$1,000 greater than the amount determined by Purchaser and if the Settlement Accountant ultimately resolves the dispute by awarding to Seller an aggregate of \$300 of the \$1,000 contested, then the costs and expenses of the Settlement Accountant will be allocated 30% to Purchaser and 70% to Seller.

Objections or Purchaser in the Post-Closing Statement. The Settlement Accountant's review and determination

(d) The Post-Closing Statement as agreed to by Seller and Purchaser or as determined by the Settlement Accountant is referred to herein as the "Final Closing Statement" and (i) the Cash set forth on such Final Closing Statement shall be deemed the final Cash as of immediately prior to the Closing, (ii) the Indebtedness set forth on such Final Closing Statement shall be deemed the final Indebtedness as of immediately prior to the Closing (except that the Income Tax Amount shall be determined as of the end of the Closing Date), (iii) the Net Working Capital set forth on such Final Closing Statement shall be deemed the final Net Working Capital

as of the Closing, (iv) the Transaction Expenses set forth on such Final Closing Statement shall be deemed the final Transaction Expenses unpaid as of immediately prior to the Closing and (v) the Purchase Price set forth on such Final Closing Statement shall be deemed the final Purchase Price (the "<u>Final Purchase Price</u>").

- (e) In the event that the Final Purchase Price is greater than the Closing Purchase Price (such excess, the "<u>Final Overage</u>"), Purchaser shall deposit, or cause to be deposited, within five (5) Business Days of the determination of the Final Overage and the Final Closing Statement, with Seller, by wire transfer of immediately available funds, an amount equal to the Final Overage.
- (f) In the event that the Closing Purchase Price is greater than the Final Purchase Price (such excess, the "Final Underage"), Seller shall deposit, or cause to be deposited, within five (5) Business Days of the determination of the Final Underage and the Final Closing Statement, with Purchaser, by wire transfer of immediately available funds, an amount equal to the Final Underage.
- (g) The parties agree to treat for all applicable Income Tax purposes any adjustment as determined pursuant to this <u>Section 2.4</u> as an adjustment to the Purchase Price.
- (h) The process set forth in this <u>Section 2.4</u> shall be the sole and exclusive remedy of Seller and its Affiliates and Purchaser and its Affiliates for any disputes between the parties related to the determination of Final Closing Statement and the calculations, items and amounts set forth therein, whether or not the underlying facts and circumstances constitute a breach of any representations or warranties contained in this Agreement; <u>provided</u>, that notwithstanding anything to the contrary herein, the foregoing shall not prohibit Seller or Purchaser, as applicable, from instituting a Proceeding to enforce any final determination of the Final Closing Statement by the Settlement Accountant pursuant to the terms and conditions of this <u>Section 2.4</u>, or to compel Seller or Purchaser, as applicable, to submit any dispute arising in connection with this <u>Section 2.4</u> to the Settlement Accountant pursuant to the terms and conditions of this <u>Section 2.4</u> in any court or other tribunal of competent jurisdiction in accordance with <u>Section 12.8</u>. Notwithstanding anything herein to the contrary, nothing in this <u>Section 2.4</u> shall be construed to affect or limit the ability of Purchaser or its Affiliates (including the Acquired Companies) to recover under the R&W Insurance Policy.
- 2.5 <u>Withholding</u>. Each of Purchaser and any other applicable withholding agent shall be entitled to deduct and withhold (or cause to be deducted and withheld) Taxes from any amounts payable to any Person pursuant to this Agreement as required by applicable Legal Requirements. Purchaser acknowledges and agrees that, provided that Seller delivers a duly executed IRS Form W-9 in accordance with <u>Section 3.2(d)</u>, no withholding in respect of U.S. federal income Taxes is expected as of the date of this Agreement to be applicable to the payment of the Purchase Price to Seller. To the extent Purchaser becomes aware of any obligation to withhold (other than any obligation to withhold (a) in respect of payments that are in the nature of compensation or (b) resulting from Seller's failure to deliver a duly executed IRS Form W-9 in accordance with <u>Section 3.2(d)</u>), it shall use commercially reasonable efforts to (i) provide reasonable notice at least ten (10) days in advance of any withholding to Seller of the amounts subject to withholding and (ii) provide Seller with a reasonable opportunity to deliver any forms, documentation or other

evidence that would reduce or eliminate such withholding Tax under Legal Requirements. Purchaser shall cooperate with Seller, upon Seller's reasonable request, to reduce or eliminate any such withholding Tax in a manner consistent with applicable Legal Requirements. To the extent any amounts are deducted and withheld by Purchaser or any other applicable withholding agent under this Section 2.5 and paid over to the applicable Governmental Authority in accordance with the applicable Legal Requirements, such amounts shall be treated for purposes of this Agreement as having been paid to the Person in respect of which such deduction and withholding was made.

ARTICLE III

CLOSING AND CLOSING DELIVERIES

- 3.1 <u>Closing; Time and Place</u>. The closing of the Transactions (the "<u>Closing</u>") shall occur at the offices of Skadden, Arps, Slate, Meagher & Flom LLP ("<u>Skadden</u>"), 155 North Wacker Drive, Chicago, Illinois 60606 (which Closing may occur by electronic exchange of documents), at 10:00 a.m. Eastern time on the third (3rd) Business Day after the date on which all of the Closing Conditions are satisfied or, if permissible, waived (other than those conditions that by their nature are to be satisfied at the Closing, but subject to the satisfaction or waiver of those conditions), or at such other date, time or place as Seller and Purchaser may agree in writing. The date on which the Closing occurs is referred to herein as the "<u>Closing Date</u>." For all purposes under this Agreement and each other Transaction Agreement, all matters at the Closing will be considered to take place simultaneously.
- 3.2 <u>Deliveries by Seller</u>. At the Closing, Seller shall deliver, or cause to be delivered, to Purchaser (or a Purchaser Designee, as applicable, with respect to <u>Section 3.2(a)</u>):
- (a) the Shares, free and clear of all Encumbrances (other than those arising under applicable securities Legal Requirements), together with certificates evidencing the Shares to the extent that such Shares are in certificate form, duly endorsed in blank or with stock powers or a similar instrument of transfer duly executed in proper form for transfer, in customary form;
- (b) a duly executed counterpart to a transition services agreement, substantially in the form attached hereto as Exhibit A (the "Transition Services Agreement");
 - (c) the certificate required to be delivered by Seller pursuant to Section 10.1(c);
 - (d) an IRS Form W-9 duly executed by Seller;
- (e) a duly executed Intellectual Property Assignment Agreement whereby Seller, on behalf of itself and its Controlled Affiliates, assigns to an Acquired Company all right, title and interest in and to all Intellectual Property (inclusive of any related rights thereto, including rights to past and future income, royalties and claims thereof) owned by Seller or any of its Controlled Affiliates (and not an Acquired Company) that (i) as of the date hereof or as of the Closing, is substantially exclusively used or held for use the conduct of the Business or (ii) was authored, created, invented or developed (in whole or part) by any Company Employee or any Former Company Employee, on a form of agreement reasonably satisfactory to Purchaser (the "IP Assignment Agreement"); and

- (f) the Seller Debt Facilities Releases.
- 3.3 <u>Deliveries by Purchaser.</u> At the Closing, Purchaser shall deliver, or cause to be delivered, to Seller:
 - (a) by wire transfer of immediately available funds, the Closing Purchase Price;
 - (b) a duly executed counterpart to the Transition Services Agreement; and
 - (c) the certificate required to be delivered by Purchaser pursuant to <u>Section 10.2(c)</u>.
- 3.4 <u>Payment Mechanics</u>. Any payment to be made pursuant to this Agreement by Purchaser shall be made to the designee and bank account or accounts designated in advance by Seller in writing to Purchaser on or before the third (3rd) Business Day prior to the due date for payment. Any payment to be made pursuant to this Agreement by Seller shall be made to the bank account designated in advance by Purchaser in writing to Seller on or before the third (3rd) Business Day prior to the due date for payment. Unless otherwise agreed in writing by Seller and Purchaser, any payments by wire transfer under this Agreement shall be in immediately available funds. All payments shall be made by electronic transfer on the due date for payment and receipt of the amount due shall be an effective discharge of the relevant payment obligation.

ARTICLE IV

REPRESENTATIONS AND WARRANTIES OF SELLER

Except as set forth in the Seller Disclosure Schedule, Seller hereby represents and warrants to Purchaser as follows:

4.1 <u>Authority; Enforceability</u>.

- (a) Seller has the requisite corporate power and authority to execute and deliver this Agreement and each other Transaction Agreement to which it is a party, to perform its obligations hereunder and under each other Transaction Agreement to which it is a party and to consummate the Transactions in accordance with the terms of this Agreement and each other Transaction Agreement to which it is a party. The execution, delivery and performance by Seller of this Agreement and each other Transaction Agreement to which it is a party and the consummation of the Transactions by Seller have been duly and validly authorized by all necessary corporate action on the part of Seller and such authorization has not been subsequently modified or rescinded. No vote or approval of the holders of any class or series of capital stock of Seller is necessary for the execution, delivery or performance by Seller of this Agreement or any other Transaction Agreement to which Seller is a party or the consummation by Seller of the Transactions in accordance with the terms of this Agreement and each other Transaction Agreement to which it is a party.
- (b) This Agreement has been duly and validly executed and delivered by Seller and constitutes, assuming due authorization, execution and delivery of this Agreement by Purchaser and Purchaser Guarantor, a valid and binding legal obligation of Seller, enforceable

against Seller in accordance with the terms hereof. Assuming due authorization, execution and delivery of each other Transaction Agreement to which Seller is a party by the other parties thereto, each such Transaction Agreement will constitute a valid and binding legal obligation of Seller at the time of execution by Seller, enforceable against Seller in accordance with the terms thereof.

4.2 Non-Contravention; Consents.

- (a) The execution and delivery of this Agreement by Seller, and each other Transaction Agreement to which Seller is a party by Seller, does not, and the performance of this Agreement by Seller, and each other Transaction Agreement to which Seller is a party by Seller, will not, require any Consent or Permit of, registration, declaration or filing with, or notification to, any Governmental Authority (other than as a party to any Government Contract or as the ultimate customer of any Government Contract), except (i) under applicable Antitrust Laws, (ii) under the applicable requirements of the Securities Exchange Act of 1934 (the "Exchange Act"), (iii) for such other Consents, Permits, filings or notifications, the failure of which to make or obtain would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect or prevent or materially delay the consummation of the Transactions, (iv) in connection with obtaining CFIUS Clearance and DCSA Approvals, (v) under Section 122.4(b) of the ITAR and (vi) those required to be obtained by Purchaser solely by reasons of the regulatory status or operations of Purchaser or its Affiliates.
- (b) Assuming the Consents, Permits, registrations, declarations, filings and notifications referred to in Section 4.2(a) are obtained or made, the execution and delivery by Seller of this Agreement and each other Transaction Agreement to which Seller is a party, and the consummation of the Transactions, will not (i) conflict with or violate any provision of the organizational documents of Seller or any applicable Acquired Company or, (ii) except as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect or prevent or materially delay the consummation of the Transactions, (A) result in a breach of, constitute a default under (with or without notice or lapse of time, or both), result in the creation or acceleration (or loss of benefit from) of any rights or obligations under, or create in any party the right to accelerate, terminate, modify or cancel, any Material Contract (other than Material Contracts entered into after the date of this Agreement that is a Pending Bid Contract, for which a Material Contract Waiver is obtained or for which Purchaser consents in writing into the entry thereof); (B) result in the creation or imposition of any Encumbrance (other than a Permitted Encumbrance) upon, or the grant, assignment or transfer to any other Person of any license or other right or interest under, any of the assets or businesses of the Acquired Companies or, to the extent related to the Business, Seller or any of its Controlled Affiliates; or (C) violate any Legal Requirement.

4.3 <u>Organization; Acquired Companies</u>.

(a) Seller is duly incorporated, validly existing and in good standing under the Legal Requirements of the jurisdiction of its incorporation, except as would not reasonably be expected to materially impair or materially delay Seller from consummating the Transactions or otherwise prevent Seller from performing in all material respects its obligations hereunder. Seller has all necessary corporate power and authority to conduct its business in the manner in which it is being conducted as of the date of this Agreement, except where the absence of such power to

conduct its business would not reasonably be expected to materially impair or materially delay Seller from consummating the Transactions or otherwise prevent Seller from performing in all material respects its obligations hereunder.

- (b) Each of the Acquired Companies (i) is duly organized, incorporated or formed, validly existing and in good standing (to the extent such concept is recognized) under the Legal Requirements of the jurisdiction of its organization, incorporation or formation, in all material respects and (ii) has all necessary organizational power and authority to conduct the Business in the manner in which it is being conducted as of the date of this Agreement and as of the Closing.
- (c) Each Acquired Company is duly qualified or licensed to do business in the jurisdictions in which the property and assets owned, leased or operated by it, or the nature of the business conducted by it, makes such qualification or licensing necessary, except where the failure to be so duly qualified or licensed would not, individually or in the aggregate, (i) be (or reasonably be expected to be) material to the Acquired Companies, taken as a whole, or (ii) prevent or materially delay the consummation of the Transactions.
- (d) Schedule 4.3(d) of the Seller Disclosure Schedule sets forth a true and complete list of the names of each Acquired Company, the jurisdiction in which each such Acquired Company is organized and the equity ownership thereof (including the authorized capitalization, number of outstanding shares of each class of capital stock or other equity interest and the record and beneficial owners thereof), in each case, as of the date of this Agreement. The Acquired Companies do not own, directly or indirectly, any capital stock, shares, membership interests, other equity or ownership rights, interests or other securities or derivatives in any Person (other than the Company Subsidiary) as of the date of this Agreement.
- (e) All of the outstanding shares of each Acquired Company are and, as of the Closing shall be, duly and validly issued and outstanding, fully paid and non-assessable and legally and beneficially owned, directly or indirectly, by Seller, free and clear of all Encumbrances (other than restrictions on transfer imposed by applicable Legal Requirements and Encumbrances that will be released at or prior to the Closing). All of the outstanding shares of each Acquired Company have been issued in all material respects in compliance with applicable Legal Requirements, the organizational documents of such Acquired Company or any Contract to which such Acquired Company is subject or bound, and are not subject to any preemptive, subscription or similar right under any provision of any of the foregoing.
- (f) None of Seller nor any of the Acquired Companies is the subject of any bankruptcy, dissolution, liquidation, reorganization or similar proceeding.
- (g) Seller has made available to Purchaser accurate and complete copies, in all material respects, of the organizational documents of each Acquired Company as in effect as of the date of this Agreement.

4.4 Title; Shares.

(a) Seller is the sole record and beneficial owner of all of the outstanding Shares, has good and valid title to the Shares and has full power and authority to transfer and

deliver the Shares to Purchaser (or a Purchaser Designee) at the Closing, free and clear of all Encumbrances (other than restrictions on transfer imposed by applicable Legal Requirements and Encumbrances that will be released at or prior to the Closing). Upon the Closing, Purchaser (or a Purchaser Designee) shall be the sole record and beneficial owner of all of the outstanding Shares, free and clear of all Encumbrances (other than restrictions on transfer imposed by applicable Legal Requirements). Except pursuant to this Agreement, there is no obligation pursuant to which Seller or any of its Controlled Affiliates have granted any option, warrant or other right to any Person to acquire, receive or vote any Shares.

- (b) The Shares are duly authorized, validly issued, fully paid and nonassessable and owned by Seller, free and clear of all Encumbrances (other than restrictions on transfer imposed by applicable Legal Requirements and Encumbrances that will be released at or prior to the Closing). Except for the Shares, there are no shares of capital stock of or other voting or equity interests in the Company that are issued, reserved for issuance or outstanding. There are no shares of capital stock of or other voting or equity interests in the Company Subsidiary that are issued, reserved for issuance or outstanding that are not directly owned by the Company.
- (c) There are no outstanding warrants, options, rights, agreements, convertible, exercisable or exchangeable securities or outstanding or authorized appreciation, phantom interest, profit participation or similar rights or other commitments (i) pursuant to which any Acquired Company is or may become obligated to issue, deliver, sell, transfer or grant (A) any shares of capital stock of or other voting or equity interests in an Acquired Company or (B) any security convertible into, or exercisable or exchangeable for, any shares of capital stock of or other voting or equity interests in an Acquired Company, (ii) pursuant to which any Acquired Company is or may become obligated to issue, deliver, sell, transfer or grant any such warrant, option, right, unit, security, commitment or undertaking described in the foregoing clause (i) or (iii) that gives any Person the right to receive any benefits or rights similar to any rights enjoyed by or accruing to the holder of the Shares or any shares of capital stock of or other voting or equity interests in any Acquired Company. There are no voting trusts, proxies or other agreements or undertakings with respect to the voting, dividend rights or disposition of the Shares or any shares of capital stock of or other voting or equity interests in any Acquired Company.

4.5 Financial Information; Liabilities.

(a) <u>Schedule 4.5(a)</u> of the Seller Disclosure Schedule sets forth the unaudited balance sheet of the Business (not including Topaz) for the fiscal period ended June 30, 2023 (the "<u>Latest Balance Sheet</u>"), along with the related unaudited statement of income of the Business and the unaudited balance sheets of the Business for the fiscal years ended December 31, 2022 and December 31, 2021, along with the related unaudited statements of income and cash flows of the Business for each of such foregoing periods (collectively, the "<u>Financial Statements</u>"). Subject to the qualifications set forth in <u>Section 4.5(b)</u>, the Financial Statements (i) have been prepared in good faith in accordance with GAAP in all material respects, (ii) present fairly in all material respects the financial condition and results of operations of the Business as of the dates and for the periods therein specified and (iii) have been derived from books and records that are regularly maintained by management of the Acquired Companies in all material respects in accordance with GAAP.

- (b) The Financial Statements are limited by the fact that the Acquired Companies have not operated as separate "stand-alone" entities apart from Seller. Purchaser acknowledges that the Financial Statements may not necessarily be indicative of the conditions that would have existed or the results of operations that would have been achieved if the Acquired Companies had been operated as an unaffiliated company.
- (c) There are no Liabilities of the Business of any nature, whether or not accrued, contingent or otherwise, that would be required to be reflected on a balance sheet or notes thereto of the Business prepared in accordance with GAAP, other than such liabilities or obligations (i) that are specifically reflected on, or specifically reserved against in, the Latest Balance Sheet, (ii) incurred in the ordinary course of business since the date of the Latest Balance Sheet, that would not, individually or in the aggregate, be (or reasonably be expected to be) material to the Acquired Companies, taken as a whole, or prevent or materially delay the consummation of the Transactions, (iii) arising out of, relating to or resulting from the Transactions or the announcement, negotiation, execution or performance of this Agreement or the other Transaction Agreements, (iv) that have been (or will be prior to the Closing) fully discharged or paid off without any ongoing Liability to the Business or any of the Acquired Companies or (v) that otherwise would not reasonably be expected to, individually or in the aggregate, be material to the Acquired Companies, taken as a whole, or prevent or materially delay the consummation of the Transactions.
- (d) The system of internal controls over financial reporting of the Acquired Companies is sufficient in all material respects to provide reasonable assurance (i) that all financial transactions are executed in accordance with management's general or specific authorization, (ii) that financial transactions are recorded as necessary to permit the accurate preparation of financial statements in accordance with GAAP and (iii) regarding prevention or timely detection of unauthorized acquisition, use or disposition of the assets of the Acquired Companies.
- (e) As of the Closing, there shall be no Retained Liabilities of the Business or any Acquired Company of any nature, whether or not accrued, contingent or otherwise.

4.6 <u>Absence of Certain Changes</u>.

- (a) (i) Since December 31, 2022, the Business has been conducted in the ordinary course consistent with past practice in all material respects, except (A) in connection with the Transactions, the negotiation and execution of this Agreement and the other Transaction Agreements and (B) as otherwise contemplated by this Agreement and (ii) since the date of this Agreement, none of the Acquired Companies (or Seller or any of its other Controlled Affiliates, with respect to the Acquired Companies or the Business) has (individually or in the aggregate) taken any action (or refrained from taking any action) that, if such action (or failure to act) were to be taken between date hereof and the Closing Date, would require Purchaser's consent pursuant to clause (i), (iii), (iv), (v), (vi), (vii), (ix), (xi), (xii), (xiii), (xiv), (xv), (xvi), (xviii), (xviii), (xix), (xx), (xx), (xxi) or (xxii) of Section 6.1(a)(2).
- (b) Since December 31, 2022, there has not been any event, change, development or effect which has had or would reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

4.7 <u>Compliance with Legal Requirements</u>. Since January 1, 2021, the Business and the Acquired Companies (and Seller and its other Controlled Affiliates with respect to the Business), (a) have been in compliance with applicable Legal Requirements and (b) have not received any written notice from any Governmental Authority alleging that any Acquired Company or, to the extent related to the Business, Seller or any of its other Controlled Affiliates is in violation of any applicable Legal Requirement, except, in the case of each of the foregoing clauses (a) and (b), for such instances of non-compliance which would not reasonably be expected to, individually or in the aggregate, be material to the Acquired Companies, taken as a whole, or prevent or materially delay the consummation of the Transactions. The Acquired Companies own, hold, possess or lawfully use in the operation of the Business all Permits which are necessary to conduct the Business as conducted as of the date of this Agreement, except as would not reasonably be expected to, individually or in the aggregate, be material to Acquired Companies, taken as a whole, or prevent or materially delay the consummation of the Transactions. Such Permits are valid and in full force and effect.

4.8 <u>Material Contracts</u>.

- (a) <u>Schedule 4.8(a)</u> of the Seller Disclosure Schedule lists, as of the date of this Agreement, all of the following Contracts (including, for the avoidance of doubt, Government Contracts) to which (x) any Acquired Company or (y) to the extent relating to the Business, Seller or any of its other Controlled Affiliates, is a party, or which the properties of any of the foregoing in clauses (x) or (y) (but with respect to clause (y), only to the extent relating to the Business) are bound (other than Benefit Plans) (such Contracts, and each other Contract entered into after the date of this Agreement that would have been required by this <u>Section 4.8(a)</u> to have been included on <u>Schedule 4.8(a)</u> of the Seller Disclosure Schedule if entered into prior to the date of this Agreement, the "<u>Material Contracts</u>"):
 - (1) any Contract with a Key Customer (other than any such Contracts with such Key Customer that are not material or any purchase orders that are consistent with any agreed override terms with such Key Customer set forth on Schedule 4.8(a)(i) of the Seller Disclosure Schedule);
 - (2) any Contract with a Key Vendor (other than any such Contracts with such Key Vendor that are not material or any purchase orders that are consistent with any agreed override terms with such Key Vendor set forth on Schedule 4.8(a)(ii) of the Seller Disclosure Schedule);
 - (3) any Contract that requires (A) the Acquired Companies or (B) to the extent relating to the Business, Seller or any of its other Controlled Affiliates to deal exclusively with a third party in connection with the sale or purchase of any product or service and such Contract involves payments or value in excess of \$35,000,000 and cannot be terminated by the Acquired Companies or Seller or any of its Controlled Affiliates, as applicable, within sixty (60) days' or less notice without penalty or other financial recourse;
 - (4) any Contract that relates to an acquisition or divestiture of the equity, assets or property (for clarity, not including ordinary course commercial arrangements) or business of any Person (whether by merger, sale of stock or other equity,

sale of assets or otherwise) with a purchase price in excess of \$10,000,000 (A) that contains covenants, indemnities or other obligations that remain in effect or (B) if such acquisition or divesture is not yet consummated other than any such acquisition or disposition of assets in the ordinary course of business and which acquisition is not material to the Business;

- (5) any Contract relating to Indebtedness of the Acquired Companies or with respect to the Business in excess of \$10,000,000, except for any Indebtedness that will be discharged at or prior to the Closing without any ongoing Liability to the Business or any Acquired Companies;
- (6) any Contract that creates any Encumbrance (other than any Permitted Encumbrance) upon any Owned Real Property, any Leased Real Property or any material asset of any Acquired Company or the Business;
- (7) any Contract pursuant to which (A) any Acquired Company or (B) to the extent relating to the Business, Seller or any of its other Controlled Affiliates has made or is required to make any advance, loan, extension of credit or capital contribution to, or other investment in, any Person (other than extensions of trade credit given in the ordinary course of business), in each case, in an amount exceeding, individually, \$10,000,000;
- (8) any material joint venture, partnership, collaboration or joint research and development Contract;
- Company or, to the extent relating to the Business, Seller or its Controlled Affiliates, (B) an inbound license is granted to an Acquired Company or, to the extent relating to the Business, Seller or its Controlled Affiliates, in each case of (A) and (B), to any Intellectual Property material to the Business or (C) the Business' ownership of, or the validity or enforceability of, any material Owned IP is otherwise materially affected, (provided, (x) any Contract to grant a license (with or without the occurrence of any condition or other event) shall be deemed to be a "license" for purposes of clauses (A) and (B), and (y) any Contract to (with or without the occurrence of any condition or other event) transfer or assign, or materially affect the validity or enforceability of, any material Owned IP shall be deemed to be a Contract referenced in clause (C)), in each case of (A) through (C), other than Contracts (1) concerning non-exclusive rights to generally commercially available Software, services, hardware or other technology entered into in the ordinary course of business, (2) in which grants of non-exclusive rights to use Intellectual Property are incidental to and not material to performance under the Contract, (3) with customers that are non-exclusive and entered into in the ordinary course of business or (4) relating to development of Owned IP by contractors or employees in the ordinary course of business;
 - (10) any Collective Bargaining Agreement;
 - (11) the Real Property Leases;
- (12) any Contract involving the resolution or settlement of any actual or threatened Proceeding which imposes monetary obligations in excess of \$5,000,000 or any

material non-monetary obligations (other than customary confidentiality obligations) and that either (A) was entered into after January 1, 2021 or (B) has any material continuing obligations on Seller or any of its Controlled Affiliates (including any Acquired Company) to the extent relating to the Business;

- (13) any Contract that requires capital expenditures (including any series of related expenditures) of more than \$10,000,000 by any of the Acquired Companies or the Business;
- (14) any Contract that by its terms limits or restrains (or purports to so limit or restrain) in any material respect any Acquired Company or the Business from competing with any Person in any location or in any business; and
- (15) any Contract that involves annual revenue or payments of greater than \$35,000,000 that (A) contains a "most favored nation" provisions or any similar requirements in favor of any Person that would be applicable to any Acquired Company or the Business from and after the Closing or (B) granting a right of first refusal, right of first negotiation, right of first offer or similar option in favor of any other Person.
- Except as would not, individually or in the aggregate, be or reasonably be expected to be material to the Acquired Companies, taken as a whole, or to prevent or materially delay the consummation of the Transactions, (i) each of the Material Contracts is in full force and effect (other than any expirations at the end of the applicable term in accordance with the terms of any such Material Contract) and enforceable by Seller or its applicable Controlled Affiliate party thereto in accordance with its terms, (ii) there exists no default under any such Material Contracts by Seller or its applicable Controlled Affiliates party thereto or, to Seller's Knowledge and as of the date of this Agreement, any other party to such Material Contracts, (iii) there exists no event or circumstance with respect to Seller or its applicable Controlled Affiliates party thereto or, to Seller's Knowledge and as of the date of this Agreement, any other party to such Material Contracts, that (with notice or lapse of time or both) would create a default under any of the Material Contracts or result in a termination right thereof or would cause or permit the acceleration of or other changes of or to any right or obligation or the loss of any benefit thereunder and (iv) as of the date of this Agreement, there exists no actual or threatened-inwriting termination or cancellation of any Material Contract. As of the date of this Agreement, neither Seller nor any of its Controlled Affiliates has served written notice on (or received written notice from) a counterparty to a Material Contract in respect of a breach of a material nature by such counterparty or Seller or any of its Controlled Affiliates, as applicable. Except as specifically noted in Schedule 4.8(b) of the Seller Disclosure Schedules, Seller has made available to Purchaser a true and correct copy of each Material Contract that is in effect as of the date of this Agreement.
- (c) Set forth on <u>Schedule 1.1(d)</u> of the Seller Disclosure Schedule is a true and complete list of the top fifty-nine (59) customers or programs of the Business, as determined by revenue of the Business for the twelve (12)-month period ended December 31, 2022. Set forth on <u>Schedule 1.1(e)</u> of the Seller Disclosure Schedule is a true and complete list of the top twenty (20) vendors or programs that provide services or products to the Business, determined on the basis of aggregate spend of the Business for the twelve (12)-month period ended December 31, 2022.

- (d) Set forth on <u>Schedule 4.8(d)</u> of the Seller Disclosure Schedule are all open customer Contracts with respect to the business of Topaz.
- (e) The general terms and conditions of purchase set forth on <u>Schedule 4.8(e)</u> of the Seller Disclosure Schedule are the general terms and conditions of purchase that govern the Contracts set forth on <u>Schedule 4.2(b)</u> of the Seller Disclosure Schedule for items (14), (15), (16), (18), (23), (24), (25), (26) and (31) and <u>Schedule 4.8(a)(i)</u> of the Seller Disclosure Schedule for items (25), (26), (27), (28), (29), (30), (31), (32) and (33) in lieu of the standard terms and conditions referenced in such Contracts.
- 4.9 <u>Litigation</u>. (a) There is no Proceeding pending (and since January 1, 2021 there has not been any Proceeding pending) or, to Seller's Knowledge, threatened in writing, against or affecting the Acquired Companies or the Business or, to the extent related to the Business, Seller or any of its other Controlled Affiliates and (b) none of the Acquired Companies, the Business or, to the extent related to the Business, Seller or any of its other Controlled Affiliates is (or has been since January 1, 2021) subject to any Orders, in the case of each of the foregoing clauses (a) and (b) that would reasonably be expected to, individually or in the aggregate, be material to the Acquired Companies, taken as a whole, or prevent or materially delay the consummation of the Transactions.
- 4.10 <u>Insurance</u>. Schedule 4.10 of the Seller Disclosure Schedule contains a true and complete list as of the date hereof of all material insurance policies covering the Business or the Acquired Companies (the "Listed Insurance Policies"), whether such policies have been issued to Seller, the Acquired Company or any of their respective Controlled Affiliates. The Listed Insurance Policies are in amounts and cover such risks as are reasonably adequate for the Business and the Acquired Companies, taken as a whole, and sufficient for compliance in all material respects by the Acquired Companies or, to the extent related to the Business, Seller or any of its other Controlled Affiliates with all Materials Contracts. All Listed Insurance Policies are in full force and effect (except for ordinary course terminations after the date of this Agreement in which substitute or replacement insurance is obtained therefor), all premiums due thereunder have been paid in full and Seller is not in default with respect to any other obligations thereunder, and no written notice of cancellation, non-renewal, termination or material limitation of coverage, in whole or in part, with respect to any such Listed Insurance Policy currently in force, has been received by Seller. As of the date hereof, no event has occurred that would reasonably be expected to result in the cancellation, non-renewal, termination or material limitation of coverage under any such Listed Insurance Policy. As of the date hereof, there are no material claims pending under any of the Listed Insurance Policies with respect to the Business or the Acquired Companies as to which the respective insurer has denied, questioned or disputed coverage or reserved rights.

4.11 <u>Intellectual Property</u>.

(a) Except as otherwise restricted by an Order by the U.S. Patent and Trademark Office's Commissioner of Patents, Schedule 4.11(a) of the Seller Disclosure Schedule sets forth a true and complete list, as of the date of this Agreement, of all (i) issued Patents and Patent applications, (ii) Trademark registrations and Trademark applications, (iii) Copyright registrations and Copyright applications and (iv) domain name registrations, in each case, that constitute Owned IP (such Intellectual Property, the "Company Registered IP"). Except for (A)

the Seller Transitional Trademarks and (B) Intellectual Property associated with services to be provided to the Acquired Companies pursuant to the Transition Services Agreement, neither Seller nor any of its Controlled Affiliates (other than the Acquired Companies) own any right, title, or interest in any Intellectual Property used in or necessary for the operation of the Business as of the date hereof or as of the Closing Date.

- (b) All Company Registered IP is subsisting, and other than applications for registration, to Seller's Knowledge, is valid and enforceable. As of the Closing Date, an Acquired Company will solely and exclusively own and possess all right, title and interest in and to each item of material Owned IP, free of all Encumbrances other than Permitted Encumbrances.
- (c) Except as would not reasonably be expected to be materially adverse to the Business, all Persons who have invented or developed (in whole or part) any material Owned IP for or on behalf of the Business have executed a written assignment to an Acquired Company of exclusive ownership of all Intellectual Property therein, or exclusive ownership in such Owned IP automatically vested in an Acquired Company by operation of law.
- (d) Except as would not reasonably be expected to be materially adverse to the Business, (i) no funding, personnel, or facilities of any Governmental Authority or any university or other educational institution were used, directly or indirectly, to develop or create any Owned IP and (ii) no Governmental Authority or any university or other educational institution has any ownership rights, or other rights other than Permitted Encumbrances, in or to any such Intellectual Property.
- (e) Since January 1, 2021, there has been no Proceeding pending or threatened in writing, against any Acquired Company or Seller or its Controlled Affiliates with respect to the Business (i) alleging that the conduct of the Business is infringing, misappropriating or otherwise violating in any material respect any Person's Intellectual Property or (ii) challenging any Acquired Company's ownership, or the validity or enforceability, of any material Owned IP. Except as would not reasonably be expected to be materially adverse to the Business, the conduct of the Business has not, since January 1, 2021, infringed, misappropriated or otherwise violated the Intellectual Property of any Person.
- (f) No material Proceeding is pending or threatened in writing, by any Acquired Company or Seller or its Controlled Affiliates, alleging that any Person is infringing, misappropriating or otherwise violating any material Owned IP. To Seller's Knowledge, no Person is infringing, misappropriating or otherwise violating any Owned IP in any material respect.
- (g) The Acquired Companies have taken commercially reasonable measures to maintain in confidence all Trade Secrets in their possession or control that are material to the operation of the Business. Since January 1, 2021, there has been no unauthorized access or use of any such Trade Secrets, except as would not reasonably be expected to be materially adverse to the Business.
- (h) Since January 1, 2021, (i) the conduct of the Business has been in compliance in all material respects with Data Protection Laws, (ii) the Business has not experienced any material Cybersecurity Incident and (iii) no Proceeding has been pending or

threatened in writing against the Business alleging a material violation of any Person's privacy or Personal Information.

- (i) None of the material Software included in the Owned IP is linked with any Open Source Software and used in a manner that would (i) require its disclosure to any Person in source code form, (ii) require the licensing thereof for the purpose of making derivative works or (iii) require the licensing thereof at no or minimal charge.
- (j) Except as has not been or as would not reasonably be expected to be materially adverse to the Business, (i) since January 1, 2021, there have been no failures of the Business Systems and (ii) the Business Systems are free from Malicious Code. The Acquired Companies have taken commercially reasonable steps and implemented reasonable procedures designed to (i) ensure that the Business Systems are free from any Malicious Code and (ii) protect the security and integrity of the Business Systems and the data hosted or processed thereby, including from Malicious Code.

4.12 Real Property.

- (i)Schedule 4.12(a)(i) of the Seller Disclosure Schedule sets forth a true and complete list of all real property owned by an Acquired Company as of the date of this Agreement and (ii)Schedule 4.12(a) (ii) of the Seller Disclosure Schedule sets forth a true and complete list of all real property which, after the consummation of the transactions contemplated by the Real Estate Reorganization Plan, will be owned by an Acquired Company ((i) and (ii) collectively, and together with all buildings, structures, improvements and fixtures located thereon, the "Owned Real Property"). Each applicable Acquired Company has (or at the Closing, will have) good and valid fee title to the applicable Owned Real Property, free and clear of any and all Encumbrances except Permitted Encumbrances. Except as would not be (or reasonably be expected to be), individually or in the aggregate, materially adverse to the Business and the Acquired Companies, taken as a whole, or except as set forth on Schedule 4.12(a) of the Seller Disclosure Schedule, (i) except for the Permitted Encumbrances, neither the Acquired Company, nor to the extent related to the Business, Seller or any of its Controlled Affiliates, has leased or otherwise granted to any Person (other than any Acquired Company) any outstanding right to use or occupy any Owned Real Property or any material portion thereof, (ii) other than the right of Purchaser pursuant to this Agreement, there are no outstanding options, rights of first offer or rights of first refusal to purchase any Owned Real Property or any material portion thereof or interest therein and (iii) no Acquired Company is a party to any outstanding Contract or option to purchase any real property or interest therein.
- (b) (i)Schedule 4.12(b)(i) of the Seller Disclosure Schedule sets forth a true and complete list of all real property leases and subleases under which an Acquired Company is a lessee or sublessee and that are, in each case, in effect as of the date of this Agreement and (ii)Schedule 4.12(b)(ii) of the Seller Disclosure Schedule sets forth a true and complete list of all real property leases and subleases under which, after the consummation of the transactions contemplated by the Real Estate Reorganization Plan, an Acquired Company will be a lessee or sublessee (any such lease or sublease in (i) and (ii), a "Real Property Lease," and such properties, the "Leased Real Property"). Except as would not be (or reasonably be expected to be), individually or in the aggregate, materially adverse to the Business and the Acquired Companies,

taken as a whole, or as set forth on <u>Schedule 4.12(b)</u> of the Seller Disclosure Schedule, (i) the applicable Acquired Company's possession and quiet enjoyment of the Leased Real Property under such Real Property Lease has not been disturbed, (ii) no Acquired Company has subleased, licensed or otherwise granted any Person (other than an Acquired Company) the right to use or occupy any Leased Real Property or any material portion thereof and (iii) no Acquired Company has collaterally assigned or granted any security interest in any Real Property Lease or any interest therein, except in favor of the landlord pursuant to the terms of the applicable Real Property Lease or applicable Legal Requirements.

4.13 Labor Matters.

- (a) Schedule 4.13(a) of the Seller Disclosure Schedule sets forth a true and complete list, as of the most recent payroll date preceding the date of this Agreement, of each Direct Employee, and for each: (i) identification number, (ii) job title, (iii) work location (by state), (iv) hire date, (v) annual base salary or hourly wage rate (as applicable), (vi) exempt or non-exempt classification (as applicable), (vii) active or inactive status, (viii) full-time or part-time status and (ix) employing entity. The Company Employees are sufficient in number and skill to operate the Business in substantially the same manner as it was operated immediately prior to the Closing. Other than the Company Employees, neither Seller nor any of its Controlled Affiliates (other than the Acquired Companies) employ or otherwise engage any individual who primarily devotes his or her working time to performing services on behalf of the Business.
- (b) There are no Collective Bargaining Agreements to which an Acquired Company is a party or bound by or to which Seller or any of its Controlled Affiliates is a party or bound by and that cover any Company Employee. With respect to Company Employees and Former Company Employees, since January 1, 2021, there has been no material labor grievance, material labor arbitration, strike, lockout, slowdown, work stoppage, picketing, hand billing, unfair labor practice charge or material labor dispute pending or, to Seller's Knowledge, threatened. Since January 1, 2021, no activities or proceedings of any labor union to organize any Company Employees or Former Company Employees have been pending or, to Seller's Knowledge, threatened. No labor union or works council represents any Company Employees in connection with their employment with Seller or any of its Controlled Affiliates (including the Acquired Companies).
- (c) Each of the Acquired Companies and Seller and its Controlled Affiliates (with respect to the Business, Company Employees and Former Company Employees) is, and since January 1, 2021 has been, in compliance in all material respects with all applicable Legal Requirements regarding labor, employment and employment practices.
- (d) To Seller's Knowledge, since January 1, 2021, the Acquired Companies and, to the extent related to the Business, Seller and its Controlled Affiliates have thoroughly and impartially investigated all sexual harassment or other harassment, discrimination, retaliation or policy violation allegations against officers, directors or managerial or supervisory-level employees. With respect to each such allegation (except those that the applicable Acquired Company or Seller or its Controlled Affiliates (as applicable) reasonably deemed to not have merit), such Acquired Company or Seller or its Controlled Affiliates (as applicable) has taken prompt corrective action reasonably calculated to prevent further improper action. There are no such

allegations of harassment or discrimination pending or, to Seller's Knowledge, threatened, that, if known to the public, would bring the Acquired Companies or the Business into material disrepute.

4.14 Employee Benefits.

- (a) Schedule 4.14(a) of the Seller Disclosure Schedule sets forth a true and complete list, as of the date of this Agreement, of each material Benefit Plan, and separately identifies each Acquired Company Benefit Plan. Each Acquired Company Benefit Plan is exclusive to the Business and does not cover employees of Seller or any of its Affiliates other than the Company Employees and Former Company Employees. With respect to each Acquired Company Benefit Plan, Seller has made available to Purchaser true and complete copies, of, as applicable, the current plan documents and summary plan descriptions and all amendments thereto, trust agreements, insurance contracts or other funding vehicles, in the case of any plan intended to be qualified under Section 401(a) of the Code, the most recent determination or opinion letter from the IRS, the most recently filed Form 5500, and any non-routine correspondence with a Governmental Authority since January 1, 2021. With respect to each Seller Benefit Plan, Seller has made available to Purchaser the plan document for, or a summary of the material terms of, such Seller Benefit Plan.
- (b) Except as would not, individually or in the aggregate, reasonably be expected to be material to the Business and the Acquired Companies, taken as a whole, each Benefit Plan has been administered, funded and operated in compliance in form and operation with applicable Legal Requirements and in accordance with its terms. No action, claim, proceeding, audit or investigation is pending or, to Seller's Knowledge, threatened with respect to any Benefit Plan (other than routine claims for benefits payable in the ordinary course, and appeals of denied claims).
- (c) Each Benefit Plan that is intended to be qualified under Code Section 401(a) is the subject of a favorable determination or opinion letter from the IRS with respect to its tax-qualified status, and to Seller's Knowledge nothing has occurred that could reasonably be expected to adversely affect such qualified status. No Acquired Company has incurred (whether or not assessed), or is reasonably expected to incur or to be subject to, any material Tax or other penalty with respect to the reporting requirements under Sections 6055 and 6056 of the Code, as applicable, or under Section 4980B, 4980D or 4980H of the Code that has not been satisfied in full.
- (d) Except as otherwise provided in this Agreement, neither the execution of this Agreement nor the consummation of the Transactions will (whether alone or together with any other event) (i) result in any payment or benefit becoming due to any Company Employee, Former Company Employee or other individual service provider of an Acquired Company (or any dependent or beneficiary thereof), (ii) increase any payment or benefit to be paid or provided to any such Person described in clause (i), (iii) result in any acceleration of the time of payment, funding (through a grantor trust or otherwise) or vesting of any payments or benefits to any such Person described in clause (i) or (iv) result in any payments or benefits under any agreement, plan or arrangement with Seller or any of its Controlled Affiliates (including the Acquired Companies) that, individually or in combination with any other payment or benefit, would constitute the payment of any "excess parachute payment" within the meaning of Section 280G of the Code or in the imposition of an excise Tax under Section 4999 of the Code.

- (e) Except for the Business Pension Plan, no Acquired Company Benefit Plan is, and none of the Acquired Companies sponsors, maintains, contributes to or is obligated to contribute to, or has a liability (contingent or otherwise) with respect to, and no liability has been incurred by an ERISA Affiliate of an Acquired Company with respect to, (i) a "defined benefit plan" (as defined in Section 3(35) of ERISA) or any other plan that is or was subject to Section 302 or Title IV of ERISA, (each, a "Pension Plan"), (ii) a "multiemployer plan" (as defined in Section 3(37) of ERISA), (iii) a "multiple employer plan" (within the meaning of Section 413 of the Code), (iv) a "multiple employer welfare arrangement" within the meaning of Section 3(40) of ERISA or (v) a plan or arrangement that provides or promises to provide post-retirement or post-termination health or life insurance or other similar benefits (other than health continuation coverage required by Part 6 of Subtitle B of Title I of ERISA or Section 4980B of the Code for which the covered Person pays the full cost of coverage).
- With respect to each Pension Plan, (i) no reportable event (within the meaning of Section 4043 of ERISA, other than an event for which the reporting requirements have been waived by regulations) has occurred or is reasonably expected to occur with respect to the Business Pension Plan, or, to the extent related to the Business, Seller or any of its other Controlled Affiliates, any other Pension Plan, except as would not reasonably be expected to result in material Liability to an Acquired Company, (ii) no Acquired Company or, except as would not reasonably be expected to result in material Liability to an Acquired Company, any ERISA Affiliate has failed to make any contributions required under Sections 412 and 430 of the Code and Section 302 of ERISA on a timely basis, and no lien on the assets of any Acquired Company has arisen under ERISA or Section 430(k) of the Code, (iii) there have been no violations of the applicable benefits restrictions under Section 436 of the Code with respect to the Business Pension Plan or, to the extent related to the Business, Seller or any of its other Controlled Affiliates, any other Pension Plan, except as would not reasonably be expected to result in material Liability to an Acquired Company, (iv) all premiums (and interest charges and penalties for late payment, if applicable) have been paid when due to the Pension Benefit Guaranty Corporation with respect to the Business Pension Plan or to the extent related to the Business, Seller or any of its other Controlled Affiliates, any other Pension Plan, except as would not reasonably be expected to result in material Liability to an Acquired Company, (v) no liability has been incurred under Section 4062(e) of ERISA with respect to the Business Pension Plan or, to the extent related to the Business, Seller or any of its other Controlled Affiliates, any other Pension Plan, except as would not reasonably be expected to result in material Liability to an Acquired Company and (vi) timely notice required under Section 204(h) of ERISA was delivered to participants and beneficiaries affected by a Business Pension Plan amendment or, to the extent related to the Business, Seller or any of its other Controlled Affiliates, any other Pension Plan amendment, resulting in a cessation or significant reduction of the rate of future benefit accruals, except as would not reasonably be expected to result in material Liability to an Acquired Company.
- (g) Each Benefit Plan that constitutes in any part a "nonqualified deferred compensation plan" (as defined under Section 409A(d)(1) of the Code) subject to Section 409A of the Code has been operated and administered in all material respects in operational compliance with, and is in all material respects in documentary compliance with, Section 409A of the Code, and no amount under any such Benefit Plan has been or is reasonably expected to be subject to any interest or additional Taxes imposed under Section 409A of the Code.

- (a) Each of the Acquired Companies has timely filed, or has caused to be timely filed on its behalf, all material Tax Returns required to be filed by it (taking into account any extensions of time in which to file), and all such Tax Returns are true, correct and complete in all material respects. All material amounts of Taxes required to be paid by or on behalf of the Acquired Companies (whether or not shown on any such Tax Return) have been timely paid in full. Each of the Acquired Companies, other than Topaz, is, and has been since its formation, properly classified as a corporation for U.S. federal (and applicable state and local) income tax purposes. Topaz is, and has been since its formation, properly classified as an entity disregarded as separate from (i) with respect to periods prior to the consummation of the Topaz Reorganization, Ball Packaging, LLC and (ii) with respect to periods following the consummation of the Topaz Reorganization, the Company Subsidiary, for U.S. federal (and applicable state and local) income tax purposes.
- (b) No claim has been made in writing by a Tax Authority in a jurisdiction where any Acquired Company does not file Tax Returns that such Acquired Company is or may be subject to taxation by or required to file a Tax Return in that jurisdiction, which claim has not been satisfied, withdrawn, settled or otherwise resolved. No Acquired Company is subject to taxation in any country, other than the country in which it was organized, by virtue of having a permanent establishment (within the meaning of an applicable income tax treaty) or other fixed place of business in such other country.
- (c) There is no pending or ongoing material dispute, audit or Proceeding concerning any material Tax liability of any Acquired Company, and no such dispute, audit or Proceeding has been threatened, claimed or raised by any Tax Authority in writing. No Acquired Company has received any written notice of a proposed adjustment, deficiency or underpayment with respect to any material Taxes, which adjustment, deficiency or underpayment has not been satisfied by payment or otherwise resolved with no further liability to any Acquired Company.
- (d) There are no outstanding Encumbrances for material Taxes other than for Taxes not yet due and payable on the assets of any Acquired Company.
- (e) Each of the Acquired Companies has (i) withheld and timely paid to the appropriate Tax Authority all material amounts of Taxes required to have been withheld and paid in connection with any amounts paid or owing to any employee, independent contractor, client, creditor, customer or other Person, (ii) remitted to the applicable Governmental Authority all material amounts required to be remitted pursuant to any applicable escheat or unclaimed property Legal Requirements and (iii) complied in all material respects with all related reporting and record-keeping requirements in respect of the matters described in the foregoing clauses (i) and (ii).
- (f) No Acquired Company has waived or extended any statute of limitations relating to any material Tax or material Tax Return, which waiver or extension is currently in effect, and no extension of time within which to file any material Tax Return of any Acquired Company is currently in effect, in each case, other than an automatic extension of the time to file an Income Tax Return.

- (g) No Acquired Company (i) is a party to, or bound by, any Tax allocation, indemnification or sharing Contract, other than pursuant to a Commercial Tax Agreement, (ii) has ever been a member of an affiliated group filing a consolidated U.S. federal income Tax Return (or any corresponding group under state, local or foreign Legal Requirements), other than such a group the common parent of which is Seller, (iii) has any liability for the Taxes of any Person, other than Seller or another Acquired Company, under Treasury Regulations Section 1.1502-6 (or any corresponding provision of state, local or foreign Legal Requirements), as a transferee or successor, by assumption, operation of any Legal Requirement or otherwise (other than pursuant to any Commercial Tax Agreement) or (iv) has participated in any "listed transaction" as described in Treasury Regulations Section 1.6011-4(b)(2) (or any corresponding transaction under state, local or foreign Legal Requirements).
- (h) No Acquired Company will be required to include any material item of income in, or exclude any material item of deduction from, taxable income for any taxable period (or portion thereof) ending after the Closing Date as a result of any: (i) change in method of accounting, or use of an improper method of accounting, with respect to a Pre-Closing Tax Period, (ii) "closing agreement" as described in Section 7121 of the Code (or any corresponding agreement under state, local or foreign Legal Requirements) executed prior to the Closing, (iii) intercompany transaction or excess loss account described in the Treasury Regulations under Section 1502 of the Code (or any corresponding transaction or account under state, local or foreign Legal Requirements) entered into or created prior to the Closing, (iv) installment sale or open transaction disposition made prior to the Closing or (v) prepaid amount received or deferred revenue accrued on or prior to the Closing Date.
- (i) No Acquired Company has distributed the stock of another Person or has had its stock distributed by another Person in the last two (2) years in a transaction that was purported or intended to be governed in whole or in part by Section 355 (or so much of Section 356 of the Code as relates to Section 355 of the Code) or Section 361 of the Code (or any corresponding provisions of state, local or foreign Legal Requirements).
- 4.16 <u>Sufficiency of Assets</u>. The assets, rights, properties and interests owned, leased or licensed by the Acquired Companies as of the Closing, together with the services to be provided by Seller to the Acquired Companies pursuant to the Transition Services Agreement, are sufficient for, and constitute all of the assets, rights, properties and interests used in or are necessary for, the conduct of the Business immediately following the Closing in substantially the same manner in all material respects as conducted as of the date hereof and as of immediately prior to the Closing. Nothing in this <u>Section 4.16</u> shall be deemed to expand the scope of any other representations or warranties made by Seller in this <u>Article IV</u>. For the avoidance of doubt, the foregoing representation shall not be deemed to be a representation as to non-infringement.
- 4.17 <u>Environmental Matters</u>. Except as would not, individually or in the aggregate, reasonably be expected to be material to the Acquired Companies, taken as a whole, or to prevent or materially delay the consummation of the Transaction, (a) since January 1, 2021, the Business and each of the Acquired Companies (and with respect to the Business, each of Seller and its other Controlled Affiliates) has been in compliance with applicable Environmental Laws, which compliance has included obtaining, maintaining, and complying with all Permits that are required pursuant to Environmental Laws; (b) since January 1, 2021, neither the Business nor any Acquired

Company (nor, with respect to the Business, Seller or any of its other Controlled Affiliates) has received any written notice, report, order, or directive and there are no Proceedings pending or, to Seller's Knowledge, threatened, against the Business, in each case alleging that the Business or any Acquired Company (or, with respect to the Business, Seller or any of its other Controlled Affiliates) is in violation of or liable under any Environmental Law, excluding any such notices, reports, orders, directives or Proceedings that have been fully and finally resolved with no further liability or obligation and (c) neither the Business nor the Acquired Companies nor, to Seller's Knowledge, any other Person, to the extent reasonably expected to result in material liability of any Acquired Company (nor, with respect to the Business, Seller or any of its other Controlled Affiliates) has treated, stored, disposed of, arranged for the disposal of, transported, manufactured, distributed, released, or exposed any Person to, and none of the facilities or real properties currently or formerly owned or operated by the Business or any Acquired Company (or, with respect to the Business, Seller or any of its other Controlled Affiliates) is or has been impacted by the release or disposal of, Hazardous Materials, in each case, as would be reasonably likely to give rise to any liabilities pursuant to any Environmental Laws.

4.18 Certain Business Practices.

- (a) Since January 1, 2018, none of the Acquired Companies or, to the extent related to the Business, Seller or any of its other Controlled Affiliates (including, to the extent related to the Business, any of their respective officers, directors, employees or, to Seller's Knowledge, agents) has used any corporate funds for any unlawful contribution, gift or entertainment or other unlawful expenses relating to political activity, made any unlawful payment to any employee of a Governmental Authority, or made any unlawful bribe, rebate, payoff, influence payment or kickback or other unlawful payment, in each case in material violation of any Anti-Corruption Law.
- (b) Since January 1, 2018, the Acquired Companies and, to the extent related to the Business, Seller and each of its other Controlled Affiliates have (i) kept books, records and accounts, which, in reasonable detail, accurately and fairly reflect their respective transactions and dispositions of their respective assets and (ii) maintained (or had maintained on their behalf) a system of internal accounting controls, in each case in material compliance with applicable Anti-Corruption Laws.
- (c) Since January 1, 2018, none of the Acquired Companies or, to the extent related to the Business, Seller or any of its other Controlled Affiliates (including, to the extent related to the Business, any of their respective officers, directors, employees or, to Seller's Knowledge, agents) have transacted business with or for the benefit of any Sanctioned Person or in any Sanctioned Territory in violation of applicable Sanctions Laws or otherwise violated any applicable Customs & Trade Laws in any material respect and applicable Sanctions Laws
- (d) As of the date of this Agreement, none of the Acquired Companies or, to the extent related to the Business, Seller and each of its other Controlled Affiliates, nor any of their respective officers, directors, employees or, to Seller's Knowledge, agents, is a Sanctioned Person.
- (e) Since January 1, 2018, no Acquired Company or, to the extent related to the Business, Seller or any of its other Controlled Affiliates, has received any allegation, inquiry,

notice or communication from any Governmental Authority that alleges that any Acquired Company or, to the extent related to the Business, Seller or any of its other Controlled Affiliates (including, to the extent related to the Business, any of their respective officers, directors, employees or, to Seller's Knowledge, agents) may have violated in any material respect, nor has made any material voluntary or directed disclosure or prior disclosure related to, any Anti-Corruption Laws, Customs & Trade Laws or Sanctions Laws.

4.19 Government Contracts.

- (a) Each Government Contract that is also a Material Contract for which the period of performance has not expired or terminated, final payment has not been received or which remains open to final audit (each, a "Material Government Contract") was legally awarded. As of the date of this Agreement, none of the Acquired Companies or, to the extent related to the Business, Seller or any of its other Controlled Affiliates have received written notice that any Material Government Contract or proposal for a Material Government Contract is the subject of any award protest proceedings.
- (b) Since January 1, 2021: (A) no Governmental Authority nor any prime contractor or higher-tier subcontractor under any Material Government Contract has notified any of the Acquired Companies or, to the extent related to the Business, Seller or any of its other Controlled Affiliates in writing of any actual or alleged violation or breach of any Legal Requirement or contract term that could be reasonably expected to adversely and materially affect the collectability of any receivable or the award of any Material Government Contract in the future, (B) none of the Acquired Companies or, to the extent related to the Business, Seller or any of its other Controlled Affiliates has received a written cure notice, show cause notice, stop work order or deficiency notice relating to any the Material Government Contract that could be reasonably expected to adversely and materially affect the collectability of any receivable or the award of any Material Government Contracts in the future and (C) no Material Government Contract awarded to any Acquired Company or, to the extent related to the Business, Seller or any of its other Controlled Affiliates has been terminated for default, cause or otherwise for failure to perform and no such Acquired Company or, to the extent related to the Business, Seller or any of its other Controlled Affiliates has been threatened in writing with termination for default or cause that remains unresolved with respect to any Material Government Contract.
- (c) Since January 1, 2021, none of the Acquired Companies (or, to the extent related to the Business, Seller or any of its other Controlled Affiliates) or any of their respective Principals (as defined in Federal Acquisition Regulation 2.101 and 52.209-5) have been debarred, suspended or proposed for suspension or debarment or otherwise excluded from participation in the award of any Government Contract.
- (d) Since January 1, 2021, except as would not reasonably be expected to, individually or in the aggregate, be material to the Business or the Acquired Companies, taken as a whole, or to prevent or materially delay the consummation of the Transactions: (i) all pricing discounts have been properly reported to and credited to the customer under any Government Contract; (ii) none of the Acquired Companies or, to the extent related to the Business, Seller or any of its other Controlled Affiliates has received any notice of any interruption or material decrease in the purchasing of products or services under any Government Contract; (iii) all

invoices and claims submitted for payment, reimbursement or adjustment submitted by each of the Acquired Companies or, to the extent related to the Business, Seller or any of its other Controlled Affiliates in connection with any Government Contract were current, accurate and complete in all material respects as of their respective submission dates; (v) there are no outstanding or unsettled allegations of fraud, false claims or overpayments nor any related investigations or audits by any Governmental Authority; (vi) all representations, certifications and statements executed, acknowledged or submitted by or on behalf of each of the Acquired Companies (or, to the extent related to the Business, Seller or any of its other Controlled Affiliates) to a Governmental Authority or any other Person in connection with any Government Contract were current, accurate and complete in all material respects as of their respective effective dates and each of the Acquired Companies and, to the extent related to the Business, Seller and its other Controlled Affiliates has provided any reasonably required updates to such representations, certifications and statements; (vii) each of the Acquired Companies has maintained adequate systems of internal controls appropriate for the operations of the Business that are in compliance in all material respects with all relevant and applicable requirements of the Government Contracts; (viii) there are no outstanding material claims or disputes with the Acquired Companies (or, to the extent related to the Business, Seller or any of its other Controlled Affiliates) arising under or relating to any Government Contract; (ix) none of the Acquired Companies (or, to the extent related to the Business, Seller or any of its other Controlled Affiliates), nor any of their respective officers, senior management or employees (to the extent related to the Business), has been under or subject to any material administrative, civil or criminal investigation, indictment, information lawsuit, subpoena, document request, administrative proceeding, or audit pertaining to an alleged or potential violation of any requirement, regulation or Legal Requirement applicable to any Government Contract; (x) other than in the ordinary course of business, none of the Acquired Companies (or, to the extent related to the Business, Seller or any of its other Controlled Affiliates) has conducted or initiated any internal investigation, made a voluntary disclosure or been under any obligation to disclose to any Governmental Authority, or any other Person, any alleged or potential irregularity, misstatement or omission arising under or relating to any Government Contract and (xi) each of the Acquired Companies (and, to the extent related to the Business, Seller and its other Controlled Affiliates) has complied in all material respects with the Legal Requirements for safeguarding covered defense information and cyber incident reporting.

certified cost or pricing data submitted in connection with each has been accurate, complete and current; (iv) all

- (e) Each of the Acquired Companies (and, to the extent related to the Business, Seller and its other Controlled Affiliates) has taken all necessary steps to preserve and protect, in all material respects, their rights in and title to all material Owned IP delivered, deliverable or otherwise provided directly or indirectly through any other Person to any Governmental Authority in connection with any Government Contract.
- 4.20 <u>Brokers</u>. Other than with respect to fees or commissions that will be borne solely by Seller and its Affiliates (other than the Acquired Companies), neither Seller nor any Acquired Company has retained any broker or finder or incurred any liability or obligation for any brokerage fees, commissions, finders', financial advisor or other similar fees with respect to this Agreement or the Transactions.
- 4.21 <u>Related Party Transactions</u>. Except as set forth on <u>Schedule 4.21</u> of the Seller Disclosure Schedule, and other than Contracts, Benefits Plans or policies for, or otherwise in

connection with, employment and benefits provided to employees and other individual service providers in the ordinary course of business consistent with past practice, no officer, director, manager or employee of Seller or its Controlled Affiliates (including the Acquired Companies): (a) has entered into any financial transaction with or is a party to any Contract with any Acquired Company; (b) has any right, title, or interest in or to, or uses, holds for use, or licenses, any of the material assets or properties used in the Business, whether tangible or intangible (including any Intellectual Property, but excluding the Seller Transitional Trademarks); or (c) provides or causes to be provided to the Business any of the material assets, properties, services or facilities used in the Business (other than those that will continue to be provided under the Transition Services Agreement), in each case of (a), (b) and (c) that is material to the Business.

- 4.22 <u>Intercompany Arrangements</u>. Other than Contracts to provide the services that are to be provided in accordance with the Transition Services Agreement, <u>Schedule 4.22</u> of the Seller Disclosure Schedule (which may be updated by Seller within thirty (30) days from the date of this Agreement, but only for items and matters existing as of the date of this Agreement) sets forth a true and complete list, as of the date hereof, of (a) all Contracts between or among any Acquired Companies, on the one hand, Seller or any of its Controlled Affiliates (other than the Acquired Company), on the other hand and (b) all Contracts relating to (i) any guaranty by any Acquired Company of any obligation of Seller or any of its Controlled Affiliates (other than the Acquired Companies) and (ii) any guarantee by Seller or any of its Controlled Affiliates (other than the Acquired Companies) of any obligation of a Acquired Company, in each case including any guarantees of borrowed money, in each case of (a) and (b) that is material to the Business.
- 4.23 Disclaimer of Seller. EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES EXPRESSLY MADE BY SELLER IN THIS ARTICLE IV (AND, AS OF THE CLOSING, THE CERTIFICATE TO BE DELIVERED PURSUANT TO SECTION 10.1(c) WITH RESPECT TO REPRESENTATIONS AND WARRANTIES), NONE OF SELLER, ITS AFFILIATES (INCLUDING THE ACQUIRED COMPANIES), OR ANY OF THEIR RESPECTIVE REPRESENTATIVES OR ANY OTHER PERSON MAKES, HAS MADE, SHALL BE DEEMED TO HAVE MADE, OR HAS BEEN AUTHORIZED TO MAKE ANY EXPRESS OR IMPLIED REPRESENTATION OR WARRANTY, AT LAW OR IN EQUITY, WHETHER WRITTEN OR ORAL, STATUTORY OR OTHERWISE, ON BEHALF OF OR WITH RESPECT TO SELLER, ITS AFFILIATES (INCLUDING THE ACQUIRED COMPANIES), OR ANY OF THEIR RESPECTIVE REPRESENTATIVES OR ANY OTHER PERSON, THEIR BUSINESSES (INCLUDING THE BUSINESS), OPERATIONS, ASSETS, LIABILITIES, FINANCIAL CONDITION, RESULTS OF OPERATIONS, FUTURE OPERATING OR FINANCIAL RESULTS, ESTIMATES, PROJECTIONS, FORECASTS, PLANS OR PROSPECTS (INCLUDING THE REASONABLENESS OF THE ASSUMPTIONS UNDERLYING SUCH ESTIMATES, PROJECTIONS, FORECASTS, PLANS OR PROSPECTS), THIS AGREEMENT, THE TRANSACTIONS, OR THE ACCURACY OR COMPLETENESS OF ANY INFORMATION REGARDING SELLER OR ITS AFFILIATES (INCLUDING THE ACQUIRED COMPANIES), FURNISHED OR MADE AVAILABLE TO (OR OTHERWISE ACQUIRED BY) PURCHASER, ITS AFFILIATES OR ITS OR THEIR RESPECTIVE REPRESENTATIVES OR ANY OTHER PERSON (INCLUDING ANY INFORMATION, DOCUMENTS OR MATERIALS MADE AVAILABLE TO PURCHASER, ITS AFFILIATES OR ITS OR THEIR RESPECTIVE REPRESENTATIVES OR ANY OTHER PERSON IN THE DATA ROOM OR OTHERWISE, IN A CONFIDENTIAL INFORMATION

MEMORANDUM OR ANY MANAGEMENT PRESENTATIONS OR IN ANY OTHER FORM), INCLUDING WITH RESPECT TO ANY ERRORS THEREIN OR OMISSIONS THEREFROM, OR AS TO THE FUTURE REVENUE, PROFITABILITY OR SUCCESS OF SELLER, ITS AFFILIATES (INCLUDING THE ACQUIRED COMPANIES) OR THE BUSINESS (INCLUDING THE FINANCIAL INFORMATION, PROJECTIONS OR OTHER FORWARD-LOOKING STATEMENTS OF SELLER, ITS AFFILIATES (INCLUDING THE ACQUIRED COMPANIES) OR THE BUSINESS, IN EACH CASE, IN EXPECTATION OR FURTHERANCE OF THE TRANSACTIONS), AND SELLER HEREBY DISCLAIMS AND SHALL HAVE NO LIABILITY FOR ANY AND ALL SUCH REPRESENTATIONS OR WARRANTIES NOT EXPRESSLY SET FORTH IN THIS ARTICLE IV (AND, AS OF THE CLOSING, THE CERTIFICATE TO BE DELIVERED PURSUANT TO SECTION 10.1(c) WITH RESPECT TO REPRESENTATIONS AND WARRANTIES). WITHOUT LIMITING THE REPRESENTATIONS OR WARRANTIES EXPRESSLY SET FORTH IN THIS ARTICLE IV (AND, AS OF THE CLOSING, THE CERTIFICATE TO BE DELIVERED PURSUANT TO SECTION 10.1(c) WITH RESPECT TO REPRESENTATIONS AND WARRANTIES), SELLER FURTHER SPECIFICALLY DISCLAIMS ANY STATEMENT, REPRESENTATION OR WARRANTY OF MERCHANTABILITY, USAGE, SUITABILITY OR FITNESS FOR ANY PARTICULAR PURPOSE WITH RESPECT TO ASSETS OF THE BUSINESS, ANY PART THEREOF, THE WORKMANSHIP THEREOF, AND THE ABSENCE OF ANY DEFECTS THEREIN, WHETHER LATENT OR PATENT, IT BEING UNDERSTOOD THAT SUCH ASSETS ARE BEING ACQUIRED "AS IS, WHERE IS" ON THE CLOSING DATE, AND IN THEIR PRESENT CONDITION. THIS SECTION 4.23 SHALL NOT LIMIT ANY RIGHT OR REMEDY OF PURCHASER WITH RESPECT TO ANY CLAIM FOR FRAUD, UNDER THE R&W INSURANCE POLICY OR IN CONNECTION WITH ANY REPRESENTATION OR WARRANTY SET FORTH IN ANY OTHER TRANSACTION AGREEMENT DELIVERED AT THE CLOSING.

4.24 No Other Representations.

- (a) IN ENTERING INTO THIS AGREEMENT, SELLER HAS RELIED SOLELY UPON ITS OWN INDEPENDENT REVIEW AND ANALYSIS AND THE REPRESENTATIONS AND WARRANTIES OF PURCHASER EXPRESSLY SET FORTH IN <u>ARTICLE V</u> (AND, AS OF THE CLOSING, THE CERTIFICATE TO BE DELIVERED PURSUANT TO <u>SECTION 10.2(c)</u> WITH RESPECT TO REPRESENTATIONS AND WARRANTIES).
- (b) SELLER UNDERSTANDS, ACKNOWLEDGES AND AGREES THAT THE REPRESENTATIONS AND WARRANTIES OF PURCHASER EXPRESSLY SET FORTH IN ARTICLE V (AND, AS OF THE CLOSING, THE CERTIFICATE TO BE DELIVERED PURSUANT TO SECTION 10.2(c) WITH RESPECT TO REPRESENTATIONS AND WARRANTIES) CONSTITUTE THE SOLE AND EXCLUSIVE REPRESENTATIONS AND WARRANTIES OF ANY KIND OF PURCHASER, ITS AFFILIATES AND ANY OF THEIR RESPECTIVE REPRESENTATIVES OR ANY OTHER PERSON TO SELLER, ITS AFFILIATES AND ITS AND THEIR RESPECTIVE REPRESENTATIVES WITH RESPECT TO PURCHASER AND ITS AFFILIATES, THIS AGREEMENT, THE TRANSACTIONS, OR THE ACCURACY OR COMPLETENESS OF ANY INFORMATION REGARDING PURCHASER OR ITS AFFILIATES, OR ANY OTHER MATTER, FURNISHED OR MADE

RESPECTIVE REPRESENTATIVES OR ANY OTHER PERSON (INCLUDING ANY INFORMATION, DOCUMENTS OR MATERIALS MADE AVAILABLE TO SELLER. ITS AFFILIATES OR ITS OR THEIR RESPECTIVE REPRESENTATIVES OR ANY OTHER PERSON), INCLUDING WITH RESPECT TO ANY ERRORS THEREIN OR OMISSIONS THEREFROM, IN EACH CASE, IN EXPECTATION OR FURTHERANCE OF THE TRANSACTIONS, AND SELLER IRREVOCABLY UNDERSTANDS, ACKNOWLEDGES AND AGREES THAT (I) EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES OF PURCHASER EXPRESSLY SET FORTH IN ARTICLE V (AND, AS OF THE CLOSING, THE CERTIFICATE TO BE DELIVERED PURSUANT TO SECTION 10.2(c) WITH RESPECT TO REPRESENTATIONS AND WARRANTIES), ALL SUCH OTHER REPRESENTATIONS AND WARRANTIES OF ANY KIND OR NATURE, EXPRESS OR IMPLIED, AT LAW OR IN EQUITY, WHETHER WRITTEN OR ORAL, STATUTORY OR OTHERWISE, ARE SPECIFICALLY AND EXPRESSLY DISCLAIMED BY PURCHASER, ITS AFFILIATES AND THEIR RESPECTIVE REPRESENTATIVES AND (II) NONE OF PURCHASER, ITS AFFILIATES OR ANY OF THEIR RESPECTIVE REPRESENTATIVES OR ANY OTHER PERSON SHALL HAVE ANY LIABILITY FOR ANY AND ALL SUCH OTHER EXPRESS OR IMPLIED REPRESENTATIONS OR WARRANTIES NOT EXPRESSLY SET FORTH IN ARTICLE V (AND, AS OF THE CLOSING, THE CERTIFICATE TO BE DELIVERED PURSUANT TO <u>SECTION 10.2(c)</u> WITH RESPECT TO REPRESENTATIONS AND WARRANTIES). THIS SECTION 4.24 SHALL NOT LIMIT ANY RIGHT OR REMEDY OF SELLER WITH RESPECT TO ANY CLAIM FOR FRAUD OR ANY REPRESENTATION OR WARRANTY SET FORTH IN ANY OTHER TRANSACTION AGREEMENT DELIVERED AT THE CLOSING.

AVAILABLE TO (OR OTHERWISE ACQUIRED BY) SELLER, ITS AFFILIATES OR ITS OR THEIR

ARTICLE V

REPRESENTATIONS AND WARRANTIES OF PURCHASER

Purchaser hereby represents and warrants to Seller as follows:

5.1 Authority; Enforceability.

- (a) Purchaser has the requisite organizational power and authority to execute and deliver this Agreement and each other Transaction Agreement to which it is a party, to perform its obligations hereunder and under each other Transaction Agreement to which it is a party and to consummate the Transactions in accordance with the terms of this Agreement and each other Transaction Agreement to which it is a party. The execution, delivery and performance by Purchaser of this Agreement and each other Transaction Agreement to which it is a party and the consummation of the Transactions by Purchaser have been duly and validly authorized by all necessary corporate action on the part of Purchaser and such authorization has not been subsequently modified or rescinded.
- (b) This Agreement has been duly and validly executed and delivered by Purchaser and constitutes, assuming due authorization, execution and delivery of this Agreement by Seller, a valid and binding legal obligation of Purchaser, enforceable against Purchaser in accordance with the terms hereof. Assuming due authorization, execution and delivery of each

other Transaction Agreement to which Purchaser is a party by the other parties thereto, each such Transaction Agreement will constitute a valid and binding legal obligation of Purchaser at the time of execution by Purchaser, enforceable against Purchaser in accordance with the terms thereof.

5.2 <u>Non-Contravention; Consents.</u>

- (a) The execution and delivery of this Agreement by Purchaser, and each other Transaction Agreement to which Purchaser is a party by Purchaser, does not, and the performance of this Agreement by Purchaser, and each other Transaction Agreement to which Purchaser is a party by Purchaser, will not, require any Consent or Permit of, registration, declaration or filing with, or notification to, any Governmental Authority (other than as a party to any Government Contract or as the ultimate customer of any Government Contract), except (i) under applicable Antitrust Laws, (ii) under the applicable requirements of the Exchange Act, (iii) for such other Consents, Permits, filings or notifications, the failure of which to make or obtain would not reasonably be expected to have, individually or in the aggregate, a Purchaser Material Adverse Effect, (iv) in connection with obtaining CFIUS Clearance and DCSA Approvals and (v) under Section 122.4(b) of the ITAR.
- (b) Assuming the Consents, Permits, registrations, declarations, filings and notifications referred to in Section 5.2(a) are obtained or made, the execution and delivery by Purchaser of this Agreement and each other Transaction Agreement to which Purchaser is a party does not, and the consummation of the Transactions will not, (i) conflict with or violate any provision of the organizational documents of Purchaser or (ii) result in a breach of, constitute a default under (with or without notice or lapse of time, or both), result in the creation or acceleration of (or loss of benefit from) any rights or obligations under, or create in any party the right to accelerate, terminate, modify or cancel any material Contract to which Purchaser or its Affiliates are party, except, in the case of the foregoing clause (ii), as would not reasonably be expected to have, individually or in the aggregate, a Purchaser Material Adverse Effect.
- 5.3 <u>Organization</u>. Purchaser is duly incorporated, validly existing and in good standing under the Legal Requirements of the jurisdiction of its incorporation, except as would not reasonably be expected to have, individually or in the aggregate, a Purchaser Material Adverse Effect. Purchaser has all necessary organizational power and authority to conduct its business in the manner in which it is being conducted as of the date of this Agreement, except where the absence of such power to conduct its business would not reasonably be expected to have, individually or in the aggregate, a Purchaser Material Adverse Effect.
- 5.4 <u>Litigation</u>. There is no Proceeding pending before any Governmental Authority or, to the knowledge of Purchaser, threatened in writing, against Purchaser which questions the validity of this Agreement or any of the other Transaction Agreements to which it is a party, and Purchaser is not subject to any Orders of any Governmental Authority, in each case, that would reasonably be expected to have, individually or in the aggregate, a Purchaser Material Adverse Effect.
- 5.5 <u>Sufficiency of Funds</u>. As of the date of this Agreement, Purchaser has, and as of immediately prior to the Closing, Purchaser will have, available to it sufficient funds (in immediately available funds, subject to, for the avoidance of doubt, delivery of a customary

borrowing notice) to (a) satisfy all of Purchaser's obligations under this Agreement, including its obligations under <u>Article II</u>, (b) pay any other amounts required to be paid by Purchaser in connection with the consummation of the Transactions and (c) pay all related fees and expenses of Purchaser.

- 5.6 <u>Solvency.</u> As of the Closing, immediately after giving effect to all of the Transactions, including, if applicable, the Debt Financing, and assuming satisfaction of the conditions to Purchaser's obligations to consummate the Closing set forth in <u>Article X</u>, Purchaser and its Subsidiaries (including, after the Closing, the Acquired Companies), on a consolidated basis, will not (a) be insolvent (either because its financial condition is such that the sum of its debts (including a reasonable estimate of the amount of all contingent liabilities) is greater than the fair value of its assets, or because the present fair saleable value of its assets will be less than the amount required to pay its probable liability on its debts as they become absolute and matured), (b) have unreasonably small capital with which to engage in its business or (c) have incurred or plan to incur debts beyond its ability to pay as they become absolute and matured.
- 5.7 <u>Brokers</u>. Other than with respect to fees or commissions that will be borne solely by Purchaser and its Affiliates, Purchaser and its Affiliates have not retained any broker or finder or incurred any liability or obligation for any brokerage fees, commissions, finders', financial advisor or other similar fees with respect to this Agreement or the Transactions.
- 5.8 Pending Transactions. As of the date of this Agreement, neither Purchaser nor any of its Affiliates is a party to any pending transaction to acquire ((a) by merging or consolidating with, by purchasing the assets of or equity in, or (b) by any other similar transaction, including a license or co-development or production agreement (for the avoidance of doubt, this clause (b) shall not include any commercial activities in the ordinary course of business)) any Person (or business division or unit thereof), where the entering into of a definitive agreement relating to or the consummation of such transaction would reasonably be expected to (i) impose any material delay in the obtaining of, or increase, in any material respect, the risk of not obtaining, any Consents, Orders or Governmental Approvals necessary to satisfy the conditions to the Closing set forth in Section 10.3, including the expiration or termination of any applicable waiting period under any Legal Requirement, (ii) increase, in any material respect, the risk of any Governmental Authority entering an Order prohibiting or restraining the consummation of the Transactions or (iii) otherwise prevent or materially delay the consummation of the Transactions.
- 5.9 <u>NTIB Entity</u>. Purchaser represents that it is a National Technology and Industrial Base (NTIB) entity as that term is defined in 32 C.F.R. Part 117.3(b).
- 5.10 <u>ITAR</u>. Purchaser represents that it is registered and in good standing under the ITAR as an exporter and a manufacturer.

5.11 <u>Inspection; No Other Representations.</u>

(a) Purchaser is an informed and sophisticated purchaser and has such knowledge and experience in financial and business matters that it is capable of evaluating the merits and risks of its purchase of the Acquired Companies. PURCHASER HAS CONDUCTED TO ITS OWN SATISFACTION AN INDEPENDENT REVIEW AND ANALYSIS OF THE

ACQUIRED COMPANIES, THE BUSINESS AND THE ASSETS, AND THE CONDITION, OPERATIONS AND PROSPECTS OF THE ACQUIRED COMPANIES AND THE BUSINESS, AND ACKNOWLEDGES AND AGREES THAT IT HAS BEEN PROVIDED ACCESS TO THE PROPERTIES, PREMISES AND RECORDS OF THE ACQUIRED COMPANIES AND THE BUSINESS FOR THIS PURPOSE. IN ENTERING INTO THIS AGREEMENT, PURCHASER HAS RELIED SOLELY UPON ITS OWN INDEPENDENT REVIEW AND ANALYSIS AND THE REPRESENTATIONS AND WARRANTIES OF SELLER EXPRESSLY SET FORTH IN <u>ARTICLE IV</u> (AND, AS OF THE CLOSING, THE CERTIFICATE TO BE DELIVERED PURSUANT TO <u>SECTION 10.1(c)</u> WITH RESPECT TO REPRESENTATIONS AND WARRANTIES).

PURCHASER UNDERSTANDS, ACKNOWLEDGES AND AGREES THAT THE REPRESENTATIONS AND WARRANTIES OF SELLER EXPRESSLY SET FORTH IN ARTICLE IV (AND, AS OF THE CLOSING, THE CERTIFICATE TO BE DELIVERED PURSUANT TO SECTION 10.1(c) WITH RESPECT TO REPRESENTATIONS AND WARRANTIES), CONSTITUTE THE SOLE AND EXCLUSIVE REPRESENTATIONS AND WARRANTIES OF ANY KIND OF SELLER, ITS AFFILIATES (INCLUDING THE ACQUIRED COMPANIES), AND ANY OF THEIR RESPECTIVE REPRESENTATIVES OR ANY OTHER PERSON TO PURCHASER, ITS AFFILIATES AND ITS AND THEIR RESPECTIVE REPRESENTATIVES WITH RESPECT TO SELLER, ITS AFFILIATES (INCLUDING THE ACQUIRED COMPANIES), THEIR BUSINESSES (INCLUDING THE BUSINESS), OPERATIONS, ASSETS, LIABILITIES, FINANCIAL CONDITION, RESULTS OF OPERATIONS, FUTURE OPERATING OR FINANCIAL RESULTS, ESTIMATES, PROJECTIONS, FORECASTS, PLANS OR PROSPECTS (INCLUDING THE REASONABLENESS OF THE ASSUMPTIONS UNDERLYING SUCH ESTIMATES, PROJECTIONS, FORECASTS, PLANS OR PROSPECTS), THIS AGREEMENT, THE TRANSACTIONS, OR THE ACCURACY OR COMPLETENESS OF ANY INFORMATION REGARDING SELLER OR ITS AFFILIATES (INCLUDING THE ACQUIRED COMPANIES), THEIR BUSINESSES (INCLUDING THE BUSINESS), FURNISHED OR MADE AVAILABLE TO (OR OTHERWISE ACQUIRED BY) PURCHASER, ITS AFFILIATES OR ITS OR THEIR RESPECTIVE REPRESENTATIVES OR ANY OTHER PERSON (INCLUDING ANY INFORMATION, DOCUMENTS OR MATERIALS MADE AVAILABLE TO PURCHASER, ITS AFFILIATES OR ITS OR THEIR RESPECTIVE REPRESENTATIVES IN THE DATA ROOM OR OTHERWISE, IN A CONFIDENTIAL INFORMATION MEMORANDUM OR ANY MANAGEMENT PRESENTATIONS OR IN ANY OTHER FORM), INCLUDING WITH RESPECT TO ANY ERRORS THEREIN OR OMISSIONS THEREFROM, OR AS TO THE FUTURE REVENUE, PROFITABILITY OR SUCCESS OF SELLER, ITS AFFILIATES (INCLUDING THE ACQUIRED COMPANIES) OR THE BUSINESS (INCLUDING THE FINANCIAL INFORMATION, PROJECTIONS OR OTHER FORWARD-LOOKING STATEMENTS OF SELLER, ITS AFFILIATES (INCLUDING THE ACQUIRED COMPANIES) OR THE BUSINESS, IN EACH CASE, IN EXPECTATION OR FURTHERANCE OF THE TRANSACTIONS), AND PURCHASER IRREVOCABLY UNDERSTANDS, ACKNOWLEDGES AND AGREES THAT (I) EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES OF SELLER EXPRESSLY SET FORTH IN ARTICLE IV (AND, AS OF THE CLOSING, THE CERTIFICATE TO BE DELIVERED PURSUANT TO SECTION 10.1(c) WITH RESPECT TO REPRESENTATIONS AND WARRANTIES), ALL SUCH OTHER REPRESENTATIONS AND WARRANTIES OF ANY

KIND OR NATURE, EXPRESS OR IMPLIED, AT LAW OR IN EQUITY, WHETHER WRITTEN OR ORAL, STATUTORY OR OTHERWISE, ARE SPECIFICALLY AND EXPRESSLY DISCLAIMED BY SELLER, ITS AFFILIATES (INCLUDING THE ACQUIRED COMPANIES) AND THEIR RESPECTIVE REPRESENTATIVES AND (II) NONE OF SELLER, ITS AFFILIATES (INCLUDING THE ACQUIRED COMPANIES), OR ANY OF THEIR RESPECTIVE REPRESENTATIVES OR ANY OTHER PERSON SHALL HAVE ANY LIABILITY FOR ANY AND ALL SUCH OTHER EXPRESS OR IMPLIED REPRESENTATIONS OR WARRANTIES NOT EXPRESSLY SET FORTH IN ARTICLE IV (AND, AS OF THE CLOSING, THE CERTIFICATE TO BE DELIVERED PURSUANT TO SECTION 10.1(c) WITH RESPECT TO REPRESENTATIONS AND WARRANTIES). THIS SECTION 5.11 SHALL NOT LIMIT ANY RIGHT OR REMEDY OF PURCHASER WITH RESPECT TO ANY CLAIM FOR FRAUD, UNDER THE R&W INSURANCE POLICY OR WITH RESPECT TO ANY REPRESENTATION OR WARRANTY SET FORTH IN ANY OTHER TRANSACTION AGREEMENT DELIVERED AT THE CLOSING.

5.12 <u>Disclaimer of Purchaser</u>. EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES EXPRESSLY MADE BY PURCHASER IN THIS ARTICLE V (AND, AS OF THE CLOSING, THE CERTIFICATE TO BE DELIVERED PURSUANT TO SECTION 10.2(c) WITH RESPECT TO REPRESENTATIONS AND WARRANTIES), NONE OF PURCHASER, ITS AFFILIATES OR ANY OF THEIR RESPECTIVE REPRESENTATIVES OR ANY OTHER PERSON MAKES. HAS MADE. SHALL BE DEEMED TO HAVE MADE, OR HAS BEEN AUTHORIZED TO MAKE ANY EXPRESS OR IMPLIED REPRESENTATION OR WARRANTY, AT LAW OR IN EQUITY, WHETHER WRITTEN OR ORAL, STATUTORY OR OTHERWISE, ON BEHALF OF OR WITH RESPECT TO PURCHASER, ITS AFFILIATES OR ANY OF THEIR RESPECTIVE REPRESENTATIVES OR ANY OTHER PERSON, THIS AGREEMENT, THE TRANSACTIONS, OR THE ACCURACY OR COMPLETENESS OF ANY INFORMATION REGARDING PURCHASER OR ITS AFFILIATES OR ANY OTHER MATTER, FURNISHED OR MADE AVAILABLE TO (OR OTHERWISE ACQUIRED BY) SELLER, ITS AFFILIATES OR ITS OR THEIR RESPECTIVE REPRESENTATIVES OR ANY OTHER PERSON, INCLUDING WITH RESPECT TO ANY ERRORS THEREIN OR OMISSIONS THEREFROM, IN EACH CASE, IN EXPECTATION OR FURTHERANCE OF THE TRANSACTIONS, AND PURCHASER HEREBY DISCLAIMS AND SHALL HAVE NO LIABILITY FOR ANY AND ALL SUCH REPRESENTATIONS OR WARRANTIES NOT EXPRESSLY SET FORTH IN THIS ARTICLE V (AND, AS OF THE CLOSING, THE CERTIFICATE TO BE DELIVERED PURSUANT TO SECTION 10.2(C) WITH RESPECT TO REPRESENTATIONS AND WARRANTIES). THIS SECTION 5.12 SHALL NOT LIMIT ANY RIGHT OR REMEDY OF SELLER WITH RESPECT TO ANY CLAIM FOR FRAUD OR ANY REPRESENTATION OR WARRANTY SET FORTH IN ANY OTHER TRANSACTION AGREEMENT DELIVERED AT THE CLOSING.

ARTICLE VI

COVENANTS OF THE PARTIES

6.1 <u>Conduct of the Business Prior to the Closing.</u>

- Except (i) as expressly contemplated by this Agreement or the other Transaction Agreements, (ii) as required by any Legal Requirement or Contract, (iii) for matters set forth on Schedule 6.1(a) of the Seller Disclosure Schedule, (iv) reasonably in response to any (A) Contagion Event, (B) Pandemic Measure or (C) Cybersecurity Measure or (v) with the written consent of Purchaser (which consent shall not be unreasonably withheld, conditioned or delayed), from the date of this Agreement until the earlier of the Closing and the termination of this Agreement pursuant to its terms, Seller shall cause the Acquired Companies to, and, to the extent relating to the Business, shall and shall cause its other Controlled Affiliates to, (1) use commercially reasonable efforts to operate the Business in the ordinary course of business consistent with past practice (provided, that (x) no action or inaction by Seller or any of its Controlled Affiliates (including any Acquired Company) with respect to any matters specifically addressed by any portion of the subsequent clause (2) of this Section 6.1(a) shall be deemed a breach of this clause (1) unless such action or inaction would constitute a breach of such portion of clause (2); (y) Purchaser's written consent with respect to any specific action or matter pursuant to clause (2) shall be deemed to constitute consent with respect to such action or matter for all purposes under this Section 6.1 and (z) nothing in this Section 6.1(a) shall be deemed to apply to any action taken or not taken by Seller or any of its Controlled Affiliates (other than the Acquired Companies) to the extent such action or inaction (A) relates to the general corporate shared services provided by Seller or any such Controlled Affiliate (and to be retained by Seller or such Controlled Affiliate in connection with the Transactions), (B) applies to all businesses of Seller and its Controlled Affiliates equally and (C) would not reasonably be expected to result in a material detriment to the Acquired Companies or the Business) and (2) not do any of the following:
 - (1) amend the organizational documents of any Acquired Company;
 - (2) except for (A) intercompany borrowings that will be repaid or settled in full or terminated or canceled at or prior to the Closing, (B) ordinary course borrowings under the Credit Facilities and (C) Indebtedness incurred to finance capital expenditures permitted under clause (xiii) below, (1) incur in excess of \$10,000,000 of indebtedness for borrowed money outstanding at any time, (2) enter into any Contract involving financing or borrowing of money or (3) assume, guarantee or endorse the obligations of any Person if, in each case, such obligations would be obligations of the Acquired Companies following the Closing;
 - (3) permit any of the material assets of the Acquired Companies or the Business to become subjected to any Encumbrance other than (A) Permitted Encumbrances or (B) Encumbrances incurred in the ordinary course of business consistent with past practice;
 - (4) with respect to any Acquired Company, fail to maintain its corporate existence, adopt or enter into any plan of complete or partial liquidation, dissolution, restructuring, recapitalization or other reorganization or merge or consolidate with any other Person, or, with respect to any Acquired Company or the Business, enter into any joint venture or similar venture with any other Person (not including joint development arrangements with customers or suppliers in the ordinary course of business);

- (5) purchase any securities or make any investment or capital contribution in any Person, either by purchase of stock or securities, contributions to capital, asset transfers or purchase of any assets, or otherwise acquire, or acquire direct or indirect control over, any Person, business, business organization or division thereof for which the aggregate consideration paid (A) in any individual transaction is in excess of \$15,000,000 or (B) in the aggregate is in excess of \$35,000,000;
- (6) loan or advance any amount other than (A) to another Acquired Company or (B) loans or advances to employees of any Acquired Company for travel and business expenses in the ordinary course of business;
- except with respect to customary rights granted to lenders to secure the (7) obligations under the Seller Debt Facilities that will be released at the Closing, redeem or otherwise acquire any shares of capital stock of or other voting or equity interests in any Acquired Company, or issue, deliver, sell, transfer or grant (A) any shares of capital stock of or other voting or equity interests in any Acquired Company or (B) any warrant, option, right, agreements, "phantom" stock right, stock appreciation right, stock-based performance unit, convertible, exercisable or exchangeable securities or rights or any other commitment or undertaking (1) pursuant to which any Acquired Company is or may become obligated to issue, deliver, sell, transfer or grant (x) any shares of capital stock of or other voting or equity interests in any Acquired Company or (y) any security convertible into, or exercisable or exchangeable for, any shares of capital stock of or other voting or equity interests in any Acquired Company, (2) pursuant to which any Acquired Company is, or may become obligated to, issue, deliver, sell, transfer or grant any such warrant, option, right, unit, security, commitment or undertaking described in the foregoing clause (1) or (3) that gives any Person the right to receive any benefit or right similar to any rights enjoyed by or accruing to the holders of any shares of capital stock of or other voting or equity interests in any Acquired Company;
- (8) split, combine or reclassify any shares of capital stock of or other voting or equity interests in any Acquired Company (including the Shares), or issue any other security in respect of, in lieu of or in substitution for any shares of capital stock of or other voting or equity interests in any Acquired Company (including the Shares);
- (9) except in the ordinary course of business consistent with past practice, sell, transfer, lease, sublease or otherwise dispose of any properties or assets of the Acquired Companies or the Business having an aggregate value in excess of \$25,000,000;
- (10) declare or set aside any dividends or distributions on any shares of capital stock of or other voting or equity interests in any Acquired Company (in cash or in kind) to the extent such dividends or distributions (A) are payable at or after the Closing or (B) payable in anything other than cash;
- (11) compromise, settle, agree to settle or grant any release of any claim relating to any pending Proceeding (A) where the amount involved is in excess of

- (12) sell, assign, transfer, exclusively lease, abandon, exclusively license or sublicense, allow to let lapse, terminate or expire, grant an Encumbrance with respect to, or otherwise dispose of any material Intellectual Property, except (A) Permitted Encumbrances, (B) expirations of contractual obligations pursuant to their terms, (C) abandonment of Company Registered IP upon the expiration of its term in the ordinary course of business and (D) non-exclusive licenses of Owned IP granted by an Acquired Company with respect to customer or supplier Contracts;
- (13) make any capital expenditure in excess of \$5,000,000 individually, or \$10,000,000 in the aggregate, other than as set forth in the capital budgets of the Business made available to Purchaser prior to the date hereof;
- (14) make any material change to any accounting method or system of internal accounting control of the Acquired Companies, except as may be appropriate to conform to changes in Legal Requirements, regulatory accounting requirements or GAAP;
- (15) except (A) as may be required by the terms as of the date of this Agreement of a Benefit Plan or (B) obligations for which Seller and its Controlled Affiliates (other than the Acquired Companies) shall be solely obligated to pay and as would not result in any liability to Purchaser or any Acquired Company (any compensation action taken in reliance of this clause (B), a "Permitted Compensation Action"), (1) increase or accelerate the funding, payment or vesting of the compensation or benefits of any Company Employee, Former Company Employee or other current or former individual service provider of the Acquired Companies or the Business, other than immaterial increases for Company Employees and individual service providers of the Acquired Companies with a gross annual base salary or other base compensation of less than \$280,000, (2) enter into, adopt, materially amend or terminate any Acquired Company Benefit Plan (or any other benefit or compensation plan, policy, program, contract, agreement or arrangement that would be an Acquired Company Benefit Plan if in effect on the date hereof), (3) hire any Person as a Company Employee or other individual service provider of the Acquired Companies or the Business, other than the hiring of Company Employees or other individual service providers (x) with a gross annual base salary or other base compensation of less than \$280,000 or (y) to replace any Company Employee or other individual service provider who resigns or whose employment or contract is terminated, or to fill any requisition open as of the date hereof (provided that, if such hiring is in reliance of this clause (y), for any open requisition not set forth on Schedule 6.1(a)(xy) of the Seller Disclosure Schedule, Seller or its applicable Controlled Affiliate shall consult in good faith with Purchaser prior to hiring any Company Employees or other individual service provider with a gross annual base salary or other compensation of more than \$280,000 if such Person is the President of the Business or any Acquired Company or a direct report of the President) on the same or substantially similar terms and conditions of employment or contract as similarly situated individuals of Seller and its Controlled Affiliates (including the Acquired Companies), including base compensation and bonus opportunity, if applicable, (4) terminate the employment, other than for cause, of any Company Employee

with a gross annual base salary equal to or in excess of \$280,000, (5) transfer the employment of any Company Employee outside of the Acquired Companies other than where such Company Employee applied for, and was selected in a competitive process not targeted at Company Employees for, a position within Seller or any of its Controlled Affiliates that is outside of the Acquired Companies; provided that, in no event may a Company Employee with a gross annual base salary equal to or in excess of \$280,000 be transferred outside of the Acquired Companies without the written consent of Purchaser or (6) grant or announce any cash or equity or equity-based incentive award, bonus, retention bonus, transaction bonus, severance, in any such case that is extraordinary in nature, or similar extraordinary compensation payable to any Company Employee, Former Company Employee or other current or former individual service provider of the Acquired Companies or the Business (for the avoidance of doubt, this clause (6) shall not limit any Permitted Compensation Action);

- (16) (A) modify, extend, terminate or enter into any Collective Bargaining Agreement covering Company Employees or (B) recognize or certify any labor union, labor organization, works council or group of employees as the bargaining representative for any Company Employees;
- (17) with respect to Company Employees, implement or announce any employee layoffs, furloughs, reductions in force, plant closings, or other similar actions, in each case, requiring notice to employees under the WARN Act in the event of a plant closing or mass layoff;
- (18) transfer or reassign the duties of (A) a Company Employee such that he or she is no longer a Company Employee but is an employee of Seller or its Controlled Affiliates or (B) any other employee of Seller or its Controlled Affiliates such that he or she would become a Company Employee, in each case other than such actions that are taken in order to fill a vacancy in the ordinary course of business (including a vacancy due to an ordinary course termination of employment) or due to death or disability, with such individual to receive the same or substantially similar terms and conditions of employment or Contract as similarly situated Company Employees, including base compensation and bonus opportunity, if applicable;
- (19) except as would not reasonably be expected to affect the Tax liability of the Acquired Companies, (A) make, change or revoke any material Tax election, (B) enter into any "closing agreement" as described in Section 7121 of the Code (or any corresponding agreement under state, local or foreign Legal Requirements), (C) settle, compromise or abandon any dispute, audit, claim or proceeding concerning any material Tax liability, (D) file any material Tax Return in a manner inconsistent with past practices or file any material amended Tax Return, (E) surrender any right to claim any material Tax refund, (F) waive or extend any statute of limitations relating to any material Tax or material Tax Return or (G) incur any material Tax liability outside the ordinary course of business, except, in each case, to the extent such actions are taken solely with respect to a Consolidated Return or Taxes reportable thereon;

- (20) conduct its cash management practices materially outside of the ordinary course of business consistent with past practice;
- (21) (A) other than in the ordinary course of business (including any specific contract or with respect to a specific program, in each case that is expressly contemplated by the operating plan of the Business made available to Purchaser), enter into, materially modify, materially amend, terminate or waive any material rights under any Material Contract, (B) other than Contracts that are for programs or projects out for bid as of the date hereof (a "Pending Bid Contract"), enter into any Contract that contains a change of control or similar provision that would require a payment to (or consent of) the other party or parties thereto in connection with the Transactions (unless a waiver or consent with respect to the Transactions is obtained from the applicable counterparty concurrently with such action and such waiver or consent does not involve the payment of a penalty or other financial recourse or consequence (a "Material Contract Waiver")) (with respect to this clause (xxi), (1) if Purchaser does not respond in writing to Seller's request for consent within three (3) Business Days, such consent shall be deemed given by Purchaser and (2) this clause (xxi) shall not apply to programs or projects (including attendant Contracts) for which Seller and its Affiliates are prohibited by applicable Legal Requirement from disclosing to or consulting with Purchaser) or (C) enter into any new customer Contract with respect to the business of Topaz that is a Government Contract;
- (22) enter into, modify, amend or waive any material rights under any Contract set forth on <u>Schedule 4.21</u> of the Seller Disclosure Schedule or that would have been required to be set forth on such schedule if entered into prior to the date of this Agreement; or
 - (23) enter into any legally binding commitment with respect to any of the foregoing.
- Notwithstanding anything to the contrary contained in Section 6.1(a), nothing in Section 6.1(a) shall prevent Seller or its Affiliates from taking or failing to take any action (i) that is substantively consistent with the policies of Seller or any of its Affiliates in effect as of the date of this Agreement in connection with any (A) Contagion Event, including to the extent reasonably necessary to comply with Pandemic Measures or (B) Cybersecurity Measures or (ii) in good faith (including the establishment of any policy, procedure or protocol or any action taken or failed to be taken to protect the Business, the health and safety of the personnel or employees of Seller or its Affiliates or other Persons with whom the personnel or employees of Seller or its Affiliates come into contact with in the ordinary course of business) (A) in response to a Contagion Event, (B) to the extent reasonably necessary to comply with Pandemic Measures, (C) in connection with Cybersecurity Measures or (D) to the extent reasonably necessary to comply with change in any Legal Requirement relating to or resulting from any Contagion Event, Pandemic Measure or Cybersecurity Measure, and none of the foregoing shall in any event be deemed to constitute a breach of Section 6.1(a); provided, that, Seller shall use commercially reasonable efforts to inform Purchaser of any such actions as promptly as reasonably practicable to the extent material to the operation of the Business or the Acquired Companies and, to the extent reasonably practicable, use commercially reasonable efforts to consult with and cooperate with Purchaser in good faith with respect to any such actions.

- (c) Subject to the other limitations of this <u>Section 6.1</u>, the bank accounts of the Acquired Companies shall continue to be subject to Seller's and its Affiliates' periodic cash sweep in their sole discretion, and the Acquired Companies may pay cash dividends and make cash distributions. Prior to the Closing, Seller shall (and shall cause its Controlled Affiliates to) take all actions necessary to ensure the Company Subsidiary has at least twenty million dollars (\$20,000,000) (after giving effect to the payment of any Indebtedness and Transaction Expenses or settlement of any Intercompany Agreements at the Closing) of freely usable cash (the "<u>Minimum Closing Cash</u>") in immediately available funds as of the Closing in bank account(s) of the Company Subsidiary located in the United States.
- (d) Nothing contained in this Agreement is intended to or shall give Purchaser or any of its Affiliates, directly or indirectly, the right to control or direct the operations of Seller, the Business or any of the Acquired Companies prior to the Closing. Prior to the Closing, Seller shall exercise, consistent with the terms and conditions of this Agreement, complete control and supervision over its operations and the operations of the Business and the Acquired Companies.
- (e) Notwithstanding anything to the contrary contained in this Agreement, nothing herein shall prevent Seller and its Affiliates from, and Seller and its Affiliates have the absolute right to, transfer the Business Pension Plan to an Acquired Company prior to the Closing.
- (f) Notwithstanding anything to the contrary contained in this Agreement (but subject to the second sentence of Section 6.1(c)), nothing herein shall prevent Seller and its Affiliates from, and Seller and its Affiliates have the affirmative right to, (i) terminate any factoring program applicable to the Business or the Acquired Companies and (ii) settle, cancel or otherwise eliminate any intercompany accounts or liabilities, using cash, through capital contributions, distributions, forgiveness, offset or any combination of the foregoing, in each case, in a manner that does not adversely affect Purchaser or any of its Affiliates (including, following the Closing, the Acquired Companies) in any material respect.

6.2 <u>Pre-Closing Access to Information</u>.

(a) Until the earlier of the Closing and the termination of this Agreement pursuant to its terms, Seller shall, and shall cause the Acquired Companies (and to the extent related to the Business, its other Controlled Affiliates) to, permit Purchaser to have reasonable access, subject to applicable Legal Requirement and upon reasonable prior notice, during normal business hours in a manner so as not to interfere with the normal business operations of Seller and its Affiliates and in accordance with the procedures established by Seller, to the books, assets, properties, Contracts and records of the Business and solely for the purposes of facilitating the consummation of the Transactions or integration planning; provided, however, that nothing in this Agreement shall (i) require Seller or its Affiliates to provide access or to disclose information where Seller reasonably believes in good faith that such access or disclosure would contravene any Legal Requirement (including Data Protection Laws), Pandemic Measure, Cybersecurity Measure or the terms of any Contract, would result in the waiver of any legal privilege or work-product protection, would expose Seller to risk of liability with respect to disclosure of sensitive, confidential or personal information or would cause significant competitive harm to Seller, its Affiliates or their respective businesses if the Transactions are not consummated, (ii) include any invasive investigations, sampling or testing of the Owned Real Property or the Leased Real

Property for or regarding any environmental matters, (iii) require Seller or any of its Affiliates to provide Purchaser, its Affiliates, its and their Representatives or other representatives with (A) any Consolidated Return (or copy thereof) or (B) information to the extent relating to businesses of Seller or any of its Affiliates other than the Business or (iv) require Seller to provide (A) information with respect to bids, the identity of any bidder, confidentiality or non-disclosure agreements, letters of intent, expressions of interest or other proposals received in connection with transactions comparable to the Transactions or any information or analysis relating to any such communications or (B) financial or operating data that has not previously been prepared by Seller or its Affiliates, or that is not otherwise prepared in the ordinary course of operating the Business; provided, however, that, if reasonably requested by Purchaser, Seller will notify Purchaser in reasonable detail of the circumstances giving rise to any non-disclosure pursuant to the foregoing and, in the case of clause (i), Seller and Purchaser shall use commercially reasonable efforts to identify and pursue a permissible method to provide such access or disclosure of such information in a manner that would not result in any of the outcomes described therein. The provision of any information pursuant to this Section 6.2(a) shall not expand the remedies available to Purchaser or its Affiliates under this Agreement in any manner. Any information disclosed will be subject to the provisions of the Non-Disclosure Agreement.

- (b) Until the earlier of the Closing and the termination of this Agreement pursuant to its terms, Purchaser shall not, and shall cause its Affiliates and its and their respective Representatives not to, communicate with any of the officers, directors, employees, landlords or customers of, or vendors or suppliers to, the Business or the Acquired Companies, to the extent such communications are related to the Business or the Acquired Companies and are in connection with this Agreement or the Transactions, without the prior written consent of Seller; provided, that nothing in this Section 6.2 shall prohibit Purchaser, its Affiliates and its and their respective Representative from communicating with (i) such Persons in the ordinary course of their respective business unrelated to this Agreement or the Transactions in connection with ongoing commercial relationships or (ii) Company Employees at such a time and in such a manner as mutually agreed by the parties in advance in connection with post-Closing matters relating to such Company Employees' employment.
- (c) Until the earlier of the Closing and the termination of this Agreement pursuant to its terms, if so reasonably requested by Seller, Purchaser will use commercially reasonable efforts to enter into a customary joint defense agreement or common interest agreement with Seller or any of its Affiliates with respect to any information provided to Purchaser or its Affiliates, or to which Purchaser or its Affiliates gain access, pursuant to this Section 6.2.
- 6.3 <u>Cooperation</u>. Subject in all cases to the other terms and conditions of this Agreement, including the provisions of <u>Section 6.4</u> and <u>Section 6.8</u>, prior to the Closing, Seller shall, and shall cause its Controlled Affiliates to, and Purchaser shall, and shall cause its Affiliates to, use reasonable best efforts to cause all Closing Conditions to be met as promptly as reasonably practicable and, in any event, on or prior to the Outside Date.

6.4 Shared Contracts and Consents.

(a) With respect to any Shared Contracts, if any, from the date hereof until the date that is twelve (12) months following the Closing Date, Seller and Purchaser shall cooperate

with each other and use their commercially reasonable efforts (i) to divide, modify or replicate (in whole or in part) the respective rights, obligations and liabilities relating to the Business under and in respect of such Shared Contract or (ii) to the extent the action contemplated in the foregoing clause (i) is not possible, novate the respective rights, obligations and liabilities relating to the Business under and in respect of such Shared Contract, such that, effective as of the Closing (to the extent practicable), (A) Purchaser or its designated Affiliate is the beneficiary of the post-Closing rights, and is responsible for the post-Closing obligations and liabilities, related to that portion of such Shared Contract to the extent related to the operation or conduct of the Business (the "Business Portion") (so that, subsequent to the Closing or, subject to the other provisions of this Section 6.4, such other time as the actions described in clauses (i) and (ii), as applicable, are effected in respect of such Shared Contract, Seller and its Affiliates (other than the Acquired Companies) shall have no post-Closing rights or post-Closing obligations and liabilities with respect to the Business Portion of such Shared Contract) and (B) Seller and its Controlled Affiliates (other than the Acquired Companies) is the beneficiary of the rights and is responsible for the obligations and liabilities related to such Shared Contract other than the Business Portion (the "Non-Business Portion") (so that, subsequent to the Closing or, subject to the other provisions of this Section 6.4, such other time the action described in clauses (i) and (ii), as applicable, are effected in respect of such Shared Contract, Purchaser and its Affiliates shall have no rights, obligations or liabilities with respect to the Non-Business Portion of such Shared Contract).

- (b) If the Closing occurs before all Shared Contracts are assigned or otherwise divided, modified, replicated or novated pursuant to Section 6.4(a), Seller and Purchaser shall use commercially reasonable efforts and cooperate with each other in any mutually agreeable and lawful arrangement under which Seller or one of its Controlled Affiliates will provide Purchaser or its designated Affiliate the economic claims, rights and benefits of the Business Portion of such Shared Contract until the earliest of (i) the expiration of the then-current term, (ii) twelve (12) months following the Closing Date and (iii) the termination of or amendment to such Shared Contract upon mutual agreement by the parties; provided that any early termination fees or similar fees incurred by Seller or any of its Controlled Affiliates in connection with the termination or amendment of such Shared Contract pursuant to this clause (iii) shall be borne by Purchaser. Such arrangement may include subcontracting, sublicensing or subleasing to Purchaser or its designated Affiliate of any and all rights of Seller or its applicable Controlled Affiliate under the Business Portion of such Shared Contract to the extent not prohibited under such Shared Contract.
- (c) Subject in all cases to the other terms and conditions set forth in this Agreement, prior to the Closing, Seller shall, and shall cause the Acquired Companies to, use commercially reasonable efforts to seek to obtain any Consents of, and make any registrations, declarations, filings and notifications to, any Persons (other than to the extent subject to Section 6.8) that may be required or appropriate in connection with the Transactions (including if such Consent is required to avoid an obligation to pay or incur any financial penalty or fee or refund any amounts in connection with, or as a result of, the Transactions). For the avoidance of doubt, except with the written consent of Seller (not to be unreasonably withheld, conditioned or delayed), Seller or its Affiliate shall be the sole party that contacts or communicates with any counterparty to third-party Consents (other than to the extent subject to Section 6.8) concerning this Agreement, the other Transaction Agreements or the Transactions; provided that, Seller and its applicable Controlled Affiliates shall use reasonable best efforts to consult in good faith with Purchaser and to keep Purchaser reasonably informed, in each case, regarding the obtaining of any third-party Consents.

Notwithstanding anything in this <u>Section 6.4(c)</u> to the contrary, (i) none of Seller or its Affiliates shall be obligated to pay any money to any Person or to offer or grant other financial or other accommodations to any Person in connection with obtaining any such Consent and (ii) the failure, in and of itself, to obtain any Consent shall not, in and of itself, constitute evidence of any breach of this <u>Section 6.4(c)</u>.

- (d) Notwithstanding anything to the contrary contained in this Agreement, neither Seller nor any of its Affiliates shall (i) be required to expend any money, commence or participate in any Proceeding, incur liabilities or offer or grant any accommodation (financial or otherwise) to any third party to obtain any Consent described in this Section 6.4 or (ii) have any obligation pursuant to this Section 6.4 with respect to any Contract that is governed by the Transition Services Agreement. No representation, warranty or covenant of Seller contained in this Agreement or the other Transaction Agreements shall be breached or deemed breached, and no condition shall be deemed not satisfied, based on (A) the failure, in and of itself, to obtain any Consent (including any third-party consent) or (B) any Proceeding commenced or threatened by or on behalf of any Person arising out of, relating to or resulting from (1) the failure, in and of itself, to obtain any Consent (including any Third-Party Consent) described in this Section 6.4 or (2) any arrangement between Seller or Purchaser, in and of itself, entered into pursuant to Section 6.4(b).
- (e) From and after the Closing, (i) Purchaser shall indemnify and hold harmless Seller and its Affiliates from and against all Losses arising from or relating to the Business Portion of any Shared Contract, (ii) Seller shall indemnify and hold harmless Purchaser and the Acquired Companies from and against all Losses arising from or relating to the Non-Business Portion of any Shared Contract, (iii) Purchaser and the Acquired Companies shall not extend the term or otherwise amend the terms of any Shared Contract (that has not been assigned or otherwise divided, modified or replicated pursuant to Section 6.4(a)) in a manner that would adversely affect Seller or any of its Affiliates in any material respect without Seller's prior written consent and (iv) Seller and its Affiliates shall not extend the term or otherwise amend the terms of any Shared Contract (that has not been assigned or otherwise divided, modified, replicated or novated pursuant to Section 6.4(a)) in a manner that would adversely affect Purchaser or the Acquired Companies in any material respect without Purchaser's prior written consent.

6.5 <u>Termination of Intercompany Agreements; Release of Guarantees.</u>

(a) Except (i)for this Agreement and the other Transaction Agreements (and each other agreement or instrument expressly contemplated by this Agreement or any other Transaction Agreement to be entered into by Seller and its Controlled Affiliates (other than the Acquired Companies), on the one hand, and any Acquired Company, on the other hand), (ii)any Contracts to which any third party is a party (including the Shared Contracts), (iii)the transactions contemplated by Section 6.18 or (iv)the Contracts listed on Schedule 6.5(a)(iv) of the Seller Disclosure Schedule, any intercompany Contract (other than commercial arrangements entered into on arms' lengths terms and listed on Schedule 6.5(a)(v) of the Seller Disclosure Schedule), arrangements, financing agreements, tax sharing agreements, intercompany loans, transactions, accounts, commitments and claims between the Acquired Companies, on the one hand, and Seller or any of its Affiliates (other than an Acquired Company), on the other hand (the "Intercompany Agreements"), shall be terminated (or deemed terminated without any further action on the part of

any party thereto) effective with respect to the Acquired Companies no later than as of the Closing without any party having any continuing obligations or liability to the other party under the Intercompany Agreements. Seller shall, and shall cause the Acquired Companies to, settle all amounts due and payable to an Acquired Company from Seller or any of its Affiliates (other than an Acquired Company), and all amounts due and payable to Seller or any of its Affiliates (other than an Acquired Company) from an Acquired Company, in each case, prior to the Closing and in a manner that does not result in an increase of any liabilities of the Acquired Companies following the Closing.

- Until the earlier of the Closing and the termination of this Agreement pursuant to its terms, each of Seller and Purchaser shall cooperate and use commercially reasonable efforts to, effective as of the Closing, terminate or cause to be terminated, or cause Purchaser or one of its Affiliates (including, as of the Closing, the Acquired Companies) to be substituted in all respects for Seller or any of its Affiliates (other than the Acquired Companies) in respect of, all liabilities of Seller or any of its Affiliates (other than the Acquired Companies) under any guarantee set forth on Schedule 6.5(b) of the Seller Disclosure Schedule (the "Intercompany Guarantees"). In the case of the failure to do so by the Closing Date, then each of Seller and Purchaser shall continue to cooperate and use commercially reasonable efforts as described in the preceding sentence, and Purchaser shall (i) use commercially reasonable efforts to provide replacement guarantees or other assurances of payment on behalf of Purchaser or one of its Affiliates (including, as of the Closing, the Acquired Companies) with respect to any outstanding Intercompany Guarantee, (ii) indemnify and hold harmless Seller and its Affiliates (other than the Acquired Companies) from and against all Losses actually suffered or incurred following the Closing arising from or relating to such Intercompany Guarantees and (iii) not permit the Acquired Companies or their Affiliates to (A) renew or extend the term of or (B) increase the obligations of the Acquired Companies or their Affiliates under, or transfer to another third party, any Contract or letter of credit or other liability or obligation for which Seller or any of its Affiliates (other than the Acquired Companies) is or would reasonably be likely to be liable under such guarantee. To the extent that Seller or any of its Affiliates (other than the Acquired Companies) has performance obligations under any such Intercompany Guarantee, Purchaser shall use its commercially reasonable efforts to (1) fully perform or cause to be fully performed such obligations on behalf of Seller or such Affiliate or (2) otherwise take such action as is reasonably requested by Seller so as to place Seller or such Affiliate in the same position as if Purchaser had performed or were performing such obligations.
- (c) Purchaser acknowledges and agrees that, without limiting Section 4.16 in any respect, (i) the Business as presently conducted receives or benefits from general corporate functions furnished by Seller and its Affiliates, including finance, accounting, tax, human resources, legal, information technology, facilities, facilities security, procurement and other ancillary or corporate shared services provided by Seller and its Affiliates (other than the Acquired Companies) or other corporate centralized functional organizations within or controlled by Seller and its Affiliates (other than the Acquired Companies), including pursuant to Excluded Shared Contracts, (ii) Seller and its Affiliates may, and have the affirmative right to, terminate, amend or modify Excluded Shared Contracts (including any portion thereof applicable to the Business and the Acquired Companies) to the extent such termination, amendment or modification (A) relates to the general corporate shared services provided by Seller or any such Controlled Affiliate (and to be retained by Seller or such Controlled Affiliate in connection with the Transactions) and (B)

would not reasonably be expected to result in a material detriment to the Acquired Companies or the Business (<u>provided</u> that this clause (ii) shall not limit Seller's obligations under the Transition Services Agreement) and (iii) effective as of the Closing, the sole obligations of Seller and its Affiliates with respect to the provision of any general corporate functions to the Business shall be if, and then to the extent, set forth in the Transaction Services Agreement.

6.6 <u>Seller Debt Facilities Releases</u>. Seller shall use reasonable best efforts to deliver to Purchaser at or prior to the Closing customary documentation evidencing the release of the Acquired Companies from each of the Seller Debt Facilities (the "<u>Seller Debt Facilities Releases</u>"). The Seller Debt Facilities Releases shall provide that all obligations of the Acquired Companies in respect of Indebtedness, all guarantees and security provided by the Acquired Companies, and in the case of any factoring programs or arrangements, all assets of the Acquired Companies subject to such programs or arrangements, in each case under the Seller Debt Facilities shall be released or terminated, as applicable, upon the consummation of the Transactions.

6.7 <u>Confidentiality</u>.

- (a) The terms of the Non-Disclosure Agreement are incorporated into this Agreement by reference and shall continue in full force and effect (and all obligations thereunder shall be binding upon Purchaser and its Representatives (as defined in the Non-Disclosure Agreement) as set forth therein) until the Closing, at which time the obligations under the Non-Disclosure Agreement shall terminate; provided, however, that Purchaser's confidentiality obligations under the Non-Disclosure Agreement shall terminate only in respect of that portion of the Evaluation Material (as defined in the Non-Disclosure Agreement) to the extent relating to the Acquired Companies and the Business and, for all other Evaluation Material (as defined in the Non-Disclosure Agreement) ("Non-Business Confidential Material"), the term of the Non-Disclosure Agreement shall continue to apply to such Non-Business Confidential Material until the termination or expiration of the Non-Disclosure Agreement in accordance with its terms. If for any reason the Closing does not occur, the Non-Disclosure Agreement shall continue in full force and effect in accordance with its terms. In the event of a conflict or inconsistency between the terms expressly set forth in this Agreement (rather than incorporated by reference herein) and the Non-Disclosure Agreement, the terms of this Agreement will govern.
- (b) During the three (3) -year period following the Closing, Seller shall keep confidential and refrain from using, and cause its Controlled Affiliates and its and their respective Representatives to keep confidential and refrain from using, all non-public, confidential or proprietary information concerning the Acquired Companies or the Business, except (i) as required or requested by a Governmental Authority or required pursuant to Legal Requirements or the rules or regulations of any securities exchange or listing authority or legal, administrative or judicial process (provided, that Seller shall, to the extent permitted by Legal Requirements, promptly notify Purchaser of such requirement or request and the disclosure that is expected to be made with respect thereto with reasonable specificity and, to the extent requested by Purchaser, shall reasonably cooperate with Purchaser to seek a protective order or other appropriate remedy to limit or obtain confidential treatment for such disclosure, and in the event no such protective order or remedy is obtained, Seller will furnish only that portion of such non-public, confidential or proprietary information which Seller is advised by counsel is required by Legal Requirements and will exercise commercially reasonable efforts to obtain reliable assurance that confidential

treatment will be accorded to such non-public, confidential or proprietary information), (ii) for information that is available as of immediately following the Closing generally to the public, or thereafter becomes generally available to the public, other than as a result of a breach of this Section 6.7(b), (iii) to the extent such use is strictly necessary in order to enable Seller to fulfil its obligations to Purchaser and the Acquired Companies under any other Transaction Agreement, (iv) for information disclosed to Seller or any of its Affiliates following the Closing Date on a non-confidential basis by any Person not known by Seller after reasonable inquiry to be bound by an obligation of confidentiality to Purchaser or any of its Affiliates or (v) is demonstrated by Seller or its Affiliates to have been independently developed following the Closing Date not in violation of its or its Representatives' obligations under this Section 6.7(b) and without reference to any non-public, confidential or proprietary information concerning the Acquired Companies or the Business or any information from a source that is subject to a confidentiality obligation to the Acquired Companies or the Business or is otherwise prohibited from furnishing such information to Seller, its Affiliates or their respective Representatives.

6.8 Reasonable Best Efforts; Cooperation; Regulatory Filings.

Each of Seller and Purchaser shall, and shall cause its respective Affiliates to, use reasonable best efforts to take, or cause to be taken, all actions, and to do, or cause to be done, and to assist and cooperate in doing, all things necessary, proper or advisable under applicable Antitrust Laws, CFIUS Authorities and other applicable Legal Requirements to consummate and make effective the Transactions, which actions include (i) using reasonable best efforts to obtain as promptly as practicable each Consent, Permit and Order of any Governmental Authority that may be, or become, necessary for the consummation of the Transactions (collectively, "Governmental Approvals"), (ii) cooperating in determining which filings are required or advisable to obtain the Governmental Approval of, or any exemption by, any Governmental Authority, (iii) furnishing all information and documents required by or advisable under applicable Legal Requirements in connection with Governmental Approvals of, or filings with, any Governmental Authority, (iv) filing, or causing to be filed, as promptly as practicable following the execution and delivery of this Agreement, applicable notifications with the necessary Governmental Authorities, including, as applicable, the U.S. Department of State, Directorate of Defense Trade Controls ("DDTC"), (v) using reasonable best efforts to obtain as promptly as practicable the expiration or termination of any waiting period or any Consent under the HSR Act and any other applicable Antitrust Laws, (vi) using reasonable best efforts to obtain as promptly as practicable CFIUS Clearance, (vii) using reasonable best efforts to obtain as promptly as practicable DCSA Approval and (viii) defending any actions, whether judicial or administrative, challenging this Agreement or the consummation of the Transactions, including seeking to have any Order entered by any court or other Governmental Authority vacated or reversed. Purchaser and Seller shall coordinate with respect to all aspects of strategy and communications regarding matters related to the HSR Act and any other Governmental Approvals, including determining which filings are required or advisable to obtain the Governmental Approval of, or any exemption by, any Governmental Authority; provided, that, without limiting Purchaser's other obligations under this Section 6.8 (including the affirmative actions Purchaser and its Affiliates are required to take). Purchaser shall be entitled to make the final determination as to the appropriate course of action. In furtherance and not in limitation of the foregoing, each party agrees that it will use its reasonable best efforts to file or cause to be made (A) as promptly as practicable, but in any event no later than ten (10) Business Days following the date of this Agreement, (w) any required

notification under the ITAR with DDTC, (x) any required notification and report forms under the HSR Act with the United States Federal Trade Commission (the "FTC") and the United States Department of Justice (the "DOJ"), (y) a draft joint voluntary notice under the CFIUS Authorities with CFIUS (and subsequently take such actions described in Section 6.8(f))) and (z) an initial notification to DCSA of the transactions pursuant to the current National Industrial Security Program Operating Manual Rule, 32 C.F.R. Part 117 (the "NISPOM Rule"), and any other applicable U.S. national industrial security regulations (and subsequently take such actions described in Section 6.8(g)) and (B) as promptly as practicable, any filing with or notification to any other competent Governmental Authorities set forth on Schedule 6.8(a) of the Seller Disclosure Schedule.

In connection with, and without limiting, the efforts referenced in Section 6.8(a), each of Seller and Purchaser shall, and shall cause its respective Affiliates to, (i) furnish to the other party such necessary information and reasonable assistance as the other may reasonably request in connection with its preparation of any filing or submission that is necessary under the HSR Act or any other applicable Antitrust Law, under the ITAR or any other applicable Customs & Trade Laws, under the CFIUS Authorities or under other applicable Legal Requirements (provided, that, each of Seller and Purchaser may, as each may determine is reasonably necessary, designate competitively sensitive or national security sensitive materials and information provided to the other pursuant to this Section 6.8(b) as "Outside Counsel Only", and such materials and information shall be given only to the outside legal counsel of the recipient and will not be disclosed by such outside counsel to directors, officers or employees of the recipient unless express permission is obtained in advance from the source of the materials (Seller or Purchaser, as the case may be) or its legal counsel, and that any materials shared may be redacted before being provided to the other party (A) to remove references concerning the valuation of the Business and (B) as necessary to comply with contractual arrangements; provided, further, that the parties need not provide each other any exhibits to communications providing the personal identifying information required by the CFIUS Authorities or any communications that are otherwise requested by CFIUS to remain confidential), (ii) permit the other party to review any filing or submission prior to forwarding to the FTC, the DOJ, CFIUS, DCSA and other Governmental Authorities (except where such material is confidential to a party, in which case it will be provided, subject to applicable Legal Requirements, to the other party's counsel on an "external counsel only" basis) and consider in good faith any reasonable comments made by that other party, (iii) keep each other apprised of the status of any communications with, and any inquiries or requests for additional information from, any Governmental Authorities and comply as promptly as practicable with any such inquiry or request (and in any event in accordance with applicable regulatory requirements) and (iv) not participate in any substantive meeting or discussion, either in person or by telephone or videoconference, with any Governmental Authority in connection with the Transactions, unless it (A) consults with the other party in advance and (B) gives the other party the opportunity to attend and participate; provided that a party shall not be required to give the other party the opportunity to attend and participate to the extent (1) prohibited by such Governmental Authority or (2) such Governmental Authority explicitly requests to communicate exclusively with one party). Whether or not the Transactions are consummated, Purchaser shall be responsible for the payment of all filing fees and disbursements to any third party expert retained by Purchaser in connection with obtaining any approvals or making the notifications or filings required pursuant to this Section 6.8, including, for the avoidance of doubt, with respect to CFIUS Clearance and the implementation of any FOCI Mitigation Plan. Purchaser

hereby agrees to use reasonable best efforts to provide, or cause its applicable Controlled Affiliates to provide, such security and assurances as to financial capability, resources and creditworthiness as may be requested by any such Governmental Authority or other third party whose Consent is sought in connection with the Transaction.

- (c) Purchaser and Seller shall not, and shall cause their respective Affiliates not to, (i) acquire or agree to acquire by merging or consolidating with, or by purchasing the assets of or equity in, or by any other manner, any Person or portion thereof, or otherwise acquire or agree to acquire any assets or (ii) agree to consummate any license or co-development or production agreement (for the avoidance of doubt, this clause (ii) shall not apply to commercial activities in the ordinary course of business), if, in each case, the entering into of a definitive agreement relating to or the consummation of such acquisition, merger, consolidation, transaction, license or agreement, would reasonably be expected to (A) impose material delay in the obtaining of, or materially increase the risk of not obtaining, any Governmental Approvals or the expiration or termination of any applicable waiting period under any Legal Requirement, (B) materially increase the risk of any Governmental Authority entering an Order prohibiting or materially delaying the consummation of the Transaction or (C) prevent or materially delay the consummation of the Transactions.
- Notwithstanding anything in this <u>Section 6.8</u> to the contrary, nothing in this Agreement shall require, or be deemed to require (i) Purchaser or Seller, or any of their respective Affiliates, to propose, negotiate, offer to commit, effect or agree to any sale, divestiture, license or disposition of assets or businesses, or to any behavioral remedy, in each case that is not conditioned upon the consummation of the Transactions or (ii) Seller, or any of its Affiliates, to propose, negotiate, offer to commit, effect or agree to any sale, divestiture, license or disposition of assets or businesses, or to any behavioral remedy, of Seller or any of its Affiliates (other than of the Acquired Companies or the Business, subject to clause (i) above that it be conditioned upon the consummation of the Transactions). Notwithstanding anything in this Section 6.8 to the contrary, none of Seller or any of its Affiliates shall under any circumstance be required to pay or commit to pay any amount or incur any obligation in favor of or offer or grant any accommodation (financial or otherwise, regardless of any provision to the contrary in the underlying Contract, including any requirements for the securing or posting of any bonds, letters of credit or similar instruments, or the furnishing of any guarantees) to any Person to obtain any Consent, including the actions set forth in this Section 6.8. None of Seller or any of its Affiliates shall have any liability whatsoever to Purchaser arising out of or relating to the failure, in and of itself, to obtain any Governmental Approvals or other Consents that may be required in connection with the Transactions or because of the termination, in and of itself, of any Contract as a result thereof. Purchaser acknowledges that no representation, warranty or covenant of Seller contained herein shall be breached or deemed breached solely as a result of (i) the failure, in and of itself, to obtain any Governmental Approval or other Consent required in connection with the Transactions, (ii) any termination, in and of itself, of a Contract or (iii) any Proceeding commenced or threatened, in and of itself, by or on behalf of any Person arising out of or relating to the failure to obtain any such Governmental Approval or Consent or any such termination.
- (e) Notwithstanding anything in this Agreement to the contrary (but subject to Section 6.8(d)), the "reasonable best efforts" of Purchaser shall be deemed to include, and Purchaser shall, and shall cause its Subsidiaries to, take any and all actions necessary or advisable

required under or in connection with any Antitrust Laws or in connection with obtaining CFIUS Clearance and DCSA Approval and to avoid or eliminate each and every impediment under any Antitrust Laws or in connection with obtaining CFIUS Clearance (including all actions necessary to mitigate any national security concerns as may be requested or required by CFIUS in connection with, or as a condition to, the receipt of CFIUS Clearance) or DCSA Approval (including all actions necessary to mitigate any national security concerns as may be requested or required by DCSA in connection with, or as a condition to, DCSA Approval), in each case, to cause the Closing and the Transactions to occur as soon as practicable following the date of this Agreement and, in any event, prior to the Outside Date, including (i) expeditiously complying with any requests or inquiries for additional information or documentation (including any "Second Request") by any Governmental Authority, (ii) offering, negotiating, committing to and effecting, by consent decree, hold separate order or otherwise, (A) the sale, divestiture, license or other disposition or encumbrance of any and all of the capital stock, assets, rights, products or businesses of Purchaser, Purchaser's Subsidiaries, the Business or the Acquired Companies, (B) the entrance into, and the assignment, amendment, modification or termination of, any Contracts or other arrangements of Purchaser, Purchaser's Subsidiaries or the Acquired Companies and (C) any behavioral limitations, conduct restrictions or other restrictions or commitments on or with respect to the activities, businesses, services, products, product lines or assets of Purchaser, Purchaser's Subsidiaries, the Acquired Companies or the Business, including terminating, amending or assigning existing relationships and contractual rights and obligations, continuing certain lines of business and agreeing to restrictions on pricing, settling any pending or threatened Proceeding (other than the type contemplated by the following clause (iii), including the payment of any Losses in settlement thereof) and (iii) contesting, defending and appealing any threatened or pending Proceeding or preliminary or permanent injunction or other Order or Legal Requirement that would adversely affect the ability of any party to consummate, or otherwise delay the consummation of, the Transactions: provided, however, that Purchaser shall not be required to take any of the actions contemplated by this Section 6.8(e), or offer or commit to take any such actions, in the event that such action contemplated by this Section 6.8(e) would reasonably be expected to result in a material adverse effect on the business. operations, financial condition or results of operations of Purchaser, Purchaser's Subsidiaries (including the Acquired Companies) or the Business, in each case measured on a scale relative to the size of the Business. For purposes of the CFIUS Clearance and the DCSA Approval, Seller shall use reasonable best efforts to cause the Acquired Companies to reasonably support Purchaser in its discussions with CFIUS and DCSA on the nature and scope of any mitigation conditions to achieve the least possible impact on the Acquired Companies and the Business; provided that, notwithstanding anything in this Section 6.8 to the contrary, neither Purchaser or the Acquired Companies shall be required to agree to any CFIUS or DCSA condition that would have the effect of altering the board and governance structure under Purchaser's existing Special Security Agreement; provided, further, that any mitigation conditions shall become effectively only from and after, the Closing.

to obtain expiration or termination of the required waiting periods and any consents, clearances or approvals

(f) Following the filing of the draft joint voluntary notice with CFIUS in accordance with Section 6.8(a), the parties shall promptly provide CFIUS with any additional or supplemental information requested by CFIUS with respect to such draft joint voluntary notice, and promptly (and in any event, no later than five (5) Business Days of receiving questions or comments from CFIUS on the draft joint voluntary notice) submit the final joint voluntary notice. The parties shall use reasonable best efforts to promptly respond (and, in any event, in accordance

with applicable Legal Requirements) to any request for additional information, documents or other materials received after filing of the joint voluntary notice. Each of the parties shall respond to any request for information from CFIUS in the timeframe set forth in the CFIUS Authorities; provided, that any party, after consultation with each such other party, may request in good faith an extension of time pursuant to the CFIUS Authorities to respond to CFIUS requests for follow-up information; provided, further, that under no circumstance may a party request any extension that would reasonably be expected to cause CFIUS to reject the joint voluntary notice.

- (g) With respect to the DCSA Approval, as promptly as practicable following the initial notification made in accordance with Section 6.8(a), the parties shall use reasonable best efforts to provide, or cause to be provided, the information necessary for DCSA to conduct a review of foreign ownership, control or influence pursuant to the current NISPOM Rule and any other applicable U.S. national industrial security regulations, and no later than fifteen (15) calendar days after the date of this Agreement, Purchaser shall submit to DCSA, and Seller shall cooperate in the submission of, a FOCI Mitigation Plan.
- 6.9 <u>Financing</u>. Purchaser shall take, or cause to be taken, all actions and do, or cause to be done, all things necessary, advisable or proper to obtain funds sufficient to pay all amounts payable in connection with the Transactions and the transactions contemplated by the other Transaction Agreements (including all of Purchaser's and its Affiliates' costs and expenses incurred in the evaluation, negotiation and execution of the Transactions and the transactions contemplated by the other Transaction Agreements). Notwithstanding anything in this Agreement to the contrary, Purchaser acknowledges and agrees that Purchaser's obligations under this Agreement are not conditioned in any manner whatsoever upon Purchaser obtaining the funds to satisfy its funding obligations contained in this Agreement, and the obtaining of the Debt Financing is not a condition to the Closing or the consummation of the Transactions.

6.10 <u>Financing Cooperation</u>.

- (a) Prior to the Closing, at Purchaser's sole cost and expense, Seller shall cause the Acquired Companies to use commercially reasonable efforts to provide to Purchaser such customary cooperation reasonably requested in writing by Purchaser in connection with the Debt Financing; <u>provided</u> that such requests are timely made so as not to delay the Closing beyond the date that it would otherwise occur. Such cooperation shall include using commercially reasonable efforts to do the following (in each case, to the extent so requested as set forth above):
 - (1) causing management teams of the Acquired Companies, with appropriate seniority and expertise, at reasonable times and locations mutually agreed and upon reasonable advance written notice, to participate in a reasonable number of meetings, conference calls, drafting sessions, due diligence sessions and similar presentations to and with prospective lenders and rating agencies (with all of the foregoing to be virtual at Seller's or such Persons' request);
 - (2) reasonably assisting with the preparation of customary rating agency presentations, bank information memoranda and other customary marketing and syndication materials (which may only be distributed to a third party to the extent permitted by the Non-Disclosure Agreement or a separate confidentiality agreement entered into

between Seller and such third party) reasonably and customarily required and reasonably requested by the Debt Financing Sources in connection with the Debt Financing, in each case solely with respect to information relating to the Acquired Companies; and

- (3) furnishing Purchaser and the Debt Financing Sources, no later than three (3) Business Days prior to the Closing Date, with all documentation and other information required by regulatory authorities under applicable "know your customer" and anti-money laundering rules and regulations, including the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001, that has been reasonably requested by Purchaser in writing, at least fifteen (15) calendar days prior to the Closing Date.
- Purchaser acknowledges and agrees that Seller, its Affiliates and their respective Representatives shall not have any responsibility for, or incur any liability to any Person under, any financing that Purchaser may raise in connection herewith, or any cooperation provided pursuant to this Section 6.10. Purchaser shall (i) promptly reimburse Seller and any of its Affiliates, and its and their respective Representatives (collectively, the "Seller Indemnitees") for all of the out-of-pocket costs and expenses (including attorneys' fees) incurred by such Seller Indemnitee in connection with this Section 6.10 and (ii) indemnify and hold harmless the Seller Indemnitees from and against all Taxes and Losses suffered or incurred, directly or indirectly, by any of them in connection with the Debt Financing (including any action pursuant to this Section 6.10), or any information used in connection therewith. Any use of Seller's or its Controlled Affiliates' logos in connection with the Debt Financing shall require Seller's prior written consent; provided, however, that in the event and to the extent Seller provides such consent, such logos shall be used solely in a manner that is reasonable and customary for such purposes and that is not intended to or reasonably likely to harm or disparage or otherwise adversely affect Seller or any of its Controlled Affiliates or the reputation or goodwill of Seller or any of its Affiliates or any of its or their respective products, services, offerings or Intellectual Property. All non-public or other confidential information provided by Seller, its Affiliates or their respective Representatives pursuant to this Section 6.10 shall be kept confidential in accordance with, and shall be subject to the terms of, the Non-Disclosure Agreement. Seller, its Affiliates and their respective Representatives shall be given a reasonable opportunity to review all presentations, bank information memoranda and similar marketing materials, materials for rating agencies and other documents prepared by or on behalf of or used by Purchaser or any of its Affiliates or used or distributed to any Debt Financing Source or any of its Affiliates in connection with the Debt Financing that include any logos of or information about or provided by the Business, Seller, its Affiliates, or their respective Representatives, and any such presentations, memoranda, materials or documents shall include a conspicuous disclaimer to the effect that none of Seller, its Affiliates or their respective Representatives has any responsibility or liability for the content of such document and that Seller, its Affiliates and their respective Representatives disclaim all responsibility therefor.
- (c) Nothing in this $\underline{\text{Section } 6.10}$ shall require any of Seller, its Affiliates or their respective Representatives to:
 - (1) waive or amend any terms of this Agreement or any Transaction Agreement or pay, agree to pay or reimburse any commitment or other fee or any expenses

in connection with any Debt Financing (other than, in the case of any such fees or expenses, those of the Acquired Companies that only take effect upon the Closing and that terminate with no liability to Seller or any of its Affiliates or their respective Representatives upon termination of this Agreement);

- (2) take any action that would, or would reasonably be expected to, result in Seller or any of its Affiliates or their respective Representatives incurring any actual or potential liability or giving or being required to give any indemnity in connection with the Debt Financing (other than those of the Acquired Companies that only take effect upon the Closing and that terminate with no liability to Seller or any of its Affiliates or their respective Representatives upon termination of this Agreement);
- (3) take any action that would require Seller or any of its Affiliates or their respective Representatives to execute, deliver, enter into or perform any document, agreement, certificate or instrument with respect to the Debt Financing (other than those of the Acquired Companies that only take effect upon the Closing and that terminate with no liability to Seller or any of its Affiliates or their respective Representatives upon termination of this Agreement), or provide (or cause any of their Representatives to provide) any accountants' comfort letter, reliance letter, legal opinion or other opinion of counsel;
 - (4) adopt resolutions or execute consents to approve or authorize the Debt Financing;
- (5) take any action that would unreasonably interfere with the Business or the other businesses or operations of Seller or any of its Affiliates or their respective Representatives;
- (6) take any action that would cause any representation or warranty in this Agreement or any Transaction Agreement to be breached or become inaccurate or that would breach any covenant in this Agreement or any Transaction Agreement;
- (7) take any action that would conflict with or violate, or that could reasonably be expected to conflict with or violate, the organizational documents of Seller or any of its Affiliates or applicable Legal Requirements;
- (8) take any action that would result in the contravention of, or that could reasonably be expected to result in a violation or breach of, or a default under, any Contract to which Seller or any of its Affiliates or their respective Representatives is a party or bound or any obligations of confidentiality binding on Seller or any of its Affiliates or their respective Representatives;
- (9) provide access to or disclose information that constitutes attorney work product or that Seller determines would jeopardize any attorney-client privilege of Seller or any of its Affiliates or their respective Representatives or which is restricted or prohibited under applicable Legal Requirements;
- (10) cause any director, officer, employee or other Representative of Seller or any of its Affiliates to incur any actual or potential personal liability; or

- (11) prepare any financial statements, projections or other similar materials.
- (d) Notwithstanding anything in this Agreement to the contrary, the parties agree that any action taken or omitted to be taken by Seller, its Affiliates or any of their respective Representatives with respect to the matters contemplated by this <u>Section 6.10</u> will not be taken into account for purposes of determining whether any of the conditions contained in <u>Section 10.1</u> have been satisfied or whether any of Purchaser's rights of termination arise under <u>Section 11.1(c)</u>.

6.11 Insurance.

- Purchaser acknowledges and agrees that (i) the coverage under all insurance policies or (a) self-insurance policies or programs, including those relating to the Acquired Companies and the Business, arranged or maintained by or for the benefit of Seller or any of its Affiliates (in each case other than insurance policies or self-insurance policies or programs that are maintained and held exclusively by the Acquired Companies) (collectively, the "Insurance Policies") shall not be available or transferred to Purchaser, the Acquired Companies or the Business, (ii) as of the Closing, the Acquired Companies and the Business shall cease to be insured by the Insurance Policies and (iii) it is Purchaser's sole responsibility to arrange for its own insurance policies or self-insurance policies or programs with respect to the Acquired Companies and the Business. Notwithstanding the foregoing, Seller shall use commercially reasonable efforts to direct any carriers for any of its third-party Insurance Policies (excluding, for clarity, self-insurance policies or programs) affording coverage for the Acquired Companies or the Business to continue to process any claims made thereunder by the Acquired Companies or the Business to the extent such claims were made prior to the Closing and reasonably cooperate with Purchaser following the Closing in connection therewith (in all cases if and to the extent permissible under the terms and conditions of the applicable Insurance Policies), and any such claims shall be further subject to Section 6.11(b) mutatis mutandis.
- Notwithstanding the foregoing in Section 6.11(a), with respect to any events or circumstances relating to the Acquired Companies or the Business that occurred or existed prior to the Closing that are covered by occurrence-based third-party Insurance Policies (excluding, for clarity, self-insurance policies or programs) (the "Occurrence-Based Policies"), Purchaser or the Acquired Companies, as applicable, may, at Purchaser's reasonable request, after the Closing and subject to prior consultation with Seller, make claims and seek coverage with respect to such events or circumstances under such Occurrence-Based Policies and retain claims made prior to the Closing subject to the terms and conditions of such Occurrence-Based Policies; provided, that (i) Purchaser and the Acquired Companies shall exclusively bear the amount of any "deductibles", retentions or premium increases associated with any such claims under such Occurrence-Based Policies and shall otherwise be liable for all uninsured or unrecovered amounts of such claims (provided this shall not be construed to affect or limit the ability of Purchaser or its Affiliates (including the Acquired Companies) to recover under the R&W Insurance Policy), (ii) Purchaser and the Acquired Companies shall be liable, and shall reimburse Seller, for any fees, costs or expenses incurred by Seller or its Affiliates through the insurers or reinsurers of such Occurrence-Based Policies relating to such claims, (iii) any amounts to be remitted to Purchaser or the Acquired Companies in respect of any insurance claim under this Section 6.11 shall be paid net of any amounts incurred by Seller or its Affiliates in accordance with clause (ii) and (iv) with respect

Occurrence-Based Policies, Seller shall use commercially reasonable efforts not to waive or settle such insurance claims without the consent of Purchaser and, at Purchaser's cost and upon Purchaser's reasonable request, shall use commercially reasonable efforts to cooperate with Purchaser and the Acquired Companies, as applicable, in filing any insurance claims and in the collection of insurance proceeds and at Purchaser's reasonable request and sole cost and expense (including, for clarity, prompt reimbursement to Seller of its reasonable and documented out-of-pocket costs and any expenses, including of its Representatives), Seller shall use commercially reasonable efforts to pursue, or cause to be pursued, collection of insurance proceeds unless Seller reasonably determines, in consultation with Purchaser, that collection of a material amount of insurance proceeds is not reasonably likely. Seller or its applicable Affiliate (other than the members of the Acquired Companies) may amend or modify any insurance policy or program (including the Occurrence-Based Policies) in the manner it deems appropriate to give effect, from and after the Closing, to this Section 6.11 or for any other purpose in the ordinary course of business; provided that (x) no such amendment or modification may disproportionately impair the rights of Purchaser and the Acquired Companies to access the Occurrence-Based Policies as set forth herein as compared to the rights of Seller and its other Affiliates with respect to the Occurrence-Based Policies, (y) Seller or its applicable Affiliates shall use commercially reasonable efforts to cause the terms of such policies, including any deductibles and self-insured retentions, to remain unaffected by the terms of this Agreement with respect to events prior to and including the Closing Date and (z) Seller or its applicable Affiliate (other than the members of the Acquired Companies) shall use commercially reasonable efforts to maintain in effect each Occurrence-Based Policy until the expiration of the term of such policy or program (and not terminate such policy or program prior to such expiration). This Section 6.11 shall not be considered as an attempted assignment of any policy of insurance or as a contract of insurance, and nothing in this Section 6.11 is intended to waive or abrogate in any way Seller's own rights to insurance coverage for any liability. Purchaser must give Seller notice of any claim to be made under the Occurrence-Based Policies in accordance with this Section 6.11 no later than the third (3rd) anniversary of this Agreement, after which Purchaser shall no longer have the right to make new claims under this Section 6.11.

to coverage claims or requests for benefits asserted by Purchaser or the Acquired Companies under such

- (c) With respect to any claims-made policies set forth on Schedule 6.11(c) of the Seller Disclosure Schedule, prior to the Closing, the Acquired Companies shall, at the written request and direction of Purchaser reasonably in advance of the Closing Date, use commercially reasonable efforts to obtain, at Purchaser's sole expense and with effect from the Closing, a "tail" extension of any existing claims-made policy (as so requested by Purchaser) issued to or maintained or held by the Acquired Companies or covering any of the Acquired Companies or the Business (other than D&O Insurance, which is addressed in Section 7.3(c)), for a claims reporting or discovery period of three years (if three years is permitted by such policy) from and after the Closing, with respect to any claims related to any period of time at or prior to the Closing, with terms, conditions, retentions and limits of liability that are no less favorable in the aggregate than the coverage provided under each such policy existing at the Closing, and with Purchaser and the Acquired Companies as named insureds.
- 6.12 <u>R&W Insurance Policy</u>. If Purchaser or any of its Affiliates elects to obtain a buy-side representations and warranties insurance policy in connection with the Transactions (the "<u>R&W Insurance Policy</u>"), Purchaser agrees that the R&W Insurance Policy shall provide that (a)

the insurer shall irrevocably waive and not pursue, directly or indirectly, any claims against Seller or any of its Affiliates or Representatives (by way of subrogation, claim for contribution or otherwise) in connection with this Agreement and the Transactions, other than in the case of Fraud and then only to the extent of such Fraud and (b) Seller and its Affiliates or Representatives shall be express third-party beneficiaries of such provision. Purchaser shall not (and shall cause its Affiliates not to) amend or modify in any material respect that is adverse to Seller and its Affiliates or Representatives, or otherwise novate, assign, waive or terminate, in each case the provisions in clauses (a) and (b) of the immediately preceding sentence without the prior written consent of Seller, which consent shall be in Seller's sole discretion. Purchaser shall be solely responsible for all costs to procure, maintain and make claims under the R&W Insurance Policy, including all premiums, broker fees, underwriting fees, retentions, Taxes, expenses and costs of any nature whatsoever. Purchaser acknowledges and agrees that the absence of coverage under the R&W Insurance Policy for any reason, including the insolvency of, or breach of any R&W Insurance Policy by, any insurer thereunder, the failure of Purchaser to file notices or claims that are timely and sufficient under any R&W Insurance Policy, or the failure by any insurer under any R&W Insurance Policy to make any payments to Purchaser under such R&W Insurance Policy, or to deny coverage, for any reason, under such R&W Insurance Policy shall not expand, alter, amend, change or otherwise affect Seller's liability under this Agreement.

6.13 <u>Litigation Support</u>.

Subject to Section 6.7, Section 6.8 and Section 12.16, in the event and for so long as (i) (a) Seller or any of its Affiliates is defending any Proceeding, charge or demand by a third party (other than an action brought against or by Purchaser or any of its Affiliates) or otherwise addressing, negotiating, disputing, investigating, complying with, mitigating, discharging or otherwise performing or managing any Loss in connection with (A) the Transactions (including in connection with recovering under the R&W Insurance Policy) or (B) any fact, situation, circumstance, status, condition, activity, practice, plan, occurrence, event, incident, action, failure to act or transaction relating to, in connection with or arising from the pre-Closing activity of the Business or the Acquired Companies, or (ii) Purchaser or any Acquired Company is defending any Proceeding, charge or demand by a third party (other than an action brought against or by Seller or any of its Affiliates) or otherwise addressing, negotiating, disputing, investigating, complying with, mitigating, discharging or otherwise performing or managing any Loss in connection with (A) the Transactions or (B) any fact, situation, circumstance, status, condition, activity, practice, plan, occurrence, event, incident, action, failure to act or transaction relating to, in connection with or arising from the pre-Closing activity of the Business or the Acquired Companies, the other party shall, and shall cause its other Affiliates and its and their officers and employees to, and shall use its reasonable best efforts to cause its and their other Representatives to, cooperate with the defending party and its Affiliates and its and their counsel in such defense (to the extent reasonably necessary for such defense), including making available its personnel, participating in meetings, providing such testimony and access to their books and records and taking such other actions as shall be reasonably necessary in connection with such defense, to the extent reasonably requested by, and at the sole cost and expense of, the defending party; provided, however, that nothing in this Agreement shall require either party or its Affiliates to provide access or to disclose information in connection with a Proceeding between the parties (or their respective Affiliates), where the disclosing party reasonably concludes in good faith (upon the advice of counsel) that there is a divergence in the parties' respective interests, or where such

party reasonably believes in good faith that such access or disclosure would contravene any Legal Requirement (including Data Protection Laws) or the terms of any Contract, would result in the waiver of any legal privilege or work-product protection, or would cause significant competitive harm to such party, its Affiliates or their respective businesses; <u>provided</u>, <u>further</u>, that, in any such case, the appliable party will, to the extent permitted by applicable Legal Requirements, notify the other in reasonable detail of the circumstances giving rise to any non-disclosure pursuant to the foregoing and the parties shall use commercially reasonable efforts to identify and pursue a permissible method to provide such access or disclosure of such information in a manner that would not result in any of the outcomes described therein; provided, further, that nothing in this Agreement shall limit any rights of discovery. Subject to <u>Section 6.2(c)</u>, nothing in this <u>Section 6.13</u> shall require any party or any their respective Affiliates to enter into any joint defense agreements.

- (b) From and after the Closing, Seller shall reimburse the Acquired Companies for the actual amount paid by the Acquired Companies in settlement or upon a judgment of a court of competent jurisdiction (the "Specified Losses") for the legal proceeding set forth on Schedule 6.13(b)(i) of the Seller Disclosure Schedule (the "Specified Litigation") as set forth on Schedule 6.13(b)(ii) of the Seller Disclosure Schedule (for the avoidance of doubt, for purposes of this Section 6.13(b), attorneys' fees of Purchaser and its Affiliates and their respective former and current Representatives in connection with the Specified Litigation are the sole responsibility of Purchaser and its Affiliates and their respective former and current Representatives).
- (c) With respect to the Specified Litigation, Purchaser and its Affiliates shall not, except with the written consent of Seller (not to be unreasonably withheld, conditioned or delayed), enter into any settlement, voluntary order or other voluntary resolution that results in Seller having any monetary payment obligations under Section 6.13(b).
- 6.14 <u>Registered Office Addresses</u>. To the extent an Acquired Company uses any facility address of Seller or any of its Affiliates (other than the Acquired Companies) as a registered office address, Purchaser shall, at Purchaser's sole cost and expense, take any and all actions to transfer the registered office address of any such Acquired Company to the registered office address of Purchaser or any of its Affiliates effective as of, and subject to the occurrence of, the Closing.
- 6.15 <u>Segregation of Email and Messaging Accounts</u>. Purchaser hereby acknowledges and agrees that prior to the Closing Date, Seller shall be entitled to undertake a review of the email accounts and any other stored messages of Company Employees set forth on <u>Schedule 6.15</u> of the Seller Disclosure Schedule to identify any emails (and other content, including attachments and contacts) and stored messages that are related exclusively to the business of Seller and its Affiliates (other than the Business) (collectively, the "<u>Excluded Emails and Messages</u>"), and to delete from such email message accounts any Excluded Emails and Messages, such that Excluded Emails and Messages are not included in such email accounts as of the Closing Date.
- 6.16 <u>DDTC 60-Day</u>. The parties shall, and shall cause their respective Affiliates to, use their respective reasonable best efforts to ensure all required filings are made with DDTC under the ITAR. The parties shall file jointly the notice, or separately the notices, required by Section 122.4(b) of the ITAR so that prior to the Closing, sixty (60) days shall have elapsed following the submission of the notice(s). The parties shall cooperate with each other in connection with the

submission of any such notice(s), including the provision of any such information necessary for submission of the notice(s) required by Section 122.4(b) of the ITAR, as well as any other filings required under the ITAR.

6.17 <u>Resignations</u>. Seller shall use reasonable best efforts to deliver to Purchaser on the Closing Date resignation letters, in form and substance reasonably acceptable to Purchaser, of such members of the board of directors of each of the Acquired Companies and such officers of each of the Acquired Companies which have been requested in writing by Purchaser at least ten (10) Business Days prior to the Closing Date, such resignation letters to be effective concurrently with, and subject to the occurrence of, the Closing.

6.18 <u>Pre-Closing Reorganization</u>.

- (a) Prior to the Closing, Seller shall, and shall cause its applicable Affiliates to, effect the contribution of all of the issued and outstanding equity interests of Topaz to the Company and thereafter to the Company Subsidiary, such that Topaz will be a wholly-owned Subsidiary of the Company Subsidiary (such transactions, the "Topaz Reorganization").
- (b) Prior to the Closing, the parties shall negotiate in good faith, acting reasonably, and mutually agree to one or more commercial and services Contracts on arms'-length terms to provide Seller and its Controlled Affiliates following the Closing the services and benefits that Topaz provides to Seller and such Controlled Affiliates as of the date of this Agreement.
- (c) Prior to the Closing, Seller shall effectuate the transfers of real estate interests held by Seller as set forth on Annex B (the "Real Estate Reorganization Plan"); provided, that, subject to Section 6.4(c) and the following proviso, but notwithstanding anything else in this Agreement to the contrary, the occurrence of the matter set forth on Schedule 6.18(c) of the Seller Disclosure Schedule, shall not, in and of itself, constitute any breach of this Section 6.18(c); provided, further, that in the event that the Closing occurs prior to the completion of the matter set forth in Item 3(a) of the Real Estate Reorganization Plan, Seller shall, and shall cause its Controlled Affiliates to, use reasonable best efforts, and Purchaser shall, and shall cause the Acquired Companies to, reasonably cooperate with Seller, to complete the Real Estate Reorganization Plan with respect to such matter as promptly as practicable after the Closing, and, until such time as the Real Estate Reorganization Plan is so completed, Seller shall, and shall cause its Controlled Affiliates to, use reasonable best efforts to provide all of the benefits of the Real Estate Reorganization Plan to Purchaser as if it had been completed, at no additional cost to Purchaser.
- (d) At least ten (10) Business Days prior to effectuating the Topaz Reorganization or the Real Estate Reorganization Plan, as applicable, Seller shall deliver to Purchaser any documentation or agreements effectuating such transactions, and such documentation or agreements shall be in form and substance reasonably acceptable to Purchaser acting in good faith (such acceptance not to be unreasonably withheld, conditioned or delayed). Without limiting the generality of the foregoing, Seller shall, at least ten (10) Business Days prior to effectuating the Real Estate Reorganization Plan, submit the proposed Purchase Price (as defined in Annex B) to Purchaser for its review and consent acting in good faith (such consent not to be unreasonably withheld, conditioned or delayed).

ARTICLE VII

ADDITIONAL COVENANTS OF THE PARTIES

7.1 <u>Transitional Trademark Rights</u>.

- It is expressly agreed that, subject to Section 7.1(b), Purchaser and its Affiliates (including, as of the Closing, the Acquired Companies), do not have any right, title or interest (whether express or implied) in, to or under any Trademark consisting of, incorporating or confusingly similar to, any Trademark of Seller or its Affiliates (other than the Acquired Companies) set forth on Schedule 7.1(a) of the Seller Disclosure Schedule, including the Seller Transitional Trademarks, but excluding the Standalone Go Beyond Mark (which the Parties acknowledge and agree is included in the Owned IP and exclusively owned by an Acquired Company as of the date hereof and the Closing) (such Trademarks, the "Seller Mark"). Subject to Section 7.1(b), as of the Closing, Purchaser (i) shall cause the Acquired Companies to cease any and all use of the Seller Marks (including in the respective corporate or other legal names of the Acquired Companies); (ii) will not, and shall cause its Affiliates not to, (A) adopt, use, apply to register or register, or authorize others to adopt, use, apply to register or register, any Seller Mark (including the Seller Transitional Trademarks) or any component part thereof or any colorable imitation thereof (including any non-English language variation thereof), or any confusingly similar name, mark, dress, number or other designation, or any confusingly similar or dilutive name, mark, dress, number or designation or (B) contest the use, ownership, validity or enforceability of any rights of Seller or any of its Affiliates in or to any Seller Mark and (iii) shall not, and shall cause its Affiliates not to, otherwise do anything inconsistent with Seller's ownership of the Seller Marks or do or cause to be done any act or thing that will in any way impair the rights of Seller in and to the Seller Transitional Trademarks or Seller's goodwill therein or have any dilutive effect thereupon.
- Notwithstanding the restrictions set forth in Section 7.1(a), Seller, on behalf of itself and its Controlled Affiliates, hereby grants the Acquired Companies a limited right to utilize the Seller Transitional Trademarks following the Closing solely in substantially the same manner of such use and solely for the administration of the Business as conducted immediately prior to the Closing Date, for a period of three (3) months following the Closing Date (the "Transitional Trademark End Date"); provided, that (i) any goodwill generated by the Acquired Companies' use of the Seller Transitional Trademarks during the period following the Closing shall inure to the benefit of Seller; (ii) any products or services provided by or on the behalf of the Acquired Companies in connection with the Seller Transitional Trademarks shall be consistent with the use of such Seller Transitional Trademarks prior to the Closing Date, including with respect to quality standards and (iii) to the extent the Seller Transitional Trademarks are being used, the Acquired Companies shall continue to display the Seller Transitional Trademarks in the same manner they were displayed prior to the Closing Date. In the event Seller identifies any breach of this Section 7.1(b), the Acquired Companies shall promptly remedy such breach. No later than the Transitional Trademark End Date, Purchaser shall cause the Acquired Companies to destroy, or remove, strike over, cover over or otherwise eliminate all Seller Transitional Trademarks (which do not include the Standalone Go Beyond Mark) from all materials (whether written, electronic or otherwise) publicly displaying the Seller Transitional Trademarks in its possession. Seller shall have the right to terminate the foregoing license, effective upon thirty (30)

days' written notice to Purchaser, if the Acquired Companies fail to materially comply with the terms and conditions regarding such license set forth herein; provided, that such license shall not terminate if the Acquired Companies cure any such default prior to the expiration of such thirty (30) day notice period. Notwithstanding the foregoing, nothing in this Section 7.1 is intended to prohibit any use (or require any destruction, removal, striking or covering over, or other elimination) by Purchaser or its Affiliates of any Seller Transitional Trademarks (A) that are included on any Business products or related materials that have been produced (or are in production) prior to the Transitional Trademark End Date, until such products or materials have been exhausted, (B) on non-public fixed assets and personal property of the Business that include an indelible Seller Transitional Trademark, (C) for internal business purposes, including in internal or archived records or systems, (D) to the extent required by Legal Requirements or (E) to factually refer to the historical relationship between Seller and its Affiliates and the Business, including in historical, tax, regulatory and similar records, or as otherwise permitted by "fair use" in accordance with applicable Legal Requirements.

- It is expressly agreed that, Seller and its Controlled Affiliates (other than the Acquired Companies), do not have any right, title or interest (whether express or implied) in, to or under any Trademark consisting of, incorporating or confusingly similar to, any Trademark of the Acquired Companies set forth on Schedule 4.11(a)(2) of the Seller Disclosure Schedule (the "Business Trademarks"). As of the Closing, Seller (i) shall, and shall cause its Controlled Affiliates to, cease any and all use of the Business Trademarks and the Combination Mark (including in the respective corporate or other legal names of Seller or its Controlled Affiliates); (ii) will not, and shall cause its Controlled Affiliates not to, (A) adopt, use, apply to register or register, or authorize others to adopt, use, apply to register or register, any such Trademark or any component part thereof or any colorable imitation thereof (including any non-English language variation thereof), or any confusingly similar name, mark, dress, number or other designation, or any confusingly similar or dilutive name, mark, dress, number or designation, provided, notwithstanding such limitation, Seller shall have the right to maintain the existing (as of the Closing) registration of, but not use, the Combination Mark or (B) contest the use, ownership, validity or enforceability of any rights of the Acquired Companies in or to any Business Trademark and (iii) shall not, and shall cause its Controlled Affiliates not to, otherwise do anything inconsistent with the Acquired Companies' ownership of such Business Trademarks or do or cause to be done any act or thing that will in any way impair the rights of the Acquired Companies in and to the Business Trademarks or the Acquired Companies' goodwill therein or have any dilutive effect thereupon, provided, in each case, that with respect to the Combination Mark, the foregoing covenants and limitations shall not apply to the word BALL as a standalone word or as used in combination with any other word, phrase, or Trademark other than a Business Trademark, including the Standalone Go Beyond Mark, and the Parties acknowledge and agree that the word Ball is a Seller Mark and exclusively owned by Seller or its Controlled Affiliates (other than the Acquired Companies) as of the date hereof and the Closing.
- (d) Notwithstanding the restrictions set forth in Section 7.1(c), subject to the terms of this Section 7.1(d), the Acquired Companies hereby grant Seller and its Controlled Affiliates a limited right to utilize the Business Trademarks following the Closing solely in substantially the same manner of such use and solely for the administration of Seller's and its Controlled Affiliates' business as conducted immediately prior to the Closing Date, from the period starting on the Closing Date until the Transitional Trademark End Date; provided, that (i)

period following the Closing shall inure to the benefit of Acquired Companies; (ii) any products or services provided by or on the behalf of Seller or its Controlled Affiliates in connection with the Business Trademarks shall be consistent with the use of such Trademarks prior to the Closing Date, including with respect to quality standards and (iii) to the extent the Business Trademarks are being used, Seller and its Controlled Affiliates shall continue to display the Business Trademarks in the same manner they were displayed prior to the Closing Date. In the event Purchaser or an Acquired Company identifies any breach of this Section 7.1(d), Seller and its Controlled Affiliates shall promptly remedy such breach. No later than the Transitional Trademark End Date, Seller shall, and shall cause its Controlled Affiliates to, destroy, or remove, strike over, cover over or otherwise eliminate all Business Trademarks (including the Combination Mark, but subject to the final proviso of Section 7.1(c)) from all materials (whether written, electronic or otherwise) publicly displaying the Business Trademarks in its possession. Purchaser shall have the right to terminate the foregoing license, effective upon thirty (30) days' written notice to Seller, if Seller or its Controlled Affiliates fail to materially comply with the terms and conditions regarding such license set forth herein; provided, that such license shall not terminate if Seller or its Controlled Affiliate, as applicable, cures any such default prior to the expiration of such thirty (30) day notice period. Notwithstanding the foregoing, nothing in this Section 7.1 is intended to prohibit any use (or require any destruction, removal, striking or covering over, or other elimination) by Seller or its Controlled Affiliates of any Business Trademarks IP (A) that are included on any products or related materials that have been produced (or are in production) prior to the Transitional Trademark End Date, until such products or materials have been exhausted, (B) on non-public fixed assets and personal property that include an indelible Business Trademark, (C) for internal business purposes, including in internal or archived records or systems, (D) to the extent required by Legal Requirements or (E) to factually refer to the historical relationship with the Acquired Companies, including in historical, tax, regulatory and similar records, or as otherwise permitted by "fair use" in accordance with applicable Legal Requirements.

any goodwill generated by Seller's or its Controlled Affiliates' use of the Business Trademarks during the

7.2 <u>Closing and Post-Closing Access to Information</u>.

- (a) Prior to or at the Closing, Seller shall, and shall cause its Controlled Affiliates (other than the Acquired Companies) to, use commercially reasonable efforts to deliver to the Acquired Companies all books, records and documents, in any form or medium, to the extent related to the Business or the Acquired Companies (in each case including any applicable attorney-client privilege, attorney work product protection and expectation of client privilege attaching to any such books, records and documents) and in the possession of Seller and its Controlled Affiliates (other than the Acquired Companies), other than books, records and documents the provision of which is subject to the Transition Services Agreement.
- (b) Without limiting Seller's obligations under Section 7.2(a), from and after the Closing for a period of seven (7) years, Seller shall, and shall cause its Controlled Affiliates to, and Purchaser shall, and shall cause its Affiliates (including the Acquired Companies) to, afford the other and its Affiliates and their respective Representatives, during normal business hours, upon reasonable request and advance notice, reasonable access to the books and records of Seller and its Controlled Affiliates or each Acquired Company and their Affiliates and the Business, as applicable, in each case, to the extent related to the Acquired Companies or the Business and for periods prior to the Closing, and to make copies of such books and records at the accessing party's

purposes, including in connection with financial statements, any potential Proceeding or investigation by or before a Governmental Authority and stock exchange, foreign securities, SEC or other Governmental Authority reporting obligations or (ii) in the case of Seller, such access is requested for reasonable business purposes, including financial statements, any potential Proceeding or investigation by or before a Governmental Authority and stock exchange, foreign securities, SEC or other Governmental Authority reporting obligations; provided, that the foregoing shall not require Seller or Purchaser or its respective Affiliates to provide access or to disclose information in connection with a Proceeding either between the parties (or their respective Affiliates), or where such party reasonably believes in good faith that such access or disclosure would contravene any Legal Requirement (including Data Protection Laws) or the terms of any Contract, would result in the waiver of any legal privilege or work-product protection, or would cause significant competitive harm to such party, its Affiliates or their respective businesses; provided, further, that, in any such case, the appliable party will, to the extent permitted by applicable Legal Requirements, notify the other in reasonable detail of the circumstances giving rise to any non-disclosure pursuant to the foregoing and the parties shall use commercially reasonable efforts to identify and pursue a permissible method to provide such access or disclosure of such information in a manner that would not result in any of the outcomes described therein; provided, further, that nothing in this Agreement shall limit any rights of discovery in any Proceeding; provided, further, that the foregoing shall not expand or otherwise affect Seller's obligations under the Transition Services Agreement. For the avoidance of doubt, this Section 7.2 shall not govern access to information relating to Taxes, which access is governed solely by Section 8.5.

expense, to the extent that (i) in the case of Purchaser, such access is requested for reasonable business

7.3 <u>D&O Indemnification</u>.

(a) From and after the Closing until the date that is six (6) years after the Closing Date, Purchaser shall cause the Acquired Companies to, (i) indemnify, defend and hold harmless all of the past and present officers and directors (in their capacities as such) of each of the Acquired Companies (collectively, the "D&O Indemnitees") from and against all costs or expenses, judgments, fines, losses, claims, damages, liabilities and amounts paid in settlement in connection with any actual or threatened claim or Proceeding arising out of, relating to or resulting from the fact that such D&O Indemnitee is or was an officer or director of any Acquired Company or is or was serving at the request of any Acquired Company as an officer, director, trustee, member, manager or employee of any other Person, in each case at or prior to the Closing, whether asserted or claimed prior to, at or after the Closing (including with respect to acts or omissions occurring in connection with the Transaction Agreements and the consummation of the Transactions) and provide advancement of expenses to the D&O Indemnitees; provided, that any D&O Indemnitee to whom expenses are advanced provides an undertaking to repay such advances to the extent required by applicable Legal Requirements, in all such cases, to the greatest extent that such Persons are indemnified or have the right to advancement of expenses prior to the Closing by any Acquired Company pursuant to (x) its organizational documents in existence as of the date of this Agreement and (y) the indemnification agreements in existence as of the date of this Agreement and set forth on Schedule 7.3 of the Seller Disclosure Schedule (the "D&O Indemnification Agreements"), (ii) without limitation of the foregoing clause (i), to the fullest extent permitted by applicable Legal Requirement, include and not amend, repeal or modify and otherwise cause to be maintained in effect the provisions regarding elimination of liability of officers and directors, and

indemnification of and advancement of expenses to officers, directors and employees contained in the organizational documents of any Acquired Company and (iii) without the consent of the applicable D&O Indemnitee (not to be unreasonably conditioned, delayed or withheld), not settle, compromise or consent to the entry of any judgment in any Proceeding or threatened Proceeding (and in which indemnification would reasonably be expected to be sought by a D&O Indemnitee hereunder), unless such settlement, compromise or consent includes an unconditional release of such D&O Indemnitee from all liability arising out of, relating to or resulting from such Proceeding.

- (b) Prior to the Closing, the Acquired Companies shall, at Purchaser's expense and with effect from the Closing, obtain a "tail" extension of the directors' and officers' liability coverage of Seller's existing directors' and officers' insurance policies and fiduciary liability insurance policies (the "D&O Insurance") covering persons who are (at or prior to the Closing) currently or who were officers or directors of any of the Acquired Companies (in their capacities as such) for a claims reporting or discovery period of six years from and after the Closing with respect to any claims related to any period of time at or prior to the Closing, with terms, conditions, retentions and limits of liability that are no less favorable in the aggregate than the coverage provided under the D&O Insurance; provided, that the aggregate cost for such "tail" insurance shall not exceed an amount to be mutually agreed by the parties (the "Maximum Amount"). If the Acquired Companies are unable to obtain such "tail" policy at an aggregate cost less than or equal to the Maximum Amount, the Acquired Companies shall instead, in consultation with Purchaser, obtain as much comparable insurance as possible for an aggregate cost equal to the Maximum Amount.
- (c) If, following the Closing until the date that is six (6) years after the Closing Date, any Acquired Company, or any of its respective successors or assigns, (i) shall consolidate with or merge into any other corporation or entity and shall not be the continuing or surviving corporation or entity of such consolidation or merger or (ii) shall transfer all or substantially all of its properties and assets to any Person, then, and in each such case, proper provisions shall be made so that the successors and assigns of such Acquired Company or any of their respective successors or assigns, as the case may be, shall assume all of the obligations set forth in this Section 7.3.
- (d) The rights of the D&O Indemnitees under this <u>Section 7.3</u> shall be in addition to any rights such D&O Indemnitees may have under the organizational documents of the Acquired Companies, or under any applicable Contracts or Legal Requirements, and Purchaser shall, and shall cause each of the Acquired Companies to honor and perform under all D&O Indemnification Agreements entered into by the Acquired Companies as in effect as of the date of this Agreement.
- (e) The obligations of Purchaser and the Acquired Companies under this Section 7.3 shall not be terminated, amended or modified in any manner so as to adversely affect any D&O Indemnitee (including such Person's successors, heirs and legal representatives) to whom this Section 7.3 applies without the written consent of such affected D&O Indemnitee (it being expressly agreed that the D&O Indemnitees to whom this Section 7.3 applies shall be third-party beneficiaries of this Section 7.3, and this Section 7.3 shall be enforceable by such D&O Indemnitees and their respective successors, heirs and legal representatives and shall be binding on all successors and assigns of Purchaser and each Acquired Company).

7.4 Non-Solicit.

- From and after the Closing until the date that is twelve (12) months after the date of the Closing, Seller will not, and will not permit any of its Controlled Affiliates to, directly or indirectly, (i) induce or encourage any Transferred Employee with a gross annual base salary equal to or in excess of \$280,000 to leave the employment of Purchaser or any of its Affiliates or (ii) solicit for employment or employ any such Transferred Employee with a gross annual base salary equal to or in excess of \$280,000; provided, that the foregoing in this Section 7.4(a) shall not preclude Seller or any of its Controlled Affiliates from (A) making general solicitations (including through the use of third-party agencies or firms) not specifically directed toward any Transferred Employee (and hiring any Transferred Employee who affirmatively contacts Seller or any of its Controlled Affiliates for employment without prior solicitation or encouragement or in response to such general solicitation and not any other form of solicitation) or (B) hiring any Transferred Employee whose employment by Purchaser or any of its Affiliates is terminated prior (if terminated by such Transferred Employee and not by Purchaser or its applicable Affiliate, or if terminated by Purchaser or its applicable Affiliate "for cause", at least three (3) months prior) to commencement of employment discussions between Seller or any of its Controlled Affiliates and such Transferred Employee. If Seller breaches any of the foregoing covenants, Purchaser will be entitled to seek injunctive relief in addition to any other remedies that may be available under applicable Legal Requirements.
- (b) Seller expressly acknowledges that (i) each of the restrictions contained in this Section 7.4 are reasonable in all respects (including with respect to subject matter, geographical scope and time period) and such restrictions are necessary to protect Purchaser's interest in, and value of, the Acquired Companies' businesses (including the goodwill inherent therein), (ii) Seller is primarily responsible for the creation of such value, (iii) the transactions contemplated by this Agreement constitute good, valid and binding consideration for Seller's obligations, covenants and agreements contained in this Section 7.4 and (iv) Purchaser would not have entered into this Agreement or any of the transactions contemplated hereby without the restrictions contained in this Section 7.4 and this Section 7.4 being in full force and effect and binding and enforceable covenants of Seller. If a court of competent jurisdiction finds that the time period of any of the foregoing covenants is too lengthy or the geographic coverage or scope of any of the covenants is too broad, the restrictive time period will be deemed to be the longest period permissible under applicable Legal Requirement and the geographic coverage and scope will be deemed to comprise the largest coverage and scope permissible under applicable Legal Requirements.

7.5 Further Assurances; Wrong Pockets.

(a) From time to time following the Closing, Seller shall, and shall cause its Controlled Affiliates to, and Purchaser shall, and shall cause its Controlled Affiliates to, at the sole cost and expense of the requesting party, execute, acknowledge and deliver all reasonable further conveyances, notices, assumptions, releases and acquittances and such instruments, and shall take such reasonable actions as may be necessary to make effective the Transactions as may be reasonably requested by the other party; provided, however, that nothing in this Section 7.5(a) shall (i) require any party or any of its respective Affiliates to expend any money, commence or participate in any Proceeding, incur liabilities or offer or grant any accommodation (financial or

otherwise) following the Closing or (ii) expand or otherwise affect Seller's obligations under the Transition Services Agreement.

- If, for a period of twenty-four (24) months after the Closing, Seller or any of its Controlled Affiliates (i) receives funds (including any refund or other amount relating to any pre-Closing claim (in respect of workers' compensation, third-party insurance or similar matters)) arising from the Business or (ii) owns or is in possession of any material asset (including Intellectual Property) primarily used or held for use in the conduct of the Business, then Seller shall promptly remit, transfer or assign, or cause its applicable Affiliate to remit, transfer or assign, such funds or asset to the Company or its designated Affiliate (and the Company or its designated Affiliate shall accept any such funds or asset), for no additional consideration and net of Seller's reasonable out-of-pocket costs incurred to effect such remittance, transfer or assignment. Until the remittance or transfer of any such funds or asset is effected, Seller shall, or shall cause its applicable Affiliate to, preserve the value of, and hold in trust for the use and benefit of, the Company or its designated Affiliate, such funds or asset and provide to the Company or its designated Affiliate all of the benefits arising from such funds or asset and otherwise cause such funds or asset to be used as reasonably instructed by the Company or its designated Affiliate. Notwithstanding the foregoing, this Section 7.5(b) is not intended to modify, and shall be subject to, any other provisions of this Agreement or any other Transaction Agreement that expressly provides that any funds or assets shall, after the Closing, be allocated to Seller or any of its Affiliates (including with respect to general corporate shared services provided by Seller or any of its Controlled Affiliates (other than the Acquired Companies) and to be retained by Seller or such Controlled Affiliate in connection with the Transactions), on the one hand, or Purchaser and its Affiliates (including the Acquired Companies), on the other hand.
- If, for a period of twenty-four (24) months after the Closing, Purchaser or any of its Affiliates (including the Acquired Companies) (i) receives any funds (including any refund or other amount relating to any pre-Closing claim (in respect of workers' compensation, third-party insurance or similar matters)) arising from the businesses of Seller or any of its Affiliates other than the Business or (ii) owns or is in possession of any material asset (including Intellectual Property) primarily used or held for use in the conduct of the businesses of Seller or any of its Affiliates other than the Business, then Purchaser shall promptly remit, transfer or assign, or cause its applicable Affiliate to remit, transfer or assign, such funds or asset to Seller or its designated Affiliate (and Seller or its designated Affiliate shall accept any such funds or asset), for no additional consideration and net of Purchaser's reasonable out-of-pocket costs incurred to effect such remittance, transfer or assignment. Until the remittance or transfer of any such funds or asset is effected, Purchaser shall, or shall cause its applicable Affiliate to, preserve the value of, and hold in trust for the use and benefit of, Seller or its designated Affiliate, such funds or asset and provide to Seller or its designated Affiliate all of the benefits arising from such funds or asset and otherwise cause such funds or asset to be used as reasonably instructed by Seller or its designated Affiliate. Notwithstanding the foregoing, this Section 7.5(c) is not intended to modify, and shall be subject to, any other provisions of this Agreement or any other Transaction Agreement that expressly provides that any funds or assets shall, after the Closing, be allocated to Seller or any of its Affiliates (including with respect to general corporate shared services provided by Seller or any of its Controlled Affiliates (other than the Acquired Companies) and to be retained by Seller or such Controlled Affiliate in connection with the Transactions), on the one hand, or Purchaser and its Affiliates (including the Acquired Companies), on the other hand.

- (d) Notwithstanding the foregoing, to the extent any provision this Agreement or any of the other Transaction Agreements specifically provides that any funds or assets shall remain or become the funds or assets of, or otherwise become the responsibility of, Purchaser, its Affiliates or the Acquired Companies, on the one hand, or Seller or its Affiliates, on the other hand, such provision shall control over Section $7.5(\underline{b})$ and Section $7.5(\underline{c})$.
- 7.6 Notifications. From and after the date of this Agreement and until the earlier of the Closing and the termination of this Agreement pursuant to its terms, each of Purchaser and Seller will give prompt notice to the other (and will subsequently keep the other informed on a reasonably current basis of any material developments related to such notice) upon its becoming aware of the occurrence or existence of any change, event, effect, occurrence or development that (a) with respect to Seller or any of its Controlled Affiliates (including the Acquired Companies and the Business), has had or would reasonably be expected to have a Material Adverse Effect, (b) with respect to Purchaser, has had or would reasonably be expected to have a Purchaser Material Adverse Effect or (c) will, or is reasonably likely to, result in any of the Closing Conditions not being able to be satisfied prior to the Outside Date. No notification given by any party pursuant to this Section 7.6 shall limit or otherwise affect any of the representations, warranties, covenants, obligations or conditions contained in this Agreement. Any party's (a) failure to comply with this Section 7.6 or (b) notice provided in accordance with this Section 7.6 will not, in and of itself, be taken into account for purposes of determining whether any Closing Conditions have been satisfied.

ARTICLE VIII

TAX MATTERS

8.1 Section 338(h)(10) Elections.

- (a) Seller and Purchaser shall (or shall cause their relevant Affiliates to) make and timely file joint elections under Section 338(h)(10) of the Code (and any corresponding elections under applicable state or local Legal Requirements) with respect to the Acquired Companies (the "Section 338(h)(10) Elections").
- (b) Seller and Purchaser shall (and shall cause their relevant Affiliates to) cooperate in the preparation and filing of all forms, attachments and schedules necessary to effectuate the Section 338(h)(10) Elections, including IRS Form 8023.
- (c) In connection with the Section 338(h)(10) Elections, as promptly as practicable after the determination of the Final Purchase Price pursuant to Section 2.4(d) but in no event later than ninety (90) days after such determination, Seller shall prepare and deliver to Purchaser an allocation of the applicable "Aggregate Deemed Sale Price" (as described in Treasury Regulations Section 1.338-4) among the assets of the Acquired Companies (the "Election Allocation"), which allocation shall be made in accordance with Section 338 of the Code and any applicable Treasury Regulations and Section 8.8. If within thirty (30) days of receiving the Election Allocation, Purchaser has not objected to such allocation in writing, the Election Allocation shall become final and binding on the parties. If within thirty (30) days of receiving the Election Allocation, Purchaser notifies Seller in writing of any objection to such allocation,

Purchaser and Seller shall cooperate in good faith to resolve any disputed items. If Purchaser and Seller fail to resolve any disputed item within thirty (30) days following Purchaser's written objection (or within such longer period as the parties may mutually agree), Purchaser and Seller shall submit such disputed items to the Settlement Accountant for resolution pursuant to the procedures set forth in Section 2.4(c), applied mutatis mutandis, provided that, for clarity, the Settlement Accountant shall apply Section 8.8. The Election Allocation, as finally determined pursuant to this Section 8.1(c), shall be final, conclusive and binding on Purchaser and Seller.

(d) Seller and Purchaser shall (and shall cause their relevant Affiliates to) (i) prepare and file all federal, state and local Tax Returns in a manner consistent with the Section 338(h)(10) Elections and the Election Allocation (it being understood that the "Adjusted Grossed-Up Basis" as described in Treasury Regulations Section 1.338-5 may differ to the extent required under such Treasury Regulations with respect to items such as Purchaser's "acquisition costs") and (ii) not take any position inconsistent therewith on any Tax Return or in connection with any Tax Proceeding, except as otherwise required pursuant to a "determination" as defined in Section 1313(a) of the Code (and any similar provision under any state, local or foreign law).

8.2 <u>Tax Returns; Allocation of Taxes.</u>

- (a) Seller and Purchaser shall prepare and file Tax Returns as follows:
- (1) Seller shall prepare and timely file, or cause to be prepared and timely filed, when due (taking into account any valid extension of a required filing date) (A) all Consolidated Returns and (B) all Tax Returns required to be filed by the Acquired Companies related to Pre-Closing Tax Periods that are due (taking into account any valid extension of a required filing date) on or before the Closing Date. Each such Tax Return shall be prepared in a manner consistent with past practices of the Acquired Companies, except as otherwise required by applicable Legal Requirements. For the avoidance of doubt, Seller shall pay, or cause to be paid, any Taxes associated with the Tax Returns described in this Section 8.2(a)(i).
- (2) Purchaser (or its Affiliates) shall prepare and timely file, or cause to be prepared and timely filed, when due (taking into account any extensions of a required filing date) all other Tax Returns required to be filed by the Acquired Companies related to Pre-Closing Tax Periods (including Straddle Periods) that are due after the Closing Date, excluding, for the avoidance of doubt, any Consolidated Returns (each such Tax Return that is material and filed on or prior to the date that the Final Closing Statement becomes final and binding, but expressly excluding a Purchaser Consolidated Return, a "Purchaser-Filed Tax Return"). Each Purchaser-Filed Tax Return shall be prepared in a manner consistent with past practices of the Acquired Companies, except as otherwise required by applicable Legal Requirements.
- (3) Any Purchaser-Filed Tax Return shall be provided in draft form to Seller (together with schedules, statements or other supporting documentation reasonably requested) at least twenty-five (25) Business Days (or, in the case of any Tax Return that is not an Income Tax Return, as soon as reasonably practicable) prior to the due date (including any applicable valid extension) of such Purchaser-Filed Tax Return. Seller shall

have the right to review and comment on such Purchaser-Filed Tax Return, and Purchaser shall consider in good faith any comments thereto that are provided by Seller to Purchaser in writing at least fifteen (15) Business Days (or, in the case of any Tax Return that is not an Income Tax Return, as soon as reasonably practicable) prior to the due date (including any applicable valid extension) of such Purchaser-Filed Tax Return. Purchaser and Seller shall cooperate in good faith to resolve any disputed items with respect to any comments that were timely provided by Seller. If Purchaser and Seller fail to resolve any disputed items within five (5) Business Days following Seller's delivery of such comments (or within such longer period as the parties may mutually agree), Purchaser and Seller shall submit such disputed items to the Settlement Accountant for resolution, and Purchaser and Seller shall instruct the Settlement Accountant to resolve such disputed items as soon as practicable prior to the due date (including any applicable valid extension) of the Purchaser-Filed Tax Return. The fees and expenses of the Settlement Accountant shall be borne in the manner contemplated by Section 2.4(c), mutatis mutandis.

- (4) Except to the extent otherwise required pursuant to a final "determination" within the meaning of Section 1313(a) of the Code (or any comparable provision of state, local or foreign Legal Requirements), from and after the Closing until the Final Closing Statement becomes final and binding, Purchaser shall not, and shall cause the Acquired Companies not to, amend any Tax Return related to a Pre-Closing Tax Period or agree to the waiver or extension of the statute of limitations relating to a Pre-Closing Tax Period of the Acquired Companies (other than as a result of ordinary course extensions of time to file Tax Returns consistent with past practice), in each case, (1) to the extent such aforementioned actions would reasonably be expected to adversely affect Seller or any of its Affiliates and (2) without the prior written consent of Seller (not to be unreasonably conditioned, delayed or withheld).
- (b) To the extent permitted by Legal Requirements, each of the Acquired Companies shall elect to close each of its respective taxable periods as of or prior to the Closing Date. Any Tax Return of any Acquired Company or Seller (to the extent related to the Acquired Companies) for a taxable period that includes the Closing Date shall, to the extent permitted by Legal Requirements, be filed on the basis that the relevant taxable period ended as of the close of business on the Closing Date. Where it is necessary for purposes of this Agreement to apportion between Seller and Purchaser Taxes with respect to the Acquired Companies for a Straddle Period, such Taxes shall be apportioned between the period deemed to end on the Closing Date and the period deemed to begin at the beginning of the day following the Closing Date on the basis of an interim closing of the books, except that (i) exemptions, allowances, and deductions that are calculated on an annual basis and (ii) Taxes (such as real or personal property Taxes) that are imposed on a periodic basis, shall, in each case, be allocated ratably across the entire Straddle Period on a per diem basis. Seller and Purchaser agree that the U.S. federal income Tax Return of the Acquired Companies for the Tax period ending on the Closing Date shall be prepared in accordance with Treasury Regulations Section 1.1502-76(b)(1)(ii) and that none of Purchaser or any Acquired Company or any of their respective Affiliates shall make a ratable allocation election under Treasury Regulations Section 1.1502-76(b)(2) or any analogous provision of state, local or foreign Legal Requirements.

- Prohibited Actions. Except as otherwise provided in Section 8.1 with respect to the Section 338(h)(10) Elections, or as otherwise contemplated by this Agreement, Purchaser shall not, and shall cause its Affiliates (including the Acquired Companies) not to, (a) take any action outside of the ordinary course of business after the Closing on the Closing Date (or pursuant to a plan in existence on the Closing Date) that would reasonably be expected to increase any Tax liability of Seller or any of its Affiliates, (b) file any ruling request with any Tax Authority relating in whole or in part to any Taxes or Tax Returns of the Acquired Companies for a Pre-Closing Tax Period or Straddle Period, (c) enter into or initiate any voluntary disclosure agreement with any Tax Authority relating in whole or in part to any Taxes or Tax Returns of the Acquired Companies for any Pre-Closing Tax Period or Straddle Period, (d) make any election with respect to any Acquired Company or change any method of Tax accounting or any Tax accounting period of any Acquired Company, which election or change would be effective on or prior to the Closing Date, in each case, without the prior written consent of Seller (not to be unreasonably conditioned, delayed or withheld) or (e) take any action outside of the ordinary course of business that would reasonably be expected to increase the amount of Specified Sales Taxes (if such Taxes are treated as Indemnified Taxes). Notwithstanding the foregoing, the prohibitions described in clauses (b), (c) and (d) of the preceding sentence shall not apply, and shall have no force or effect, after the Final Closing Statement becomes final and binding.
- 8.4 <u>Consolidated Returns and Purchaser Consolidated Returns; Tax Proceedings.</u> For the avoidance of doubt, but subject to <u>Section 8.1</u>, Seller shall control any Tax Proceeding, Tax election or other Tax-related decision or determination in respect of a Consolidated Return, and Purchaser shall control any Tax Proceeding, Tax election or other Tax-related decision or determination in respect of a Purchaser Consolidated Return. Purchaser shall promptly notify Seller of the commencement of any Tax Proceeding in respect of Specified Sales Taxes (if such Taxes are treated as Indemnified Taxes), and Seller shall have the right to control, at Seller's expense, any such Tax Proceeding to the extent that such Tax Proceeding relates to Specified Sales Taxes. Purchaser shall have the right to participate in any such Tax Proceeding at Purchaser's expense, and Seller shall not settle or compromise any such Tax Proceeding without the prior written consent of Purchaser (not to be unreasonably withheld, conditioned or delayed).
- 8.5 Tax Matters Cooperation. Seller shall, and shall cause its Controlled Affiliates to, and Purchaser shall, and shall cause its Controlled Affiliates to, cooperate in good faith to the extent reasonably requested by the other party in connection with the filing of any Tax Returns in connection with any Tax Proceeding or in connection with determining a liability for Taxes, in each case, related to the Acquired Companies. Such cooperation shall include, (a) with respect to any Tax Proceeding, reasonable cooperation with the requests of the applicable Tax Authority, (b) the retention and (upon the other party's request) the provision of records and information reasonably relevant to any of the foregoing Tax matters described in the preceding sentence and (c) making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder; provided, however, that notwithstanding anything to the contrary contained in this Agreement, (i) Seller and its Affiliates shall not be required to provide Purchaser or any of its Affiliates or its or their respective Representatives with any Consolidated Return (or copy thereof), (ii) Purchaser and its Affiliates shall not be required to provide Seller or any of its Affiliates or its or their respective Representatives with any Purchaser Consolidated Return (or copy thereof) and (iii) neither party shall be required to provide any information or materials the provision of which it reasonably believes in good faith would result

in the waiver of any legal privilege or work-product protection. Purchaser shall cause the Acquired Companies to retain all books and records with respect to Tax matters pertinent to the Acquired Companies related to any taxable period beginning before the Closing Date until the expiration of the applicable statute of limitations, and to abide by all record retention agreements entered into with any Tax Authority. Subject to the limitations described in clauses (ii) and (iii) above, Purchaser shall cause the Acquired Companies to furnish, at Seller's sole cost and expense and in the ordinary course of business of the Acquired Companies, any Tax information related to a Consolidated Return reasonably requested by Seller for any taxable period of the Acquired Companies that includes the Closing Date; provided, that Seller shall not be required to reimburse Purchaser for any costs or expenses pursuant to this paragraph that are not reasonable and documented out-of-pocket costs or expenses.

- Return to the other party. Seller shall prepare and timely file such Tax Returns and promptly provide a copy of such Tax Returns to the other party. Seller shall, and shall cause its Affiliates to, cooperate in connection with the preparation and filing of any such Tax Returns.
- 8.7 <u>Indemnified Taxes</u>. Seller shall indemnify and hold harmless Purchaser and its Affiliates (including, following the Closing, the Acquired Companies) from and against any Taxes and Losses (without duplication) attributable to or arising from any Indemnified Taxes, other than the amount of any such Taxes resulting from or attributable to (a) any breach or violation by Purchaser or its Affiliates (including, after the Closing, the Acquired Companies) of a covenant, agreement, undertaking or obligation in this <u>Article VIII</u> or (b) any action taken on the Closing Date after the Closing by Purchaser or its Affiliates (including, after the Closing, the Acquired Companies) outside the ordinary course of business and not otherwise contemplated under this Agreement. Seller's obligation to indemnify Purchaser and its Affiliates hereunder shall be reduced by the proceeds of any recovery under any insurance policy with respect to such Taxes (after reduction for any costs and expenses, including Taxes, imposed in connection with the receipt of such proceeds). The parties agree to treat for all applicable Income Tax purposes any indemnity payment made under this <u>Section 8.7</u> as an adjustment to the Final Purchase Price, except as otherwise required by applicable Legal Requirements.
- 8.8 <u>Deferred Revenue</u>. Purchaser and Seller agree that, for Tax purposes, Purchaser will not be deemed to receive a payment from Seller in exchange for assuming any liability in respect of deferred revenue of Seller or the Acquired Companies.
- 8.9 <u>Survival</u>. This <u>Article VIII</u> shall survive the Closing until ninety (90) days after the expiration of the statute of limitations (including extensions) applicable to the relevant Tax matter.

ARTICLE IX

EMPLOYEES

- 9.1 <u>Transferred Employees</u>. Prior to the Closing, Seller shall transfer, or cause to be transferred, to an Acquired Company, the employment of each Internal Transfer Employee other than an Inactive Employee. Purchaser shall cause one of its Subsidiaries to offer employment to an Inactive Employee to the extent such employee is able to commence and presents himself for active employment within a ninety (90) day period immediately following the Closing Date. Each Company Employee who is employed by an Acquired Company as of the Closing Date and who remain so employed immediately following the Closing shall be referred to herein as a "<u>Transferred Employee</u>". To the extent an Inactive Employee commences active employment with Purchaser or a Subsidiary of Purchaser after the Closing Date pursuant to the second sentence of this <u>Section 9.1</u>, he or she will be considered a Transferred Employee for applicable purposes of this Agreement as of such date.
- 9.2 <u>Continuation Period</u>. Subject to any applicable Legal Requirements or Contract that provide for greater benefits:
- For the period commencing on the Closing Date and ending on the first (1st) anniversary of the Closing Date, or for such shorter period of employment, as the case may be (the "Continuation Period"), Purchaser shall cause to be provided to the Transferred Employees (i) a base salary (or hourly base wage rate) that is at least equal to the base salary (or hourly base wage rate) provided to such Transferred Employee immediately prior to the Closing Date, (ii) a target annual cash bonus or commission opportunity that is at least equal to the target annual cash bonus or commission opportunity provided to such Transferred Employee immediately prior to the Closing Date, (iii) long-term incentive opportunities (including equity or equity-based incentives, which shall not be considered benefits for purposes of clause (iv)) in accordance with the terms and conditions (including with respect to eligibility) of Purchaser's long-term incentive compensation plans and (iv) employee health, welfare, retirement, fringe benefits and other benefits (excluding any retention, transaction or nonqualified deferred compensation (the "Excluded Benefits")) that are no less favorable in the aggregate (including a combination of benefits and compensation) than the employee health, welfare, retirement, fringe benefits and other benefits (other than the Excluded Benefits) provided to the Transferred Employees immediately prior to the Closing Date under the Benefit Plans set forth on Schedule 4.14(a) of the Seller Disclosure Schedule and the Benefit Plans as may be adopted or entered into following the date hereof, as permitted under this Agreement;
- (b) In the event of termination of the employment of any Transferred Employee during the Continuation Period, Purchaser shall provide, or shall cause to be provided, to such Transferred Employee severance pay and benefits no less favorable than the greater of the severance pay and benefits to which such Transferred Employee (i) would have been entitled immediately prior to the Closing Date under any applicable Benefit Plan set forth on Schedule 4.14(a) of the Seller Disclosure Schedule and (ii) is entitled under any applicable severance plan, policy, practice or arrangement of Purchaser or any of its Affiliates on the actual date of termination of the Transferred Employee's employment;

- (c) Purchaser shall cause its applicable Controlled Affiliates (including any Acquired Company) to, give each Transferred Employee credit for purposes of eligibility to participate, level of paid time off benefits, and vesting (but not benefit accruals, except for benefit accruals under the Business Pension Plan) under each employee benefit plan, policy or arrangement maintained and made available for the benefit of Transferred Employee's service prior to the Closing Date by Purchaser or any of its Subsidiaries, for such Transferred Employee's service prior to the Closing Date with Seller and its applicable Controlled Affiliates and their respective predecessors, to the same extent and for the same purpose as such service is recognized by Seller and its applicable Controlled Affiliates immediately prior to the Closing Date under the Acquired Company Benefit Plan or Seller Benefit Plan; provided that such credit shall not be given to the extent that it would result in a duplication of benefits or coverage for the same period of service; and
- (d) Purchaser shall cause its applicable Controlled Affiliates (including any Acquired Company) to, for the plan year in which the Closing Date occurs, (i) use commercially reasonable efforts to waive any limitation on health and welfare coverage of such Transferred Employees due to pre-existing conditions, waiting periods, active employment requirements, and requirements to show evidence of good health under any applicable group health plan of Purchaser or any of its Controlled Affiliates (including any Acquired Company) to the extent such Transferred Employees were covered under a comparable Benefit Plan that is a group health plan immediately prior to the Closing Date and such limitations did not apply to such Transferred Employee under the applicable Benefit Plan and (ii) use commercially reasonable efforts to credit each such Transferred Employee with all deductible payments, co-payments and co-insurance paid by (and credited to) such employee under any Benefit Plan that is a group medical plan prior to the Closing Date during the year in which the Closing Date occurs for the purpose of determining the extent to which any such employee has satisfied the corresponding applicable deductible and whether such employee has reached the corresponding out-of-pocket maximum under any comparable group medical plan of Purchaser or any of its Subsidiaries for such year.
- Seller Benefit Plan Participation; M&A Qualified Beneficiaries; Certain Benefits for Transferred 9.3 Employees. Effective as of the Closing Date (or such later date as provided under the terms of the Seller Benefit Plans or contemplated under Section 9.1), the Transferred Employees shall no longer actively participate in any Seller Benefit Plan. Notwithstanding the foregoing, Seller or its Controlled Affiliates shall retain all liabilities with respect to claims incurred by each Transferred Employee prior to the Closing Date under those Seller Benefit Plans that provide medical, dental, vision and prescription drug coverage, life, accidental death and dismemberment, disability and business travel accident insurance. For purposes of this Section 9.3, the following claims shall be deemed to be incurred as follows: (a) with respect to life, disability and accidental death and dismemberment benefits, upon the event giving rise to such benefits and (b) with respect to medical, dental, vision care, prescription and health-related benefits, upon provision of medical, dental, vision, prescription and health-related services, materials or supplies. From and after the Closing Date, Purchaser and its ERISA Affiliates shall be solely responsible for any and all obligations arising under the Consolidated Omnibus Budget Reconciliation Act of 1985 (or state law equivalents) with respect to each Transferred Employee (and qualifying dependents thereof) who is an "M&A qualified beneficiary" (as defined in Treasury Regulations Section 54.4980B-9) in connection with the Transactions. Effective as of or as soon as practicable after the Closing Date, Purchaser or its applicable Controlled Affiliate shall provide, under its

applicable health and life insurance plans, health and life insurance benefits to eligible Transferred Employees who are receiving salary replacement benefits under Seller's long-term disability insurance policy for the same duration as such health and life insurance benefits would have been provided to such Transferred Employee under the terms of the applicable Seller Benefit Plans ("Assumed Disability Health Benefits").

9.4 Qualified Retirement Plans.

- (a) Effective as of the Closing Date, Purchaser shall, or shall cause one of its Affiliates to, have in effect a defined contribution retirement plan that is tax-qualified under Section 401(a) of the Code and includes a qualified cash or deferred arrangement within the meaning of Section 401(k) of the Code (the "Purchaser 401(k) Plan"). Each Transferred Employee who is a participant immediately prior to the Closing Date in a Seller Benefit Plan that is a defined contribution retirement plan and is tax-qualified under Section 401(a) of the Code and includes a qualified cash or deferred arrangement within the meaning of Section 401(k) of the Code (each, a "Seller 401(k) Plan") shall have the opportunity to enroll in the Purchaser 401(k) Plan as of, or as soon as administratively practicable after (but no later than the third payroll following), the Closing Date. The parties shall take all actions necessary to permit the direct rollover in cash (and outstanding loan promissory notes for participant loans, if applicable) of account balances of Transferred Employees from the Seller 401(k) Plan to the Purchaser 401(k) Plan, by each Transferred Employee who elects such direct rollover in accordance with the terms of the Seller 401(k) Plan immediately prior to the Closing shall be permitted to enroll in the Purchaser 401(k) Plan on or after the Closing Date in accordance with the terms of the Purchaser 401(k) Plan.
- (b) Seller and its Controlled Affiliates shall take all actions necessary and appropriate to, (i) cause all Transferred Employees who participate in any Seller Benefit Plans that are intended to be qualified under Section 401(a) of the Code to be fully vested in their account balances and accrued benefits under such Seller Benefit Plans as of the Closing Date, (ii) cause the employees of Seller or its Controlled Affiliates (other than an Acquired Company) set forth on Schedule 9.4(b) of the Seller Disclosure Schedule who are not Transferred Employees to be fully vested in their accrued benefits under the Business Pension Plan as of the Closing Date and (iii) as soon as administratively practicable following the Closing Date, make all employer contributions to the Seller 401(k) Plan that would have been made on behalf of such Transferred Employees had the transactions contemplated by this Agreement not occurred, regardless of any service or end of year employment requirements, but prorated for the portion of the plan year that ends on the Closing Date; provided that the Seller 401(k) Plan performance sharing match contribution shall be calculated as if it was an Assumed Incentive Amount and, for the avoidance of doubt, will be contributed to the Seller 401(k) Plan by Seller.
- (c) Purchaser and Seller shall cooperate to take any and all commercially reasonable measures needed to prevent, to the extent reasonably possible, a deemed distribution or loan offset with respect to outstanding loans under the Seller 401(k) Plan with respect to Transferred Employees, including without limitation (i) permitting Transferred Employees with an outstanding loan balance under the Seller 401(k) to continue to make scheduled loan payments to the Seller 401(k) Plan after the Closing or (ii) at the discretion of the Seller, allowing such

Transferred Employees to elect to rollover their loan balances in-kind to the Purchaser 401(k) Plan during a limited rollover window.

- (d) With respect to the Business Pension Plan, (i) Seller shall timely make to the trust under the Business Pension Plan any payments of minimum required contributions (as determined by the actuaries of the Business Pension Plan in accordance with the requirements of Sections 412 and 430 of the Code and Section 302 of ERISA) that have payment due dates that fall prior to the Closing Date and (ii) Purchaser or its applicable Controlled Affiliate (including, after the Closing, an Acquired Company) shall timely make to the trust under the Business Pension Plan any payments of minimum required contributions (as determined by the actuaries of the Business Pension Plan in accordance with the requirements of Section 412 and 430 of the Code and Section 302 of ERISA) that have payment due dates after the Closing Date, in each case, regardless of the plan year for which such minimum required contribution payment relates.
- 9.5 <u>FSAs</u>. The parties hereto agree to make reasonable, good faith efforts to implement, as of or as soon as administratively practicable after the Closing Date, a transfer of Transferred Employees' health and dependent care flexible spending accounts from the health and dependent care flexible spending account plans of Seller to the health and dependent care flexible spending account plans of Purchaser, taking into account the complexity of transferring flexible spending accounts, the date Closing occurs in the calendar year, and whether the costs and administrative complexities of such transfer outweigh the benefit to the Transferred Employees taken as a whole.
- 9.6 Annual Cash Bonuses; Similar Benefits. Purchaser shall, or shall cause one of its Affiliates to, assume and pay on a pro rata basis all unpaid cash bonuses earned or accrued as of the Closing Date under the Benefit Plans set forth on Schedule 9.6 of the Seller Disclosure Schedule (the "Incentive-Based Programs") for and in respect of each Transferred Employee (the "Assumed Incentive Amount"), which shall be paid at such time as such amounts would have been paid to the Transferred Employees under the applicable Incentive-Based Program. The Assumed Incentive Amount shall be determined for each Transferred Employee (a) based on target performance levels, if the Closing occurs in the first three (3) months of the applicable fiscal year and (b) based on the most recently approved performance projections available as of the Closing Date if the Closing occurs at any time other than the first three (3) months of the applicable fiscal year. Purchaser's obligations under this Section 9.6 are subject to the inclusion of the Assumed Incentive Amount (together with the employer portion of any payroll, social security, unemployment or similar Taxes associated with the Assumed Incentive Amount) in Net Working Capital. Purchaser shall provide such bonus eligible Transferred Employees with the opportunity to earn cash incentives for the remainder of the fiscal year after the Closing Date, which cash incentives shall be governed by the cash incentive plans or programs maintained by Purchaser and its Affiliates (including the Acquired Companies) in Purchaser's discretion, subject to Purchaser's obligations under Section 9.2(a).
- 9.7 <u>Vacation and Paid Time Off.</u> To the extent permitted pursuant to applicable Legal Requirements, Purchaser and its Affiliates shall (a) with respect to each (i) Direct Employee who becomes a Transferred Employee, recognize and honor and (ii) Internal Transfer Employee who becomes a Transferred Employee, credit with, and assume all liabilities for, in each case, the amount of accrued but unused vacation time, paid time off and other time-off benefits, if any, as

such Transferred Employee had with any Acquired Company, Seller or any of its Controlled Affiliates, as applicable, as of immediately prior to the Closing Date to the extent such amounts are reflected in Net Working Capital and (b) permit each Transferred Employee to use such accrued but unused vacation time, paid time off and other time-off benefits in the same manner and upon the same terms and conditions as the Transferred Employee would have been so permitted under the terms and conditions of the applicable policies of any Acquired Company, Seller or any of its Controlled Affiliates, as applicable, in effect for the year in which such Closing Date occurs.

- 9.8 <u>Communications</u>. Prior to the Closing Date, except as otherwise approved in advance and in writing by Seller, Purchaser shall not make any written or oral communications to Company Employees pertaining to the transfer of Company Employees, any compensation or benefits matters, or any redundancy and layoff plans, in each case, that may affect Company Employees in connection with the Transactions.
- 9.9 <u>Seller Long-Term Incentive Awards</u>. Seller and its Controlled Affiliates shall remain solely responsible for all liabilities and obligations with respect to any equity awards relating to shares of Seller's common stock and any long-term cash-based awards that are held by Transferred Employees (or any director, officer, employee, consultant, or independent contractor of Seller or any of its Controlled Affiliates who is not a Transferred Employee) and outstanding immediately prior to the Closing Date to the extent such equity awards and long-term cash-based awards are not forfeited upon the Closing pursuant to the terms and conditions of any Seller Benefit Plan.
- 9.10 <u>Deferred Compensation Plans</u>. Prior to the Closing, Seller shall establish a mirror plan (each, a "Mirror Plan") for each Benefit Plan that is a non-tax qualified deferred compensation plan set forth on Schedule 9.10(a) of the Seller Disclosure Schedule and in which the Transferred Employees participate as of the Closing Date (each, a "Seller Deferred Compensation Plan"). Each Mirror Plan shall be subject to Purchaser's review and comment, and Seller shall take Purchaser's comments into account in good faith. Prior to the Closing, Seller shall transfer all liabilities and obligations related to the Transferred Employees from a Seller Deferred Compensation Plan to a Mirror Plan. At the Closing, Purchaser shall assume each Mirror Plan, and all liabilities and obligations related to the Transferred Employees under each Mirror Plan, which liabilities are set forth on Schedule 9.10(b), and which schedule shall be updated by Seller to reflect current liabilities and obligations as of the Closing (the "Assumed Deferred Compensation"). Prior to the Closing, Seller shall provide to Purchaser details relating to the accounts and liabilities of each Transferred Employee who participates in a Mirror Plan and all information necessary to administer the Mirror Plans, including the terms relating to distribution of such Transferred Employees' accounts. From and following the Closing, Purchaser or one of its Affiliates shall be liable for such Transferred Employees' accounts and shall be responsible for payment of account balances under the Mirror Plans when payable pursuant thereto. Purchaser's obligations under this Section 9.10 are subject to the inclusion of the Assumed Deferred Compensation (together with the employer portion of any payroll, social security, unemployment or similar Taxes associated with the Assumed Deferred Compensation calculated as if all such amounts were paid at the Closing without regard to any Social Security wage base limits) in Indebtedness. Seller shall remain responsible for all liabilities associated with the Seller Deferred Compensation Plans other than the Assumed Deferred Compensation.

- Employee Liabilities. Seller and Purchaser hereby acknowledge and agree that, effective as of the Closing Date, (a) Purchaser shall, or shall cause its Affiliates to, assume all liabilities and obligations (i) arising out of, relating to or resulting from the employment or termination of employment of any Company Employee and Former Company Employee (solely with respect to any period during which such Former Company Employee's regular employment duties or responsibilities were primarily dedicated or primarily related to the Business), in each case, whenever incurred, but other than any such liabilities and obligations related to any Seller Benefit Plan, whenever incurred (unless otherwise expressly assumed by Purchaser or its Affiliates under this Article IX, (ii) arising out of, relating to or resulting from any Acquired Company Benefit Plan whenever incurred and (iii) expressly assumed by Purchaser or its Affiliates in accordance with the provisions of this Article IX and (b) other than the liabilities assumed by Purchaser or its Affiliates in the foregoing clause (a), Seller shall, or shall cause its Controlled Affiliates to, retain and perform (i) all liabilities, claims and obligations arising under any Seller Benefit Plan or any other benefit or compensation plan, program, agreement, policy, contract or arrangement of any kind at any time maintained, sponsored or contributed to or required to be contributed by Seller or any ERISA Affiliate thereof, whenever incurred and (ii) all other liabilities and obligations expressly retained by Seller or its Controlled Affiliates in accordance with the provisions of this Article IX. For the avoidance of doubt, neither Purchaser nor any of its Affiliates (including, after the Closing, any Acquired Company) shall have any obligation to provide retiree medical or retiree life insurance benefits to any Transferred Employee, Company Employee, Former Company Employee, or dependent or beneficiary of any such individual, nor shall Purchaser or any of its Affiliates (including, after the Closing, any Acquired Company) assume any of Seller's liabilities or obligations related to such retiree benefits, including under a Seller Benefit Plan.
- 9.12 No Third-Party Beneficiaries. Nothing contained in this Article IX, express or implied, is intended to confer upon any Person not a party hereto (including any Company Employee or any beneficiary thereof) any right, benefit or remedy of any nature whatsoever, including any right to employment or continued employment for any period of time by reason of this Agreement, any right to a particular term or condition of employment or any right to any specific compensation or benefits. Notwithstanding anything to the contrary contained in this Agreement, no provision of this Agreement is intended to, or does: (i) constitute the establishment of, termination of or an amendment to any Benefit Plan or other benefit or compensation plan, agreement or arrangement or (ii) limit the right of any party to this Agreement or any of their respective Affiliates to establish, terminate or amend any Benefit Plan or other benefit or compensation plan, agreement or arrangement.

ARTICLE X

CONDITIONS TO THE CLOSING

10.1 <u>Conditions of Purchaser</u>. The obligations of Purchaser to consummate the Transactions shall be subject to the satisfaction, or waiver by Purchaser (in its sole discretion), to the extent permitted by applicable Legal Requirements, at or prior to the Closing, of each of the following conditions:

- Representations and Warranties of Seller. (i) The representations and warranties of Seller (a) (other than the Seller Fundamental Representations and the representations and warranties set forth in Section 4.6(b)) contained in Article IV shall be true and correct as of the Closing as if made on the Closing Date (other than any such representation and warranty that is made as of a specific date, which representation and warranty shall have been true and correct as of such date), except for breaches or inaccuracies, as the case may be, as to matters that, individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect; provided, however, that for purposes of determining the satisfaction of the condition in this clause (i), no effect shall be given to the exceptions or qualifications of "material" or "Material Adverse Effect", or similar "materiality" based exceptions or qualifications, in such representations and warranties, (ii) the Seller Fundamental Representations shall be true and correct in all material respects as of the Closing as if made on the Closing Date (other than any such representation and warranty that is made as of a specific date, which representation and warranty shall have been true and correct in all material respects as of such date); provided, however, that for purposes of determining the satisfaction of the condition in this clause (ii), no effect shall be given to the exceptions or qualifications of "material" or "Material Adverse Effect" or similar "materiality" based exceptions or qualifications, in such representations and warranties and (iii) the representations and warranties of Seller contained in Section 4.6(b) shall be true and correct in all respects as of the Closing as if made on the Closing Date.
- (b) <u>Covenants of Seller</u>. The covenants contained in this Agreement required to be complied with by Seller at or prior to the Closing shall have been complied with in all material respects.
- (c) <u>Certificate of Seller</u>. Purchaser shall have received a certificate signed by an authorized officer of Seller, dated as of the Closing Date, certifying that the conditions set forth in <u>Section 10.1(a)</u> and <u>Section 10.1(b)</u> have been satisfied.
- 10.2 <u>Conditions of Seller</u>. The obligations of Seller to consummate the Transactions shall be subject to the satisfaction, or waiver by Seller (in its sole discretion), to the extent permitted by applicable Legal Requirements, at or prior to the Closing, of each of the following conditions:
- (a) Representations and Warranties of Purchaser. (i) The representations and warranties of Purchaser (other than the Purchaser Fundamental Representations) contained in Article V shall be true and correct as of the Closing as if made on the Closing Date (other than any such representation and warranty that is made as of a specific date, which representation and warranty shall have been true and correct as of such date), except for breaches or inaccuracies, as the case may be, as to matters that, individually or in the aggregate, would not reasonably be expected to have a Purchaser Material Adverse Effect; provided, however, that for purposes of determining the satisfaction of the condition in this clause (i), no effect shall be given to the exceptions or qualifications of "material" or "Purchaser Material Adverse Effect", or similar "materiality" based exceptions or qualifications, in such representations and warranties and (ii) the Purchaser Fundamental Representations shall be true and correct in all material respects as of the Closing as if made on the Closing Date (other than any such representation and warranty that is made as of a specific date, which representation and warranty shall have been true and correct in all material respects as of such date); provided, however, that for purposes of determining the

satisfaction of the condition in this clause (ii), no effect shall be given to the exceptions or qualifications of "material" or "Purchaser Material Adverse Effect", or similar "materiality" based exceptions or qualifications, in such representations and warranties.

- (b) <u>Covenants of Purchaser</u>. The covenants contained in this Agreement required to be complied with by Purchaser at or prior to the Closing shall have been complied with in all material respects.
- (c) <u>Certificate of Purchaser</u>. Seller shall have received a certificate signed by an authorized officer of Purchaser, dated as of the Closing Date, certifying that the conditions set forth in <u>Section 10.2(a)</u> and <u>Section 10.2(b)</u> have been satisfied.
- 10.3 <u>Mutual Conditions</u>. The respective obligations of each party to consummate the Transactions shall be subject to the satisfaction, or waiver by mutual consent of, to the extent permitted by applicable Legal Requirements, at or prior to the Closing, of each of the following conditions:
- (a) <u>HSR Approval</u>. Any waiting period (and any extension thereof) applicable to the consummation of the Transactions under the HSR Act shall have been terminated or shall have expired, and there shall be no timing agreement (entered into with the mutual consent of the parties) in effect with any Governmental Authority in relation to any Antitrust Law prohibiting the consummation of the sale of the Shares pursuant to this Agreement.
- (b) <u>No Orders or Proceedings</u>. No Governmental Authority of competent authority and jurisdiction shall have (i) issued an Order or enacted a Legal Requirement that remains in effect and makes illegal or prohibits the sale of the Shares pursuant to this Agreement or (ii) commenced a Proceeding that shall remain pending seeking issuance of an Order or enactment of a Legal Requirement that makes illegal or prohibits the sales of the Shares pursuant to this Agreement.
 - (c) <u>DCSA Approval</u>. DCSA Approval shall have been received and remains in effect.
 - (d) <u>CFIUS Clearance</u>. CFIUS Clearance shall have been received and remains in effect.
- (e) <u>DDTC</u>. The applicable notice period under ITAR Section 122.4(b) with respect to the Transactions shall have concluded or DDTC shall have consented to the Transactions.
- 10.4 <u>Waiver of Conditions</u>. The conditions set forth in <u>Section 10.1</u> may only be waived by written notice from Purchaser. The conditions set forth in <u>Section 10.2</u> may only be waived by written notice from Seller. The conditions set forth in <u>Section 10.3</u> may only be waived by written notice from both Seller and Purchaser.

ARTICLE XI

TERMINATION

- 11.1 <u>Termination</u>. This Agreement may be terminated at any time prior to the Closing:
 - (a) by the mutual written consent of Seller and Purchaser;
- (b) by Seller, if Purchaser shall have breached any representation or warranty or failed to comply with any covenant or agreement applicable to Purchaser that in each case would cause any Closing Condition set forth in Section 10.2 not to be satisfied and such (i) Closing Condition is incapable of being satisfied by the Outside Date or (ii) breach or failure to perform has not been cured on or prior to the earlier of (A) the date that is thirty (30) days from the date that Purchaser is notified in writing by Seller of such breach or failure to perform and (B) the day prior to the Outside Date; provided, however, that the right to terminate this Agreement under this Section 11.1(b) shall not be available to Seller if Seller is then in material breach of any covenant or agreement contained in this Agreement that would result in a failure of a Closing Condition set forth in Section 10.1 not to be satisfied;
- (c) by Purchaser, if Seller shall have breached any representation or warranty or failed to comply with any covenant or agreement applicable to Seller that in each case would cause any Closing Condition set forth in Section 10.1 not to be satisfied and such (i) Closing Condition is incapable of being satisfied by the Outside Date or (ii) breach or failure to perform has not been cured on or prior to the earlier of (A) the date that is thirty (30) days from the date that Seller is notified in writing by Purchaser of such breach or failure to perform and (B) the day prior to the Outside Date; provided, however, that the right to terminate this Agreement under this Section 11.1(c) shall not be available to Purchaser if Purchaser is then in material breach of any covenant or agreement contained in this Agreement that would result in a failure of a Closing Condition set forth in Section 10.2 not to be satisfied;
- (d) by either Seller or Purchaser, if the Closing has not occurred by the date that is twelve (12) months after the date hereof (as may be extended pursuant to this proviso that follows, the "Outside Date"); provided, however, that the right to terminate this Agreement under this Section 11.1(d) shall not be available to any party whose failure to fulfill any material obligation under this Agreement has been the primary cause of the failure of the Closing to occur by the Outside Date; provided, further, that, if on the date that would have been the Outside Date, the conditions set forth in Section 10.3 (with respect to Section 10.3(b), solely to the extent such condition has not been satisfied due to an Order or Legal Requirement arising under Antitrust Laws or the Legal Requirements covered under Section 10.3(a), Section 10.3(c), Section 10.3(d) or Section 10.3(e)) are the only conditions in Article X (other than those conditions that by their nature are to be satisfied by actions taken at the Closing, provided that each such condition would be satisfied if the Closing were to occur at such time) that shall not have been satisfied on or before such date, then Seller or Purchaser may extend the Outside Date by six (6) months upon delivering written notice to the other party;
- (e) by either Seller or Purchaser in the event that any Governmental Authority of competent authority and jurisdiction shall have issued an Order or enacted a Legal Requirement

that permanently enjoins or otherwise makes illegal or prohibits the sale of the Shares pursuant to this Agreement and such Order or other Legal Requirement shall have become final and non-appealable; <u>provided</u>, <u>however</u>, that the right to terminate this Agreement under this <u>Section 11.1(e)</u> shall not be available to any party whose failure to fulfill any material obligation under this Agreement has been the primary cause of the issuance of such Order or the imposition of such other Legal Requirement; or

(f) by Seller, if:

- (1) all of the Closing Conditions set forth in <u>Section 10.1</u> and <u>Section 10.3</u> have been satisfied (or duly waived by Purchaser) (other than those conditions that by their nature are to be satisfied by actions taken at the Closing, <u>provided</u> that each such condition would be satisfied if the Closing were to occur at such time), and Purchaser fails to consummate the Transactions at the time the Closing was required to occur under <u>Section 3.1</u>;
- (2) Seller has irrevocably certified in writing to Purchaser that (A) all of the conditions set forth in Section 10.2 and Section 10.3 have been satisfied (or duly waived by Seller) (other than those conditions that by their nature are to be satisfied by actions taken at the Closing, provided that each such condition would be satisfied if the Closing were to occur at such time) and (B) Seller is ready, prepared, willing and able to consummate the Transactions on such date of notice and at all times during the three (3) Business Day period immediately thereafter; and
- (3) Purchaser fails to consummate the Closing within the three (3) Business Day period following the date of delivery of such written certification by Seller pursuant to clause (ii) above.
- 11.2 <u>Notice of Termination</u>. If this Agreement is terminated pursuant to <u>Section 11.1</u>, written notice of such termination shall be given by the terminating party to the other party (setting forth a reasonably detailed description of the basis on which such party is terminating the Agreement).
- 11.3 <u>Effect of Termination</u>. If this Agreement is terminated in accordance with <u>Section 11.1</u> and <u>Section 11.2</u>, all rights and obligations of the parties shall terminate without any liability of any party or other Person; provided that (a) the rights and obligations of the parties under <u>Section 6.7</u> (Confidentiality), <u>Section 6.10</u> (Financing Cooperation) (with respect to the confidentiality, reimbursement and indemnification obligations of Purchaser therein), this <u>Section 11.3</u> (Effect of Termination), <u>Section 11.4</u> (Purchaser Termination Fee), <u>Article XII</u> (Miscellaneous Provisions) and the Non-Disclosure Agreement shall survive termination of this Agreement and (b) nothing herein shall relieve any party from liability for Willful Breach of any covenant or agreement contained herein occurring prior to termination or for Fraud, in which case the aggrieved party shall be entitled to all rights and remedies available at law or in equity.

11.4 Purchaser Termination Fee.

(a) In the event that this Agreement is validly terminated by Seller or Purchaser pursuant to Section 11.1(d) or Section 11.1(e) (but, with respect to Section 11.1(e), only if the

applicable Order or Legal Requirement arises under Antitrust Laws or the Legal Requirements covered under Section 10.3(a), Section 10.3(c), Section 10.3(d) or Section 10.3(e)) if, in each case, at the time of such termination, at least one condition set forth in Section 10.3 (with respect to Section 10.3(b), solely to the extent such condition has not been satisfied due to an Order or Legal Requirement arising under Antitrust Laws or the Legal Requirements covered under Section 10.3(a), Section 10.3(c), Section 10.3(d) or Section 10.3(e) shall not have been satisfied; then, in each case, Purchaser shall pay to Seller a termination fee of one hundred million dollars (\$100,000,000) in cash (the "Purchaser Termination Fee").

- (b) Any payment required to be made pursuant to <u>Section 11.4(a)</u> shall be made to Seller promptly following termination of this Agreement (and in any event no later than three (3) Business Days following the termination of this Agreement). Such payment shall be made by wire transfer of immediately available funds to an account designated in writing by Seller at least three (3) Business Days prior to the payment date.
- (c) Purchaser acknowledges and agrees that the agreement contained in this <u>Section 11.4</u> is an integral part of the Transactions and that, without this agreement, Seller would not enter into this Agreement; accordingly, if Purchaser fails to promptly pay any amount due pursuant to this <u>Section 11.4</u>, and, in order to obtain such payment, Seller commences a Proceeding against Purchaser for the Purchaser Termination Fee, Purchaser shall pay to Seller its reasonable and documented out-of-pocket costs and expenses (including attorneys' fees and expenses) in connection with such Proceeding, together with interest on the amount of the Purchaser Termination Fee from the date such payment was required to be made until the date of payment at a rate per annum equal to the prime rate (as published in The Wall Street Journal).
- Except for Seller's right of specific performance to the extent permitted by Section 12.10. (x) Seller's right to receive the Purchaser Termination Fee when payable pursuant to Section 11.4(a), together with any related costs, expenses and interest payable pursuant to Section 11.4(c), shall constitute the sole and exclusive remedies of Seller and its Affiliates and their respective Representatives against Purchaser or any of its former, current and future Affiliates, their respective Representatives, and the respective successors and assigns of the foregoing Persons, in respect of this Agreement, the other Transaction Agreements or the Transactions, including for any Proceeding, Liabilities and Losses of any kind (whether in tort, contract or otherwise) suffered or incurred by Seller arising out of this Agreement, the other Transaction Agreements or the Transactions, including the termination of this agreement or any breach of this Agreement by Purchaser), Seller shall not bring nor permit any of its Controlled Affiliates or their respective Representatives, or any successor or assign of any of the foregoing Persons, to bring any Proceeding against Purchaser or any of its former, current and future Affiliates, their respective Representatives, and the respective successors and assigns of the foregoing Persons seeking such Liabilities or Losses suffered or incurred as a result of or under, or otherwise relating to or arising out of, this Agreement, the other Transaction Agreements or the Transactions, including the termination of this agreement or any breach of this Agreement by Purchaser and (y) upon the payment of the Purchaser Termination Fee, together with any related costs, expenses and interest payable pursuant to Section 11.4(c), none of Purchaser or any of its former, current and future Affiliates, their respective Representatives, and the respective successors and assigns of the foregoing Persons, shall have any further Liability or other obligation relating to or arising out of this Agreement, the other Transaction Agreements or the Transactions; provided, that Purchaser

shall remain obligated with respect to the Non-Disclosure Agreement, pursuant to Section 12.1 and pursuant to Section 6.10 (with respect to the confidentiality, reimbursement and indemnification obligations of Purchaser therein). Purchaser's former, current and future Affiliates, their respective Representatives, and the respective successors and assigns of the foregoing Persons are intended third party beneficiaries of this Section 11.4(d).

(e) The parties acknowledge and agree that (i) in no event shall Purchaser be required to pay the Purchaser Termination Fee on more than one occasion and (ii) any payment of the Purchaser Termination Fee is not a penalty but is liquidated damages in a reasonable amount that will compensate Seller in the circumstances in which such fees are payable for the efforts and resources expended and the opportunities foregone while negotiating this Agreement and in reliance on this Agreement and on the expectation of the consummation of the transactions contemplated by this Agreement, which amount would otherwise be impossible to calculate with precision.

ARTICLE XII

MISCELLANEOUS PROVISIONS

- 12.1 <u>Expenses.</u> Whether or not the Transactions are consummated, unless otherwise expressly provided herein, and except as otherwise specified in the Transaction Agreements, each party shall pay its own costs and expenses in connection with this Agreement and the Transactions, including the fees and expenses of its advisors, accountants and legal counsel.
- 12.2 <u>Survival</u>. Except as set forth below in this <u>Section 12.2</u>, none of the representations and warranties contained in this Agreement or in any other agreement, certificate or other document executed in connection herewith shall survive the Closing and all such representations and warranties, including any claim arising from or related thereto, shall terminate automatically upon the Closing; provided, that such termination shall not release any Person from liability for Fraud. Except for in the case of Fraud, the sole and exclusive remedy of Purchaser in respect of any and all rights and claims for any breach of representation or warranty is the right to terminate this Agreement prior to the Closing pursuant to <u>Article XI</u> and not consummate the Transactions. The covenants and agreements contained in this Agreement and to be performed or complied with at or prior to the Closing shall not survive the Closing and no party shall have any liability with respect thereto from and after the Closing. Subject to <u>Section 8.9</u> with respect to the Tax matters described in <u>Article VIII</u>, the covenants and agreements contained in this Agreement and to be performed or complied with after the Closing shall survive the Closing in accordance with their respective terms. Notwithstanding anything herein to the contrary, nothing in this <u>Section 12.2</u> shall be construed to affect or limit the ability of Purchaser or its Affiliates (including the Acquired Companies) to recover under the R&W Insurance Policy.
- 12.3 <u>Interpretation</u>. Except as otherwise explicitly specified to the contrary, (a) references to a Section, Article, schedule, annex or exhibit means a Section or Article of, or schedule, annex or exhibit to, this Agreement, unless another agreement is specified, (b) the word "including" (and words of similar import) means "including without limitation," (c) references to a particular statute or regulation include all rules and regulations thereunder and any predecessor or successor statute, rules or regulation, in each case, as amended or otherwise modified from time

singular form, respectively, and words of one gender shall be held to include the other gender as the context requires, (e) references to the parties or a party means the parties hereto or a party hereto, respectively, unless another agreement is specified, (f) references to a particular Person include such Person's successors and assigns to the extent not prohibited by this Agreement, (g) "extent" in the phrase "to the extent" means the degree to which a subject or other thing extends, and such phrase does not mean simply "if," (h) the headings contained in this Agreement, in any schedule, annex or exhibit hereto and in the table of contents to this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement, (i) references to "\$" shall mean United States dollars, (j) the word "or" is not exclusive, and shall be read to mean "and/or", (k) the words "hereof," "herein," "hereby," "hereto" and derivative or similar words refer to this entire Agreement, including the schedules, annexes and exhibits hereto, (1) the word "any" means "any and all," (m) the words "writing," "written" and comparable terms refer to printing, typing and other means of reproducing words (including electronic media) in a visible form, (n) if the last day of the time period for the giving of any notice or the taking of any action required under this Agreement falls on a day that is not a Business Day, the time period for giving such notice or taking such action shall be extended through the next Business Day following the original expiration date of such, (o) unless otherwise specified, the words "made available to," "delivered to," "provided to" or "furnished to" Purchaser (or words of similar import) means such documents have been posted to, or provided in, the Data Room and made accessible to Purchaser or one or more of its Representatives at least one (1) hour prior to the execution and delivery of this Agreement, (p) Seller and Purchaser have each participated in the negotiation and drafting of this Agreement and the other Transaction Agreements and if an ambiguity or question of interpretation should arise, this Agreement and the other Transaction Agreements shall be construed as if drafted jointly by the parties thereto and no presumption or burden of proof shall arise favoring or burdening either party by virtue of the authorship of any of the provisions in this Agreement or the other Transaction Agreements and (a) any accounting term not specifically defined within the Agreement shall have the meaning ascribed to such term set forth under GAAP.

to time (unless otherwise expressly provided), (d) words in the singular or plural form include the plural and

- 12.4 <u>Entire Agreement</u>. This Agreement, the Seller Disclosure Schedule, the Non-Disclosure Agreement, the Clean Team Agreement and the other Transaction Agreements, including the other documents, agreements, exhibits and schedules specifically referred to herein and therein, constitute the entire agreement between and among the parties with regard to the subject matter hereof, and supersede all prior agreements and understandings with regard to such subject matter.
- 12.5 Amendment and Waivers. This Agreement shall not be amended or modified, in whole or in part, except by supplemental agreement or amendment signed by Seller and Purchaser. No failure or delay by a party in exercising any right or remedy provided by Legal Requirement or under this Agreement shall impair such right or remedy or operate or be construed as a waiver or variation of it or preclude its exercise at any subsequent time and no single or partial exercise of any such right or remedy shall preclude any further exercise of it or the exercise of any other remedy. The waiver by any party of a breach of any term or provision of this Agreement shall not be construed as a waiver of any other provision or any subsequent breach. Notwithstanding anything to the contrary contained herein, this Section 12.5, Section 12.6, Section 12.8(c), Section 12.9 and Section 12.12(c) (collectively, the "DFS Provisions") may not be amended in a

manner that is adverse to a Debt Financing Source, without the prior written consent of such Debt Financing Source (not to be unreasonably withheld, conditioned or delayed).

- 12.6 <u>Successors and Assigns</u>. This Agreement shall bind and inure to the benefit of the parties and their respective successors and permitted assigns; provided, however, that no party may assign any right or obligation hereunder without the prior written consent of the other party, and any assignment in violation of this <u>Section 12.6</u> shall be null and void. Notwithstanding the foregoing, and subject to the last sentence of this <u>Section 12.6</u>, from and after the Closing (a) Seller may assign this Agreement or all of its rights or obligations hereunder to any of its Affiliates without Purchaser's prior written consent (but with notice to Purchaser), (b) Purchaser (or its Purchaser Designee) shall have the right to assign this Agreement or all of its rights or obligations hereunder to any of its Affiliates without Seller's prior written consent (but with notice to Seller) and (c) from and after the Closing Date, Purchaser (or its Affiliate) shall have the right to assign all or any portion of its rights and obligations pursuant to this Agreement to any Debt Financing Source (so long as any such assignment does not relieve Purchaser of its obligations hereunder) under terms of the Debt Financing solely for the purpose of creating a security interest herein or otherwise assigning collateral with respect to the Debt Financing. Notwithstanding anything to the contrary in this <u>Section 12.6</u>, no assignment shall relieve the assigning party of its obligations hereunder.
- 12.7 <u>Governing Law.</u> This Agreement, the rights of the parties and all Proceedings arising in whole or in part under or in connection herewith, will be governed by and construed and enforced in accordance with the laws of the State of Delaware, without giving effect to any choice or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the laws of any other jurisdiction.

12.8 <u>Jurisdiction; Venue; Service of Process</u>.

Each of the parties, by its execution hereof, hereby (i) irrevocably submits to the exclusive jurisdiction of the Delaware Chancery Court (or, if the Delaware Chancery Court declines to accept jurisdiction, any United States District Court located in the State of Delaware or any state court of the State of Delaware) for the purpose of any Proceeding among any of the parties relating to or arising in whole or in part under or in connection with this Agreement, any other Transaction Agreement or the Transactions, (ii) waives to the extent not prohibited by applicable Legal Requirements, and agrees not to assert, by way of motion, as a defense or otherwise, in any such Proceeding, any claim that it is not subject personally to the jurisdiction of the above-named courts, that its property is exempt or immune from attachment or execution, that any such Proceeding brought in one of the above-named courts should be dismissed on grounds of *forum non conveniens*, should be transferred or removed to any court other than one of the above-named courts, or should be stayed by reason of the pendency of some other Proceeding in any other court other than one of the above-named courts or that this Agreement, any other Transaction Agreement or the subject matter hereof or thereof may not be enforced in or by such court and (iii) agrees not to commence any such Proceeding other than before one of the above-named courts. Notwithstanding the previous sentence, a party may commence any Proceeding in a court other than the above-named courts solely for the purpose of enforcing an Order issued by one of the abovenamed courts.

- (b) Each of the parties (i) consents to service of process in any Proceeding among any of the parties relating to or arising in whole or in part under or in connection with this Agreement, any other Transaction Agreement or the Transactions in any manner permitted by Delaware law, (ii) agrees that service of process made in accordance with the foregoing clause (i) or made by registered or certified mail, return receipt requested, at its address specified pursuant to Section 12.14, will constitute good and valid service of process in any such Proceeding and (iii) waives and agrees not to assert (by way of motion, as a defense or otherwise) in any such Proceeding any claim that service of process made in accordance with the foregoing clause (i) or (ii) does not constitute good and valid service of process. This Section 12.8 shall not apply to any dispute that is required to be decided by the Settlement Accountant.
- (c) Notwithstanding anything in this Agreement to the contrary, no party, nor any of its Controlled Affiliates, shall bring, or support, any action, whether at law or in equity, whether in contract or in tort or otherwise, against any Debt Financing Source in any way relating to this Agreement or any of the Transactions, including any dispute arising out of or relating in any way to the Debt Commitment Letter or the definitive agreements executed in connection therewith or the transactions contemplated thereby, anywhere other than in a court of England and any such action shall be governed by English law. In furtherance of the foregoing, each of the parties (on behalf of itself and its respective Controlled Affiliates) (A) submits to the exclusive jurisdiction of the English courts for the purpose of any action described in this Section 12.8(c) and (B) irrevocably waives, and agrees not to assert by way of motion, defense or otherwise, in any such action, any claim that it is not subject personally to the jurisdiction of such courts, that its property is exempt or immune from attachment or execution, that the action is brought in an inconvenient forum, that the venue of the action is improper or that this Agreement or the transactions contemplated hereby may not be enforced in or by such courts in any such action described in this Section 12.8(c).
- 12.9 Waiver of Jury Trial. TO THE EXTENT NOT PROHIBITED BY APPLICABLE LEGAL REQUIREMENTS THAT CANNOT BE WAIVED, THE PARTIES IRREVOCABLY AND UNCONDITIONALLY WAIVE, AND COVENANT THAT THEY WILL NOT ASSERT (WHETHER AS PLAINTIFF, DEFENDANT OR OTHERWISE), ANY RIGHT TO TRIAL BY JURY IN ANY PROCEEDING ARISING IN WHOLE OR IN PART UNDER OR IN CONNECTION WITH THIS AGREEMENT, ANY OTHER TRANSACTION AGREEMENT OR ANY OF THE TRANSACTIONS (INCLUDING ANY LITIGATION AGAINST ANY DEBT FINANCING SOURCE ARISING OUT OF THIS AGREEMENT OR THE DEBT COMMITMENT LETTER), WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER SOUNDING IN CONTRACT, TORT OR OTHERWISE. THE PARTIES AGREE THAT ANY OF THEM MAY FILE A COPY OF THIS PARAGRAPH WITH ANY COURT AS WRITTEN EVIDENCE OF THE KNOWING, VOLUNTARY AND BARGAINED-FOR AGREEMENT BETWEEN THE PARTIES UNCONDITIONALLY AND IRREVOCABLY TO WAIVE THEIR RESPECTIVE RIGHTS TO TRIAL BY JURY IN ANY PROCEEDING WHATSOEVER BETWEEN OR AMONG THEM RELATING TO THIS AGREEMENT, ANY OTHER TRANSACTION AGREEMENT OR ANY OF THE TRANSACTIONS AND THAT SUCH PROCEEDINGS WILL INSTEAD BE TRIED IN A COURT OF COMPETENT JURISDICTION BY A JUDGE SITTING WITHOUT A JURY.

12.10 Specific Performance.

- Each party acknowledges and agrees that, in the event that any of the provisions of this Agreement are not performed in accordance with their specific terms or are otherwise breached, (i) the other party will be irreparably damaged and (ii) such party would not have any adequate remedy at law and would not be adequately compensated by monetary damages. Accordingly, in addition to any other right or remedy to which a party may be entitled, at law or in equity, that party shall be entitled to an injunction or injunctions to prevent breaches or threatened breaches of this Agreement and to compel specific performance of the terms and provisions of this Agreement (including to require the other party to consummate the Closing as contemplated hereby, which shall include the obligation of Purchaser to bring a Proceeding against any Debt Financing Sources, to the extent necessary to cause such Debt Financing Sources to provide funds in accordance with the Debt Commitment Letter if such Debt Financing Sources' failure to provide such funds is preventing the Closing) without (A) the need for proof of actual damages and (B) the requirement of securing or posting any bond or other security or indemnity. Furthermore, each party irrevocably waives and agrees not to raise any objections or defenses that the equitable remedy of specific performance is (1) unavailable to prevent or restrain breaches of this Agreement or to specifically enforce the terms of this Agreement (including to require the other party to consummate the Closing as contemplated hereby), (2) invalid, (3) unenforceable, (4) contrary to law or (5) inequitable for any reason, or that a remedy of monetary damages would provide an adequate remedy.
- (b) For the avoidance of doubt, while Seller may pursue both a grant of specific performance in accordance with this <u>Section 12.10</u> and the payment of the Purchaser Termination Fee pursuant to <u>Section 11.4</u>, under no circumstances shall Seller be permitted or entitled to receive both a grant of specific performance for the consummation of the Transactions and any such Purchaser Termination Fee.
- 12.11 <u>Severability</u>. Each of the provisions of this Agreement is severable, such that if any such provision is held to be or becomes invalid or unenforceable in any respect under the Legal Requirements of any jurisdiction, it shall have no effect in that respect in such jurisdiction and the parties shall use all reasonable efforts to replace it in that respect with a valid and enforceable substitute provision the effect of which is as close to its intended effect as possible.

12.12 Certain Releases.

(a) From and after, and subject to the occurrence of, the Closing, and to the maximum extent permitted by Legal Requirement: (i) Purchaser, for itself and on behalf of its Affiliates (including, after the Closing, the Acquired Companies) and its and their respective successors and assigns of the foregoing, in each case in their capacity as such (each, a "Purchaser Releasing Party"), acknowledges and agrees that, from and after the Closing, to the fullest extent permitted under applicable Legal Requirements, any and all rights, claims, demands, obligations, liabilities, defenses, setoffs, counterclaims, actions and causes of action whatsoever (in each case, whether accrued, absolute, contingent or otherwise, known or unknown, or due or to become due, express or implied, in law or in equity, or based on contract, tort or otherwise) it has, had, may have or may have had against Seller or any of its Affiliates (other than the Acquired Companies), or any of its or their respective past, present and future Representatives or other representatives, and all successors, assigns, executors and heirs of the foregoing, in each case in their capacity as such (each, a "Seller Releasee") and (ii) Seller, for itself and on behalf of its Affiliates and its and

their respective successors and assigns of the foregoing, in each case in their capacity as such (each, a "Seller Releasing Party"), acknowledges and agrees that, from and after the Closing, to the fullest extent permitted under applicable Legal Requirements, any and all rights, claims, demands, obligations, liabilities, defenses, setoffs, counterclaims, actions and causes of action whatsoever (in each case, whether accrued, absolute, contingent or otherwise, known or unknown, or due or to become due, express or implied, in law or in equity, or based on contract, tort or otherwise) it has, had, may have or may have had against Purchaser or any of its Affiliates (including the Acquired Companies), or any of its or their respective past, present and future Representatives or other representatives, and all successors, assigns, executors and heirs of the foregoing, in each case in their capacity as such (each, a "Purchaser Releasee"), now or in the future, in case of each of the foregoing clauses (i) and (ii), to the extent relating to, arising out of or in connection with Seller and its Controlled Affiliates' ownership or operation of, or involvement with, in each case in respect of periods at or prior to the Closing, the Acquired Companies or the Business (collectively, the "Released Matters"), whether arising under, or based upon, any Legal Requirement (including common law) or otherwise (including any right, whether arising at law or in equity, to seek indemnification, contribution, cost recovery, damages or any other recourse or remedy, including as may arise under common law) are, in each case, hereby knowingly, voluntarily, irrevocably and unconditionally waived, released, discharged and relinquished; provided, however, that nothing contained in this Section 12.12 or otherwise shall limit in any respect the rights of any of Purchaser, Seller or any of their respective Affiliates under this Agreement or any of the other Transaction Agreements or any such other schedule, annex, exhibit, Seller Disclosure Schedule, certificate or other document or agreement entered into, made, delivered or made available in connection herewith or entered into after the Closing. Furthermore, without limiting the generality of this Section 12.12, from and after the Closing, no demand, claim or cause of action will be brought or maintained by, or on behalf of, Purchaser or any of its Affiliates (including, after the Closing, the Acquired Companies) against any Seller Releasee, or Seller or any of its Affiliates against any Purchaser Releasee, and no recourse will be sought or granted against any of them, by virtue of, or based upon, the Released Matters; provided, however, that nothing contained in this Section 12.12 or otherwise shall limit in any respect the rights of any of Purchaser, Seller or any of their respective Affiliates under this Agreement or any of the other Transaction Agreements or any such other schedule, annex, exhibit, Seller Disclosure Schedule, certificate or other document or agreement entered into, made, delivered or made available in connection herewith or entered into after the Closing. Purchaser acknowledges, for itself and on behalf of the other Purchaser Releasing Parties, and Seller acknowledges, for itself and on behalf of the other Seller Releasing Parties, that the Legal Requirements of certain states provide substantially the following: "A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS THAT THE CREDITOR OR RELEASING PARTY DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING THE RELEASE AND THAT, IF KNOWN BY HIM OR HER, WOULD HAVE MATERIALLY AFFECTED HIS OR HER SETTLEMENT WITH THE DEBTOR OR RELEASED PARTY." Purchaser acknowledges, for itself and on behalf of the other Purchaser Releasing Parties, and Seller acknowledges, for itself and on behalf of the other Seller Releasing Parties, that such Legal Requirements are designed to protect a Person from waiving claims which it does not know exist or may exist. Nonetheless, Purchaser knowingly and irrevocably acknowledges and agrees, for itself and on behalf of the other Purchaser Releasing Parties, that, from and after the Closing, Purchaser and the other Purchaser Releasing Parties shall be deemed to waive their rights under any such Legal Requirements as and to the extent set forth

in this <u>Section 12.12(a)</u>, and Seller knowingly and irrevocably acknowledges and agrees, for itself and on behalf of the other Seller Releasing Parties, that, from and after the Closing, Seller and the other Seller Releasing Parties shall be deemed to waive their rights under any such Legal Requirements as and to the extent set forth in this <u>Section 12.12(a)</u>; <u>provided</u>, <u>however</u>, that nothing contained in this <u>Section 12.12</u> or otherwise shall limit in any respect the rights of any of Purchaser, Seller or any of their respective Affiliates under this Agreement or any of the other Transaction Agreements or any such other schedule, annex, exhibit, Seller Disclosure Schedule, certificate or other document or agreement entered into, made, delivered or made available in connection herewith or entered into after the Closing.

- (b) Notwithstanding anything in this Agreement to the contrary (except Section 12.21), Purchaser and Seller each acknowledges and agrees that no recourse under this Agreement or any documents or instruments delivered in connection with this Agreement shall be had against any Seller Releasee (other than Seller) or any Purchaser Releasee (other than Purchaser), as applicable, whether by the enforcement of any assessment or by any legal or equitable proceeding, or by virtue of any statute, regulation or other applicable Legal Requirement, it being expressly agreed and acknowledged that no personal liability whatsoever shall attach to, be imposed on or otherwise be incurred by any Seller Releasee (other than Seller) for any obligation of Seller under this Agreement or any documents or instructions delivered in connection with this Agreement or any documents or instructions delivered in connection with this Agreement or any documents or instructions delivered in connection with this Agreement, as applicable, for any claim based on, in respect of or by reason of such obligations or their creation.
- Notwithstanding anything in this Agreement to the contrary, Seller, for itself and on behalf of its Controlled Affiliates, and its and their respective Representatives, hereby waives any rights, claims, demands, obligations, liabilities, defenses, setoffs, counterclaims, actions and causes of action against any Debt Financing Source in connection with this Agreement, the Debt Commitment Letter or the Debt Financing, in each case, in respect of any of the transactions contemplated hereby or thereby, whether at law or in equity, and Seller, for itself and on behalf of its Controlled Affiliates, and its and their respective Representatives, agrees not to commence any Proceeding against any Debt Financing Source in connection with this Agreement, the Debt Commitment Letter or the Debt Financing, in each case, in respect of any of the transactions contemplated hereby or thereby, whether at law or in equity. In furtherance and not in limitation of the foregoing waiver, Seller, for itself and on behalf of its Controlled Affiliates, and its and their respective Representatives, hereby acknowledges and agrees that no Debt Financing Source shall have any liability for any claims or damages to Seller or its Controlled Affiliates or its or their respective Representatives in connection with this Agreement, the Debt Commitment Letter, the Debt Financing or the transactions contemplated hereby or thereby. Notwithstanding the foregoing, nothing in this Section 12.12(c) shall limit, impair or otherwise modify (i) the rights of any of the parties to the Debt Commitment Letter (including Purchaser or its Affiliates party to the Debt Commitment Letter and their respective successors and assigns) set forth in the Debt Commitment Letter in accordance with the terms and conditions thereof or (ii) any liability or obligation of any of the Debt Financing Sources, or any of the rights of Purchaser or its Affiliates under any of the definitive documentation with respect to the Debt Financing.

12.13 The Seller Disclosure Schedule, Schedules, Annexes and Exhibits. The Seller Disclosure Schedule and the schedules, annexes and exhibits attached to this Agreement shall be construed with and as an integral part of this Agreement to the same extent as if the same had been set forth verbatim herein. The representations and warranties of Seller set forth in this Agreement are made and given subject to the disclosures contained in the Seller Disclosure Schedule, subject to this Section 12.13 and as supplemented by the front matters of the Seller Disclosure Schedule. Inclusion of information in the Seller Disclosure Schedule will not be construed as an admission that such information is required to be disclosed, that the matter underlying such information did not arise in the ordinary course of business or in a manner consistent with past practice, that such information is material to the business, operations or condition (financial or otherwise) of Seller, any Acquired Company or the Business or that any Material Adverse Effect has occurred, nor shall it establish any standard of materiality for any purpose whatsoever. Each item set forth in any Schedule or portion of a Schedule, of the Seller Disclosure Schedule shall be deemed to be disclosed only against the corresponding Section, or portion of a Section, of this Agreement; provided, that disclosure of any matter in any Schedule or portion of a Schedule of the Seller Disclosure Schedule shall be deemed to be disclosure of such matter with respect to any Section, or portion of a Section, of this Agreement to which such matter is specifically cross referenced or to which such matter relates to the extent it is reasonably apparent on its face that such disclosure applies. The headings contained in the Seller Disclosure Schedule are inserted for convenience only and shall not be considered in interpreting or construing any of the provisions contained in this Agreement.

12.14 Notices. Any notice required or permitted to be given hereunder shall be sufficient if in writing and shall be deemed to have been duly given or made (a) when personally delivered, (b) on the date sent by electronic transmission (if between 9:00 a.m. and 6:00 p.m. Eastern Time on a Business Day, or, if after 6:00 p.m. Eastern Time on a Business Day or if not on a Business Day, the next Business Day) (provided, that the sending party does not receive an automatically generated message from the recipient's email server that such email could not be delivered to such recipient) or (c) one (1) Business Day after deposit with an overnight courier service, in each case to the addresses, email addresses and attention parties indicated below (or such other address, email address or attention party as the recipient party has specified by prior notice given to the sending party in accordance with this Section 12.14):

To Purchaser at:

BAE Systems, Inc. 1101 Wilson Blvd, Suite 2000 Arlington, VA 22209

Attention: SVP, General Counsel and Corporate Secretary

VP and Associate General Counsel alice.eldridge@baesystems.com

Email: alice.eldridge@baesystems.com katherine.h.brown@baesystems.com With copies (which shall not constitute notice) to:

Kirkland & Ellis LLP 601 Lexington Avenue New York, NY 10022

Attention: Sarkis Jebejian, P.C.

Edward J. Lee, P.C.

Steven Y. Li

Email: sarkis.jebejian@kirkland.com

edward.lee@kirkland.com steven.li@kirkland.com

To Seller at:

Ball Corporation 9200 W. 108th Circle

Westminster, Colorado 80021 Attention: General Counsel

Associate General Counsel

Email: CBaker@ball.com

Kate.Kimball@ball.com

With copies (which shall not constitute notice) to:

Skadden, Arps, Slate, Meagher & Flom LLP 155 North Wacker Drive Chicago, Illinois 60606

Attention: Shilpi Gupta and David R. Clark Email: Shilpi.Gupta@skadden.com; David.Clark@skadden.com

12.15 No Third-Party Beneficiaries. Except for Section 6.10(b) (with respect to Seller's Affiliates and Seller's and its Affiliates' respective Representatives), Section 6.13(b) (with respect to Purchaser's Affiliates and Purchaser and its Affiliates' respective former and current Representatives), Section 7.3 (with respect to the D&O Indemnitees and their respective successors, heirs and legal representatives), the DFS Provisions (to the extent they apply to the Debt Financing Sources), the provisions of Section 11.4(d) (which shall be enforceable by Purchaser's former, current and future Affiliates, their respective Representatives, and the respective successors and assigns of the foregoing Persons), the provisions of Section 12.12 (which shall be enforceable by the Seller Releasees and the Purchaser Releasees) and Section 12.16 (with respect to Skadden and Axinn), a Person who is not a party to this Agreement shall have no right to enforce any of its terms and this Agreement is not intended to give any Person other than the parties to this Agreement and their permitted assigns any rights hereunder.

12.16 <u>Provision Regarding Legal Representation</u>. It is acknowledged by each party that Seller has retained Skadden and Axinn to act as its counsel in connection with the Transactions

Affiliates, on the one hand, and Purchaser, any Acquired Company or their respective Affiliates, on the other hand, Skadden or Axinn (or both) may represent Seller and its Affiliates in such dispute even though the interests of Seller and its Affiliates may be directly adverse to Purchaser, the Acquired Companies or their respective Affiliates, and even though Skadden or Axinn may have represented any of the Acquired Companies or any of their Affiliates in a matter substantially related to such dispute prior to the Closing or may be handling ongoing matters for Purchaser, any Acquired Company or any of their respective Affiliates. Purchaser further agrees that all communications among Seller, the Acquired Companies or any of their respective Affiliates (in the case of the Acquired Companies and their Affiliates, solely prior to the Closing), on the one hand, and their counsel, including in-house counsel, Skadden and Axinn, on the other hand, that relate in any way to the Transactions shall be deemed attorney-client privileged communications (collectively, the "Privileged Communications") and the attorney-client privilege and the expectation of client confidence belongs to Seller and may be controlled by Seller and, notwithstanding anything to the contrary contained in this Agreement, shall not pass to or be claimed by Purchaser, any Acquired Company or any of their Affiliates. The Privileged Communications are (and upon the Closing shall remain) the property of Seller. As to any such Privileged Communications made prior to the Closing Date, Purchaser, together with its Affiliates (including the Acquired Companies), successors and assigns, further agrees that no such party may access, obtain, use or rely on any of the Privileged Communications for any purpose without a waiver of the attorney-client privilege, which waiver shall be in Seller's sole discretion. The Privileged Communications may be used by Seller in connection with any dispute that relates in any way to the Transactions. Notwithstanding the foregoing, in the event that a dispute arises between Purchaser, any Acquired Company or any of their respective Affiliates, on the one hand, and any other Person or Persons (other than a party to this Agreement or any of its respective Affiliates), on the other hand, after the Closing, such Acquired Company and its Affiliates may assert the attorney-client privilege to prevent disclosure of the Privileged Communications to such Person or Persons; provided, however, that none of the Acquired Companies nor their Affiliates may waive such privilege without the prior written consent of Seller.

and that neither Skadden nor Axinn has acted as counsel for any other party in connection with such Transactions. The parties agree that, in the event that a dispute arises after the Closings between Seller or its

- 12.17 <u>No Other Duties</u>. The only duties and obligations of the parties under this Agreement are as specifically set forth in this Agreement, and no other duties or obligations shall be implied in fact, Legal Requirement or equity, or under any principle of fiduciary obligation.
- 12.18 <u>Reliance on Counsel and Other Advisors</u>. Each party has consulted such legal, financial, technical or other experts as it deems necessary or desirable before entering into this Agreement.
- 12.19 <u>Public Announcements</u>. None of Seller or its Controlled Affiliates, Purchaser or its Affiliates or any Representative of any such party shall issue or cause the publication of any press release, public announcement or other public communication in respect of this Agreement or the Transactions without the prior written consent of the other party (which consent shall not be unreasonably withheld, conditioned or delayed), except (a) as may be required by Legal Requirement or stock exchange rules or as the disclosing party deems necessary or advisable to comply with its SEC (or applicable foreign securities laws) filing requirements, in which case the party seeking to publish such press release, public announcement or other public communication

shall use reasonable efforts to provide the other party a reasonable opportunity to comment on such press release, public announcement or other public communication in advance of such publication; provided that the foregoing will not restrict or prohibit Purchaser or Seller from making any announcement in compliance with the terms and conditions of this Agreement to its respective employees, customers and other business relations (in each case, in their capacities as such) to the extent that such party reasonably determines in good faith that such announcement is necessary or advisable, but only to the extent the content of which is consistent with that of any prior public announcement made in compliance with this Section 12.19 or (b) to the extent the contents of such press release, public announcement or other public communication or filing have previously been released publicly by a party or are consistent in all material respects with materials or disclosures that have previously been released publicly, in each case, without violation of this Section 12.19. Notwithstanding the foregoing, this Section 12.19 shall not apply to the disclosure of the express terms of this Agreement in any public filings required by Legal Requirement, stock exchange rules, U.K. listed company rules or SEC (or applicable foreign securities laws) filing requirements.

12.20 <u>Counterparts</u>. This Agreement may be signed in any number of counterparts, including electronic scan copies thereof delivered by electronic mail, each of which shall be deemed an original, with the same effect as if the signatures thereto and hereto were upon the same instrument.

12.21 Purchaser Guarantor.

To induce Seller to enter into this Agreement, Purchaser Guarantor, intending to be legally bound, hereby absolutely, irrevocably and unconditionally guarantees to Seller, as primary obligor and not merely as surety, the due and punctual payment of the Purchase Price by Purchaser (or any Purchaser Designee) under Article II of this Agreement, in each case in accordance with and solely if, as and when required by the terms of this Agreement (the "Obligations"). Purchaser Guarantor agrees that its Obligations under this Section 12.21 are irrevocable, continuing, absolute and unconditional and shall not be discharged or impaired or otherwise affected by, and Purchaser Guarantor hereby irrevocably waives any defenses to enforcement it may have (now or in the future) by reason of (i) any illegality, invalidity or unenforceability of any Obligation or this Agreement or any related agreement or instrument, or any Legal Requirement or Order of any jurisdiction or any other event affecting any term of the Obligations; (ii) any change in the time, place or manner of payment or performance of, or in any other term of the Obligations, or any rescission, waiver, release, assignment, amendment or other modification of this Agreement except to the extent Seller has consented thereto; (iii) any taking, exchange, substitution, release, impairment, amendment, waiver, modification or non-perfection of any collateral or any other guaranty for the Obligations, or any manner of sale, disposition or application of proceeds of any collateral or other assets to all or part of the Obligations; (iv) any default, failure or delay, willful or otherwise, in the performance of the Obligations; (v) any change, restructuring or termination of the corporate structure, ownership or existence of Purchaser Guarantor or Purchaser (or any Purchaser Designee) or any insolvency, bankruptcy, reorganization or other similar proceeding affecting Purchaser (or any Purchaser Designee) or its (or their) assets or any resulting restructuring, release or discharge of any Obligations; (vi) the failure of Seller to assert any claim or demand or to exercise or enforce any right or remedy under the provisions of this Agreement or otherwise; or (vii) the existence of any claim, set-off, counterclaim, recoupment

or other rights that Purchaser Guarantor or Purchaser (or any Purchaser Designee) may have against Seller (other than a defense of payment or performance); <u>provided</u>, <u>however</u>, that, notwithstanding anything in this <u>Section 12.21</u> to the contrary, Purchaser Guarantor does not waive any defenses to the payment of the Obligations that are available to Purchaser under the express terms of this Agreement.

- (b) Purchaser Guarantor hereby represents and warrants to Seller as follows:
- (1) Purchaser Guarantor is duly incorporated, validly existing and in good standing under the Legal Requirements of the jurisdiction of its incorporation.
- (2) Purchaser Guarantor has the requisite corporate power and authority to execute and deliver this Agreement and each other Transaction Agreement to which it is a party, to perform its obligations hereunder and under each other Transaction Agreement to which it is a party and to consummate the Transactions in accordance with the terms of this Agreement and each other Transaction Agreement to which it is a party. The execution, delivery and performance by Purchaser Guarantor of this Agreement and each other Transaction Agreement to which it is a party and the consummation of the Transactions have been duly and validly authorized by all necessary corporate action on the part of Purchaser Guarantor and such authorization has not been subsequently modified or rescinded.
- Guarantor and constitutes, assuming due authorization, execution and delivery of this Agreement by Seller, a valid and binding legal obligation of Purchaser Guarantor, enforceable against Purchaser Guarantor in accordance with the terms hereof. Assuming due authorization, execution and delivery of each other Transaction Agreement to which Purchaser Guarantor is a party by the other parties thereto, each such Transaction Agreement will constitute a valid and binding legal obligation of Purchaser Guarantor, enforceable against Purchaser Guarantor in accordance with the terms thereof.
- (4) The execution, delivery and performance by Purchaser Guarantor of this Agreement and each other Transaction Agreement to which Purchaser Guarantor is a party and the consummation of the Transactions does not conflict with, violate or result in a breach of any Contract, Legal Requirement or Order applicable to Purchaser Guarantor, except for such conflicts, violations or breaches that would not materially adversely affect or delay the ability of Purchaser Guarantor to pay the Obligations.
 - (c) Purchaser Guarantor further acknowledges and agrees as follows:
- (1) Purchaser Guarantor hereby unconditionally and irrevocably waives any right to revoke this <u>Section 12.21</u> and acknowledges that this <u>Section 12.21</u> is continuing in nature and applies to all presently existing and future Obligations, until the complete, irrevocable and indefeasible payment and satisfaction in full of the Obligations or the termination of this Agreement pursuant to <u>Section 11.1</u>.
- (2) This <u>Section 12.21</u> is a direct guaranty and independent of the obligations of Purchaser (or any Purchaser Designee) under this Agreement. Seller may

resort to Purchaser Guarantor for payment and performance of the Obligations whether or not Seller shall have proceeded against Purchaser (or any Purchaser Designee) with respect to the Obligations. Seller may, at Seller's option, proceed against Purchaser Guarantor and Purchaser (or any Purchaser Designee), jointly and severally, or against Purchaser Guarantor only without having obtained a judgment against Purchaser (or any Purchaser Designee).

- (3) Purchaser Guarantor agrees that its guaranty hereunder shall continue to be effective or be reinstated, as the case may be, if at any time all or part of any payment of any Obligation is voided, rescinded or recovered or must otherwise be returned by Seller upon the insolvency, bankruptcy or reorganization of Purchaser (or any Purchaser Designee).
- (d) Purchaser Guarantor waives and shall not exercise any rights that it may acquire by way of subrogation, contribution, reimbursement or indemnification for payments made under this <u>Section 12.21</u> until all Obligations shall have been indefeasibly paid and discharged in full. For the avoidance of doubt, Purchaser Guarantor shall not be deemed to be a party to this Agreement for purposes of any provisions other than this <u>Section 12.21</u>.

[Signature pages follow.]

IN WITNESS WHEREOF, the parties have caused this Agreement to be duly executed as of the date first written above.

BAE SYSTEMS, INC.

By: /s/Alice M. Eldridge Name: Alice M. Eldridge Title: SVP, General Counsel

BALL CORPORATION

By: <u>/s/ Charles E. Baker</u> Name: Charles E. Baker

Title: Vice President & Corporate Secretary

BAE SYSTEMS PLC, solely for purposes of Section 12.21

By: <u>/s/ David Parkes</u> Name: David Parkes Title: Company Secretary The following list identifies contents of schedules and similar attachments omitted from the copy of the Stock Purchase Agreement, dated as of August 16, 2023, by and among BAE Systems, Inc., Ball Corporation and, solely for purposes of Section 12.21 thereof, BAE Systems plc (the "Agreement") contained in this Exhibit 2.1 pursuant to Item 601(a)(5) of Regulation S-K, other than those schedules and similar attachments as to which information is otherwise included within this Exhibit 2.1 in a manner that conveys the subject matter thereof (capitalized terms in this list have the respective meanings ascribed to them in the Agreement):

Annex A	Certain Financial Definitions and Matters
Annex B	Real Estate Reorganization Plan
Exhibit A	Form of Transition Services Agreement
Schedule 1.1(a) of the Seller Disclosure Schedule	Credit Facilities
Schedule 1.1(b) of the Seller Disclosure Schedule	Data Rooms
Schedule 1.1(c) of the Seller Disclosure Schedule	Internal Transfer Employees
Schedule 1.1(d) of the Seller Disclosure Schedule	Key Customers
Schedule 1.1(e) of the Seller Disclosure Schedule	Key Vendors
Schedule 1.1(f) of the Seller Disclosure Schedule	Seller Debt Facilities
Schedule 1.1(g) of the Seller Disclosure Schedule	Knowledge Persons
Schedule 4.2 of the Seller Disclosure Schedule	Non-Contravention; Consents
Schedule 4.3 of the Seller Disclosure Schedule	Organization; Acquired Companies
Schedule 4.4 of the Seller Disclosure Schedule	Title; Shares
Schedule 4.5 of the Seller Disclosure Schedule	Financial Information; Liabilities
Schedule 4.7 of the Seller Disclosure Schedule	Compliance with Legal Requirements
Schedule 4.8 of the Seller Disclosure Schedule	Material Contracts
Schedule 4.9 of the Seller Disclosure Schedule	Litigation
Schedule 4.10 of the Seller Disclosure Schedule	Insurance
Schedule 4.11 of the Seller Disclosure Schedule	Intellectual Property
Schedule 4.12 of the Seller Disclosure Schedule	Real Property
Schedule 4.13 of the Seller Disclosure Schedule	Labor Matters
Schedule 4.14 of the Seller Disclosure Schedule	Employe Benefits
Schedule 4.15 of the Seller Disclosure Schedule	Taxes

4.18 of the Seller Disclosure Schedule Certain Business Practices				
4. To of the Selier Disclosure Schedule General Dusiness Fractices				
4.19 of the Seller Disclosure Schedule Government Contracts				
4.20 of the Seller Disclosure ScheduleBrokers				
4.21 of the Seller Disclosure ScheduleRelated Party Transactions				
4.22 of the Seller Disclosure ScheduleIntercompany Arrangements				
6.1 of the Seller Disclosure ScheduleConduct of the Business Prior to Closing				
Schedule 6.5 of the Seller Disclosure ScheduleTermination of Intercompany Agreements; Release of Guarantees				
6.8(a) of the Seller Disclosure Schedule Governmental Filings or Notices				
6.11(c) of the Seller Disclosure Schedule				
6.13(b) of the Seller Disclosure ScheduleSpecified Litigation				
6.15 of the Seller Disclosure ScheduleSegregation of Email and Messaging Accounts				
6.18(c) of the Seller Disclosure ScheduleWaived Real Estate Reorganization Plan Matter				
7.1 of the Seller Disclosure ScheduleSeller Transitional Trademarks				
7.3 of the Seller Disclosure Schedule D&O Indemnification				
9.4 of the Seller Disclosure Schedule Qualified Retirement Plans				
9.6 of the Seller Disclosure Schedule Annual Incentive-Based Programs				
9.10 of the Seller Disclosure ScheduleSeller Deferred Compensation Plans				

Bylaws of Ball Corporation (As of October 23, 2024)

Article 1 Capital Stock

- **Section A.** Classes of Stock. The capital stock of the corporation shall consist of shares of such kinds and classes, with such designations and such relative rights, preferences, qualifications, limitations and restrictions, including voting rights, and for such consideration as shall be stated in or determined in accordance with the Amended Articles of Incorporation and any amendment or amendments thereof, or the Indiana Business Corporation Law. Consistent with the Indiana Business Corporation Law, capital stock of the corporation owned by the corporation may be referred to and accounted for as treasury stock.
- **Section B.** Certificates for Shares. All share certificates shall be consecutively numbered as issued and shall be signed by the chairman and the corporate secretary.
- **Section C.** Transfer of Shares. The shares of the capital stock of the corporation shall be transferred only on the books of the corporation by the holder thereof, or by his attorney, upon the surrender and cancellation of the stock certificate, whereupon a new certificate shall be issued to the transferee. The transfer and assignment of such shares of stock shall be subject to the laws of the State of Indiana. The board of directors shall have the right to appoint and employ one or more stock registrars and/or transfer agents in the State of Indiana or in any other state.
- **Section D. Control Share Acquisition Statute Inapplicable.** Chapter 42 of the Indiana Business Corporation Law (IC 23-1-42) shall not apply to control share acquisitions of shares of the corporation.

Article 2 Shareholders

- **Section A.** Annual Meetings. The regular annual meeting of the shareholders of the corporation shall be held on the fourth (4th) Wednesday after the first (1st) Wednesday in April of each year, or on such other date within a reasonable interval after the close of the corporation's last fiscal year as may be designated from time to time by the board of directors, for the election of directors of the corporation, and for the transaction of such other business as is authorized or required to be transacted by the shareholders.
- **Section B. Special Meetings.** Special meetings of the shareholders may be called by the chairman of the board or by the board of directors or as otherwise may be required by law.
- **Section C.** Time and Place of Meetings. All meetings of the shareholders shall be held at the principal office of the corporation or at such other place within or without the State of Indiana and at such time as may be designated from time to time by the board of directors.

Section D. Notice of Shareholder Nominations of Directors. Only persons who are nominated in accordance with the following procedures shall be eligible for election as directors of the corporation, except as may be otherwise provided in the Amended Articles of Incorporation of the corporation with respect to the right of holders of preferred stock of the corporation to nominate and elect a specified number of directors in certain circumstances. Nominations of persons for election to the board of directors may be made at any annual meeting of shareholders (a) by or at the direction of the board of directors (or any duly authorized committee thereof) or (b) by any shareholder of the corporation (i) who is a shareholder of record on the date of the giving of the notice provided for in this Section D and on the record date for the determination of shareholders entitled to vote at such annual meeting and (ii) who complies with the notice procedures set forth in this Section D.

In addition to any other applicable requirements, for a nomination to be made by a shareholder, such shareholder must have given timely notice thereof in proper written form to the Secretary of the corporation.

To be timely, a shareholder's notice to the Secretary must be delivered to or mailed and received at the principal executive offices of the corporation not less than ninety (90) days nor more than one hundred twenty (120) days prior to the anniversary date of the immediately preceding annual meeting of shareholders; provided, however, that in the event that the annual meeting is called for a date that is not within thirty (30) days before or after such anniversary date, notice by the shareholder in order to be timely must be so received not later than the close of business on the tenth (10th) day following the day on which such notice of the date of the annual meeting was mailed or such public disclosure of the date of the annual meeting was made, whichever first occurs. In no event shall the public disclosure of an adjournment of an annual meeting commence a new time period for the giving of a shareholder's notice as described above.

To be in proper written form, a shareholder's notice to the Secretary must set forth as to each person whom the shareholder proposes to nominate for election as a director and as to the shareholder giving the notice and any Shareholder Associated Person (as defined below) (i) the name, age, business address, residence address and record address of such person, (ii) the principal occupation or employment of such person, (iii) the class or series and number of shares of capital stock of the corporation which are owned beneficially or of record by such person, (iv) any information relating to such person that would be required to be disclosed in a proxy statement or other filings required to be made in connection with solicitations of proxies for election of directors pursuant to Section 14 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and the rules and regulations promulgated thereunder, (v) the nominee holder for, and number of, shares owned beneficially but not of record by such person, (vi) whether and the extent to which any hedging or other transaction or series of transactions has been entered into by or on behalf of, or any other agreement, arrangement or understanding (including any derivative or short positions, profit interests, options or borrowed or loaned shares) has been made, the effect or intent of which is to mitigate loss to or manage risk or benefit of share price changes for, or to increase or decrease the voting power of, such person with respect to any share of stock of the corporation, (vii) to the extent known by the shareholder giving the notice, the name and address of any other shareholder supporting the nominee for election or reelection as a director on the date of such shareholder's notice, (viii) a description of

all arrangements or understandings between or among such persons pursuant to which the nomination(s) are to be made by the shareholder and any relationship between or among the shareholder giving notice and any Shareholder Associated Person, on the one hand, and each proposed nominee, on the other hand, (ix) a representation that the shareholder intends to appear in person or by proxy at the meeting to nominate the persons named in its notice, (x) all information that would be required to be set forth in a Schedule 13D filed pursuant to Rule 13d-1(a) or an amendment pursuant to Rule 13d-2(a) if such a statement were required to be filed under the Exchange Act and the rules and regulations promulgated thereunder by the shareholder or any Shareholder Associated Person, and (xi) a representation that the shareholder intends to solicit proxies in support of director nominees other than the corporation's nominees in accordance with Rule 14a-19 promulgated under the Exchange Act. Any information required by this paragraph shall be supplemented by the shareholder giving the notice not later than ten (10) days after the record date for the meeting as of the record date. Such notice must be accompanied by a written consent of each proposed nominee to being named as a nominee and to serve as a director for a full term if elected. The corporation may require any proposed nominee to furnish such other information as may reasonably be required by the corporation to determine the eligibility of such proposed nominee to serve as an independent director of the corporation or that could be material to a reasonable shareholder's understanding of the independence, or lack thereof, of such nominee.

Additionally, without limiting the other provisions and requirements of this Section D, unless otherwise required by law, if any shareholder (i) provides notice pursuant to Rule 14a-19(b) under the Exchange Act, and (ii) subsequently fails to comply with the requirements of Rule 14a-19(a)(2) and Rule 14a-19(a)(3) under the Exchange Act, then the corporation shall disregard any proxies or votes solicited for such shareholder's nominees. Upon request by the corporation, if any shareholder provides notice pursuant to Rule 14a-19(b) under the Exchange Act, such shareholder shall deliver to the corporation, no later than five business days prior to the applicable meeting, reasonable evidence that it has met the requirements of Rule 14a-19(a)(3) under the Exchange Act.

No person shall be eligible for election as a director of the corporation unless nominated in accordance with the procedures set forth in this Section D (including the provision of the information required pursuant to the immediately preceding paragraph). If the Chairman of the meeting determines that a nomination was not made in accordance with the foregoing procedures, the Chairman shall declare to the meeting that the nomination was defective and such defective nomination shall be disregarded.

Notwithstanding anything in the third paragraph of this Section D to the contrary, in the event that the number of directors to be elected to the board of directors of the corporation is increased and there is no public disclosure by the corporation naming all of the nominees for director or specifying the size of the increased board of directors at least one hundred (100) days prior to the first anniversary of the preceding year's annual meeting, a shareholder's notice required by this Bylaw shall also be considered timely, but only with respect to nominees for any new positions created by such increase, if it shall be delivered to the Secretary at the principal executive offices of the corporation not later than the close of business on the tenth (10th) day following the day on which such public disclosure is first made by the corporation.

Section E. Notice of Shareholder Proposals of Business. No business may be transacted at an annual meeting of shareholders, other than business that is either (a) specified in the notice of meeting (or any supplement thereto) given by or at the direction of the Board of Directors (or any duly authorized committee thereof), (b) otherwise properly brought before the annual meeting by or at the direction of the board of directors (or any duly authorized committee thereof) or (c) otherwise properly brought before the annual meeting by any shareholder of the corporation (i) who is a shareholder of record on the date of the giving of the notice provided for in this Section E and on the record date for the determination of shareholders entitled to vote at such annual meeting and (ii) who complies with the notice procedures set forth in this Section E.

In addition to any other applicable requirements, for business to be properly brought before an annual meeting by a shareholder, such shareholder must have given timely notice thereof in proper written form to the Secretary of the corporation.

To be timely, a shareholder's notice to the Secretary must be delivered to or mailed and received at the principal executive offices of the corporation not less than ninety (90) days nor more than one hundred twenty (120) days prior to the anniversary date of the immediately preceding annual meeting of shareholders; provided, however, that in the event that the annual meeting is called for a date that is not within thirty (30) days before or after such anniversary date, notice by the shareholder in order to be timely must be so received not later than the close of business on the tenth (10th) day following the day on which such notice of the date of the annual meeting was mailed or such public disclosure of the date of the annual meeting was made, whichever first occurs. In no event shall the public disclosure of an adjournment of an annual meeting commence a new time period for the giving of a shareholder's notice as described above.

To be in proper written form, a shareholder's notice to the Secretary must set forth as to each matter such shareholder proposes to bring before the annual meeting a brief description of the business desired to be brought before the annual meeting and the reasons for conducting such business at the annual meeting and as to the shareholder giving the notice and any Shareholder Associated Person, (i) the name and record address of such person, (ii) the class or series and number of shares of capital stock of the corporation which are owned beneficially or of record by such person, (iii) the nominee holder for, and number of, shares owned beneficially but not of record by such person, (iv) whether and the extent to which any hedging or other transaction or series of transactions has been entered into by or on behalf of, or any other agreement, arrangement or understanding (including any derivative or short positions, profit interests, options or borrowed or loaned shares) has been made, the effect or intent of which is to mitigate loss to or manage risk or benefit of share price changes for, or to increase or decrease the voting power of, such person with respect to any share of stock of the corporation, (v) to the extent known by the shareholder giving the notice, the name and address of any other shareholder supporting the proposal of business on the date of such shareholder's notice, (vi) a description of all arrangements or understandings between or among such persons in connection with the proposal of such business by such shareholder and any material interest in such business and (vii) a representation that the shareholder giving the notice intends to appear in person or by proxy at the annual meeting to bring such business before the meeting. Any information required pursuant to this paragraph shall be supplemented by the shareholder giving the notice not later than ten (10) days after the record date for the meeting as of the record date.

No business shall be conducted at the annual meeting of shareholders except business brought before the annual meeting in accordance with the procedures set forth in this Section E (including the provision of the information required pursuant to the immediately preceding paragraph); provided, however, that, once business has been properly brought before the annual meeting in accordance with such procedures, nothing in this Section E shall be deemed to preclude discussion by any shareholder of any such business. If the Chairman of an annual meeting determines that business was not properly brought before the annual meeting in accordance with the foregoing procedures, the Chairman shall declare to the meeting that the business was not properly brought before the meeting and such business shall not be transacted.

Section F. Definitions.

For purposes of Article Two of these Bylaws:

"Public disclosure" shall mean disclosure in a press release reported by the Dow Jones News Service, Associated Press or comparable national news service or in a document publicly filed by the corporation with the Securities and Exchange Commission pursuant to Section 13, 14 or 15(d) of the Exchange Act.

"Shareholder Associated Person" of any shareholder shall mean (i) any person acting in concert, directly or indirectly, with such shareholder and (ii) any person controlling, controlled by or under common control with such shareholder or any Shareholder Associated Person.

Section G. Proxies.

Any shareholder directly or indirectly soliciting proxies from other shareholders must use a proxy card color other than white, which color shall be reserved for the exclusive use for solicitation by the board of directors of the corporation.

Article 3 Directors

- **Section A. Number and Terms of Office.** The business of the Corporation shall be controlled and managed in accordance with the Indiana Business Corporation Law by a board of eleven directors. The corporation elects not to be governed by IND. CODE §23-1-33-6(c).
- **Section B. Eligibility.** No person shall be eligible for election or reelection as a director after having attained the age of seventy-five prior to or on the day of election or reelection. A director who attains the age of seventy-five during his or her term of office shall be eligible to serve only until the annual meeting of shareholders of the corporation next following such director's seventy-fifth birthday, or until his or her successor is elected and qualified.

Section C. Director Resignation Policy.

In an uncontested election of directors of the corporation, any nominee who receives a greater number of votes "withheld" from his or her election than votes "for" his or her election will, within ten (10) days following the certification of the shareholder vote, tender his or her written resignation to the chairman of the board for consideration by the Nominating/Corporate

Governance Committee (the "Committee"). As used in this Section C, an "uncontested election of directors of the corporation" is an election in which the only nominees are persons nominated by the board of directors of the corporation.

The Committee will consider such tendered resignation and, within sixty (60) days following the certification of the shareholder vote, will make a recommendation to the board of directors concerning the acceptance or rejection of such resignation. In determining its recommendation to the board, the Committee will consider all factors deemed relevant by the members of the Committee.

The Committee also will consider a range of possible alternatives concerning the director's tendered resignation as the members of the Committee deem appropriate, including, without limitation, acceptance of the resignation, rejection of the resignation or rejection of the resignation coupled with a commitment to seek to address and cure the underlying reasons reasonably believed by the Committee to have substantially resulted in the "withheld" votes.

The board of directors of the corporation will take formal action on the Committee's recommendation no later than ninety-five (95) days following the certification of the shareholder vote. In considering the Committee's recommendation, the board will consider the information, factors and alternatives considered by the Committee and such additional information, factors and alternatives as the board deems relevant.

Following the board's decision on the Committee's recommendation, the corporation, within four (4) business days after such decision is made, will publicly disclose, in a Current Report on Form 8-K filed with the Securities and Exchange Commission, the board's decision, together with an explanation of the process by which the decision was made and, if applicable, the board's reason or reasons for its decision.

No director who, in accordance with this Section C, is required to tender his or her resignation, shall participate in the Committee's deliberations or recommendation, or in the board's deliberations or determination, with respect to accepting or rejecting his or her resignation as a director. If a majority of the members of the Committee received a greater number of votes "withheld" from their election than votes "for" their election, then the independent directors then serving on the board of directors who received a greater number of votes "for" their election than votes "withheld" from their election, and the directors, if any, who were not standing for election, will appoint an ad hoc board committee from among themselves (the "Ad Hoc Committee"), consisting of such number of directors as they may determine to be appropriate, solely for the purpose of considering and making a recommendation to the board with respect to the tendered resignations. The Ad Hoc Committee shall serve in place of the Committee and perform the Committee's duties for purposes of this Section C. Notwithstanding the foregoing, if an Ad Hoc Committee would have been created but fewer than three directors would be eligible to serve on it, the entire board of directors (other than the director whose resignation is being considered) will make the determination to accept or reject the tendered resignation without any recommendation from the Committee and without the creation of an Ad Hoc Committee.

This director resignation policy set forth in this Section C, as it may from time to time be amended, will be summarized or included in the corporation's proxy statement for each meeting of shareholders (annual or special) at which directors are to be elected.

- **Section D.** Regular Meetings. The regular annual meeting of the board of directors shall be held immediately after the adjournment of each annual meeting of the shareholders. Regular quarterly meetings of the board of directors shall be held on the fourth (4th) Wednesday after the first (1st) Wednesday of January, July, and October of each year, or on such other date as may be designated from time to time by the board of directors.
- **Section E. Special Meetings.** Special meetings of the board of directors may be called at any time by the chairman of the board or by the board, by giving to each director an oral or written notice setting the time, place and purpose of holding such meetings.
- **Section F.** Time and Place of Meetings. All meetings of the board of directors shall be held at the principal office of the corporation, or at such other place within or without the State of Indiana and at such time as may be designated from time to time by the board of directors.
- **Section G. Notices.** Any notice, of meetings or otherwise, which is given or is required to be given to any director may be in the form of oral notice.
- **Section H.** Committees. The board of directors is expressly authorized to create committees and appoint members of the board of directors to serve on them, as follows:
- (1) Temporary and standing committees, including an executive committee, and the respective chairmen thereof, may be appointed by the board of directors, from time to time. The board of directors may invest such committees with such powers and limit the authority of such committees as it may see fit, subject to conditions as it may prescribe. The executive committee shall consist of three or more members of the board. All other committees shall consist of one or more members of the board. All committees so appointed shall keep regular minutes of the transactions of their meetings, shall cause them to be recorded in books kept for that purpose in the office of the corporation, and shall report the same to the board of directors at its next meeting. Within its area of responsibility, each committee shall have and exercise all of the authority of the board of directors, except as limited by the board of directors or by law, and shall have the power to authorize the execution of an affixation of the seal of the corporation to all papers or documents which may require it.
- (2) Neither the designation of any of the foregoing committees or the delegation thereto of authority shall operate to relieve the board of directors, or any member thereof, of any responsibility imposed by law.
- **Section I. Loans to Directors.** Except as consistent with the Indiana Business Corporation Law, the corporation shall not lend money to or guarantee the obligation of any director of the corporation.

Article 4 Officers

Section A. Election and Term of Office. The officers of the corporation shall be elected by the board of directors at the regular annual meeting of the board, unless the board shall otherwise determine, and shall consist of a chairman of the board of directors, if so designated as an officer by the board, a chief executive officer, a president, one or more vice presidents (any one or more of whom may be designated "corporate," "group," or other functionally described vice president), a corporate secretary, a treasurer, a controller, and may include a vice-chairman of the board of directors. The board of directors may, from time to time, designate a chief operating officer and a chief financial officer from among the officers of the corporation. At any one time a person may hold more than one office of the corporation. Only the chairman and any vice-chairman of the board must be a director of the corporation. Each officer shall continue in office until his successor shall have been duly elected and qualified or until removed with or without cause by the board of directors. Vacancies in any of such offices may be filled for the unexpired portion of the term by the board of directors.

Section B. Chairman of the Board. The chairman of the board shall preside at all meetings of the board of directors and of the shareholders. He shall confer from time to time with members of the board and the officers of the corporation and shall perform such other duties as may be assigned to him by the board. Except where by law the signature of another officer is required, the chairman of the board shall possess the power to sign all certificates, deeds, mortgages, bonds, contracts and other instruments of the corporation which may be authorized by the board of directors. During the absence or inability to act of the chief executive officer, the chairman of the board shall act as the chief executive officer of the corporation and shall exercise all the powers and discharge all the duties of the chief executive officer.

Section C. Vice-Chairman of the Board. The vice-chairman of the board, if elected, shall, in the absence of the chairman of the board, preside at all meetings of the board of directors and of the shareholders. He shall have and exercise the powers and duties of the chairman of the board in the event of the chairman's absence or inability to act or during a vacancy in the office of chairman of the board. He shall possess the same power as the chairman to sign all certificates, contracts, and other instruments of the corporation which may be authorized by the board of directors. He shall also have such other duties and responsibilities as shall be assigned to him by the board of directors or the chairman.

Section D. The Chief Executive Officer. The chief executive officer shall have general charge, supervision and management of the business, affairs and operations of the corporation in all respects, subject to such directions as the board of directors may from time to time provide. The chief executive officer shall be the senior executive officer of the corporation, shall perform such other duties as are customarily incident to such office and shall have full power and authority to see that all directions and resolutions of the board of directors are carried out and, without limitation, the power and authority to determine and direct:

	(a)	The management, supervision and coordination of all business divisions and functional
areas;		

- (b) The implementation of strategic objectives, the setting of operating priorities and the allocation of human and material resources;
- (c) The management, supervision and coordination of all other executive officers and all business division heads; and
- (d) The briefing of the directors at meetings of the board of directors concerning the corporation's business, affairs and operations.

The chief executive officer shall have the power to sign and execute all certificates, deeds, mortgages, bonds, contracts, and other instruments of the corporation as authorized by the board of directors, except in cases where the signing and execution thereof shall be expressly designated by the board of directors or by these bylaws to some other officer or agent of the corporation.

- **Section E.** The President. The president shall perform such duties as the board of directors or the chief executive officer shall from time to time specify and other duties incident to the office of president and as are required of him by these bylaws. The president shall have the power to sign and execute all certificates, deeds, mortgages, bonds, contracts and other instruments of the corporation as authorized by the board of directors, except in cases where the signing and execution thereof shall be expressly designated by the board of directors or by these bylaws to some other officer or agent of the corporation.
- **Section F.** The Vice Presidents. The vice presidents shall possess the same power as the president to sign all certificates, contracts, and other instruments of the corporation which may be authorized by the board of directors, except where by law the signature of the president is required. All vice presidents shall perform such duties as may from time to time be assigned to them by the board of directors, the chairman of the board, and the president. In the event of the absence or disability of the president, and at the request of the chairman of the board, or in his absence or disability, at the request of the vice-chairman of the board, or in his absence or disability at the request of the board of directors, the vice presidents in the order designated by the chairman of the board, or in his absence or disability by the vice-chairman of the board, or in his absence or disability by the board of directors, shall perform all of the duties of the president, and when so acting they shall have all of the powers of and be subject to the restrictions upon the president and shall act as a member of, or as a chairman of, any standing or special committee of which the president is a member or chairman by designation or ex officio.

Section G. The Corporate Secretary. The corporate secretary of the corporation shall:

- (1) Keep the minutes of the meetings of the shareholders and the board of directors in books provided for that purpose.
- (2) See that all notices are duly given in accordance with the provisions of these bylaws and as required by law.

- (3) Be custodian of the records and of the seal of the corporation and see that the seal is affixed to all documents, the execution of which on behalf of the corporation under its seal is duly authorized in accordance with the provisions of these bylaws.
- (4) Keep a register of the post office address of each shareholder, which shall be furnished to the corporate secretary at his request by such shareholder, and make all proper changes in such register, retaining and filing his authority for all such entries.
- (5) See that the books, reports, statements, certificates and all other documents and records required by law are properly kept, filed, and authenticated.
- (6) In general, perform all duties incident to the office of corporate secretary and such other duties as may from time to time be assigned to him by the board of directors.

Section H. The Treasurer. The treasurer of the corporation shall:

- (1) Give bond for the faithful discharge of his duties if required by the board of directors.
- (2) Have the charge and custody of, and be responsible for, all funds and securities of the corporation, and deposit all such funds in the name of the corporation in such banks, trust companies, or other depositories as shall be selected in accordance with the provisions of these bylaws.
- (3) At all reasonable times, exhibit his books of account and records, and cause to be exhibited the books of account and records of any corporation a majority of whose stock is owned by the corporation, to any of the directors of the corporation upon application during business hours at the office of this corporation or such other corporation where such books and records are kept.
- (4) Render a statement of the conditions of the finances of the corporation at all regular meetings of the board of directors, and a full financial report at the annual meeting of the shareholders, if called upon so to do.
- (5) Receive and give receipts for monies due and payable to the corporation from any source whatsoever.
- (6) In general, perform all of the duties incident to the office of treasurer and such other duties as may from time to time be assigned to him by the board of directors.
- (7) All acts affecting the treasurer's duties and responsibilities shall be subject to the review and approval of the corporation's chief financial officer.

Section I. The Controller. The controller of the corporation shall:

(1) Direct the financial closings and the preparation of monthly, quarterly and annual consolidated historical financial statements and reports to executive and operating management.

- (2) Direct the preparation of financial reports required by federal, state and local regulatory agencies and the preparation of quarterly and annual financial statements and reports to shareholders, the Securities and Exchange Commission and other interested parties.
- (3) Provide primary contact for the corporation's independent accountants and all of its consolidated domestic and foreign subsidiaries and represent management to the corporation's domestic and international independent accountants.
- (4) Perform and/or direct technical accounting and financial reporting research and monitor developments in accounting and regulatory standards (e.g., FASB, SEC, EITF, IRS).
 - (5) Direct the corporation's domestic and foreign tax planning, preparation and compliance.
- (6) In general, perform all of the duties incident to the office of controller and such other duties as may from time to time be assigned by the board of directors.
- (7) In case of absence or disability of the controller, the assistant controllers, in the order designated by the chief financial officer, shall perform the duties of controller.
- (8) All acts affecting the controller's duties and responsibilities shall be subject to the review and approval of the corporation's chief financial officer.

Article 5 Indemnification

- **Section A. Indemnification of Directors and Officers General.** Certain of the terms used herein are more specifically defined in Section F of this Article Five.
- (1) The corporation shall indemnify an individual made a party to a proceeding because he is or was a director or officer of the corporation against liability incurred in connection with a proceeding to the fullest extent permitted by the Indiana Business Corporation Law (the "IBCL"), as the same now exist or may hereafter be amended (but only to the extent any such amendment permits the corporation to provide broader indemnification rights than the IBCL permitted the corporation to provide prior to such amendment).
- (2) The termination of a proceeding by judgment, order, settlement, or conviction, or upon a plea of nolo contendere or its equivalent is not, of itself, determinative that the director or officer did not meet the standard of conduct set forth in the IBCL.
- (3) To the extent that a director or officer has been wholly successful, on the merits or otherwise, in the defense of any proceeding to which he was a party, or in defense of any claim, issue, or matter therein, because he is or was a director or officer of the corporation, the corporation shall indemnify the director or officer against reasonable expenses incurred by him in connection therewith regardless of whether the director or officer has met the standards set forth in the IBCL and without any action or determination under Section D of this Article Five.

Section B. Advancement of Expenses.

- (1) The corporation shall pay for or reimburse the reasonable expenses incurred by a director or officer who is a party to a proceeding in advance of final disposition of the proceeding if:
- (a) The director or officer furnishes the corporation a written affirmation of his good faith belief that he has met the standard of conduct set forth in the IBCL;
- (b) The director or officer furnishes the corporation a written undertaking, executed personally or on his behalf, to repay any advances if it is ultimately determined that he is not entitled to indemnification under this Article Five; and
- (c) A determination is made that the facts then known to those making the determination would not preclude indemnification under the IBCL.
- (2) The undertaking required by paragraph (b) of subsection (1) of this Section B must be an unlimited general obligation of the director or officer but need not be secured and may be accepted without reference to financial ability to make repayment.

Section C. Limitations on Indemnification.

- (1) The corporation shall not indemnify a director or officer under Section A of this Article Five unless a determination has been made in the specific case that indemnification of the director is permissible in the circumstances because he has met the standard of conduct set forth in the IBCL. Such determination shall be made within sixty (60) days of the request for indemnification:
- (a) By the board of directors by majority vote of a quorum consisting of directors not at the time parties to the proceeding;
- (b) If a quorum cannot be obtained under paragraph (a) of this subsection, by majority vote of a committee duly designated by the board of directors (in which designation directors who are parties may participate), consisting solely of two or more directors not at the time parties to the proceeding;
 - (c) By special legal counsel:
- (i) Selected by the board of directors or its committee in the manner prescribed in paragraph (a) or (b) of this subsection; or
- (ii) If a quorum of the board of directors cannot be obtained under paragraph (a) of this subsection and a committee cannot be designated under paragraph (b) of this subsection, selected by majority vote of the full board of directors (in which selection directors who are parties may participate); or
- (d) By the shareholders, but the shares owned by or voted under the control of the officers and directors who are at the time parties to the proceeding may not be voted on the

determination; provided, however, that following a change of control of the corporation, with respect to all matters thereafter arising out of acts, omissions or events prior to the change of control of the corporation concerning the rights of any person seeking indemnification under this Article Five, such determination shall be made by special legal counsel selected by such person and approved by the board of directors or its committee in the manner described in Section C(1)(c) above (which approval shall not be unreasonably withheld), which counsel has not otherwise performed services (other than in connection with similar matters) within the five (5) years preceding its engagement to render such opinion for such person or for the corporation or any affiliates (as such term is defined in Rule 405 under the Securities Act of 1933, as amended) of the corporation (whether or not they were affiliates when services were so performed) ("Independent Counsel"). Unless such person has theretofore selected Independent Counsel pursuant to this Section C and such Independent Counsel has been approved by the corporation, legal counsel approved by a resolution or resolutions of the board of directors of the corporation prior to a change of control of the corporation shall be deemed to have been approved by the corporation as required. Such Independent Counsel shall determine as promptly as practicable whether and to what extent such person would be permitted to be indemnified under applicable law and shall render its written opinion to the corporation and such person to such effect. In making a determination under this Section C, the special legal counsel and Independent Counsel referred to above shall determine that indemnification is permissible unless clearly precluded by this Article Five or the applicable provisions of the IBCL. The corporation agrees to pay the reasonable fees of the Independent Counsel referred to above and to fully indemnify such Independent Counsel against any and all expenses, claims, liabilities and damages arising out of or relating to this Article Five or its engagement pursuant hereto.

- (2) Authorization of indemnification or an obligation to indemnify and evaluation as to reasonableness of expenses shall be made as set forth in paragraph (a) above, except that if the determination is made by special legal counsel (pursuant to Section C(1)(c) above), authorization of indemnification and evaluation as to reasonableness of expenses shall be made by those entitled under Section C(1)(c) above to select counsel.
- (3) Indemnification under this Article Five in connection with a proceeding by or in the right of the corporation shall be limited to reasonable expenses incurred in connection with the proceeding.
- **Section D.** Enforceability. The provisions of this Article Five shall be applicable to all proceedings commenced after its adoption, whether such arise out of events, acts, omissions or circumstances which occurred or existed prior or subsequent to such adoption, and shall continue as to a person who has ceased to be a director or officer and shall inure to the benefit of the heirs, executors and administrators of such person. This Article Five shall be deemed to grant each person who is entitled to indemnification hereunder rights against the corporation to enforce the provisions of this Article Five, and any repeal or other modification of this Article Five or any repeal or modification of the IBCL or any other applicable law shall not limit any rights of indemnification then existing or arising out of events, acts, omissions, circumstances occurring or existing prior to such repeal or modification, including, without limitation, the right to indemnification for proceedings commenced after such repeal or modification to enforce this Article Five with regard to acts, omissions, events or circumstances occurring or existing prior to such repeal or modification.

Section E. Severability. If this Article Five or any portion hereof shall be invalidated on any ground by any court of competent jurisdiction, then the corporation shall nevertheless indemnify each director or officer of the corporation as to liabilities incurred in connection with any proceeding, including an action by or in the right of the corporation, to the full extent permitted by any applicable portion of this Article Five that shall not have been invalidated and to the full extent permitted by the Articles and by applicable law.

Section F. Definitions.

As used in this Article, the term:

- (1) "Change of control," for purposes of this Article Five, means (a) an acquisition by any person of 30 percent (30%) or more of the corporation's voting shares; (b) a merger in which the shareholders of the corporation before the merger own 50 percent (50%) or less of the corporation's (or the ultimate parent corporation's) voting shares after the merger; (c) shareholder approval of a plan of liquidation or to sell or dispose of substantially all of the assets of the corporation; and (d) if, during any two-(2) year period, directors at the beginning of the period (and any new directors nominated by a majority of the directors at the beginning of such period) fail to constitute a majority of the board of directors. Notwithstanding the foregoing, a change of control shall not be deemed to occur solely because 30 percent (30%) or more of the then outstanding voting securities is acquired by (i) a trustee or other fiduciary holding securities under one or more employee benefit plans maintained by the corporation or any of its subsidiaries or (ii) any corporation which, immediately prior to such acquisition, is owned directly or indirectly by the shareholders of this corporation in the same proportion as their ownership of shares in this corporation immediately prior to such acquisition.
- (2) "Corporation" includes Ball Corporation and any domestic or foreign predecessor entity of the corporation or a corporation in a merger or other transaction in which the predecessor's existence ceased upon consummation of the transaction.
- (3) "Director" means an individual who is or was a director of the corporation or an individual who, while a director of the corporation, is or was serving at the corporation's request as a director, officer, partner, member, manager, trustee, employee, or agent of another foreign or domestic corporation, partnership, limited liability company, joint venture, trust, employee benefit plan, or other enterprise, whether for profit or not. A director is considered to be serving an employee benefit plan at the corporation's request if his duties to the corporation also impose duties on, or otherwise involve services by, him to the plan or to participants in or beneficiaries of the plan. Director includes, unless the context requires otherwise, the estate or personal representative of a director.
 - (4) "Expenses" include attorneys' fees.
- (5) "Liability" means the obligation to pay a judgment, settlement, penalty, fine (including an excise tax assessed with respect to an employee benefit plan), or reasonable expenses incurred with respect to a proceeding.
- (6) "Party" includes an individual who was, is, or is threatened to be made a named defendant or respondent in a proceeding.

(7) "Proceeding" means any threatened, pending, or completed action, suit, or proceeding, whether civil, criminal, administrative, or investigative and whether formal or informal, except for a proceeding (or part thereof) initiated by a person against the corporation or any director, officer, employee or agent thereof (other than to enforce his rights under this Article Five) and not consented to by the corporation.

Article 6 Corporate Seal

The corporate seal of the corporation shall be a round, metal disc with the words "Ball Corporation" around the outer margin thereof, and the words "Corporate Seal," in the center thereof, so mounted that it may be used to impress words in raised letters upon paper.

Article 7 Amendment

These bylaws may be altered, added to, amended, or repealed by the board of directors of the corporation at any regular or special meeting thereof or by the majority of the outstanding shares of stock entitled to vote generally in the election of directors.

Article 8 Adjudication of Certain Disputes

Section A. Forum for Adjudication of Certain Disputes. Consistent with the Indiana Business Corporation Law (the "IBCL"), unless the corporation consents in writing to the selection of an alternative forum (an "Alternative Forum Consent"), the circuit or superior courts of the State of Indiana shall be the sole and exclusive forum for (a) any derivative action brought on behalf of, or in the name of the corporation; (b) any action asserting a claim for breach of a fiduciary duty owed by any director, officer, employee, or agent of the corporation to the corporation or any of the corporation's constituents identified in Chapter 35 of the IBCL (IC 23-l-35-l(d)); (c) any action asserting a claim arising under any provision of the IBCL, the corporation's Amended Articles of Incorporation and any amendment or amendments thereof, or these bylaws; or (d) any actions otherwise relating to the internal affairs of the corporation; provided, however, that, in the event that the circuit or superior courts of the State of Indiana lack subject matter jurisdiction over any such action or proceeding, the sole and exclusive forum for such action or proceeding shall be a federal court located within the State of Indiana, in each such case, unless a circuit or superior court of the State of Indiana (or federal court located within the State of Indiana, as applicable) has dismissed a prior action by the same plaintiff asserting the same claims because such court lacked personal jurisdiction over an indispensable party named as a defendant therein. Any person or entity purchasing or otherwise acquiring any interest in shares of capital stock of the corporation shall be deemed to have notice of and consented to the provisions of this Article 8. The existence of any prior Alternative Forum Consent shall not act as a waiver of the corporation's ongoing consent right as set forth above in this Section A of Article 8 with respect to any current or future actions or claims.

Section B. Consent to Jurisdiction and Service. If any action the subject matter of which is within the scope of Section A of this Article 8 is filed in a court other than a court located within the State of Indiana (a "Foreign Action") in the name of any shareholder, such

shareholder shall be deemed to have consented to (a) the personal jurisdiction of the state and federal courts located within the State of Indiana in connection with any action brought in such court to enforce Section A of this Article 8 (an "FSC Enforcement Action") and (b) having service of process made upon such shareholder in any such FSC Enforcement Action by service upon such shareholder's counsel in the Foreign Action as agent for such shareholder.

OBLIGOR GROUP SUBSIDIARIES OF BALL CORPORATION $\underline{\text{September 30, 2024}}$

The following is a list of Obligor Group subsidiaries of Ball Corporation (an Indiana Corporation)

Name	State or Country of Incorporation or Organization	Percentage Ownership Direct & Indirect
Ball Advanced Aluminum Technologies Corp. (f/k/a Neuman USA Ltd.)	Delaware	100%
Ball Asia Services Limited	Delaware	100%
Ball Beverage Can Americas Inc. (f/k/a Rexam Beverage Can Americas Inc.)	Delaware	100%
Ball BP Holding Company (f/k/a Rexam BP Holding Company)	Delaware	100%
Ball Container LLC	Delaware	100%
Ball Corporation	Indiana	100%
Ball Glass Containers, Inc.	Delaware	100%
Ball Holdings LLC	Delaware	100%
Ball Inc. (f/k/a Rexam Inc.)	Delaware	100%
Ball Metal Beverage Container Corp.	Colorado	100%
Ball Metal Container Corporation	Indiana	100%
Ball Packaging, LLC (f/k/a Ball Packaging Corp., f/k/a Ball Packaging Holdings Corp.)	Colorado	100%
Ball Pan-European Holdings, LLC (f/k/a Ball Pan-European Holdings, Inc.)	Delaware	100%
Latas De Aluminio Ball, Inc.	Delaware	100%
Rexam Beverage Can Company	Delaware	100%
USC May Verpackungen Holding Inc.	Delaware	100%

Certification

- I, Dan W. Fisher, certify that:
- 1. I have reviewed this Quarterly Report on Form 10-Q of Ball Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our
 conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this
 report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: October 31, 2024

/s/ Dan W. Fisher

Dan W. Fisher

Chairman and Chief Executive Officer

Certification

I, Howard H. Yu, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Ball Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: October 31, 2024

/s/ Howard H. Yu

Howard H. Yu

Executive Vice President and Chief Financial Officer

Certification of Chief Executive Officer Pursuant to 18 U.S.C. Section 1350 and Rule 13a-14(b) or Rule 15d-14(b)

My name is Dan W. Fisher and I am the Chairman and Chief Executive Officer of Ball Corporation (the "Company").

I hereby certify pursuant to 18 U.S.C. Section 1350 as adopted by Section 906 of the Sarbanes—Oxley Act of 2002 that to the best of my knowledge and belief:

- (1) the Quarterly Report on Form 10-Q for the quarter ended September 30, 2024, filed with the U.S. Securities and Exchange Commission on October 31, 2024 ("Report"), fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 as amended; and
- (2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of the operations of Ball Corporation as of, and for, the periods presented in the Report.

/s/ Dan W. Fisher
Dan W. Fisher
Chairman and Chief Executive Officer

Date: October 31, 2024

Ball Corporation

This certification, which accompanies the Form 10-Q to which it relates, is not deemed filed with the Securities and Exchange Commission and is not to be incorporated by reference into any filing of the Company under the Securities Act of 1933, as amended, or the Securities Exchange Act of 1934, as amended (whether made before or after the date of the Form 10-K), irrespective of any general incorporation language contained in such filing.

A signed original of this written statement required by Section 906, or other document authenticating, acknowledging, or otherwise adopting the signature that appears in typed form within the electronic version of this written statement required by Section 906, has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.

Certification of Chief Financial Officer Pursuant to 18 U.S.C. Section 1350 and Rule 13a-14(b) or Rule 15d-14(b)

My name is Howard H. Yu and I am the Executive Vice President and Chief Financial Officer of Ball Corporation (the "Company").

I hereby certify pursuant to 18 U.S.C. Section 1350 as adopted by Section 906 of the Sarbanes—Oxley Act of 2002 that to the best of my knowledge and belief:

- (1) the Quarterly Report on Form 10-Q for the quarter ended September 30, 2024, filed with the U.S. Securities and Exchange Commission on October 31, 2024 ("Report"), fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 as amended; and
- (2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of the operations of Ball Corporation as of, and for, the periods presented in the Report.

/s/ Howard H. Yu

Howard H. Yu Executive Vice President and Chief Financial Officer Ball Corporation

Date: October 31, 2024

This certification, which accompanies the Form 10-Q to which it relates, is not deemed filed with the Securities and Exchange Commission and is not to be incorporated by reference into any filing of the Company under the Securities Act of 1933, as amended, or the Securities Exchange Act of 1934, as amended (whether made before or after the date of the Form 10-K), irrespective of any general incorporation language contained in such filing.

A signed original of this written statement required by Section 906, or other document authenticating, acknowledging, or otherwise adopting the signature that appears in typed form within the electronic version of this written statement required by Section 906, has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.

SAFE HARBOR STATEMENT UNDER THE PRIVATE SECURITIES LITIGATION REFORM ACT OF 1995

In connection with the safe harbor provisions of the Private Securities Litigation Reform Act of 1995 (the Reform Act), Ball is hereby filing cautionary statements identifying important factors that could cause Ball's actual results to differ materially from those described in forward-looking statements made by or on behalf of Ball. Forward-looking statements may be made in several different contexts; for example, in the company's Form 10-K, 10-Q, 8-K and other filings with the Securities and Exchange Commission ("SEC"), quarterly and annual earnings news releases, quarterly earnings conference calls hosted by the company, public presentations at investor and credit conferences, the company's Annual Report and in other periodic communications with investors. As time passes, the relevance and accuracy of forward-looking statements may change; however, except as required by law, the company undertakes no obligation to publicly update or revise any forward-looking statements, whether as a result of new information, future events, or otherwise. You are advised to consult any further disclosures and cautionary statements Ball makes on related subjects in our Form 10-K, 10-Q and 8-K reports and other filings with the SEC. The Reform Act defines forward-looking statements as statements that express or imply an expectation or belief and contain a projection, plan or assumption with regard to, among other things, future revenues, income, earnings per share, cash flow or capital structure. Words such as "expects," "anticipates," "estimates," "believes," "targets," "likely," "foresees", "positions" and similar expressions typically identify forward-looking statements, which are generally any statements other than statements of historical fact. These forward-looking statements are not guarantees of future performance, and you should therefore not place undue reliance upon such statements. Rather, these statements involve estimates, assumptions uncertainties and known and unknown risks, many of which are outside our control, and such statements are therefore qualified in their entirety by reference to the following important factors, among others (including those described in any "Risk Factors" section of our most current Form 10-K, 10-Q or other filings with the SEC), that could cause Ball's actual results or performance to differ materially from those expressed or implied in forward-looking statements made by or on behalf of Ball:

- Fluctuation in customer and consumer growth, spending, demand or preferences, and changes in consumption patterns, both on a seasonal basis and those that may be longer-term or structural in nature, including any effect on demand for our products as a result of the enactment of laws and programs aimed at discouraging the consumption or altering the package or portion size of certain of our customers' products.
- Customer, competitor or supplier consolidation and potential correspondent supply chain influence.
- Loss of one or more major customers or suppliers or changes to contracts with one or more customers or suppliers.
- Failure to achieve anticipated productivity improvements or cost reductions including those associated with capital expenditures; failure to achieve
 an appropriate or optimal level of maintenance and capital expenditures; and failure to achieve expectations with respect to expansion plans,
 accretion to reported earnings, working capital improvements and investment income or cash flow projections.
- Changes in the environment and in climate, including the increasing frequency of severe weather events such as drought, wildfires, storms, hurricanes, tornadoes and floods; virus and disease outbreaks and responses thereto; acts of war, terrorism or other significant or catastrophic geopolitical events or natural disasters, or the catastrophic loss of one of our key manufacturing or operating facilities.
- Financial risks, including inflation and changes in interest rates affecting our debt or our ability to comply with the terms of our debt instruments; changes in the hedging markets or our inability or failure to economically hedge or insure against certain risks or potential exposures; changes in international currency exchange rates of the currencies in the countries in which the company and its joint ventures carry on business; counterparty risk; liquidity risk; inflation or deflation; and changes in capital availability and our access to financing, including the risk of constraints on financing in the event of a credit rating downgrade.

- Competition in each line of business, including with respect to pricing and the possible decrease in, or loss of, sales or margins resulting
 therefrom; product development and introductions by our competitors; and technology changes, including the effect on us of technological or
 product advances made by our competitors.
- The ability or inability to achieve and protect technological and product extensions or new technological and product advances in the company's
 businesses, including our ability to maintain, develop, and capitalize on competitive technologies for the design and manufacture of products and
 to withstand competitive and legal challenges to the proprietary nature of such technology (or protect any unpatented proprietary know-how and
 trade secrets).
- Ball's ability or inability adapt to fluctuating supply and demand and to have available sufficient production capacity, or have such capacity
 available in the right locations, in a timely manner, as well as footprint adjustments and other manufacturing changes.
- Overcapacity or undercapacity of Ball or in the metal container industry generally, and its potential impact on costs, pricing and financial results.
- Regulatory action or issues, or changes in federal, state, local or international laws, including those related to tax, environmental, health and workplace safety, including in respect of climate change, pollution, environmental, social and governance (ESG) reporting, greenhouse gas emissions, or chemicals or substances used in raw materials or in the manufacturing process, particularly concerning Bisphenol-A (BPA), a chemical used in the manufacture of epoxy coatings applied to many types of containers (including certain of those products produced by the company), as well as laws relating to recycling, unfavorable mandatory deposit or packaging legislation, or to the effects on health of ingredients or substances in, or attributes of, certain of our customers' products.
- The effect of any antitrust, intellectual property, consumer, employee or other litigation, investigations or governmental proceedings.
- The extent to which sustainability-related opportunities arise and can be capitalized upon.
- The availability and cost of raw materials, commodities, supplies, energy, logistics and natural resources needed for the production of metal containers, supply chain disruptions, widespread ocean and shipping constraints, and our ability or inability to pass on to customers changes in freight and raw material costs, particularly aluminum.
- Changes in senior management; strikes and other labor issues; increases and trends in various employee benefits and labor costs, including pension, medical and health care costs incurred in the countries in which Ball has operations; the ability to attract and retain skilled labor; rates of return projected and earned on assets and discount rates used to measure future obligations and expenses of the company's defined benefit retirement plans; and changes in the company's pension plans.
- International business and market risks and economic conditions; political and economic instability in various markets, including periodic sell-offs
 on global or regional debt or equity markets; restrictive trade practices of national governments; the imposition of duties, trade actions, taxes or
 other government charges by national governments; exchange controls; trade sanctions; and ongoing uncertainties and other effects surrounding
 geopolitical events and governmental policies and actions, both in the U.S. and in other countries.

- Undertaking successful or unsuccessful acquisitions, divestitures, joint ventures or strategic realignments; and the effect of acquisitions, divestitures, joint ventures or strategic realignments on our business relationships, operating results and business generally.
- The company's ability to protect its information technology network, systems and data and those of its customers and suppliers from attacks or catastrophic failure, and the strength of the company's cyber-security.
- The timing and extent of regulation or deregulation, or changes to regulations and standards, including changes in generally accepted accounting principles or their interpretation.
- Changes to unaudited results due to statutory audits of our financial statements or management's evaluation of the company's internal controls
 over financial reporting.
- Loss contingencies related to income and other tax matters, including those arising from audits performed by national and local tax authorities.